

1	· I	BEFORE THE FLORIDA PUBLIC SERVICE COMMISSION
2		FLORIDA POWER & LIGHT COMPANY
3		TESTIMONY OF TIMOTHY GERRISH
. 4		DOCKET NO.
5		November 10, 2008
6		
7	Q.	Please state your name and business address.
8	Α.	My name is Timothy Gerrish. My business address is 700
9		Universe Blvd., Juno Beach, FL 33408.
10	Q.	By whom are you employed and what is your position?
11	A.	i am employed by Florida Power & Light Company ("FPL" or
12		the "Company") as the Director of Origination.
13	Q.	What are your present job responsibilities?
14	Α.	My current responsibilities include: managing the long term
15		marketing and purchasing of energy commodities to support
16		FPL's System of Generation. Through this function, I assess
17		and negotiate opportunities to enter into procurement or sales
18		of power, natural gas and coal contracts.
19	Q.	Would you please give a brief description of your
20		educational background and professional experience?
21	Α.	I received a Bachelor of Science Degree in Electrical
22		Engineering from Michigan Technological University in 1989,
23		and a Masters of Business Administration Degree in 2001 from 1

the University of North Carolina - Chapel Hill. I have been 1 2 employed in my current position at FPL since August 24, 2007. From 1998 until 2007, I was employed by Progress Energy's 3 4 unregulated affiliate, Progress Ventures, LLC. In 2007, 5 Progress Ventures sold a portion of its business to 6 Constellation Energy, where I briefly worked prior to joining FPL. Throughout my employment at Progress Ventures | held 7 a number of positions in Energy Marketing and Trading, where I 8 9 had responsibility for various regulated and unregulated power including responsibility for projects, negotiation and 10 administering power purchase agreements. Prior to joining 11 Progress Ventures, I was a United States Naval Submarine 12 Officer for nine years. 13

14 Q. What is the purpose of your testimony?

My testimony is provided in support of FPL's request for a Α. 15 Commission finding that entering into the Lee County Electric 16. Cooperative, Inc. ("LCEC") Long-Term Agreement for Full 17 Requirement Electric Service dated August 21, 2007 18 ("Agreement") is prudent and consistent with the interests of 19 FPL's retail customers. The load obligation represented by the 20 Agreement is expected to average approximately 1,100 MW 21 over the term of the Agreement. My testimony describes the 22 Agreement, identifies its principal benefits, and explains why 23

<b>ب</b>			
	1		the Commission should find that entering into the Agreement is
	2		prudent and consistent with the interests of FPL's retail
	3		customers.
	4	Q.	Have you prepared, or caused to be prepared under your
	5		direction or supervision, any exhibits to be used in this
	6		proceeding?
	7	А	. Yes. The following exhibits are included in Appendix I
	8		Exhibit TWG - 1 Long-Term Agreement for Full Requirements
	9		Electric Service
	10		Exhibit TWG – 2 – Retail Impact Analysis (July 2007)
	11	Q.	Please describe the contract and summarize its key
	12		elements.
	13	A.	FPL negotiated the Agreement (my Exhibit TWG – 1) to sell
	14		Full Requirements Electric Service to LCEC. Under the
	15		Agreement, FPL would supply all of LCEC's electrical energy
	16		needs from January 1, 2014 until December 31, 2033. LCEC
	17		would pay FPL a monthly capacity and energy payment, both of
	18		which are derived through cost-of-service formulas that are tied
	19		to actual FPL System Capital and Operating costs. The
	20		Agreement has provisions that allow the term to extend an
	21		additional 20 years (to 2053) unless one of the parties
	22		terminates the agreement prior to December 31, 2026.

### 1 Q. What is FPL's purpose in entering into the Agreement?

A. FPL was contacted by LCEC and advised that they were
 interested in pursuing alternative power supply solutions to
 meet the needs of their customers. The Agreement will allow
 LCEC and its customers to benefit from FPL's reliable and cost
 effective electrical service, in a manner that is not detrimental to
 FPL's retail customers.

8 Q. How does LCEC currently meet its load obligations?

LCEC is a member-owner of Seminole Electric Cooperative, 9 Α. Inc. ("Seminole"). Currently, LCEC receives 100% of their 10 capacity and energy through power supply arrangements with 11 Seminole. LCEC has provided notice to Seminole terminating 12 their power supply arrangement. Beginning in 2010, FPL will 13 supply up to 300 MW of Partial Requirements service to LCEC 14 under a separate Short Term Agreement to meet a portion of 15 LCEC's capacity and energy needs prior to the commencement 16 of this Agreement. 17

18 Q. Please provide an overview of the Agreement.

A. Under the Agreement, FPL would serve the capacity and
 energy needs of LCEC just as it would serve FPL's retail load.
 FPL would forecast LCEC's load requirements and incorporate
 LCEC's load in its total load serving obligations. FPL would
 plan future generation additions around meeting this total load

.

1 obligation. FPL would deliver the capacity and energy to 2 LCEC's load through FPL's transmission system pursuant to FPL's Open Access Transmission Tariff. LCEC has requested 3 4 and subsequently been granted firm Network Transmission 5 Service for the term of the Agreement. LCEC is responsible for paying the Network Transmission Service Tariff Rate and is in 6 7 the process of entering into Transmission Service Agreements with FPL Transmission. 8

10 On a monthly basis, LCEC would pay FPL for the capacity and energy that is required to meet LCEC's load. FPL has 11 designed a cost-based formula rate that captures FPL system 12 costs associated with providing the capacity and energy 13 consistent with Federal Energy Regulatory Commission 14 ("FERC") guidelines. These costs include but are not limited to 15 capital, capital recovery, O&M, property taxes, emissions, 16 corporate overhead, purchased power and fuel (nuclear, gas, 17 coal and oil). 18

9

Q. Does the Agreement contain any terms relating to
 Commission approval?

A. Yes. The Agreement makes Commission approval of the
 Agreement a condition precedent. If the Commission does not
 grant approval satisfactory to FPL by December 31, 2009, then

FPL will have the right to terminate the Agreement. There is also a condition precedent relating to FERC approval of the Agreement, which also gives FPL the right to terminate the Agreement if FERC does not grant satisfactory approval by December 31, 2009.

6 Q. Why is FPL seeking Commission approval for the
 7 Agreement?

The Agreement represents a large, long-term, discretionary Α. 8 commitment of FPL's resources to serving load outside its own 9 retail service territory. LCEC and FPL have concluded that the 10 Agreement will be in the interests of both utilities and their 11 customers, and is consistent with Florida's interest in 12 diversifying the access to and utilization of generating 13 resources within the state. Because of the size and duration of 14 the commitment, however, FPL feels that it is important to 15 confirm that the Commission concurs with our conclusions. 16

Q. Is the LCEC load associated with the Agreement included
 in FPL's current Ten Year Site Plan?

A. Yes. The Agreement was executed on August 21, 2007 and
 thus FPL's 2008 Ten Year Site Plan contemplates serving the
 LCEC load. Entering into the Agreement would be consistent
 with that plan.

1

Q.

# What are the key benefits of entering into this Agreement?

A. This Agreement offers several important benefits to both FPL
and LCEC. In conjunction with the Agreement:

FPL will be able to leverage its economies of scale. The
 LCEC load will represent approximately 1,100 MW of additional
 capacity and energy to which FPL can allocate certain fixed
 costs (i.e. overhead) through the cost-of-service rate structure.

2) LCEC will receive reliable and cost effective electrical
 service from the largest utility in the State with a substantial
 number of generating resources and significant fuel diversity.

3) In view of the benefits both to LCEC customers and FPL 11 customers, this Agreement will enhance the use of Florida's 12 generating resources. FPL has demonstrated over time that 13 we are an efficient, cost effective and environmentally friendly 14 builder of new generation in Florida. The future generation that 15 FPL builds to meet load growth, including LCEC load, will 16 leverage those core competencies to serve the greater needs 17 of all Floridians and not just retail customers of FPL. 18

Q. What has FPL done to evaluate the impact of the
 Agreement on its retail customers?

A. FPL conducted an extensive analysis to determine the impact
 to FPL's retail customers as a result of serving LCEC load
 under this Agreement. FPL analyzed two separate scenarios to

support this analysis. The first scenario, or "Base Case," . 1 2 consisted of determining the retail generation cost responsibility 3 associated with serving FPL's existing load obligation without 4 the LCEC load. Certain assumptions and forecasts were utilized such as load forecasts, fuel forecasts, costs of existing 5 facilities, generation additions etc. The second scenario, or 6 7 "LCEC Case," consisted of determining the retail generation cost responsibility associated with serving FPL's existing load 8 obligation with the addition of LCEC's load. The assumptions 9 10 and forecasts utilized in the LCEC Case were consistent with the Base Case, with the exception that certain additional 11 generation resources were identified for FPL's system as a 12 13 result of serving the LCEC load. The difference in retail cost responsibility between the LCEC Case and the Base Case 14 represented the effect on retail customers associated with 15 serving the LCEC load. If the retail cost responsibility in the 16 LCEC Case is not greater than in the Base Case then one can 17 conclude that the FPL retail customers were not negatively 18 impacted by entering into the Agreement. 19

20

21 FPL performed this analysis in July of 2007, prior to executing 22 the Agreement with LCEC. The analysis showed a favorable 23 impact to FPL's retail customers of approximately \$110 million

(nominal) through 2020. Although the term of the Agreement extends out to 2033, there is considerable uncertainty beyond 2020 as to the amount and type of generation that will be needed to meet FPL's total load obligations. Thus, FPL focused its analysis on the years from 2010 to 2020 as a representative time period: it covers the full duration of the current (2008) Ten Year Site Plan; and it also includes all of FPL's currently identified unit additions (the last being the proposed Turkey Point nuclear generation additions in 2018 and 2020).

11

12

1

2

3

4

5

6

7

8

9

10

The results of FPL's analysis are summarized in the following

13

table:

9

. . .

and the second second

والمراجع وال

	·····		Retail Impact		
_	Base Case Resource plan	LCEC Case Resource Plan	Yearly	Cumulative	
Year			Millions \$	Millions \$	
2010	WCEC 2	WCEC 2	(7)	(7)	
2011	104 MW Nuclear Upgrade	WCEC 3, 104 MW Nuclear Upgrade	36	29	
2012	WCEC 3, 304 MW Nuclear Upgrade	304 MW Nuclear Upgrade	(22)	7	
2013			7	14	
2014		CC	28	43	
2015	CC	CC	(36)	4	
2016	CC CC	CC	16	21	
2017	CC	CC	2	23	
2018	TP 6 nuclear	TP 6 nuclear	12	35	
2019			32	68	
2020	TP 7 nuclear	TP 7 nuclear	42	110	

# 2 Q. Did FPL's analysis evaluate the impact of the Agreement 3 on retail customers beyond 2020?

1

Α. Yes. As shown in my Exhibit TWG-2, the analysis evaluated 4 the impact of the Agreement on retail customers through 2033, 5 assuming a mix of generation additions beyond 2020 that FPL 6 considered to be potentially viable technologies in that time 7 period. This included assumed additions of gas-fired combined 8 cycle units, integrated gasification combined cycle units and 9 nuclear units. The longer term results shown in Exhibit TWG -10 2 reinforce the conclusion one can draw from the results 11 through 2020: the Agreement is consistent with the interests of 12 FPL's retail customers. I want to caution again, however, that 13

the results of the analysis are necessarily more uncertain the
 further one goes out in time, especially into periods beyond the
 time horizon of FPL's currently identified unit additions.

4 Q. Has FPL updated the July 2007 analysis to reflect current
 assumptions?

Α. Yes. In preparation for this filing, FPL performed an updated 6 7 analysis in August of 2008 of the retail rate impact through 2020, using the most recent forecasts for load, fuel and 8 9 generation plans. The updated analysis continues to show a favorable impact to FPL's retail customers, with the cumulative 10 benefit increasing to approximately \$435 million (nominal) 11 through 2020 from the \$110 million nominal cumulative benefit 12 shown in the July 2007 analysis. 13

14 The results of the updated analysis are summarized in the 15 following table:

		LCEC Case Resource Plan	Retail Impact		
	Base Case Resource Plan		Yearly	Cumulative	
Year			Millions \$	<ul><li>Millions \$</li></ul>	
2010	WCEC 2	WCEC 2	(22)	(22)	
2011	WCEC 3	WCEC 3	8	(15)	
2012		·	10	(4)	
2013P	CC conversion	PCC conversion	1,6	12	
2014	PRV conversion	PRV conversion	78	90	
2015			86	175	
2016			73	248	
2017	·		63	311	
2018	TP 6 nuclear	TP 6 nuclear	55	366	
2019			40	405	
2020	TP 7 nuclear	TP 7 nuclear	29	435	

This updated analysis provides additional confidence that the Agreement is consistent with the interests of FPL's retail customers.

5 Q. Please summarize your testimony.

The Agreement benefits LCEC customers by providing them A. 6 with reliable, cost-effective power and increased diversity in the 7 sources of that power, without being disadvantageous to FPL's 8 own retail customers. The Agreement is thus a "win - win" 9 proposition for LCEC, FPL's customers and the state of Florida 10 The Commission should confirm that it concurs as a whole. 11 with these conclusions so that FPL can move forward with 12 implementing the Agreement. 13

14 Q. Does this conclude your testimony?

15 A. Yes.

1

Long-Term Agreement for Full Requirements Electric Service

1

EXHIBIT TWG-1

DOCKET NO.

November 7, 2008

PAGES 1-168

### LONG-TERM AGREEMENT FOR

5

# FULL REQUIREMENTS ELECTRIC SERVICE

# DATED AS OF AUGUST 21, 2007

### BY AND BETWEEN

### FLORIDA POWER & LIGHT COMPANY

# AND

# LEE COUNTY ELECTRIC COOPERATIVE, INC.

# TABLE OF CONTENTS

PAGE
ARTICLE 1 DEFINITIONS
ARTICLE 2 TERM, SERVICE, DELIVERY PROVISIONS AND CONDITIONS
PRECEDENT
ARTICLE 3 SALE AND PURCHASE
ARTICLE 4 MONTHLY CHARGES AND BILLING
ARTICLE 5 CHANGE OF TRANSMISSION
ARTICLE 6 CREDITWORTHINESS
ARTICLE 7 DEFAULT AND REMEDIES
ARTICLE 8 CURTAILMENT, TEMPORARY INTERRUPTIONS AND FORCE
MAJEURE
ARTICLE 9 NOTICES, REPRESENTATIVES OF THE PARTIES
ARTICLE 10 LIABILITY, INDEMNIFICATION, AND RELATIONSHIP OF PARTIES49
ARTICLE 11 REPRESENTATIONS, WARRANTIES AND COVENANTS
ARTICLE 12 ASSIGNMENT
ARTICLE 13 CONFIDENTIALITY
ARTICLE 14 REGULATORY AUTHORITIES
ARTICLE 15 DISPUTE RESOLUTION
ARTICLE 16 CONSOLIDATION, MERGER, CONVEYANCE AND TRANSFER
ARTICLE 17 GENERAL PROVISIONS64
ARTICLE 18 TAXES
ARTICLE 19 RULES OF CONSTRUCTION

- APPENDIX A List of Delivery Points
- APPENDIX B Generation Demand Charge and Generation Energy Charge Formula Rate

APPENDIX C Fuel Charge Factor and Fuel Adjustment Charge Factor

APPENDIX D Financial Covenant Compliance Worksheet

APPENDIX E FPL Generating Resources by Location

APPENDIX F Load Management

APPENDIX G Dedicated Line of Credit

APPENDIX H Form of Letter of Credit

APPENDIX I Form of Consent and Agreement

APPENDIX J Not Used

APPENDIX K Billing and CP Examples

APPENDIX L Loss of Members Charge Calculation

#### LONG-TERM AGREEMENT FOR FULL REQUIREMENTS ELECTRIC SERVICE

This AGREEMENT is dated as of August 21, 2007 ("Effective Date") and is by and between Florida Power & Light Company (together with its successors and permitted assigns, hereinafter referred to as "Company"), and Lee County Electric Cooperative, Inc. (together with its permitted successors and permitted assigns, hereinafter referred to as "Customer") (Company and Customer each individually referred to herein as a "Party," or collectively, the "Parties").

#### **RECITALS:**

WHEREAS, Company is a corporation organized and existing under the laws of the State of Florida, with its principal place of business at 700 Universe Blvd., Juno Beach, Florida 33408, and owns and operates facilities for the generation, transmission and distribution of electric power and energy in the State of Florida;

WHEREAS, Customer is a rural electric cooperative, with its principal place of business at 4980 Bayline Drive, North Fort Myers, Florida 33917-3910;

WHEREAS, Customer desires to purchase Full Requirements Electric Service (as hereinafter defined) from Company to meet Customer's Retail Load at the Delivery Points (as hereinafter defined);

WHEREAS, Company has proposed to supply Full Requirements Electric Service to Customer, subject to the terms and conditions set forth herein; and

NOW THEREFORE, in consideration of the mutual covenants and agreements herein contained, the Parties hereby agree that this Agreement, together with the Appendices attached hereto, sets forth the terms under which Company will supply Full Requirements Electric Service to Customer during the Delivery Period.

#### **ARTICLE 1 DEFINITIONS**

The following words and terms shall be understood to have the following meanings when used in this Agreement or in any associated documents entered into in conjunction with this Agreement, unless a different meaning is plainly required by the context.

1.1 "Affiliate" means, with respect to any Person, any other Person (other than an individual) that, directly or indirectly, through one or more intermediaries, controls, or is controlled by, or is under common control with, such Person. For this purpose, "control" means the direct or indirect ownership of fifty percent (50%) or more of the outstanding capital stock or other equity interests having ordinary voting power.

1.2 "After-Tax Basis" shall mean, with respect to any payment to be received by any Person, the amount of such payment (the base payment) supplemented by a further payment (the additional payment) to that Person so that the sum of the base payment plus the additional payment shall, after deduction of the amount of all Taxes required to be paid by such Person in respect of the receipt or accrual of the base payment and the additional payment (taking into account the net present value of any reduction in such Taxes resulting from Tax benefits realized by the recipient as a result of the payment or the event giving rise to the payment), be equal to the amount required to be received. Such calculations shall be made on the basis of the highest generally applicable Tax rates applicable to the Person for whom the calculation is being made for all relevant periods, and shall take into account the deductibility of Taxes for federal income tax purposes.

1.3 "Agreement" means this Long-Term Agreement for Full Requirements Electric Service, including the Appendices, all as amended, modified or supplemented from time to time in accordance with the provisions hereof.

1.4 "Ancillary Services" means those services as defined in the OATT.

1.5 "Annual Capacity Availability Factor" has the meaning set forth in Section 3.3(b).

**1.6** "Billing Demand" has the meaning set forth in Section 4.8(a).

1.7 "Billing Energy" has the meaning set forth in Section 4.8(b).

1.8 "Billing Period" means the calendar month, which shall be the standard period for all payments and metering measurements under this Agreement unless otherwise agreed to by the Parties or specifically required by the OATT of the Transmission Provider.

1.9 "Business Day" means a day ending at 5:00 p.m. Eastern Prevailing Time, other than Saturday, Sunday and any day which is a legal holiday or a day designated as a holiday by NERC.

1.10 "Calendar Year" means a twelve-consecutive-month period commencing at the start of hour ending ("HE") 0100 on January 1 and ending at the conclusion of HE 2400 on December 31.

1.11 "CFO" means the chief financial officer or similar official.

1.12 "Change in Control" has the meaning set forth in Section 16.1(e).

1.13 "Claims" means all third party claims or actions, threatened or filed and, whether groundless, false, fraudulent or otherwise, that directly or indirectly relate to the subject matter of this Agreement, and the resulting losses, damages, expenses, reasonable attorneys' fees and court costs, whether incurred by settlement or otherwise, and whether such claims or actions are threatened or filed prior to or after the termination of this Agreement.

1.14 "Claiming Party" has the meaning set forth in Section 8.2.

1.15 "Clean Air Act" has the meaning set forth in 42 USC 7401, et seq.

1.16 "Company" has the meaning set forth in the Preamble.

1.17 "Company's Customers" means Company's retail electric customers,

1.18 "Company Generation Resources" means, during the Term, the generation assets owned by Company and Company's share of any jointly-owned units, as changed from time to time during the Term; capacity and energy purchases by Company pursuant to power purchase agreements; and to the extent of the sale of electric power to Company therefrom, all generating plants of co-generators, qualifying facilities, and independent power producers that are not owned by Company, but that produce electric power and sell it to Company. The individual elements of Company Generation Resources are expected to change during the Term of this Agreement and any such changes shall be automatically taken into account in determining the elements that comprise Company Generation Resources. The generation assets owned by Company and Company's share of any jointly-owned units are set forth in Appendix E. Whenever there is a change in such generation assets, Appendix E automatically shall be amended to reflect such change and, if necessary, be submitted to FERC.

1.19 "Confidential Information" means such information about a Party's business or future plans and reports or other information compiled or prepared by a Party which is identified as confidential at the time of disclosure. The terms of this Agreement shall not be confidential except for Section 6.3(d), Section 16.5 and Appendices D and L, which shall be confidential to the extent redacted with the consent of FERC. Notwithstanding the foregoing, the following shall not constitute Confidential Information:

(a) Information which was already in a Party's possession prior to its receipt from another Party and not subject to a requirement of confidentiality;

(b) Information which is obtained from a third Person who, insofar as is known to the receiving Party, is not prohibited from transmitting the information to the Party by a contractual, legal or fiduciary obligation to a Party; and

(c) Information which is or becomes publicly available through no fault of the Party.

1.20 "CP Demand" has the meaning set forth in Section 4.8(c).

1.21 "CPR" has the meaning set forth in Section 15.4(a).

1.22 "Creditworthy Bank" means a bank which satisfies the requirements of being (i) a member bank of the Farm Credit System, provided such Farm Credit System continues to be a Government Sponsored Enterprise or (ii) a U.S. commercial bank or a foreign bank with a U.S. branch, or (iii) the National Rural Utilities Cooperative Finance Corporation, with such banks or entities described in (ii) and (iii), above, having and maintaining a Credit Rating of at least A- from S&P or A3 from Moody's.

1.23 "Credit Rating" means, with respect to any Person on any date of determination, the respective rating then assigned to its unsecured senior long-term debt obligations (not supported by third party credit enhancement) by S&P or Moody's; and if no rating is assigned to such Person's unsecured, senior long-term debt obligations by such agency, the lower of the general corporate credit rating or issuer rating, as applicable, assigned by such rating agency to such Person. 1.24 "Customer" has the meaning set forth in the Preamble.

1.25 "Customer Adjusted Load Management Capability" has the meaning set forth in Section 3.7(a).

1.26 "Customer Charge" has the meaning set forth in Section 4.2.

1.27 "CWIP" has the meaning set forth in Section 4.7(a).

1.28 "Debt Service Coverage Ratio" has the meaning set forth in Appendix D.

1.29 "Dedicated Line of Credit" means the line of credit equal to or greater than the Required Amount that is in a form mutually acceptable to Customer, Company and the Creditworthy Bank providing the line of credit, and that is established by Customer pursuant to and in compliance with the terms of this Agreement, including without limitation Article 6 and Appendix G, and that is provided and maintained by a Creditworthy Bank.

**1.30** "Defaulting Party" means the Party which has caused an Event of Default or with respect to which an Event of Default has occurred.

**1.31** "Delivery Period" means the period defined in Section 2.2(a).

1.32 "Delivery Points" means the point or points designated on Appendix A, as amended from time to time in accordance with Section 2.4.

1.33 "Direct Load Management" has the meaning set forth in Section 3.7(d).

1.34 "Dispute" has the meaning set forth in Section 15.1.

1.35 "Early Termination Date" has the meaning set forth in Section 7.2.

**1.36** "Eastern Prevailing Time" means the prevailing time in the City of Juno Beach, Florida.

1.37 "Effective Date" has the meaning set forth in the Preamble.

1.38 "Energy" means three-phase, 60-cycle alternating current electric energy, expressed in kWhs.

1.39 "Equity Ratio" has the meaning set forth in Appendix D.

**1.40** "Event of Default" means those events by or with respect to the Defaulting Party, as set forth in Section 7.1.

1.41 "FPA" means the Federal Power Act, as amended from time to time.

1.42 "Federal Funds Rate" has the meaning set forth in Section 6.4(d),

1.43 "FERC" means the Federal Energy Regulatory Commission, and any successor entity.

"Force Majeure" means an event or circumstance that is not within the reasonable 1.44 control of, or the result of the negligence of, the Claiming Party, and which, by the exercise of due diligence, the Claiming Party is unable to overcome, avoid, or cause to be avoided in a commercially reasonable manner, and without limiting the scope of this definition, includes acts of God or of the public enemy, or insurrection, war, riot, acts of terrorism, acts of sabotage, civil disobedience or disorder, strikes, lockouts, work stoppages, freezes, fires, earthquakes, lightning, tornadoes, floods, hurricanes, storms or other natural disasters, explosions, failure or imminent threat of failure of equipment, or actions or restraints by court order or governmental authority or arbitration award. Nothing contained herein shall be construed so as to require a Party to settle any strike. lockout, work stoppage or other dispute in which it may be involved, or to seek review of or take an appeal from any administrative, arbitral or judicial action. Force Majeure shall not include (i) the loss or failure of Company's generation in the absence of a Force Majeure event, (ii) Company's ability to sell Energy to another market at a more advantageous price, (iii) the loss of any load by Customer, or (iv) Customer's inability economically to use the Energy.

1.45 "FPSC" means the Florida Public Service Commission and any successor entity.

1.46 "FRCC" means the Florida Reliability Coordinating Council and any successor entity.

1.47 "Fuel Adjustment Charge" has the meaning set forth in Section 4.5.

1.48 "Fuel Charge" has the meaning set forth in Section 4.5.

1.49 "Full Requirements Electric Service" means the supply of firm Energy Company is required to supply to Customer at the Receipt Points during the Delivery Period to serve all of Customer's Retail Load, as the same may fluctuate in real time, as measured at the Delivery Points, and as grossed-up for Losses from the Receipt Points to the Delivery Points based on the demand loss factor for all transmission delivery points on the Company's system for the applicable period as set forth in Appendix B, together with all associated Generation-Related Services and the capacity related thereto.

1.50 "Generation Demand Charge" means the monthly charge determined as the product of the Billing Demand and the Generation Demand Charge Rate.

1.51 "Generation Demand Charge Rate" means the rate in \$/kW-month determined from Schedule A-1 of the Generation Demand Charge and Generation Energy Charge Formula Rate set forth in Appendix B that is used to calculate the Generation Demand Charge.

**1.52** "Generation Energy Charge" means the monthly charge determined as the product of the Billing Energy and the Generation Energy Charge Rate.

**1.53** "Generation Energy Charge Rate" means the rate in \$/kWh determined from Schedule A-2 of the Generation Demand Charge and Generation Energy Charge Formula Rate set forth in Appendix B that is used to calculate the Generation Energy Charge.

1.54 "Generation-Related Services" means those ancillary generation services set forth in Schedules 2 through 6 of the OATT required to maintain reliability within and among the control areas affected by the transmission service in the transmission of Full Requirements Electric Service; provided, however, such services shall be excluded to the extent purchased by Customer from the Transmission Provider.

1.55 "Good Utility Practice" means any of the practices, methods, techniques and standards (including the practices, methods, techniques and standards approved by a significant portion of the electric power generation industry, the Transmission Provider and/or NERC) that, in the exercise of reasonable judgment in light of the facts known or that should reasonably have been known at the time a decision was made and having due regard for, among other things, contractual obligations, applicable Laws and equipment manufacturer's recommendations, could have been expected to accomplish the desired result in a manner consistent with good business practices, reliability, safety, environmental protection, economy and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, technique or standard to the exclusion of all others, but rather to be a range of possible practices, methods, techniques or standards within which a Party shall conduct its activities under this Agreement.

**1.56** "Government Sponsored Enterprise" means an entity that is chartered by the United States Congress to serve a public purpose and is officially sponsored by the United States of America.

1.57 "Investment Grade Credit Rating" means Customer has a current and continuous Credit Rating from each of S&P and Moody's equal to or higher than "BBB-" (with a neutral/stable or higher outlook) by S&P (or its equivalent under any successor rating category of S&P) and/or "Baa3" (with a neutral/stable or higher outlook) by Moody's (or its equivalent under any successor rating category of Moody's). In determining whether Customer has an Investment Grade Rating, if Customer has a Credit Rating from both S&P and Moody's, Customer's Credit Rating shall be an Investment Grade Credit Rating only if the above minimum ratings requirements are satisfied with respect to both S&P and Moody's.

1.58 "kW" means kilowatt.

1.59 "kWh" means kilowatt-hour.

**1.60** "Law" means any statute, rule, regulation, ordinance, order, writ, judgment, decree or other legal or regulatory determination, authorization or approval by a court, regulatory agency, or any other governmental authority of competent jurisdiction.

**1.61** "Lenders" means the National Rural Utilities Cooperative Finance Corporation, CoBank A.C.B. or any other financial institution or bank in the United States or a foreign financial institution or bank with a branch in the United States, in each case extending credit to Customer.

1.62 "Letter(s) of Credit" means one or more irrevocable, transferable standby letters of credit, substantially in the form attached hereto as Appendix H, equal to or greater than the Required Amount, and issued by a Creditworthy Bank.

1.63 "Limited Services Agreement" means that certain limited services agreement between the Parties dated as of the date of this Agreement.

**1.64** "Load Management" means programs which allow the applicable service provider the ability and right to control the load of customers participating in the programs in order to temporarily reduce overall system requirements.

1.65 "Loss of Members" has the meaning set forth in Section 16.5(e).

1.66 "Losses" means any transmission loss, transformation loss, sub-transmission and/or distribution loss or other loss of electrical energy incurred in providing Full Requirements Electric Service hereunder, including without limitation the demand loss factor for all transmission delivery points on Company's system for the applicable period as set forth in Appendix B; provided, however, except for such demand loss factors set forth in Appendix B, Losses shall only include losses incurred at and from the Receipt Points to the Delivery Points. In no case shall Losses include losses that may be incurred from the Delivery Points to the Members.

1.67 "Members" means the Persons taking retail electric service from Customer, any retail customer or member of Customer, any Person that takes retail electric service from Customer pursuant to a franchise or similar arrangement granted to Customer or any Person that Customer has a statutory, contractual or other right or obligation to provide retail electric service, in each case irrespective of whether such Persons are members of Customer.

**1.68** "Monthly Bill" shall have the meaning set forth in Section 4.9 of this Agreement.

**1.69** "Monthly Charges" means the monthly charges set forth in Article 4 of this Agreement or that Company is otherwise permitted to charge Customer pursuant to this Agreement.

1.70 "Moody's" means Moody's Investors Service, Inc. and its successors.

1.71 "MW" means Megawatt.

1.72 "MWh" means Megawatt-hour.

1.73 "NERC" means the North American Electric Reliability Corporation and any successor entity.

1.74 "Non-Defaulting Party" means, with respect to an Event of Default, the Party that is not the Defaulting Party.

1.75 "OATT" means the Transmission Provider's Open Access Transmission Tariff on file at FERC, as amended from time to time.

1.76 "Party" or "Parties" has the meaning set forth in the Preamble.

1.77 "Performance Assurance" means security in the form of either a Dedicated Line of Credit, Letter(s) of Credit, weekly prepayment of charges pursuant to Section 6.4(a), or other security acceptable to Company in its sole discretion.

1.78 "Person" means an individual, partnership, corporation, business trust, joint stock company, trust, unincorporated association, joint venture, governmental entity or authority, limited liability company or any other entity of whatever nature.

1.79 "Points of Receipt" or "Receipt Points" means the points at which Full Requirements Electric Service is first delivered to the Transmission System from Company Generation Resources interconnected with the Transmission System or from points of interconnection with other transmission systems.

**1.80** "Qualifying Facility" shall mean a qualifying small power production facility or a qualifying cogeneration facility as defined in Section 210 of the Public Utility Regulatory Policies Act of 1978 and FERC's implementing regulations, as amended from time to time.

**1.81** "Renewable Benefits" has the meaning set forth in Section 3.6(a).

**1.82** "Required Amount" has the meaning set forth in Section 6.3(b).

1.83 "Retail Competition" means that Company no longer has the exclusive right to serve Company's Customers as a class in its service territory and that instead alternative suppliers of capacity and energy are entitled to make sales directly to such retail end-use customers.

1.84 "Retail Load" means Energy metered at the Delivery Points to meet all requirements of Customer's Members for Energy, taking into account Losses (as provided in Section 4.8), that Customer has a statutory, contractual or other right or obligation to serve.

**1.85** "ROE" has the meaning set forth in Section 4.7(b).

**1.86** "RPS Compliance Costs" has the meaning set forth in Section 3.6(a).

1.87 "RPS Requirement" has the meaning set forth in Section 3.6(a).

1.88 "Rules" has the meaning set forth in Section 15.4(a).

1.89 "RUS" means the Rural Utilities Service of the United States Department of Agriculture or any agency succeeding to the functions of RUS.

1.90 "S&P" means Standard & Poor's Rating Group (a division of McGraw-Hill, Inc.), and its successors.

1.91 "Security Interest" has the meaning set forth in Section 6.3(b)(1).

1.92 "Short-Term Agreement" has the meaning set forth in Section 2.5.

**1.93** "SLM Ratio" has the meaning set forth in Section 3.7(a).

1.94 "Summer Months" means the months of April through October of the current Calendar Year.

1.95 "Tax" or "Taxes" means any tax, charge, assessment, duty, levy, or fee (including any interest, additions to tax or penalties thereon) of the United States or any state, county or local taxing jurisdiction therein, or of any other nation or any jurisdiction therein, including without limitation any income, net worth, franchise, margin, ad valorem, property, occupational, business, severance, emissions, carbon based fee, generation, first use, conservation, energy, transmission, gross or net receipts, utility, privilege, sales, use, excise, capital gain, transfer, license, permit, production, employment, social security, payroll, registration, governmental pension or insurance, withholding, royalty, stamp or documentary, value added, or other tax, charge, assessment, duty, levy or fee (including any interest, additions to tax or penalties thereon) of the United States or any state, county or local taxing jurisdiction therein, or of any other nation or any jurisdiction therein.

1.96 "Term" has the meaning set forth in Section 2.1.

**1.97** "Times Interest Earned Ratio" has the meaning set forth in Appendix D.

**1.98** "Total Debt to EBITDA" has the meaning set forth in Appendix D.

1.99 "Transmission Provider" means the entity or entities that transmit the Full Requirements Electric Service, and provide certain Ancillary Services associated with the delivery of Full Requirements Electric Service, from the Receipt Points to the Delivery Points.

1.100 "Transmission System" means the transmission system of Transmission Provider.

1.101 "Winter Months" means the months of November and December of the immediately prior Calendar Year and the months of January through March of the current Calendar Year.

1.102 "WLM Ratio" has the meaning set forth in Section 3.7(a).

1.103 "Worksheet" has the meaning set forth in Section 6.2.

#### ARTICLE 2 TERM, SERVICE, DELIVERY PROVISIONS AND CONDITIONS PRECEDENT

2.1 Term. The term of this Agreement shall begin as of the Effective Date and extend through and include the end of the Delivery Period (the "Term"); provided, however, Company shall have no obligation to provide Full Requirements Electric Service unless and until the conditions precedent set forth in Section 2.5 have been satisfied or waived. Prior to the commencement of the Delivery Period, Company shall have no obligation to provide Full Requirements Electric Service until the Delivery Period commences and the rights and obligations of the Parties shall be limited to those rights and obligations set forth in the Sections and Articles enumerated in the first paragraph of Section 2.5. To the extent Full Requirements Electric Service has been provided and the Delivery Period has commenced, whenever the Term of this Agreement ends, subject to Section 17.10, the applicable provisions of this Agreement shall continue in effect for one (1) year to the extent necessary to provide for final accounting, billing (including any "true-up" billing provided for in this Agreement), billing adjustments, resolution of any billing disputes, realization of any collateral or other security, set-off, final payments, or payments pertaining to liability and indemnification obligations arising from acts or events that occurred in connection with this Agreement during the Term. Nothing in this Section is to be construed as extending the time permitted to raise Disputes or as extending the period for providing Full Requirements Electric Service.

#### 2.2 Delivery Period.

(a) The "Delivery Period" means the period commencing January 1, 2014, and extending thereafter through December 31, 2033, from HE 0100 through HE 2400 Eastern Prevailing Time, and extending further thereafter through 2053 (an additional period of twenty (20) Calendar Years) unless (i) this Agreement is terminated earlier in accordance with Section 2.2(b), hereof, in which event the Delivery Period shall automatically terminate and expire by its terms as of HE 2400 Eastern Prevailing Time on the effective termination date set forth in the notice provided for in Section 2.2(b); or (ii) an Early Termination Date becomes effective in accordance with the provisions of Section 7.2 of this Agreement, in which event the Delivery Period shall automatically terminate and expire by its terms as of HE 2400 Eastern Prevailing Time on the Early Termination Date. In no event shall the Delivery Period commence on or before the satisfaction or waiver of the conditions precedent set forth in Section 2.5.

(b) This Agreement may be terminated effective on the last day of any Calendar Year on or after December 31, 2033 at HE 2400 by either Party giving the other Party at least seven (7) years prior written notice. For example, in order to terminate this Agreement on December 31, 2033 at HE 2400, a Party must give the other Party written notice of such termination on or before December 31, 2026. In addition, as a further example, in order to terminate this Agreement on December 31, 2038, a Party must give the other Party written notice of such termination on or before December 31, 2031.

(c) Except to the extent set forth in Section 3.6(b) and subject to Section 17.10, any notice given under Section 2.2(b) of this Agreement, for any reason, shall irrevocably

terminate and cancel forever both this Agreement and any obligation on Company's part to provide any level or amount of service (including Full Requirements Electric Service) to Customer under this Agreement effective upon the termination date set forth in the notice given under Section 2.2(b). Customer further expressly waives any and all rights to raise in any forum a claim that Company must provide any level or amount of service (including Full Requirements Electric Service) to Customer on any basis other than pursuant to the terms of the Short-Term Agreement, including without limitation:

(i) any terms or provisions of this Agreement, once this Agreement is terminated or cancelled;

(ii) any previous agreements, if any, between Customer and Company, including, but not limited to, any previous electric service agreements, any settlement agreements resolving FERC proceedings or any settlement agreements resolving any state or federal court suits;

- (iii) any federal Law;
- (iv) any Florida Law; or
- (v) on any other basis.

#### 2.3 Planning.

(a) The Parties understand and agree that, as a result of this Agreement and subject to the satisfaction or waiver of the conditions precedent set forth in Section 2.5, Company shall plan to provide Full Requirements Electric Service to Customer during the Delivery Period of this Agreement. In recognition of this fact, the Parties have entered into this Agreement. In order for Company to continue to plan to serve Customer beyond December 31, 2053, the Parties understand and agree that, absent a written mutual agreement to the contrary, the Parties (although not obligated to do so) would need to negotiate and execute a new agreement for a new delivery period extending beyond the last day of the Term of this Agreement at least seven (7) years before December 31, 2053. Upon the termination or expiration of this Agreement, in addition to such rights and obligations enumerated elsewhere in this Agreement, the grant of any and all rights and interest to Company to supply Full Requirements Electric Service under this Agreement shall cease. Customer and Company shall cooperate, in advance to the extent possible, to make all necessary filings with the Transmission Provider and to perform all other acts necessary to transfer all such rights and interests back to Customer in a timely manner.

(b) At the end of the Term of this Agreement (whenever such event shall occur), Company's obligation to serve Customer under this Agreement shall automatically terminate by the terms of this Agreement, and Customer expressly waives any and all rights to raise in any forum a claim that Company must provide any level or amount of service (including Full Requirements Electric Service) to Customer on any basis other than pursuant to the terms of the Short-Term Agreement, including without limitation: (i) any terms or provisions of this Agreement;

(ii) any previous agreements, if any, between Customer and Company, including, but not limited to, any previous electric service agreements, any settlement agreements resolving FERC proceedings or any settlement agreements resolving any state or federal court suits;

(iii) any federal Law;

(iv) any Florida Law; or

(v) on any other basis.

(c) In order to allow Company to plan to meet the requirements of Customer's Retail Load, Customer agrees to provide Company, by September 15, 2008, and each September 15 thereafter of each Calendar Year during the Term of this Agreement, a forecast of Customer's expected Retail Load and adjustments to such forecast (if any) related to any and all Load Management programs for the following ten (10) Calendar Years for each Delivery Point or for such period remaining in the Term of this Agreement, whichever is shorter. Upon request by Company, Customer shall provide such reasonable supporting detail and assumptions underlying the forecast.

2.4 Delivery Points. The Delivery Points for the Full Requirements Electric Service to be provided hereunder are set forth on Appendix A. In coordination with the Transmission Provider, if necessary, and subject to the execution of any necessary agreements, the Parties may mutually agree to add or delete Delivery Points and to make other changes regarding Delivery Points. Any costs imposed by the Transmission Provider to implement the change in Delivery Points shall be borne by Customer. Consent by either Party to changes regarding Delivery Points shall not be unreasonably withheld. Whenever there is any change in Delivery Points, Appendix A hereto automatically shall be amended to reflect such change and, if necessary, be submitted to FERC.

2.5 Conditions Precedent. Except for Customer's obligations under this Agreement to provide information or data and for each Party's rights and obligations expressly set forth in this Section 2.5, Section 2.6, Section 2.7, Section 2.8, Section 6.3(b), Section 16.1, Article 10, Article 12, Article 13, Article 17 and Article 19 (and, with respect to such rights and obligations, the disclaimers, limitations, rights, remedies and dispute resolution and other provisions set forth in Article 7 and Article 15), neither Party shall have any rights or obligations under this Agreement unless and until each of the following conditions precedent are satisfied or waived by the Parties; provided, however, the conditions precedent set forth in Sections 2.5(a), 2.5(c), 2.5(d), 2.5(e) and 2.5(f) shall apply only to Company and may be waived only by Company in its sole discretion:

(a) On or before December 31, 2009, FERC shall have issued an order accepting or approving the Short-Term Agreement for Partial Requirements Electric Service Dated as of August 21, 2007, By and Between Company and Customer for supply of up to 300

MW of Energy by Company to Customer with Delivery Points at Belle Meade and Buckingham and for a term of four (4) years (the "Short-Term Agreement").

(b) Customer has been granted transmission service from the Receipt Points to the Delivery Points by the Transmission Provider reasonably acceptable to Customer, including the designation of this Agreement as a "Network Resource" (as such term is defined in the OATT) on or before January 1, 2009.

(c) On or before December 31, 2009, the FPSC shall have issued one or more final orders authorizing or approving this Agreement (or the supply of Full Requirements Electric Service as contemplated by this Agreement) which order(s) shall also include without limitation the approval of any additional generating resources that would be used to support the Retail Load, all in a form acceptable to Company in its sole discretion, and any governmental approvals and authorizations required in order to construct, develop, operate or maintain any such additional generating resources, whether with respect to need, siting, emissions, or otherwise, shall have been obtained in a form acceptable to Company in its sole discretion and any conditions of such final order(s) and/or governmental approvals and authorizations shall have been satisfied in a form acceptable to Company in its sole discretion by such date.

On or before December 31, 2009, Company shall have filed this Agreement with (d) FERC; provided, however, that Company shall have no obligation to file this Agreement with FERC if: (i) in its sole discretion, Company determines that modifications or conditions, if any, imposed by FERC on its acceptance or approval of the Short-Term Agreement, do, in the judgment of the Company exercised in its sole discretion, (A) adversely affect the commercial, economic, legal and/or other benefits reasonably anticipated to be derived by Company from the Short-Term Agreement; (B) adversely affect the commercial, economic, legal and other benefits reasonably anticipated to be derived by Company from this Agreement assuming that such modifications or conditions imposed by FERC on the Short-Term Agreement also were to be placed on FERC's acceptance of this Agreement, or (C) otherwise have an adverse impact on Company's Customers, Company or its shareholders (or otherwise have an adverse impact on Company's Customers, Company or its shareholders assuming that such modifications or conditions imposed by FERC on the Short-Term Agreement also were to be placed on FERC's acceptance or approval of this Agreement); or (ii) the conditions precedent set forth in Sections 2.5(a) and (c) have not been satisfied and have not been waived by the Company.

(e) On or before December 31, 2009, FERC shall have issued an order accepting or approving this Agreement for filing and permitting it to become effective as filed without modification, suspension, investigation or any condition (including setting this Agreement, or part thereof, for hearing) unacceptable to Company in its sole discretion; and

(f) The Board of Directors of Company has approved any modifications or conditions to this Agreement required by FERC and the performance of Company's obligations thereunder on or prior to December 31, 2009. If the Board of Directors has

not approved such modifications or conditions to this Agreement by December 31, 2009, Company may waive this condition precedent by providing written notice of waiver to Customer by such date; otherwise this Agreement shall be null and void.

In the event any of the above conditions precedent set forth in this Section 2.5 are not satisfied or are not waived in writing by the dates provided above, Company's obligations under this Agreement shall automatically terminate by the terms of this Agreement, and Customer expressly waives any and all rights to raise in any forum a claim that Company must provide service (including without limitation Full Requirements Electric Service) to Customer on any basis other than pursuant to the terms of the Short-Term Agreement, including without limitation:

(i) any terms or provisions of this Agreement;

(ii) any previous agreements, if any, between Customer and Company, including, but not limited to, any previous electric service agreements, any settlement agreements resolving FERC proceedings or any settlement agreements resolving any State or Federal court suits;

(iii) any federal Law;

(iv) any Florida Law; or

(v) on any other basis.

2.6 FERC Acceptance. With respect to the condition set forth in Section 2.5(e), Company agrees to accept any modifications or conditions imposed by FERC with respect to this Agreement that, in the judgment of Company exercised in its sole discretion, do not (i) adversely affect the commercial, economic, legal and other benefits reasonably anticipated to be derived by Company from this Agreement or (ii) otherwise have an adverse impact on Company's Customers, Company or its shareholders; provided, however, Company will be under no obligation to accept any modifications to this Agreement or conditions imposed by FERC in connection with the acceptance or approval of this Agreement, if in the judgment of Company, exercised in its sole discretion, such modifications to this Agreement or conditions imposed by FERC in connection with the acceptance or approval of this Agreement would be adverse to Company as set forth in subsections (i) and/or (ii), above.

2.7 Cooperation. If requested by Company, Customer shall undertake commercially reasonable efforts to cooperate with and assist Company in Company's efforts to secure acceptance of this Agreement by FERC and request FERC action on these filings and, upon Company's request, shall make a timely submittal at FERC affirmatively supporting the acceptance of this Agreement by FERC in each case without modification, suspension, investigation, or condition. In addition, if requested by Company, Customer shall submit letters to the FPSC and take such other actions as reasonably requested by Company in support of Company's effort to secure authorization or approval from the FPSC pursuant to Section 2.5(c). To the extent that FERC requires modifications to this

Agreement that are acceptable to Company pursuant to Section 2.6, Customer shall promptly execute an amendment to this Agreement that implements such modifications.

#### 2.8 Termination for Failure of Condition Precedent.

(a) In the event any of the above conditions precedent set forth in Section 2.5 are not satisfied or are not waived in writing by the dates provided above, this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(b) In the event that FERC issues an order requiring modifications or conditions on its acceptance or approval of the Short-Term Agreement and Company does not, within the earlier of (i) December 31, 2009, or (ii) six (6) months after the later of (x) the date of such FERC order or (y) the satisfaction of the conditions set forth in Section 2.5(c), provide written notice to Customer that such modifications are acceptable to Company in its sole discretion as set forth in Section 2.5(d), then this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(c) In the event that FERC issues an order requiring modifications or conditions on its acceptance or approval of this Agreement and Company does not, by December 31, 2009, provide written notice to Customer that such modifications are acceptable to Company in its sole discretion as set forth in Section 2.5(e) and Section 2.6, then this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(d) Notwithstanding any provision of this Agreement to the contrary, in the event this Agreement terminates pursuant to this Section 2.8, subject to Section 17.10, the Parties shall be released and discharged from any and all obligations arising or accruing hereunder from and after such date and shall not incur any additional liability to each other as a result thereof, except to the extent of any claims arising with respect to Section 2.7, Article 10 or Article 13. With respect to any such claims, Articles 7, 10, 13, 15, 17 and 19 shall survive and continue to apply.

#### ARTICLE 3 SALE AND PURCHASE

#### 3.1 Full Requirements Electric Service.

(a) During the Delivery Period, except as permitted by Sections 3.3, 3.4, 3.5 and 3.7, Company shall sell and deliver to Customer and Customer shall receive and purchase from Company, Full Requirements Electric Service at the Receipt Points sufficient to serve Customer's Retail Load at the Delivery Points, as such load may be verified by Company. As a provider of Full Requirements Electric Service, Company is solely responsible for satisfying all requirements and paying all costs incurred or to be incurred to provide Full Requirements Electric Service to the Receipt Points, and Customer shall pay for such Full Requirements Electric Service as provided in this Agreement, including without limitation as provided in Article 4.

(b) Subject to Section 16.5, Company understands that Customer's Retail Load may change from time to time. Subject to Sections 3.3, 3.4, 3.5 and 3.7, except as specifically provided herein, at no time during the Delivery Period shall Customer use, directly or indirectly, including through any of its Affiliates (i) constructed or purchased generation resources, (ii) power purchases or power purchase agreements, or (iii) any other source of generation, energy or capacity to reduce Customer's Retail Load. Except to the extent arising from a Force Majeure or as set forth in Section 16.3, Company is responsible for Full Requirements Electric Service regardless of changes in Retail Load arising from daily fluctuations, increased or decreased usage, extreme weather and/or similar events.

(c) Customer shall promptly provide notice to Company as soon as practical of any new Member to be included within Retail Load that by itself would reasonably be expected to cause a load increase exceeding 6 MW to be served by Company. Notwithstanding the foregoing and except as provided in Section 16.3, Company shall be obligated to serve the requirements of such Member included within Customer's Retail Load.

(d) Except as set forth in Section 16.5(d), Customer shall not resell the Full Requirements Electric Service purchased hereunder except to its Retail Load.

#### 3.2 Transmission Service and Ancillary Services.

(a) It is Customer's responsibility to arrange and pay for, enter into and maintain all necessary agreements for the delivery and transmission of Full Requirements Electric Service under this Agreement at and from the Points of Receipt to the Delivery Points (and at and from the Delivery Points) for the Term and to arrange and pay for transmission and Ancillary Services for (and enter into and maintain all necessary agreements for) the delivery of Energy under this Agreement at and from the Points of Receipt to the Delivery Points (and at and from the Delivery Points), including without limitation the designation of this Agreement as a "Network Resource" (as such term is defined in the OATT). Notwithstanding any provision in this Agreement to the contrary, there shall be no reduction in Customer's payment or performance obligations under this Agreement as a result of curtailments, interruptions, or reductions of transmission service or Ancillary Services at and from the Points of Receipt to the Delivery Points (or at and from the Delivery Points), whether as a result of Force Majeure or otherwise. If the Billing Demand or CP Demand is zero for any applicable calendar month as a result of curtailments, interruptions, or reductions of transmission service or Ancillary Services. the Billing Demand or CP Demand (as applicable), for purposes of calculating the Generation Demand Charge, shall be the average Billing Demand or CP Demand (as applicable), for the most recent Summer Months or Winter Months depending on whether the applicable month is itself a Summer Month or Winter Month.

(b) Customer shall enter into, and file with the Transmission Provider and other applicable entities, all documents necessary for Customer to fulfill its obligations to take Full Requirements Electric Service at and from the Points of Receipt.

(c) Notwithstanding any contrary provision set forth in this Agreement, in the event of a failure to perform, breach or default by Company of any of its obligations in its capacity as the Transmission Provider or as Customer's limited services provider under the Limited Services Agreement, such failure, breach or default shall not constitute a failure to perform, breach or default of Company's obligations under this Agreement. Any liability or obligations of Company in its capacity as Transmission Provider shall be solely as set forth in any transmission services agreement between Customer and Transmission Provider or as set forth under the OATT, and any liability or obligation of Company in its capacity as limited services provider shall be solely as set forth in the Limited Services Agreement.

(d) Each Party shall provide the other Party access to information the other Party reasonably requests to facilitate the administration of this Agreement with respect to the Transmission Provider.

#### 3.3 Qualifying Facility Purchases.

(a) So long as Customer is receiving Full Requirements Electric Service from Company, if Customer has an obligation under Law, and only if it has an obligation under Law, to purchase electrical output from a duly licensed and operating Qualifying Facility connected directly to Customer's electric system, Customer may purchase the electrical output from such Qualifying Facility connected to Customer. Company shall not be obligated to purchase from any Qualifying Facility on Customer's electric system.

If Customer is receiving electrical output from a Qualifying Facility connected **(b)** directly to Customer's electric system, Customer shall pay the Qualifying Facility directly for all such services. Customer's monthly Billing Energy calculated pursuant to Section 4.8(b) of this Agreement shall not include any amounts of energy purchased by Customer from the Qualifying Facility; but the calculation of the Generation Demand Charges that Customer is obligated to pay Company shall be computed as though Company were serving that portion of the Retail Load displaced by Customer's receipt of electric output from any Qualifying Facility (other than with respect to the 5MW aggregate adjustment permitted pursuant to Sections 4.8(a) and (c)). In calculating the Generation Demand Charge paid by Customer, the Billing Demand (and the Customer Billing Demand schedule set forth as part of Appendix B) and the CP Demand (and the Customer CP Demand schedule set forth as part of Appendix B) shall include any purchases made or electrical output received by Customer from any Qualifying Facility, reduced by an amount of up to five (5) MW in the aggregate for the amount of any purchases made or electrical output received by Customer from all Qualifying Facilities as further described in Sections 4.8(a) and 4.8(c). The calculation of the Generation Demand Charge that Customer is obligated to pay Company shall continue to be computed as though Company were serving that portion of the Retail Load displaced by Customer's receipt of output from any Qualifying Facility until such time that the FPSC,

in a proceeding, recognizes Company's generation that has been displaced by any such Qualifying Facility as being needed and serving to satisfy Company's system wide generation requirements for the benefit of Company's Customers and permits Company to recover the costs (including ROE) associated with any such displaced generation through the retail rates payable by Company's Customers. To the extent such a finding has been made by the FPSC, in the event that the Qualifying Facility fails to maintain an Annual Capacity Availability Factor of at least eighty-five percent (85%) over the previous twelve-month period during the Term, Company shall have the right to reinstitute the Generation Demand Charge until such time, if ever, as the Qualifying Facility again achieves an Annual Capacity Availability Factor of at least eighty-five percent (85%). The term "Annual Capacity Availability Factor" shall mean, for purposes of this provision, (x) the sum of the capacity that is deemed available by generator (subject to certification or audit rights of Company) from the applicable Qualifying Facility during each hour of the previous twelve months, divided by (y) the sum of the rated capacity as recognized by the FRCC for each hour of the previous twelve months for the applicable Qualifying Facility. Until a full twelve months have elapsed for purposes of the aforementioned calculation, the months that have occurred to date shall be used for purposes of defining the Annual Capacity Availability Factor. As a result, the Annual Capacity Availability Factor formula can be summarized as follows:

Annual Capacity Availability Factor =

(Sum of hourly available capacity for previous 12 months)

(Sum of hourly rated capacity for previous 12 months)

(c) Upon a request from any Qualifying Facility, Customer shall notify Company of the proposed connection of any Qualifying Facility to Customer's electric system. Notification shall be in writing as soon as practical. Company shall have the right, but not the obligation, to act as Customer's operations and/or administrative services provider (in whole or in part) with respect to any Qualifying Facility, including without limitation with respect to the negotiation of applicable terms and conditions and operational requirements, instructions and interface with Customer.

(d) Customer shall not engage in activities to promote the construction or installation of any Qualifying Facility.

**3.4** Renewable Energy Connected Directly to Customer's System. Customer shall notify Company of the proposed purchase or receipt of energy and/or capacity by Customer to be delivered to Customer's electric system from any renewable energy resource of a Member of Customer that is connected directly to Customer's electric system on Customer's side of the Delivery Point. Notification shall be in writing as soon as practical, but at least ninety (90) days prior to the commencement of the purchase. Company shall not be obligated to purchase from any renewable energy resource on Customer's system. Customer's monthly Billing Energy calculated pursuant to Section 4.8(a) of this Agreement shall not include any amounts of energy purchased by Customer from such renewable resources; but, the calculation of the Generation Demand Charges that Customer is obligated to pay Company shall be computed as though Company were serving that portion of the Retail Load displaced by Customer's receipt of electrical output from any renewable energy resource (other than with respect to the 2 MW aggregate adjustment permitted pursuant to Sections 4.8(a) and (c)). In calculating the Generation Demand Charge paid by Customer, the Billing Demand (and the Customer Billing Demand schedule set forth as part of Appendix B) and the CP Demand (and the Customer CP Demand schedule set forth as part of Appendix B) shall include any purchases made or electrical output received by Customer from any renewable energy resource, reduced by an amount of up to two (2) MW in the aggregate for the amount of any purchases made or electrical output received by Customer from all such renewable energy resources as further described in Section 4.8(a) and 4.8(c). For purposes of clarification, Company and Customer agree that any renewable energy described in this Section 3.4 shall not be included in any calculation for distributed generation described in Section 3.5.

#### 3.5 Distributed Generation.

(a) Customer shall notify Company of the proposed purchase of energy and/or capacity to be delivered to Customer's electric system from any distributed generation resource of a Member of Customer that is connected directly to Customer's electric system on Customer's side of the Delivery Point. Notification shall be in writing as soon as practical, but at least ninety (90) days prior to the commencement of the purchase. Company shall not be obligated to purchase from distributed generation on Customer's system. Customer's monthly Billing Energy calculated pursuant to Section 4.8(a) of this Agreement shall not include any amounts of energy purchased by Customer from such distributed generation resources; but, the calculation of the Generation Demand Charges that Customer is obligated to pay Company shall be computed as though Company were serving that portion of the Retail Load displaced by Customer's receipt of electrical output from any distributed generation resource (other than with respect to the 2 MW aggregate adjustment permitted pursuant to Sections 4.8(a) and (c)). In calculating the Generation Demand Charge paid by Customer, the Billing Demand (and the Customer Billing Demand schedule set forth as part of Appendix B) and the CP Demand (and the Customer CP Demand schedule set forth as part of Appendix B) shall include electrical output received by Customer from any distributed generation resource, reduced by an amount of up to two (2) MW in the aggregate for the amount of any purchases made or electrical output received by Customer from all such distributed generation resources as further described in Section 4.8(a) and 4.8(c).

(b) The provisions of Section 3.5(a) requiring the payment of Generation Demand Charges attributable to the capacity from any such distributed generation resources and as adjusted by the up to 2 MW credit to Customer's monthly Billing Demand shall not apply to the purchase or the receipt of energy from any distributed generation resource during any period when the FRCC (or other successor entity with similar responsibilities) declares a "Generating Capacity Emergency", as such term is defined in the FRCC Generating Capacity Shortage Plan adopted by the FPSC in April 2003, or during the period when the Transmission Provider has requested a curtailment of Customer's Retail Load for any reason other than economic curtailment.
## 3.6 Renewable Portfolio Standards.

(a) If Company is required by Law to meet the requirements of any renewable (including without limitation clean energy) portfolio standards, including but not limited to the obligation to maintain a specific amount of renewable energy and/or renewable energy credits ("RPS Requirement"), Company shall comply (as described in this Section 3.6(a)) with such RPS Requirement and, in addition, Customer shall comply with the same terms and conditions of such RPS Requirement applicable to the Company in order to facilitate Company's compliance with Company's RPS Requirement. Subject to the conditions set forth in Sections 3.6(b) through 3.6(d), during the Delivery Period, Customer shall comply with the RPS Requirement applicable to Company by paying its share of any and all costs and expenses associated with Company's compliance with the RPS Requirement, including but not limited to: (i) any investments by Company in renewable generation, including a return on equity equal to the ROE, and related operation and maintenance expenses; (ii) any capacity, energy and/or renewable credits acquired through purchase power agreements; (iii) the purchase of any renewable energy credit(s) and/or renewable energy; and (iv) any costs, expenses, charges, fees, assessments and/or Taxes of the type recoverable through Article 4 and Appendices B and/or C (collectively, "RPS Compliance Costs"). Customer's share of the costs associated with Company's compliance with the RPS Requirement shall be charged to Customer as part of the Monthly Bill for Full Requirements Electric Service, provided, however, in its Monthly Bill, Customer shall receive a credit for its share, if any, of income received by Company from the sale of Renewable Benefits to third parties. The cost and credit allocation to Customer shall be made pursuant to the cost-of-service formulas set forth in Article 4 and the related Appendices. Unless and until the RPS Requirement is implemented in Florida, both Parties acknowledge that it is unknown as to whether such a requirement will apply only to generators, to load serving entities or to both generators and load serving entities. As a result, to the extent that the RPS Requirement applies to Customer as well as Company such that Customer is required to make a separate demonstration of compliance, Company and Customer shall use commercially reasonable efforts to transfer to Customer its share of any renewable energy credits and similar related renewable portfolio credits that are created by statute as a result of the enactment of the RPS Requirement ("Renewable Benefits") such that Company and Customer are both able to mutually satisfy their respective renewable portfolio standard obligations; provided, however, that Company (following any such transfer) is released of any obligation to make a separate demonstration of compliance for Customer's share of Company's generation or load (whether due to the generation required to support Retail Load or otherwise). Notwithstanding anything contained in this Agreement to the contrary, to the extent that the FPSC determines that Company's Customers are incurring an incremental cost burden as a result of Customer's RPS Compliance Costs allocation and/or RPS Requirement's benefits allocation under this Section 3.6, and the FPSC disallows (either during or following the expiration of the Term) the recovery of such incremental costs, all such disallowed costs shall be paid by Customer to Company. The preceding sentence shall survive the termination or expiration of this Agreement and shall continue to apply as a Customer obligation pursuant to subsection (b), below, following the termination or expiration of this Agreement.

Company's obligations to Customer with respect to the RPS Requirement as set **(b)** out in this Section 3.6 shall terminate upon expiration or termination of this Agreement; provided, however, Customer shall continue to pay Company Customer's share of the RPS Compliance Costs following the expiration or termination of the Term for so long as Company continues to incur such RPS Compliance Costs, which share shall be allocated in a commercially reasonable manner consistent with the applicable cost-of-service formulas set forth in Article 4 and Appendices B and C; and provided further, that Customer shall receive a credit for its share, if any, of income received by Company from the sale of Renewable Benefits to third parties. Customer's share of the RPS Compliance Costs shall exclude RPS Compliance Costs attributable to any renewable energy power purchase agreements entered into by Company following the expiration or termination of the Term and any renewable energy generating resources acquired or constructed by Company following the expiration or termination of the Term. In calculating Customer's share of RPS Compliance Costs post-termination or post-expiration, that portion of Company's renewable generation that was previously allocated to Customer shall be reduced by the amount of such generation that the FPSC recognizes as being used to satisfy Company's own RPS Requirements, and with respect to which, the FPSC permits Company to recover the costs (including ROE) associated with such generation through the retail rates payable by Company's Customers. Company shall use commercially reasonable efforts to obtain such FPSC recognition. Following the expiration or termination of the Term, to the extent Company is not subject to a continuing RPS Requirement with respect to that portion of the renewable energy generating resources acquired or constructed by Company to serve Retail Load during the Term, Company shall use commercially reasonable efforts to transfer to Customer, as permitted by Law. its allocable share of Renewable Benefits associated with its allocable share of such renewable energy generating resources referenced in this sentence, which share allocation shall be only to the extent and for the period of Company's continuing RPS Compliance Costs obligations and shall take into account any reductions resulting from FPSC recognition of such renewable energy generating resources as described above in this subsection (b). In addition, to the extent that the RPS Requirement applies to Customer as well as Company such that Customer is required to make a separate demonstration of compliance, Company and Customer shall use commercially reasonable efforts to transfer to Customer its share of any Renewable Benefits such that Company and Customer are both able to mutually satisfy their respective renewable portfolio standard obligations, which share allocation shall be only to the extent and for the period of Company's continuing RPS Compliance Costs obligations and shall take into account any reductions resulting from FPSC recognition of such renewable energy generating resources as described above in this subsection (b); provided, however, that Company (following any such transfer) is released of any obligation to make a separate demonstration of compliance for Customer's share of Company's generation or load (whether due to the generation required to support Retail Load or otherwise). Upon sixty (60) days prior written notice to Customer, Company may elect to cease providing Customer with such Renewable Benefits post-termination or post-expiration of the Term and, in the event Company makes such election, Customer's payment obligations pursuant to this Section 3.6(b) shall cease. For so long as Customer shall have an obligation to pay Company pursuant to this Section 3.6(b), the terms of this Agreement.

1.

to the extent applicable, shall continue to apply to the obligations of the Parties under this Section 3.6(b); provided, however, Company shall not have any obligation to provide Full Requirements Electric Service or any other level or amount of electric service after the expiration or termination of this Agreement.

(c) In no event shall Company be obligated to meet any renewable portfolio standards imposed upon or adopted by Customer that are different from those standards that are imposed by Law upon or adopted by Company. Company may at its option decline to meet any incremental renewable portfolio standards imposed upon Customer following the delivery by Company of a termination notice pursuant to Section 2.2(b) of this Agreement; provided, however, that in such event, Customer shall have the rights set forth in Section 3.6(d), below solely with respect to such incremental renewable portfolio standards that have not been satisfied by Company as of such date.

(d) If Company determines, in its sole discretion (at any time during or after the Term), that it cannot accommodate Customer's renewable portfolio standard requirements due to (i) any inability to recover costs (including without limitation ROE) associated with Customer's share of Company's RPS Requirements, or (ii) any other legal or regulatory reasons, then Company and Customer shall meet to negotiate a basis upon which Customer may pursue its own renewable portfolio standards up to an amount of renewable energy and/or Renewable Benefits equivalent to the percentage of renewable energy and/or Renewable Benefits that Customer would have received credit for had it been part of the Company's RPS Requirement. Notwithstanding anything in this Agreement to the contrary and unless otherwise agreed in writing by Company, in the event Customer pursues its own renewable energy and/or Renewable Benefits pursuant to the foregoing sentence, the calculation of the Generation Demand Charges that Customer is obligated to pay Company shall continue to be computed as though Company were serving that portion of the Retail Load displaced by Customer's receipt of electrical output from any renewable energy resource (which calculation shall be performed in the same manner as provided in Section 3.4, above) until such time that the FPSC, in a proceeding, recognizes Company's generation that has been displaced as a result of Customer pursuing its own renewable portfolio standard as being needed and serving to satisfy Company's system wide generation requirements for the benefit of Company's Customers and permits Company to recover the costs (including ROE) associated with such generation through the retail rates payable by Company's Customers.

## 3.7 Load Management.

(a) For the year 2006, during the Summer Months, and coincident with the peak hour of Customer's Summer Months, Customer had 33.50 MW of Load Management capability as compared to 782.70 MW of peak load for the Summer Months. Also, for the year 2006, during the Winter Months, and coincident with the peak hour of Customer's Winter Months, Customer had 44.10 MW of Load Management capability as compared to 899.13 MW of peak load for the Winter Months. The effects of any Customer Load Management activated during the peak hour of Company's Summer and Winter Months are removed in determining the above values of Customer peak load

coincident with the peak hour of Company's Summer and Winter Months during 2006. During the year 2006, the ratio of summer Load Management capability to summer peak load was 4.28% ("SLM Ratio") and the ratio of winter Load Management capability to winter peak load was 4.91% ("WLM Ratio"). The SLM Ratio and WLM Ratio shall remain fixed for the Term of this Agreement. Customer and Company agree that during the Term of this Agreement, Customer Adjusted Load Management Capability (as defined below) for each month of the applicable Calendar Year shall be used to calculate the adjustments set forth in subparagraph (b). For purposes of this Agreement, the term "Customer Adjusted Load Management Capability" shall be an amount equal to the lower of (x) or (y), where (x) is, for the Summer Months, the SLM Ratio times Customer's actual peak load coincident with the peak hour of Company's Summer Months with the effect of any Customer Load Management activated removed; and where (x) is, for the Winter Months, the WLM Ratio times Customer's actual peak load coincident with the peak hour of Company's Winter Months with the effect of any Customer Load Management activated removed; and where (y) is Customer's actual Load Management capability for the Summer Months or Winter Months, as applicable, depending on whether such month is a Summer Month or a Winter Month, as established during a verification test as set forth in Section 3.7(g), below, or pursuant to an actual request by Company to implement all of Customer's Load Management capability during such month taking into account actual load conditions at the time of implementation.

Customer and Company agree that, in determining the Load Management impact (b) on the Generation Demand Charge Rate in Appendix B, the following adjustments shall be made to the sum of the twelve (12) monthly coincident peaks of Company and Customer. First, the sum of the Customer monthly coincident peaks for the applicable Calendar Year shall be adjusted by the sum of the monthly Customer Adjusted Load Management Capability for the applicable Calendar Year. Second, the sum of Company's monthly coincident peaks (which includes the sum of Customer's monthly coincident peaks adjusted for losses) for the applicable Calendar Year shall be adjusted by the sum of the monthly Customer Adjusted Load Management Capability for the applicable Calendar Year and the sum of the monthly Company Load Management capability for the applicable calendar year (which does not include Customer's Adjusted Load Management Capability). For clarification, in calculating the impact of the Customer Adjusted Load Management Capability on the Generation Demand Charge Rate for the applicable Calendar Year, the value of Customer Adjusted Load Management Capability used for each month shall be the value that was actually in effect for that month.

For example, if in 2014, with no Load Management capability activation (i) the sum of Customer's monthly coincident peaks adjusted for losses totaled 2400 MW; (ii) Customer's Adjusted Load Management Capability was 7MW for the Summer Months and 12MW for the Winter Months; (iii) the sum of Company's monthly coincident peaks (which includes the sum of Customer's monthly coincident peaks adjusted for losses) totaled 240,000 MW; and (iv) Company's Load Management capability for the Summer Months was 1800MW and 2280MW for the Winter Months (which does not include Customer's Adjusted Load Management Capability); then in determining the Generation Demand Charge Rate in Appendix B, schedule A-1, the value for Customer monthly coincident peaks would be 2291MW (2400 MW - 7x7 MW - 5x12 MW). The value for Company (total of 12 monthly coincident peak firm MWs @ generator) would be 215,891 MW (240,000 MW - 7x7 MW - 5x12 MW - 7x1800 MW - 5x2280 MW).

(c) Customer and Company agree that during the Delivery Period Company shall have exclusive use of all Customer's Load Management capability, and if Company and/or Customer has not installed equipment with the ability to implement remote electronic Load Management as desired, then Customer will implement Load Management as requested by Company within those time-frames described in Section 3.7(d) below. During the Delivery Period, Customer will not implement or utilize any Customer Load Management capability at any time unless requested by Company or in the case of an emergency as mutually agreed. During the Delivery Period, Company agrees to utilize Customer Load Management with respect to frequency, duration and devices deployed, consistent with how Company utilizes Company's Load Management.

(d) Customer agrees to activate all load reduction that can be activated remotely by Customer electronically ("Direct Load Management") within five (5) minutes of Company's electronic request and to have the Direct Load Management reduction ramped in at the applicable Delivery Points within ten (10) minutes following request by Company, and to maintain such load reduction for the full duration requested by Company.

(e) Customer agrees to provide load reduction available from Customer's Members from whom Customer has a contractual right to request a reduction of load ("Scheduled Load Management") at the Delivery Points within one (1) hour of Company's electronic request and to maintain such load reduction for the full duration requested by Company.

(f) Customer agrees to be responsible for all costs incurred to implement Load Management on Customer's side of the Delivery Points.

(g) Customer agrees that Company may conduct verifications tests, in accordance with Appendix F, to ascertain the amount of actual Load Management capability to be used in determining the Customer Adjusted Load Management Capability in Section 3.7(a), above. Such verification tests shall be conducted by Company and shall take into account load conditions at the time of the test. Company may schedule one test during the Summer Months and one test during the Winter Months at Customer's expense each Calendar Year for continuing verification of said capability. Company shall use commercially reasonable efforts to minimize the costs of verification tests. Company shall provide Customer at least ten (10) Business Days notice of the proposed date and time for any test. Customer shall, within five (5) Business Days of receiving notice, advise Company whether it agrees to the proposed date and time for the test or, if not, Customer shall agree to a date and time (Monday through Friday, excluding holidays, between 7:00 A.M. and 6 P.M.) within five (5) Business Days of Company's proposed test date.

(h) The Parties recognize that local weather conditions will impact Customer's Load Management capability. Customer will be notified of the results of a verification test or a

Load Management request and Customer may elect to have a confirmatory test performed by Company, at Customer's expense, within thirty (30) calendar days of such notification in order to demonstrate its Load Management capability. Any such test will be conducted and scheduled consistent with Section 3.7(g) and the results of such test shall be used to calculate the Customer Adjusted Load Management Capability. Company shall use commercially reasonable efforts to minimize the costs of such confirmatory tests.

**3.8 Billing Information.** To the extent Company, pursuant to its tariff, offers billing rate information to Company's Customers covering a shorter time interval than as is provided as of the Effective Date of this Agreement, then Company shall, at the request of Customer, provide Customer with billing rate information on a similar basis, but Customer shall be responsible for any costs associated with providing such information to Customer.

3.9 Environmental. Except with respect to its share of any renewable energy credits pursuant to Section 3.6, Customer shall acquire no rights, title or interest in any emission allowances or other environmental credits that may be issued, under the Clean Air Act or otherwise, and whether pursuant to state or federal law, as a result of purchases of Full Requirements Electric Service from Company under this Agreement; provided, however, in its Monthly Charges for Full Requirements Electric Service, Customer shall receive a credit for its share, if any, of income received by Company and bear a share of the burden, if any, of any costs incurred by Company with respect to emission allowances or other environmental credits as allocated pursuant to the cost-of-service formulas set forth in Article 4, Appendix B and Appendix C. Customer shall also be responsible for and pay its share of any other costs, charges or expenses incurred by Company with respect to applicable environmental compliance and emissions standards, regardless of whether such costs, charges or expenses are incurred in the form of a fee, levy, license charge, permit charge, penalty, Tax, carbon tax, allowance or any other cost, charge or expense or similar assessment. The allocation to Customer shall be pursuant to the cost-of-service formulas set forth in Article 4, Appendix B and Appendix C.

**3.10** Merchant Generating Facility. Upon a request to Customer to interconnect to Customer's electric system from a generating facility owned and/or operated by an entity whose retail rates are not directly or indirectly subject to the jurisdiction of the FPSC, Customer shall notify Company of the proposed connection of such new generating facility to Customer's system. Notification shall be in writing as soon as practical. Any such connection shall not reduce, affect or otherwise excuse Customer's obligations under this Agreement, including without limitation its Full Requirements Electric Service obligations, or otherwise reduce or affect Customer's Billing Energy, Billing Demand or CP Demand.

**3.11** Monthly Information Obligation. Customer shall provide Company with monthly detailed information on an hourly basis related to available capacity and energy produced from Qualifying Facilities, renewable energy and/or distributed generation sources in a manner which will enable the Company to make adjustments, if any, to the Generation Demand Charges as specified in Section 4.8(a) and 4.8(c). Customer further

agrees, at the request of Company, to provide Company with access to its books and records and to provide such supporting documentation as reasonably requested by Company in order to verify generation amounts from such sources.

# ARTICLE 4 MONTHLY CHARGES AND BILLING

General Principles Regarding Monthly Charges For Full Requirements 4.1 Electric Service. The Monthly Charges for Full Requirements Electric Service supplied to Customer by Company hereunder during the Delivery Period of this Agreement shall include a Generation Demand Charge, a Generation Energy Charge (collectively, the "Generation Demand and Generation Energy Charges"), Customer Charge and a Fuel Charge, all calculated in accordance with the terms of this Article, together with other charges set forth in this Agreement and subject to all other terms and conditions contained in this Agreement. For each Calendar Year, the Generation Demand and Generation Energy Charges for which Customer is responsible shall be determined using data from Company's FERC Form No. 1 for that Calendar Year, the costs as identified in the cost-of-service formulas set forth in Appendix B hereto and the ROE as specified in Section 4.7(b). Because of the time lag in availability of the FERC Form No. 1 data, Generation Demand and Generation Energy Charges will initially be assessed based on estimated monthly charges, and a "true-up" will subsequently be conducted for each Calendar Year, as provided in Section 4.4 hereof. The Generation Energy Charge shall exclude costs recovered through the Fuel Charge and the Fuel Adjustment Charge that are included in Appendix C attached hereto. All fuel-related costs shall be recovered through the Fuel Charge, which shall be "trued-up" to more current actual fuel-related costs by means of the Fuel Adjustment Charge set forth in Section 4.5 hereof. The Generation Demand Charge Rate used to calculate the "Generation Demand Charge" and the Generation Energy Charge Rate used to calculate the "Generation Energy Charge" shall be determined pursuant to the cost-of-service formulas set forth in Appendix B hereto. The monthly charges used to calculate the "Fuel Charge" and the "Fuel Adjustment Charge" shall be determined pursuant to the formulas set forth in Appendix C hereto. These formulas will be used for calculation of both estimated and actual ("trued-up") charges. In no event shall the costs for Generation Related Services be recovered twice from Customer through inclusion both in the costs used to derive the Generation Demand Charge or the Generation Energy Charge in Appendix B and also in the payments by Customer to the Transmission Provider for Ancillary Services.

**4.2** Customer Charge. Customer shall pay a monthly charge of \$2,500.00 per Delivery Point ("Customer Charge").

4.3 Estimated Generation Demand Charge and Generation Energy Charge. The estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate shall be developed for each Calendar Year. Each Calendar Year the estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate used to calculate the Generation Demand Charge and Generation Energy Charge applicable for the months of June through December of the current Calendar Year and the months of January through May of the following Calendar Year shall be determined annually by Company, pursuant to Appendix B, prior to the 31st day of May, and be provided to

Customer by June 1st for its review. The estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate in effect during each Calendar Year shall be based upon the results for the most recent Calendar Year for which FERC Form No. 1 Customer shall have the rights set forth in Section 4.6 of this data is available. Agreement regarding the calculation of the estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate. For example, in Calendar Year 2014, for the first five months (January through May) the estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate used to bill Customer will be based upon Calendar Year 2012 FERC Form No. 1 data, which estimate shall have been provided to Customer by June 1, 2013; beginning on June 1, 2014, the estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate for June through December of Calendar Year 2014 and January through May of Calendar Year 2015 shall be based upon Calendar Year 2013 FERC Form No. 1 data and provided to Customer by June 1, 2014 (and will remain in effect until June 1, 2015); and all the Generation Demand and Generation Energy Charges based on these estimated charges for Calendar Year 2014 will be subject to "true-up" of the Generation Demand Charge and Generation Energy Charge in 2015, pursuant to Section 4.4. Customer agrees, commencing in Calendar Year 2012 and continuing for the Term of this Agreement, to provide Company with good faith estimates of the sum of its system monthly maximum non-coincident peaks and the amount of Customer Adjusted Load Management Capability as provided in Section 3.7, and good faith estimates of its expected purchases or electric output received from the following: Qualifying Facilities as provided in Section 3.3(b), renewable energy resources as provided in Section 3.4 and/or distributed generation resources as provided in Section 3.5. Such amounts shall be in total and at each Delivery Point beginning in 2012 and shall be used in preparing the estimated Generation Demand Charge Rate and the billing determinants for such rate, including the estimated Billing Demand and CP Demand. Such good faith estimates shall be based on the prior year's actuals for such amounts. Such good faith estimates shall be trued-up based on actuals pursuant to Section 4.4.

4.4 True-Up of Generation Demand and Generation Energy Charges. Because the Generation Demand and Generation Energy Charges initially collected during a Calendar Year will be the product of estimated charges based on data from previous Calendar Years as provided in Section 4.3, above, the Generation Demand and Generation Energy Charges shall be adjusted (or "trued-up") based upon (i) FERC Form No. 1 cost data from the Calendar Year in which the service was provided, (ii) the actual costs or other amounts as identified in the cost-of-service formulas set forth in Appendix B hereto and (iii) the Return on Common Equity specified in Section 4.7(b). The true-up for Generation Demand and Generation Energy Charges shall be conducted by Company and provided to Customer by June 30th of each Calendar Year once the FERC Form No. 1 data for the prior Calendar Year is available, and shall be accomplished by recalculating the sums due to Company from Customer for service provided during the prior Calendar Year. The difference between the sum of the Generation Demand and Generation Energy Charges originally billed to Customer for service provided during the Calendar Year. based on the estimated Generation Demand Charge Rate and the estimated Generation Energy Charge Rate, and the sum of the Generation Demand and Generation Energy Charges for which Customer is ultimately responsible shall be billed to Customer or

credited to Customer, as appropriate, in twelve (12) equal monthly amounts beginning with the month of July of the year immediately following the Calendar Year for which the "true-up" is calculated, unless otherwise agreed by the Parties. The amount to be billed or credited for any such over-collections or under-collections will include interest determined in accordance with Section 35.19a of FERC's regulations and charged or applied from the date of original payment to the date when the over-collections or undercollections are paid or credited, but excluding the date paid or credited. Customer may unilaterally elect to prepay any amounts owed to Company, and Company may unilaterally elect to credit, in advance, any amounts due to Customer, and any such prepayments or advance credits shall be reflected in any interest calculations hereunder. Customer shall have the rights set forth in Section 4.6 of this Agreement regarding the calculation of the "trued-up" Monthly Charges. Trued-up Generation Demand and Generation Energy Charges shall always be based on Company's actual costs.

For example, the "true-up" of Generation Demand and Generation Energy Charges initially collected for service provided during Calendar Year 2014 shall be conducted by Company by June 30, 2015. The "trued-up" charges shall be calculated using (i) 2014 FERC Form No. 1 cost data and (ii) the ROE, as specified in Section 4.7(b). The difference between the sum of the Generation Demand and Generation Energy Charges originally billed to Customer for service provided during 2014 and the sum of the "truedup" Generation Demand and Generation Energy Charges for Calendar Year 2014 will be divided into twelve (12) equal amounts and billed to Customer or credited to Customer, as appropriate and with interest, in invoices sent to Customer from July 2015 through June 2016, covering services provided during January 2014 through December 2014, unless otherwise agreed by the Parties or except to the extent a full prepayment is made by Customer or a full advance credit is refunded by Company.

Fuel Charge and Fuel Adjustment Charge. Customer shall pay Company a 4.5 monthly Fuel Charge and Fuel Adjustment Charge based upon the fuel factors and the fuel adjustment factors determined pursuant to Appendix C. The "Fuel Charge" for each month shall be an amount equal to (i) the product of the estimated On-peak Fuel Charge Factor determined pursuant to Appendix C for the applicable month and the Billing Energy for the applicable month, plus (ii) the product of the estimated Off-peak Fuel Charge Factor determined pursuant to Appendix C for the applicable month and the Billing Energy for the applicable month. The "Fuel Adjustment Charge" shall be an amount equal to (i) the product of the actual On-peak Fuel Adjustment Charge Factor determined pursuant to Appendix C for the applicable month that the Fuel Charge is being trued-up and the Billing Energy for the applicable month that the Fuel Charge is being trued-up, plus (ii) the product of the actual Off-peak Fuel Adjustment Charge Factor determined pursuant to Appendix C for the applicable month that the Fuel Charge is being trued-up and the Billing Energy for the applicable month that the Fuel Charge is being trued-up. The fuel factors used to bill the Fuel Charges shall be projected by November 30th of every year for the following Calendar Year. The Fuel Charge shall then be subject to true-up through the Fuel Adjustment Charge. Customer shall be billed an estimated Fuel Charge each month for the energy delivered in the preceding month. Any difference between the estimated Fuel Charges and the Fuel Charges based on actual fuel costs shall be billed or credited to Customer through the Fuel Adjustment Charge on

the first bill rendered after such actual fuel costs have been determined. If the Fuel Adjustment Charge is positive, such amount shall be billed to Customer and if the Fuel Adjustment Charge is negative, such amount shall be credited to Customer. The amount to be billed or credited for any over-collections or under-collections based on such estimates versus actual costs shall include interest accrued at the average of the Prime Rate as published in the Wall Street Journal for the last business day of the current and prior month and charged or applied to the average of the beginning and ending true-up balance for the month. Company shall use reasonable diligence when estimating monthly fuel charges to Customer. Fuel Adjustment Charges shall always be based on Company's actual costs for fuel and purchased power.

## 4.6 Review Process.

(a) The following information shall be provided to Customer by Company on or before June 30th: (i) Company's FERC Form No. 1 Report for the Calendar Year in which the service was provided; (ii) the calculation of the trued-up Generation Demand Charge and the trued-up Generation Energy Charge; and (iii) work papers showing the source of all data utilized in the calculation(s) and any other supporting documentation. The accuracy of Company's calculations, as well as the data used in those calculations (including FERC Form No. 1 data), shall be subject to review and adjustment only in accordance with the following procedure:

(i) Customer shall have until September 30 of the same Calendar Year to review the information provided by Company. During that time Customer may ask Company questions, but such questions shall be limited to determining if Company properly applied the cost-of-service formulas in Appendix B; if the data used in Appendix B was accurate; and if Company's calculations were consistent with this Agreement. Company shall respond to such questions within twenty (20) Business Days. If the FERC Form No. 1 data or other Company data used in the calculations is changed, Company shall make the appropriate adjustment.

(ii) If Company and Customer do not resolve any dispute as to the accuracy of the data used by Company or the application of the cost-of-service formulas in Appendix B by September 30 of the same Calendar Year, Customer may file a complaint at FERC within thirty (30) calendar days regarding the accuracy of Company's calculations or the data used in those calculations (including FERC Form No. 1 data), or both. The Parties agree that the complaint proceeding will be limited to determining if Company properly applied the cost-of-service formulas in Appendix B, if the data used in Appendix B was accurate, and if Company's calculations were consistent with this Agreement. Company agrees to bear the burden of proof regarding these matters in any such complaint proceeding. Any refund obligation will extend for the entire Calendar Year that is the subject of review as a result of the complaint proceeding, with Company having no right to seek suspension of the refund effective date.

(iii) The Generation Demand and Generation Energy Charges being collected from Customer by Company under this Agreement for a Calendar Year shall become final and not subject to adjustment on the latest of (i) September 30 of the same Calendar Year in which the FERC Form No. 1 Report for the subject Calendar Year is provided during the Term of this Agreement, if at such time, there has been no complaint filed at FERC under this Section 4.6(a); (ii) the final resolution of any complaint filed pursuant to this Section 4.6(a); or (iii) the day any required corrections have been made by Company. Company shall make any necessary corrections required pursuant to this Section 4.6 as soon as possible and shall make any adjustments to Customer's Monthly Bill as provided in Section 4.4.

(b) The following information shall be provided to Customer by Company on or before December 1 of each Calendar Year: (i) the calculation of the fuel factors used to determine the monthly Fuel Charge for the following Calendar Year; and (ii) work papers showing the source of all data utilized in the calculation of the fuel factors used to determine the monthly Fuel Charge for the following Calendar Year and any other supporting documentation. The accuracy of Company's calculations, as well as the data used in those calculations, shall be subject to review and adjustment only in accordance with the following procedure:

(i) Customer shall have until December 31 of the same Calendar Year in which the information described in the above paragraph is provided to Customer to review the information provided by Company. During that time Customer may ask Company questions, but such questions shall be limited to determining if Company properly applied the formula in Appendix C, if the data used in Appendix C was accurate, and if Company's calculations were consistent with this Agreement. Company shall respond to such questions within twenty (20) Business Days.

(ii) If Company and Customer do not resolve any dispute as to the accuracy of the data used by Company or the application of the formula in Appendix C by February 28 of the succeeding Calendar Year, Customer may file a complaint at FERC within thirty (30) calendar days regarding the accuracy of Company's calculations or the data used in those calculations, or both. The Parties agree that the complaint proceeding will be limited to determining if Company properly applied the formula in Appendix C, if the data used in Appendix C was accurate, and if Company's calculations were consistent with this Agreement. Company agrees to bear the burden of proof regarding these matters in any such complaint proceeding. Any refund obligation will extend for the entire Calendar Year that is the subject of review as a result of the complaint proceeding, with Company having no right to seek suspension of the refund effective date.

(iii) The Fuel Charges being collected from Customer by Company under this Agreement shall become final and not subject to adjustment on the latest of (i) February 28 of each succeeding Calendar Year during the Term of this Agreement (as described in (ii) above), if at such time, there has been no complaint filed at FERC under this Section 4.6(b); (ii) the final resolution of any complaint filed pursuant to this Section 4.6(b); or (iii) the day any required corrections have been made by Company. Company shall make any necessary corrections as soon as possible and shall make any adjustments to Customer's bill on the next monthly billing or as soon as commercially reasonable. (c) Each month following the determination of the actual fuel costs for such month by Company for use in Fuel Adjustment Charge billings to Customer, the following information shall be provided to Customer by Company on or before the last day of the second month following the month for which the Fuel Adjustment Charge applies: (i) the calculation of the fuel factors used to determine the monthly Fuel Adjustment Charge for the applicable month; and (ii) work papers showing the source of all data utilized in the calculation of the monthly Fuel Adjustment Charge for the applicable month and any other supporting documentation. The accuracy of Company's calculations, as well as the data used in those calculations, shall be subject to review and adjustment only in accordance with the following procedure:

(i) Customer shall have until the end of the fourth month following the month for which the Fuel Adjustment Charge applies to review the information provided by Company. During that time Customer may ask Company questions, but such questions shall be limited to determining if Company properly applied the formulas in Appendix C, if the data used in Appendix C was accurate, and if Company's calculations were consistent with this Agreement. Company shall respond to such questions within twenty (20) Business Days.

(ii) If Company and Customer do not resolve any dispute as to the accuracy of the data used by Company or the application of the formula in Appendix C by the fourth month following the month for which the Fuel Adjustment Charge applies, Customer may file a complaint at FERC within thirty (30) calendar days regarding the accuracy of Company's calculations or the data used in those calculations, or both. The Parties agree that the complaint proceeding will be limited to determining if Company properly applied the formula in Appendix C, if the data used in Appendix C was accurate, and if Company's calculations were consistent with this Agreement. Company agrees to bear the burden of proof regarding these matters in any such complaint proceeding. Any refund obligation will extend for the entire month that is the subject of review as a result of the complaint proceeding, with Company having no right to seek suspension of the refund effective date.

(iii) The Fuel Adjustment Charges being collected from Customer by Company under this Agreement shall become final and not subject to adjustment on the latest of (i) the fourth month following the month for which the Fuel Adjustment Charge applies, if at such time, there has been no complaint filed at FERC under this Section 4.6(b); (ii) the final resolution of any complaint filed pursuant to this Section 4.6(b); or (iii) the day any required corrections have been made by Company. Company shall make any necessary corrections as soon as possible and shall make any adjustments to Customer's bill on the next monthly billing or as soon as commercially reasonable.

(d) Except as otherwise specifically provided in Section 4.6 of this Agreement, Customer shall not oppose, directly or indirectly, Company or its positions at FERC with regard to the formula itself or charges assessed by Company for Full Requirements Electric Services under this Agreement and hereby waives its right to do so. 4.7 Cost-of-Service Formulas. The cost-of-service formulas set forth in Appendix B were designed, by mutual agreement, primarily to use Company's FERC Form No. 1 costs. Certain costs that either are not reflected in Company's FERC Form No. 1, or that the Parties agreed were not representative of Company's costs, were determined by negotiation. The specific elements of the cost-of-service formulas set forth below were negotiated as an integrated and integral part of this Agreement:

(a) For purposes of calculating Company's charges used in Appendix B, the Parties agree that Company will include one hundred percent (100%) of its expenditures for Pollution Control Facilities and Fuel Conversion Facilities (as defined in Section 35.25 of the FERC's regulations) recorded on Company's books and records as construction work in progress ("CWIP"), and fifty percent (50%) of its expenditures for all other CWIP.

(b) The Return on Common Equity ("ROE") is 11.75%.

Company's total production costs as shown in Appendix B shall be reduced for a (c) portion of its gains on non-separated wholesale power sales, excluding emergency sales. Non-separated wholesale power sales shall refer to wholesale power sales that are shortterm (less than one year in duration) and/or non-firm in nature. Consistent with the rate treatment applied to Company's Customers, an incentive shall apply to the gains on all non-separated wholesale power sales, firm and non-firm, excluding emergency sales. A three year moving average of gains on all non-separated wholesale power sales, firm and non-firm, excluding emergency sales, shall be established each year as the threshold for application of the incentive. All gains below this threshold shall be credited as a reduction to Company's total production costs as shown on Appendix B. Eighty percent (80%) of the gains above this threshold shall also be shown as a reduction to Company's total production costs as shown on Appendix B. Twenty percent (20%) of the gains above this threshold shall be retained by Company's shareholders. To the extent that the FPSC approves a different formula for sharing the gains on non-separated sales with Company's Customers, Company reserves the right to file at FERC to modify provisions of this paragraph, consistent with the FPSC approved methodology, and Customer agrees to support such filing. To the extent that the FPSC approves a different formula for sharing the gains on non-separated sales with Company's Customers which would reduce the gains to be retained by Company, Company shall notify Customer and shall make such filing at FERC if requested by Customer. Company shall apply all allocations associated with the above referenced formula consistently across all Company's total load requirements and Customer shall only receive its share of these benefits.

(d) The method set forth in Appendix B to allocate costs as production-related and/or transmission-related shall be used throughout the Term of this Agreement; provided, however, that to the extent that any of the following circumstances occur during the Term of this Agreement, the Parties will meet and attempt to determine a mutually agreeable modification to Appendix B: (i) FERC determines in a rulemaking proceeding, or in a proceeding regarding Company's costs to be included in the Transmission Provider's OATT, or it is agreed in a FERC-approved settlement of any such proceeding, that items currently included in Appendix B as production-related should be treated as transmission-related; or (ii) FERC determines in a rulemaking proceeding, or in a proceeding regarding

Company's costs to be included in the Transmission Provider's OATT, or it is agreed in a FERC-approved settlement of any such proceeding, that items currently excluded from Appendix B as transmission-related should be treated as production-related. If either or both of the circumstances in this Section 4.7(d) occurs and the Parties cannot agree upon a modification to Appendix B, then either Party may seek FERC review of Appendix B's treatment of the items addressed in such proceedings under the "just and reasonable" standard of Sections 205 and 206 of the FPA.

### 4.8 Billing Demand and Energy.

The monthly "Billing Demand" shall be an amount equal to the following for the (a) applicable Billing Period: (1) the sum of the single highest 60-minute kW demand at all of the Delivery Points during the same 60-minute period; plus (2) any purchases made or electrical output received (expressed in kWs) at or on Customer's side of the Delivery Points in that 60-minute period from Qualifying Facilities as provided in Section 3.3, renewable energy resources as provided in Section 3.4 and distributed generation resources as provided in Section 3.5; and (3) grossed-up for Losses (expressed in kWs) at and from the Receipt Points to the Delivery Points based on the demand loss factors for all transmission delivery points on Company's system for the applicable period as set forth in Appendix B. In determining the monthly "Billing Demand", the purchases described in Section 4.8 (a)(2), above, shall be reduced as follows: (i) by up to 5 MW for the first five (5) MW, in the aggregate, that are purchased or received by Customer from all Qualifying Facilities, (ii) by up to two (2) MW for the first two (2) MW, in the aggregate, that are purchased or received by Customer from all renewable energy resources, (iii) by up to two (2) MW for the first two (2) MW, in the aggregate, that are purchased or received from all distributed generation resources. Examples regarding the calculation of monthly Billing Demand are set forth in Appendix K.

(b) The monthly "Billing Energy" shall be the total kWhs of Full Requirements Electric Service, as measured during the applicable Billing Period, at all of the Delivery Points and grossed-up for Losses (expressed in kWs) at and from the Receipt Points to the Delivery Points based on the energy loss factors for all transmission delivery points on the Company's system for the applicable period as set forth in Appendix B.

(c) The monthly "CP Demand" shall be the total 60-minute kW demand measured at the time of Company's highest hourly system demand during the Billing Period: (1) at all of the Delivery Points; plus (2) any purchases made or electrical output received (expressed in kWs) in that 60-minute period at or on Customer's side of the Delivery Points from Qualifying Facilities as provided in Section 3.3, renewable energy resources as provided in Section 3.4 and distributed generation resources pursuant to Section 3.5; and (3) grossed-up for Losses (expressed in kWs) at and from the Receipt Points to the Delivery Points based on the demand loss factor for all transmission delivery points on the Company's system for the applicable period as set forth in Appendix B. In determining the monthly "CP Demand", the purchases or amounts received described in Section 4.8(c)(2), above, shall be reduced as follows: (i) by up to 5 MW for the first five (5) MW, in the aggregate, that are purchased or received by Customer from all Qualifying Facilities, (ii) by up to two (2) MW for the first two (2) MW, in the aggregate,

that are purchased or received by Customer from all renewable energy resources and (iii) by up to two (2) MW for the first two (2) MW, in the aggregate, that are purchased or received from all distributed generation resources. In addition, the CP Demand shall be adjusted for Load Management as set forth in Section 3.7(b) and the Calculation of Customer's CP Demand Schedule attached to Appendix B. Examples regarding the calculation of monthly CP Demand are set forth in Appendix K.

(d) The meters at the Delivery Points shall be tested from time to time, and the results shall be reported to Company and Customer. The Transmission Provider's OATT shall govern the process for testing the meters at the Delivery Points. In the event such testing results in a need to correct prior meter readings, such corrections shall be made in accordance with Transmission Provider's OATT and Company's FERC Form No. 1 data shall be appropriately amended to reflect the effects of such corrections.

(e) Customer shall provide by e-mail to Company on the second Business day of the month following the month of delivery all electronic metering information relating to electrical output received during the applicable Billing Period by Customer at and on Customer's side of the Delivery Points. Such metering information shall be sufficient to enable Company to calculate the adjustments, if any, to the monthly Billing Demand and monthly CP Demand resulting from the receipt by Customer of such electrical output on Customer's side of the Delivery Points from Qualifying Facilities as provided in Section 3.3, renewable energy resources as provided in Section 3.4 and distributed generation resources as provided in Section 3.5.

# 4.9 Determination of Monthly Bill.

The Monthly Bill shall set forth charges for Full Requirements Electric Service, charges to reimburse costs incurred by Company on behalf of Customer in connection with Full Requirements Electric Service, and other charges associated with Full Requirements Electric Service. For purposes of this Agreement, the "Monthly Bill" shall be the total sum of the following:

- (a) The Generation Demand Charge;
- (b) The Generation Energy Charge;
- (c) The Fuel Charge;
- (d) The Fuel Adjustment Charge;
- (e) True-up of estimated versus actual charges as described in Section 4.4;
- (f) Customer Charge;

(g) All amounts due to Company for reimbursement of charges incurred by Company for delivery of Full Requirements Electric Service in accordance with the OATT or related agreements for transmission-related schedules or services for which Customer is responsible pursuant to this Agreement, to the extent that such charges are not included in Items (a) through (l);

(h) Amounts due as a result of a Loss of Members pursuant to Section 16.5;

(i) Any applicable governmental Taxes, fees and assessments attributable or related to Full Requirements Electric Service to the extent not included in items (a) through (l);

(j) Environmental and/or emissions charges, costs and expenses as further described in Section 3.9 of this Agreement to the extent not included in items (a) through (l);

(k) Customer's share of RPS Compliance Costs as further described in Section 3.6 to the extent not included in items (a) through (l); and

(1) Any other amounts payable or due to Company pursuant to this Agreement.

4.10 Payment Date. Customer shall pay Company any amounts due and payable hereunder on or before the later of the fifteenth (15th) calendar day of each month following the period of delivery of Full Requirements Electric Service, or the tenth (10th) calendar day after receipt of invoice or, if such due date is not a Business Day, then on the next Business Day. All invoices shall be delivered electronically to Customer on a Business Day, and all invoices shall be paid by electronic funds transfer, or by other mutually agreeable methods, to the account designated by Company. If all or any part of any amount due and payable pursuant to this Agreement shall remain unpaid thereafter, interest shall thereafter accrue and be payable to Company on such unpaid amount at a rate determined in accordance with Section 35.19a of FERC's regulations.

Payment in Event of Billing Disputes. If Customer, in good faith, disputes an 4.11 invoice, Customer shall immediately notify Company of the basis for the dispute and pay the undisputed portion of such invoice no later than the due date; provided, however, Customer shall not withhold amounts in connection with any dispute under Section 4.6 (or payments that would be subject to the provisions of Section 4.6); provided, further, however, with respect to any amounts Company passed through from the Transmission Provider to Customer pursuant to item (g) in Section 4.9, above, the full amount of the disputed bill shall be paid when due and any billing dispute shall be handled in accordance with the procedures set forth in the OATT. Upon resolution of the dispute, any required payment shall be made within five (5) Business Days of such resolution along with any interest accrued at the rate determined in accordance with Section 35.19a of the FERC's regulations from and including the due date to, but excluding the date paid. Inadvertent overpayments shall be returned by Company or deducted at the option of Customer, in either case with interest accrued at the rate determined in accordance with Section 35.19a of the FERC's regulations until the date paid or deducted from and including the date of such overpayment to but excluding the date repaid or deducted by Company.

4.12 Retail Competition. In the event the State of Florida enacts a Law providing for the implementation of Retail Competition in the State of Florida, Company and Customer shall have the right to renegotiate the price of Full Requirements Electric Service hereunder as well as other terms hereof. If Company and/or Customer elect(s) to renegotiate this Agreement, the Parties shall renegotiate in good faith to attempt to reach an agreement on the proposed changes to this Agreement. In the event the Parties do not agree on new terms hereunder that restore the economic and other benefits reasonably anticipated by Company and Customer under this Agreement as of the date prior to Retail Competition within a commercially reasonable period of time after the effective date of such Law (but not to extend beyond the commencement date of Retail Competition in the State of Florida), Company or Customer shall provide fifteen (15) days notice to the other Party that negotiations of a modified cost of capacity and energy are terminated. Upon termination of such negotiations, this Agreement will continue under the then current charges and the Term of this Agreement shall be automatically modified to extend until the date such Retail Competition commences in the State of Florida. Company has the right to file with FERC for changes to this Agreement pursuant to Section 205 of the Federal Power Act to incorporate any changes to the Full Requirements Electric Service pricing and other terms and conditions hereunder that are agreed to by both Parties and to incorporate the change in Term, if applicable. In addition to any other rights specified in this Section 4.12, Customer agrees that Company shall have the right to recover its stranded costs in accordance with Section 35.26 of FERC's regulations (or any successor provision) in the event Retail Competition is implemented in Florida and the Term is modified or this Agreement is terminated pursuant to this Section 4.12.

### **ARTICLE 5 CHANGE OF TRANSMISSION**

# 5.1 Nodal Market.

In the event that a nodal or similar market design is implemented that is reviewed (a) or approved by FERC or the applicable governing body, Customer shall be responsible for all charges imposed under such market design between the Receipt Points and the Delivery Points, including, but not limited to congestion and Losses, any locational marginal pricing differentials, locational capacity pricing differences, and incremental losses and any other similar charges, and Company shall maintain the right to allocate the output of Company Generation Resources to meet the requirements of Company. The Receipt Points shall, for such purposes, be the points that most closely resemble the geographic characteristics of the Receipt Points as of the date prior to the implementation of the nodal or similar market design. In such case, Company's transmission costs, including transmission losses and other applicable fees to the Receipt Points shall be added to the charges set forth in Article 4. In the event that a regional transmission organization or similar independent transmission entity is established in Florida, Customer shall be responsible for all additional costs of providing service hereunder resulting from such independent transmission entity and Company shall maintain the right to allocate the output of the Company Generation Resources.

(b) In the event that a nodal or similar market design is implemented, to the extent permitted by applicable Laws, Company shall meet with Customer on an annual basis in order to review the dispatch characteristics of the Company Generation Resources.

### 5.2 Regional Transmission Organization.

(a) If an ISO, RTO, ITC or other future organization agency or authority is formed, created or otherwise implemented in Florida that has been approved by FERC to serve as the Transmission Provider, then Company and Customer shall reasonably cooperate to make or enter into arrangements with such entity to assist such entity with implementation of this Agreement.

(b) Nothing herein shall be construed in any way to relieve Customer of, or impose upon Company, the responsibility for any fees, costs, or charges (including but not limited to congestion costs, transmission losses, or the costs or charges to secure financial transmission rights or the equivalent thereof) that may be imposed on Customer by an ISO, RTO, ITC or other future organization, agency or authority that has been approved by FERC to serve as the Transmission Provider. Company shall have no right or interest in any financial transmission rights or the equivalent thereof that are allocated, assigned, transferred to, or acquired by Customer.

# ARTICLE 6 CREDITWORTHINESS

Financial Information. If requested by a Party ("Party X"), the other Party 6.1 ("Party Y") shall deliver (i) within one hundred fifty (150) days following the end of each fiscal year, a copy of the annual report containing audited consolidated financial statements for such fiscal year with respect to Party Y, and (ii) within sixty (60) days after the end of each of its first three (3) fiscal quarters of each fiscal year, a copy of a quarterly report containing unaudited consolidated financial statements for such fiscal guarter with respect to Party Y, to the extent that such reports are prepared in Party Y's ordinary course of business on such schedule. In all cases the statements shall be for the most recent accounting period and shall be prepared in accordance with generally accepted accounting principles, except for accounting adjustments that the Party customarily makes only at fiscal year end in accordance with generally accepted accounting principles; provided, however, that should any such statements not be available on a timely basis due to a delay in preparation or certification, such delay shall not be an Event of Default for a period of not more than forty-five (45) days so long as during such period Party Y diligently pursues the preparation, certification and delivery of the statements. Company's obligation to deliver financial statements as set forth in this Section 6.1 shall continue only for so long as Company has a legal obligation to prepare and publicly disclose such financial statements. In the event Company no longer has a legal obligation to prepare and publicly disclose such financial statements, Company shall deliver to Customer in satisfaction of the requirements of this Section 6.1, financial statements of its ultimate parent company.

6.2 Financial Covenant Compliance Worksheet. If Customer is preparing financial covenant worksheets for its Lenders or if Customer does not have a credit rating from either S&P or Moody's, Customer shall prepare each month, on a rolling twelve month basis, a Financial Covenant Compliance Worksheet ("Worksheet") that contains the calculations of the covenants set forth in Section 6.3(d), a sample of which is attached hereto as Appendix D. If Customer is preparing financial covenant worksheets for its Lenders or if Customer does not have a credit rating from either S&P or Moody's, Customer shall continue to prepare such Worksheet in the same or similar format and

with substantially the same type of content during and for the Term of this Agreement and shall continue to provide Company on or before the fifteenth (15th) day of each month, but no earlier than five (5) days after the receipt of an invoice by Customer (or, if later, immediately after Customer's Board of Trustees approves the Worksheet if the Board of Trustees does not meet until after the fifteenth (15<sup>th</sup>) day of each month; provided that in no event shall the Worksheet be delivered later than the last calendar day of the applicable month), with a copy of the Worksheet covering the prior month period, certified by its chief financial officer ("CFO") as being true and correct and prepared in accordance with generally accepted accounting principles. The Worksheet for the month of December shall be delivered on or before the last calendar day of January. In addition, Customer shall provide to Company, along with the Worksheet, Customer's monthly financial statements used to prepare such Worksheet. No later than the Effective Date, Customer shall provide Company a copy of its loan documents for secured and unsecured loans with each of its Lenders, certified by its CFO as being true and correct copies thereof. Customer shall provide Company a copy of any new loan documents entered into by Customer during the Term and any amendments, supplements, modifications, restatements or replacements of any loan documents in effect as of the date of this Agreement and new loan documents, which shall be certified by its CFO as being true and correct copies thereof, within fourteen (14) Business Days after execution thereof.

# 6.3 Credit Assurances of Customer.

(a) Customer covenants that it shall maintain, at all times in which it has a Credit Rating, a minimum of an Investment Grade Credit Rating. If Customer does not have a Credit Rating from S&P or Moody's, Customer shall at all times meet or exceed each of the minimum financial covenants that are determined in accordance with Section 6.3(d).

**(b)** Beginning on December 2, 2013, and continuing throughout the Term, Customer's obligations to Company arising under this Agreement shall at all times be secured by a Letter of Credit, or if mutually agreed as to form, structure and content by Customer and Company, by a Dedicated Line of Credit, in each case, in an amount equal to or greater than the highest total monthly bill incurred by Customer under this Agreement over the most recent prior twelve (12) month period (such amount, as the same is required herein to be adjusted from time to time, the "Required Amount"); provided, however, for purposes of this Section, the minimum Required Amount for Calendar Year 2014 shall be \$35 million. In lieu of the Letter of Credit, Company shall agree to Customer securing its obligations to Company under this Agreement by use of a Dedicated Line of Credit based on terms and conditions satisfactory to Company, including those set forth in Appendix G, if, but only if, Company, Customer and Customer's Lender reach a mutual written agreement on all the terms and conditions of such Dedicated Line of Credit; otherwise (and until such time) a Letter of Credit will continue to be required pursuant to the terms and conditions of this Agreement.

(1) If Customer, its Lender and Company reach a mutual written agreement on the form and use of a Dedicated Line of Credit to secure Customer's obligations to Company, such Dedicated Line of Credit shall, among other things, permit Company to make a drawing for the account of Customer of the full amount of the Dedicated Line of

Credit in the event that (a) Customer fails to renew or replace the Dedicated Line of Credit at least thirty (30) calendar days prior to the stated expiration of the Dedicated Line of Credit, (b) if, within the applicable period of grace provided therefor in this Article 6, Customer fails to provide Company with a Letter of Credit or additional Performance Assurance as required pursuant to the provisions of this Article 6 or this Agreement, (c) an Event of Default by Customer has occurred and is continuing, or (d) Company otherwise has the right to draw upon any Performance Assurance of Customer pursuant to the terms of this Agreement. The documents evidencing and securing any Dedicated Line of Credit shall be in form and substance satisfactory to Company, and the Lender providing such Dedicated Line of Credit shall provide to Company a Consent and Agreement in form and substance satisfactory to Company (for the avoidance of doubt, the Consent and Agreement attached as Appendix I is not intended to dictate or limit the form or substance of the Consent and Agreement contemplated by this Section 6.3(b)(1). The Dedicated Line of Credit shall be for the sole and exclusive benefit of Company, and its sole purpose shall be to secure Customer's obligations under this Agreement. Customer shall have no right to draw under the Dedicated Line of Credit, and Company shall have the sole and exclusive right to draw under the Dedicated Line of Credit for the account of Customer, without Customer's consent, by using a power of attorney. Upon notice by Company to Customer from time to time advising Customer of the amount of the highest monthly bill incurred by Customer over the most recent twelve (12) month period, Customer shall adjust the amount of the Dedicated Line of Credit within thirty (30) calendar days to the Required Amount if the available amount of the Dedicated Line of Credit does not then equal or exceed the Required Amount and provide Company evidence of such change. If Company draws upon the Dedicated Line of Credit, Customer shall restore and maintain the amount available for draw under the Dedicated Line of Credit to equal or exceed the Required Amount within five (5) Business Days after any such drawing so that the Required Amount is available at all times. Customer shall cooperate with Company and take such actions as are required for Company to perfect a first priority security interest in the Dedicated Line of Credit ("Security Interest"), including but not limited to entering into a security agreement to establish such Security Interest and to grant Company a power of attorney, each such security agreement and power of attorney to be in form and substance satisfactory to Company. Customer agrees that the Dedicated Line of Credit shall not be subject to termination or amendment without the express prior written consent of Company. All costs of a Dedicated Line of Credit shall be borne by Customer. If a Dedicated Line of Credit is established pursuant to this Section 6.3(b)(1), Company shall have the option, at any time, upon thirty (30) calendar days prior notice, to require Customer to replace the Dedicated Line of Credit with a Letter of Credit equal to the Required Amount. Should Company exercise its option to cause Customer to replace the Dedicated Line of Credit with a Letter of Credit as provided in this Section 6.3(b)(1), Company shall pay one half of the Letter of Credit fees required to post and maintain such a Letter of Credit. The obligation of Company to pay one-half of the Letter of Credit fees shall not apply when a Letter of Credit is posted by Customer pursuant to Section 6.3(b)(2), Performance Assurance is required by Company pursuant to Section 6.4 or Article 7 of this Agreement, or when a Dedicated Line of Credit is not permitted to be used (or the conditions for use have not been satisfied) pursuant to this Section 6.3(b).

If Customer secures its obligations to Company by Letter of Credit, the Letter of (2)Credit shall be substantially in the form set forth in Appendix H naming Company as the sole beneficiary, and shall at all times be in an amount equal to or greater than the Required Amount. The Letter of Credit shall, among other things, permit Company to make a drawing for the full amount of the Letter of Credit in the event that (a) Customer fails to renew or replace the Letter of Credit at least thirty (30) calendar days prior to the stated expiration of the Letter of Credit, (b) if, within the applicable period of grace provided therefor in this Article 6, Customer fails to provide Company with additional Performance Assurance as required pursuant to the provisions of this Article 6 or this Agreement, (c) an Event of Default by Customer has occurred and is continuing, or (d) Company otherwise has the right to draw upon any Performance Assurance of Customer pursuant to the terms of this Agreement. Upon notice by Company to Customer from time to time of the amount of the highest monthly bill incurred by Customer over the most recent twelve (12) month period, Customer shall adjust the amount of the Letter of Credit to the Required Amount within thirty (30) calendar days if the available amount of the Letter of Credit does not then equal or exceed the Required Amount and provide Company evidence of such change. If Company draws upon the Letter of Credit, Customer shall restore and maintain the amount available for draw under the Letter of Credit to equal or exceed the Required Amount within five (5) Business Days after any such drawing so that the Required Amount is available at all times. All costs of a Letter of Credit shall be borne by Customer.

(c) If, at any time, Customer's Chief Executive Officer or its CFO has reasonable grounds to believe or has actual knowledge that Customer has experienced a business condition or event that has caused Customer to fail to either (i) maintain an Investment Grade Credit Rating, or (ii) if Customer does not have a Credit Rating, to satisfy the minimum financial covenants determined in accordance with Section 6.3(d), below, in accordance with the applicable requirements of Section 6.3(a), Customer shall immediately provide written notice to Company advising Company of such event and shall further provide Company a summary of the details of such event to the extent available.

(d) Customer agrees to comply with each of the financial covenants set forth in this Section 6.3(d) and represents and warrants that such financial covenants are the most stringent criteria among Customer's Lenders for these four financial covenants. If at any time during the Term of this Agreement, Customer does not have a Credit Rating, Customer shall at all times meet or exceed each of the following minimum financial covenants that are included in the Worksheet and defined and calculated in accordance with Appendix D:

- (i) Debt Service Coverage Ratio of greater than 1.75.
- (ii) Times Interest Earned Ratio of greater than 1.50.
- (iii) Equity Ratio equal to or greater than .35 to 1.0.
- (iv) Total Debt to EBITDA of not greater than 8.0 to 1.0.

Customer acknowledges that the minimum financial covenants and related definitions established by Customer's Lenders (or trustee under a trust indenture or similar instrument securing holders of obligations secured by a senior lien on Customer's assets), as referenced in the Worksheet, may be changed from time to time by Customer's Lenders or trustee under a trust indenture or similar instrument securing holders of obligations secured by a senior lien on Customer's assets. In the event the most stringent financial covenants and/or related definitions as set forth in the Worksheet are modified by Customer's Lender or trustee, as the case may be, imposing the most stringent covenant(s) such that the applicable covenant becomes more or less stringent than the corresponding financial covenant and/or related definition set forth in this Section 6.3(d) and the Worksheet, Customer agrees that the applicable covenant set forth in this Section 6.3(d) shall be automatically revised, as of the effective date of such change in such financial covenant, to match the most stringent covenant among Customer's Lenders or trustees, as the case may be, for each of these four financial covenants; provided that in the event any of these covenants is deleted by Customer's Lenders or trustees, such covenant shall not, without Company's consent, be deleted from and shall continue to apply with respect to this Agreement at the level and with the definition in effect prior to such deletion. Customer shall notify Company in writing within five (5) days of any such modification, adjustment or amendment and the Parties shall, within ten (10) days thereof, amend the Worksheet to conform to such modifications, adjustments or amendments, provided that, until such Worksheet is amended, the financial covenants and related definitions shall be automatically revised as provided in the preceding sentence. As a result, the most stringent financial covenant imposed by any one of Customer's Lenders or trustees, as the case may be, shall be the applicable financial covenant that applies for purposes of this Agreement.

### 6.4 Remedies.

In the event that (1) Customer provides notice to Company pursuant to Section (a) 6.3(c), above, (2) an Event of Default by Customer occurs and is continuing, (3) Customer fails to maintain an Investment Grade Credit Rating (if it has a Credit Rating), (4) Customer fails to satisfy any one or more of the financial covenants set forth in Section 6.3(d), or (5) Customer otherwise fails to comply with any of the other provisions of Section 6.3 (including, without limitation, the failure of Customer to replace any bank issuing a Dedicated Line of Credit or Letter of Credit with a Creditworthy Bank and provide a new Dedicated Line of Credit (that satisfies the requirements of this Agreement) or Letter of Credit, as applicable, within thirty (30) calendar days of the failure of such bank to satisfy the requirements set forth in the definition of Creditworthy Bank), upon the earlier to occur of (i) notice from Customer or Company to the other of any such event or (ii) actual knowledge of any such event by Customer, Customer shall (A) pay all amounts outstanding under this Agreement as of the date of notice or such knowledge within five (5) Business Days of receipt of such notice or such knowledge, (B) post and maintain a Letter of Credit in favor of Company for an amount equivalent to or greater than two times the highest total monthly bill incurred by Customer over the most recent prior twelve (12) month period, and (C) commence prepaying Company weekly in advance for all amounts due as reasonably computed and invoiced by Company until such time as Customer satisfies the applicable requirements of Section

6.3(a) and each of the applicable events described in this Section 6.4(a) has been cured and is no longer continuing, unless otherwise agreed by Company. Company agrees that simultaneously with the establishment of a Letter of Credit pursuant to this Section 6.4(a), any previously established Dedicated Line of Credit shall be extinguished and of no further effect and any previously established Letter of Credit shall be amended to equal an amount equivalent to or greater than two times the highest total monthly bill incurred by Customer over the most recent prior twelve (12) month period as provided above or replaced upon the establishment of a new Letter of Credit pursuant to the preceding sentence. In the event that Company draws on the Letter of Credit from time to time, Customer shall post and maintain an additional Letter of Credit, or amend the existing Letter of Credit to reinstate the available amount thereunder by an amount equal to two times the Required Amount, for the amount drawn within three (3) Business Days after such drawing so that a Letter of Credit in an amount equal to two times the Required Amount remains available at all times.

(b) Failure of Customer to comply with the requirements of Article 6 shall be an Event of Default unless remedied within five (5) days; provided, however, the cure period set forth in this Section 6.4(b) shall not apply to the failure to post Performance Assurance under Section 6.4(a).

(c) Company shall apply the proceeds of the Performance Assurance realized upon the exercise of any such rights or remedies to reduce Customer's obligations under this Agreement (with Customer remaining liable for any amounts owing to Company after such application). Company shall return any surplus proceeds remaining after such obligations are satisfied in full provided that all required Performance Assurance has been posted, and provided further that no Event of Default or event, which with the giving of notice or the lapse of time, or both, would constitute an Event of Default, shall have occurred and then be continuing under this Agreement.

(d) Prepayments made by Customer pursuant to this Article 6 shall bear interest at a rate equal to the Federal Funds Rate minus 0.25% as from time to time in effect. "Federal Funds Rate" means the rate, for the relevant determination date opposite the caption "Federal Funds (Effective)", as set forth in the weekly statistical release designated as H.15 (519), published by the Board of Governors of the Federal Reserve System. Such prepayment shall bear interest at the foregoing rate commencing on the date the prepayment is received by Company, but excluding the date prepayment is applied to a Customer's obligations, and shall be calculated on a 365- or 366-day year, as the case may be, for the actual number of days elapsed.

(e) Customer shall invoice Company monthly setting forth the calculation of the interest amount due, and, provided that no Event of Default, or event which with the giving of notice or the lapse of time, or both, would constitute an Event of Default, shall have occurred and be continuing, Company shall credit Customer's account with the amount of interest due by the later of (i) the fifteenth day of the first month after the last month to which such invoice relates, or (ii) the fifteenth day after the day on which such invoice is received.

6.5 Credit Assurances of Company. Customer hereby waives any and all rights it may have at law or otherwise to require Company to provide financial assurances or security (including cash, letters of credit or other security) in respect of Company's obligations under this Agreement.

## **ARTICLE 7 DEFAULT AND REMEDIES**

7.1 Events of Default. Any one or more of the following shall constitute an Event of Default hereunder with respect to the Defaulting Party:

(a) The failure to make, when due, any payment required pursuant to this Agreement (other than payments disputed under Section 4.11) if such failure is not remedied within five (5) Business Days after written notice;

(b) Any representation or warranty made by a Party herein is false or misleading in any material respect when made, if such failure is not remedied within thirty (30) calendar days after written notice;

(c) Any certification made by Customer pursuant to Article 6, the Worksheet or the financial covenants is false or misleading in any material respect when made, if such failure is not remedied within five (5) Business Days after the earlier to occur of (i) receipt of written notice and (ii) Customer's knowledge of such failure;

(d) The failure by Customer to comply with the provisions of Article 6 if such failure is not remedied within five (5) Business Days (without duplication of the time period set forth in Section 6.4(b)), except that in the case of the failure to perform, maintain or replenish Performance Assurance within the time period set forth in Article 6, such failure shall be an Event of Default with no further cure period permitted;

(e) The failure to perform any material covenant or obligation set forth in this Agreement (except to the extent constituting a separate Event of Default as specified above, and except to the extent Company's obligations to deliver Full Requirements Electric Service are excused by the provisions of Article 8 or this Agreement), if such failure is not remedied within thirty (30) calendar days after written notice; provided, however, if a period in excess of thirty (30) calendar days is required to cure such failure, the Defaulting Party shall have an additional amount of time not to exceed ninety (90) calendar days, as may be necessary to cure such failure, provided that the Defaulting Party uses reasonable diligence to remedy such failure;

(f) A Party: (i) files a petition or otherwise commences, authorizes or acquiesces in the commencement of a proceeding or cause of action under any bankruptcy, insolvency, reorganization or similar Law, or has any such petition filed or commenced against it, which is not dismissed within ninety (90) days; (ii) makes an assignment or any general arrangement for the benefit of creditors; (iii) otherwise becomes bankrupt or insolvent (however evidenced); (iv) has a liquidator, administrator, receiver, trustee, conservator or similar official appointed with respect to it or any substantial portion of its property or assets; or (v) is generally unable to pay its debts as they fall due; (g) With respect to Customer, a default of Customer occurs under Customer's loan documents with any Lender and such Lender provides a notice of default to Customer or, if no such notice is provided, a default under Customer's loan documents that (if uncured or due to the lapse of time or both) could lead to an acceleration of the outstanding indebtedness by such Lender;

(h) With respect to Customer, a breach of any of the requirements set forth in Article 16; or

(i) An Event of Default of Customer occurs under the Short-Term Agreement and such event of default is not cured within the applicable cure period.

## 7.2 Declaration of an Early Termination by Company.

(a) If an Event of Default by Customer as the Defaulting Party shall have occurred and be continuing, Company as the Non-Defaulting Party shall have the right to take one or more of the following actions or combinations of actions: (i) to designate a Business Day, no earlier than the day such notice is effective and no later than forty-five (45) days after such notice is effective, as an early termination date ("Early Termination Date"); (ii) withhold any payments due to the Defaulting Party under this Agreement; (iii) suspend performance; and/or (iv) pursue any remedy at Law, in equity and/or under this Agreement. The declaration of an Early Termination Date shall, as of the Early Termination Date, automatically terminate this Agreement by its terms, Company shall have no obligation to provide any level or amount of service to Customer under this Agreement on any basis and Customer waives any and all rights to raise in any forum a claim that Company must provide service to Customer on any basis other than pursuant to the terms of the Short-Term Agreement, including without limitation:

(i) Any terms or provisions of this Agreement;

(ii) Any previous agreements between Customer and Company, including, but not limited to, any previous electric service agreements, any settlement agreements resolving FERC proceedings, or any settlement agreements resolving any state or federal court suits;

(iii) any federal Law;

(iv) any Florida Law; or

(v) on any other basis.

(b) If, but only if, an Event of Default as specified in Section 7.1(f) occurs with respect to Company, Customer shall have the right to designate a Business Day, no earlier than the day such notice is effective and no later than forty-five (45) days after such notice is effective, as an Early Termination Date; and/or suspend performance.

7.3 Other Remedies of Company. Notwithstanding any other provision of this Agreement, if an Event of Default by Customer as the Defaulting Party shall have

occurred and be continuing, or an Early Termination Date has occurred, Company as the Non-Defaulting Party, upon written notice to Customer, shall have the right to take one or more of the following actions or combinations of actions:

(a) To accelerate all amounts owed by Customer to Company to be due and payable immediately upon receipt of notice from Company;

(b) Exercise any remedy available at Law, subject to the limitations set forth in Section 10.1 hereof, which remedies shall include without limitation the right to recover direct damages for the remaining Term (as if this Agreement had not been terminated (if terminated)), and, in addition, Company shall have the right to recover its stranded costs in accordance with Section 35.26 of FERC's regulations;

(c) To exercise any remedy available in equity;

(d) To suspend performance under this Agreement, but only if the Event of Default is as specified in Sections 7.1(a), 7.1(d), 7.1(f), 7.1(g), 7.1(h) or 7.1(i) and only if, after such Event of Default, additional Performance Assurance as requested by Company to secure such Event of Default is not provided by Customer within two (2) Business days of Company's written demand therefor;

(e) exercise its rights of setoff against the Short-Term Agreement or any and all property of Customer in the possession of Company or its agent;

(f) draw on any outstanding Performance Assurance issued for Company's benefit (whether issued under the Short-Term Agreement or this Agreement); and/or

(g) exercise any of the rights and remedies of Company with respect to Performance Assurance issued for Company's benefit (whether issued under the Short-Term Agreement or this Agreement) including liquidation of all Performance Assurance then held by or for the benefit of Company free from any claim or right of any nature whatsoever of Customer, including any equity or right of purchase or redemption by Customer.

7.4 Right of Specific Performance and Injunctive Relief by Customer. If an Event of Default by Company shall have occurred and be continuing, Customer as the Non-Defaulting Party, upon written notice to Company, shall have the right to pursue specific performance of Company's obligations hereunder and injunctive relief pursuant to Article 15. Notwithstanding any other provision of this Agreement, absent fraud, such specific performance and injunctive relief and the remedies set forth in Section 8.1(c) (if applicable) and Section 13.1(d) (in the case of a breach of the confidentiality provisions set forth in Section 13.1) shall be Customer's sole and exclusive remedies under this Agreement, and Customer hereby waives all other rights, damages and remedies, including without limitation direct damages, cover damages, damages at law, and any consequential or other damages or remedies waived or limited by Section 10.1. In no event shall such injunctive relief require the payment of money or property to Customer. Customer agrees and acknowledges that Company shall have no liability for damages or other responsibility for other losses hereunder (except as set forth in Sections 8.1(b) and

10.2). Without limiting the provisions of Section 17.5, if any provision or provisions of this Section 7.4 shall be held to be invalid, illegal, or unenforceable, the validity, legality, and enforceability of the remaining provisions of this Section 7.4 shall in no way be affected or impaired thereby; and the Parties hereby agree to effect such modifications to this Agreement as shall be reasonably necessary in order to give effect to the original intention of the Parties with respect to this Section 7.4.

7.5 Sole and Exclusive Remedies. ABSENT FRAUD, THE REMEDIES SET FORTH IN THIS AGREEMENT, INCLUDING WITHOUT LIMITATION IN ARTICLE 6, THIS ARTICLE 7 AND SECTIONS 4.11, 4.12, 8.1(c) AND 13.1(d), CONSTITUTE THE SOLE AND EXCLUSIVE REMEDIES AGAINST THE OTHER FOR EVENTS OF DEFAULT, BREACH OF CONTRACT OR ANY FAILURE TO PERFORM ANY OF THE OBLIGATIONS UNDER THIS AGREEMENT.

7.6 Defenses and Counterclaims; Dispute Resolution Authority. Subject to the terms and conditions of this Agreement, including without limitation the limitation on remedies and liability set forth in Sections 7.4 and 7.5 and Articles 8 and 10 of this Agreement, in the event of a Dispute (including but not limited to any Dispute concerning Company's right to take any action under Section 7.2), each Party reserves the right to assert affirmative defenses and compulsory counterclaims pursuant to the provisions of Article 15 of this Agreement. In addition, notwithstanding any contrary provision set forth in this Agreement, the Parties agree that FERC's, the arbitrators' or any court's (as the case may be) authority to grant remedies (including the award of money damages) pursuant to Article 15 shall be limited to the remedies expressly set forth in this Agreement (including without limitation those set forth in Articles 7 and 8) and any such remedies (including any award of money damages) shall be made subject to the limitations of liability and remedies and disclaimer of damages set forth in Sections 7.4 and 7.5 and Articles 8 and 10 of this Agreement. In the Event of a Default, breach of contract or other failure to perform by Company, money damages may be recovered by Customer only to the extent provided in Section 8.1 of this Agreement.

7.7 Obligations At Expiration or Termination. Upon the termination or expiration of this Agreement, in addition to such rights and obligations enumerated elsewhere in this Agreement, the grant of any and all right and interest to Company to supply Full Requirements Electric Service or any other level or amount of electric service under this Agreement shall automatically cease, except to the extent otherwise required under Section 3.6. In such event, Customer and Company shall cooperate, in advance to the extent possible, to make all necessary filings with the Transmission Provider and to perform all other acts necessary to transfer all such rights and interests back to Customer in a timely manner.

## ARTICLE 8 CURTAILMENT, TEMPORARY INTERRUPTIONS AND FORCE MAJEURE

8.1 Curtailment and Temporary Interruptions.

If there is a shortage of capacity and/or energy requiring the curtailment of (a) Company's Full Requirements Electric Service deliveries, then upon being notified by the Transmission Provider or Company, Customer shall institute procedures which will cause a corresponding curtailment of the use of Full Requirements Electric Service by its Retail Load, including without limitation curtailment by Customer of its non-firm loads. It is the express intention of this provision that any curtailment of Full Requirements Electric Service shall fall equitably upon all firm loads served by Customer and all firm loads served by Company in the area affected by the curtailment, after, to the extent within Company's control, the curtailment of Company's and Customer's non-firm loads. and Company agrees that it will not curtail the supply of Full Requirements Electric Service to Customer in an undue discriminatory manner as compared with Company's Customers in the area affected by the curtailment. If upon notification of a requirement to curtail, Customer fails to institute such curtailment, Company shall be entitled to limit deliveries of Full Requirements Electric Service to Customer in order to effectuate reductions in Energy deliveries, in the smallest amount that is operationally practical. equivalent to or greater than the reduction which would have been effected had Customer fulfilled its curtailment obligation hereunder during the period any shortage exists, and, in such event, Company shall not incur any liability to Customer in connection with any such action so taken by Company.

Company will use reasonable diligence in furnishing Full Requirements Electric (b) Service to Customer, but Company does not guarantee that the supply of Full Requirements Electric Service furnished to Customer will be uninterrupted, or that voltage and frequency will be at all times constant. Company will not unduly discriminate against Customer, as among Company's Customers, when handling the impact of temporary interruptions that affect delivery of energy in the area of Customer. Temporary interruptions and fluctuations in voltage or frequency of Full Requirements Electric Service deliveries hereunder shall not constitute a breach of the obligations of Company under this Agreement. Regardless of any cause other than undue discrimination against Customer as described above in this Section 8.1(b), Company shall not in any case be liable under this Agreement (whether due to Company's own negligence, or strict liability), (a) for any such temporary interruptions of service, or (b) for complete or partial failure or interruption of service, or for fluctuations in voltage or frequency.

(c) It is the express intention of this provision that Company shall not be liable to Customer for the occurrence without regard to the cause (whether due to Company's own negligence, or strict liability), of any shortage of capacity and/or energy which will require or results from curtailment of any of the firm load served by Company, provided that Company curtails its available capacity and/or energy in a non-discriminatory manner in the manner provided above in this Section 8.1. In the event of any shortage of or failure to provide capacity and/or energy by Company to Customer (whether pursuant to Section 8.1(a), (b) or otherwise) that is not excused pursuant to this Agreement (including without limitation pursuant to this Section 8.1) and that results from undue discrimination by Company, Company shall pay Customer its actual direct damages, if any, resulting from any such shortage or failure, not to exceed five million dollars (\$5,000,000) in the aggregate for all such shortages or failures to provide capacity and/or energy during any one Calendar Year. Except for the recovery of such direct damages in the event of undue discrimination by Company in breach of Company's obligations under this Section 8.1, Company shall not be liable to Customer for any damages, including direct damages, cover damages, damages at law, or consequential or other damages, and Customer's rights and remedies shall be limited to the extent provided in Article 7 for any other Claims under this Agreement, whether relating to any shortage of or failure to provide capacity and/or energy by Company to Customer or otherwise. Without limiting the preceding sentence, in no event shall Company be liable to Customer for curtailments of capacity and/or energy, whether discriminatory or non-discriminatory, to the extent resulting (either in whole or in part) from an event or circumstance described in Section 8.2 or 8.3, below. Notwithstanding any provision of this Agreement to the contrary, the liability of Company to Customer pursuant to this Agreement shall not exceed five million dollars (\$5,000,000) in the aggregate, in any one Calendar Year, and such liability is limited solely to the direct damages described in this Section 8.1(c).

8.2 Force Majeure. To the extent either Party is prevented by Force Majeure (or the effects of a Force Majeure) from carrying out, in whole or part, its obligations under this Agreement and such Party (the "Claiming Party") gives notice and details of the Force Majeure to the other Party as soon as practicable, then the Claiming Party shall be excused from the performance of its obligations with respect to this Agreement (other than obligations to pay money). The Claiming Party shall remedy the Force Majeure with all reasonable dispatch. Until remedied by the Claiming Party, the non-Claiming Party shall not be required to perform or resume performance of its obligations (other than obligations to pay money) to the Claiming Party corresponding to the obligations of the Claiming Party excused by Force Majeure.

**8.3** Transmission. Customer recognizes that the Transmission Provider may curtail service to Customer's Retail Load and that upon notification of such a requirement to curtail, Customer and Company shall be obligated to do so, and if Customer fails to institute the required curtailment, the Transmission Provider will be entitled to limit deliveries during the period any shortage of capacity and/or energy exists. In no event shall Company be liable under this Agreement for any shortage of capacity and/or energy to the extent resulting from the transmission and/or distribution of capacity and/or energy, any acts or omissions of Company under the Limited Services Agreement or any acts or omissions of Company in its capacity as the Transmission Provider.

#### **ARTICLE 9 NOTICES, REPRESENTATIVES OF THE PARTIES**

9.1 Notices. Any notice, demand, or request required or authorized by this Agreement to be given by one Party to another Party shall be in writing. Such notice shall be sent by facsimile, courier, personally delivered or mailed, postage prepaid, to the representative of the other Party designated in this Article 9. Unless provided otherwise in this Agreement, any such notice, demand, or request shall be deemed to be given and effective (i) when received by facsimile; (ii) when actually received if delivered by courier, overnight mail or personal delivery; or (iii) three (3) days after deposit in the United States mail, if sent by first class mail.

(a) Notices and other communications to Company from Customer shall be addressed to:

Florida Power & Light Company 700 Universe Blvd. Mail Stop EMT/JB Juno Beach, FL 33408 Attention: Vice President

(b) With a copy to:

Florida Power & Light Company 700 Universe Blvd. Mail Stop EMT/JB Juno Beach, FL 33408 Attention: General Counsel Office

(c) Notices and other communications to Customer from Company shall be addressed to:

Lee County Electric Cooperative, Inc. 4980 Bayline Drive North Fort Myers, Florida 33917-3910 Attention: Chief Executive Officer

(d) With a copy to:

Henderson, Franklin, Starnes and Holt, P.A. 1715 Monroe Street P.O. Box 280 Fort Myers, FL 33902 Attention: General Counsel for Lee County Electric Cooperative, Inc.

(e) Any Party may change its representative by written notice to the other Parties.

**9.2** Authority of Representative. The Parties' representatives designated in Section 9.1 shall have full authority to act for their respective principals in all technical matters relating to the performance of this Agreement. The Parties' representatives shall not, however, have the authority to amend, modify or waive any provision of this Agreement unless they are duly authorized officers of their respective entities and such amendment, modification or waiver is made pursuant to Section 17.6.

# ARTICLE 10 LIABILITY, INDEMNIFICATION, AND RELATIONSHIP OF PARTIES

**10.1 Limitation on Consequential, Incidental and Indirect Damages.** THE LIMITATIONS ON DAMAGES CONTAINED HEREIN ARE IN ADDITION TO AND NOT IN LIEU OF ANY OTHER LIMITATIONS OF LIABILITY PROVIDED

FOR ELSEWHERE IN THIS AGREEMENT. TO THE FULLEST EXTENT PERMITTED BY LAW, NEITHER CUSTOMER NOR COMPANY, NOR THEIR RESPECTIVE OFFICERS, DIRECTORS, AGENTS, EMPLOYEES, MEMBERS, CUSTOMERS, PARENTS, SUBSIDIARIES OR AFFILIATES, SUCCESSORS OR ASSIGNS, OR THEIR RESPECTIVE OFFICERS, DIRECTORS, AGENTS, EMPLOYEES, MEMBERS, CUSTOMERS, SUCCESSORS, SUBSIDIARIES, AFFILIATES OR ASSIGNS, SHALL BE LIABLE TO THE OTHER PARTY OR THEIR RESPECTIVE MEMBERS, CUSTOMERS, PARENTS, SUBSIDIARIES, AFFILIATES, OFFICERS, DIRECTORS, AGENTS, EMPLOYEES, SUCCESSORS OR ASSIGNS, FOR CLAIMS, SUITS, ACTIONS OR CAUSES OF ACTION FOR INCIDENTAL, INDIRECT, SPECIAL, PUNITIVE, MULTIPLE, EXEMPLARY OR CONSEQUENTIAL DAMAGES IN CONNECTION WITH OR RESULTING FROM PERFORMANCE OR NON-PERFORMANCE OF THIS AGREEMENT, OR ANY ACTIONS UNDERTAKEN IN CONNECTION WITH OR RELATED TO THIS AGREEMENT, INCLUDING WITHOUT LIMITATION, ANY SUCH DAMAGES WHICH ARE BASED UPON CAUSES OF ACTION FOR BREACH OF CONTRACT, TORT (INCLUDING NEGLIGENCE AND MISREPRESENTATION). BREACH OF WARRANTY. LIABILITY, STATUTE, OPERATION STRICT OF LAW. INDEMNITY (EXCLUDING THIRD PARTY INDEMNIFIABLE CLAIMS UNDER SECTION 10.2) OR ANY OTHER THEORY OF RECOVERY. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN FOR COMPANY, CUSTOMER'S LIABILITY TO COMPANY SHALL BE LIMITED TO DIRECT DAMAGES ONLY, AND SUCH DIRECT DAMAGES SHALL BE THE SOLE AND EXCLUSIVE MEASURE OF DAMAGES AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. THE PROVISIONS OF THIS SECTION 10.1 SHALL APPLY REGARDLESS OF FAULT AND SHALL SURVIVE TERMINATION, CANCELLATION, SUSPENSION, COMPLETION OR EXPIRATION OF THIS AGREEMENT. NOTHING CONTAINED IN THIS AGREEMENT SHALL BE DEEMED TO BE A WAIVER OF A PARTY'S RIGHT TO SEEK INJUNCTIVE RELIEF.

## 10.2 Indemnification.

(a) Except for indemnifiable Claims pursuant to Section 10.2(c) and/or Section 10.2(d), to the extent permitted by Law, each Party shall indemnify, defend and hold harmless, on an After-Tax Basis, the other Party from and against any Claims arising from or out of any event, circumstance, act or incident, occurring or existing during the period when control and title to Full Requirements Electric Service is vested in such Party as provided in Section 10.3 of this Agreement.

(b) Each Party shall indemnify, defend and hold harmless, on an After-Tax Basis, the other Party from and against any and all Claims for injuries to person or property arising in any manner directly or indirectly by reason of the acts of such Party's authorized representatives while on the premises of the other Party under any rights of access provided herein to the extent of the indemnified Party's self-insured retention or deductible under its insurance policies.

(c) Customer shall indemnify, defend and hold harmless, on an After-Tax Basis, Company from and against any and all Claims by any Member or Person arising in any manner directly or indirectly by reason of this Agreement, including without limitation (i) a failure, interruption, curtailment, or deficiency in Company's supply of Full Requirements Electric Service under this Agreement for any reason or (ii) a breach or default of this Agreement for any reason.

(d) Neither Party assumes any responsibility of any kind with respect to the construction, maintenance, or operation of the system or other property owned or used by the other Party. To the extent permitted by Law, each Party agrees to indemnify, defend and hold harmless, on an After-Tax Basis, the other Party from any and all Claims for injuries to person or property by any Member or Person in any way resulting from, growing out of, or arising from or in connection with the construction, maintenance or operation of the other Party's system or other property. Customer further agrees to indemnify, defend and hold harmless Company from any and all Claims for injuries to persons or property by any Member or Person in any way resulting from, growing out of, or arising in or in connection with the use of, or contact with, Energy or Full Requirements Electric Service delivered hereunder after it is delivered to Customer and while it is flowing through the lines of Customer, or is being distributed by Customer, or is being used by Retail Load.

(e) If a Party intends to seek indemnification under this Section 10.2 from the other Party with respect to any Claim, the Party seeking indemnification shall give such other Party notice of such Claim within thirty (30) calendar days of the commencement of, or actual knowledge of such Claim, whichever is earlier. Such Party seeking indemnification shall have the right, at its sole cost and expense, to participate in the defense of any such Claim. The Party seeking indemnification shall not compromise or settle any such Claim without the prior consent of the other Party, which consent shall not be unreasonably withheld.

(f) The provisions of this Section 10.2 shall not apply to Company in its capacity as Transmission Provider or to Company as limited services provider under the Limited Services Agreement.

10.3 Title; Risk of Loss. Title to and risk of loss related to the Full Requirements Electric Service provided hereunder shall transfer from Company to Customer at the Points of Receipt. Company represents and warrants that it will deliver Full Requirements Electric Service to Customer free and clear of all claims or any interest therein or thereto by any person arising prior to the Points of Receipt.

## ARTICLE 11 REPRESENTATIONS, WARRANTIES AND COVENANTS

11.1 Company and Customer Representations and Warranties. As of the date of this Agreement and assuming the satisfaction of the conditions precedent set forth in Section 2.5 that are applicable to Company or Customer (as the case may be), Company and Customer each represent and warrant to the other that:

(a) It is duly organized, validly existing and in good standing under the Laws of the State of Florida;

(b) It has all corporate (or cooperative in the case of Customer) and regulatory authorizations, consents, notices and approvals necessary for it to legally perform its obligations under this Agreement;

(c) The execution, delivery and performance of this Agreement are within its powers, have been duly authorized by all necessary action and do not violate any of the terms and conditions in its governing documents, including, but not limited to, any organizational documents, charters, by-laws, indentures, mortgages or any material contracts to which it is a party, or any Law applicable to it;

(d) It has the necessary power and authority to enter into and deliver this Agreement and to perform its obligations under this Agreement;

(e) This Agreement, and each other document executed and delivered in accordance with this Agreement, constitutes its legally valid and binding obligation enforceable against it in accordance with its terms, subject to any equitable defenses;

(f) It is not bankrupt or insolvent and there are no proceedings pending or being contemplated by it or, to its knowledge, threatened against it, which would result in it being or becoming bankrupt or insolvent;

(g) As of the Effective Date of this Agreement, there is not pending or, to its knowledge, threatened against it any legal proceedings or Claims that could materially and/or adversely affect its ability to perform its obligations under this Agreement;

(h) There is no Event of Default or events which, with the giving of notice or lapse of time or both, would constitute an Event of Default with respect to it, and no such event or circumstance would occur as a result of its entering into or performing its obligations under this Agreement;

(i) It has no rights of sovereign immunity; and

(j) It is acting for its own account, has made its own independent decision to enter into this Agreement and as to whether this Agreement is appropriate or proper for it based upon its own judgment, is not relying upon the advice or recommendations of the other Party in so doing, and is capable of assessing the merits of and understands and accepts the terms, conditions and risks of this Agreement.

11.2 Additional Company and Customer Representations and Warranties.

(a) Company represents and warrants to Customer that this Agreement has been duly approved by the board of directors of Company and the board is authorized under the organizational documents of Company to enter into this Agreement.

(b) Customer represents and warrants to Company that this Agreement has been duly approved by the board of trustees of Customer and the board is authorized under the organizational documents of the Customer to enter into this Agreement without further action or consent of the Members.

(c) Customer represents and warrants that it is a rural electric cooperative that is subject to, and entitled to the powers provided by, Chapter 425 of the Florida Statutes (the "Act").

## 11.3 Additional Customer Covenants.

(a) Customer shall establish, maintain, and revise from time to time its rates, fees, and other charges for electric energy and/or capacity and distribution and other facilities, suppliers, equipment or services furnished by Customer so that Customer's cash flow shall be sufficient at all times to enable Customer to satisfy all of its obligations under this Agreement.

(b) Customer and Company hereby waive any rights of sovereign immunity, if any, for liability under contract or for torts.

(c) The Term of this Agreement does not extend beyond any applicable limitation on Customer imposed by Chapter 425 of the Florida Statutes or other relevant constitution, organic or other governing documents and applicable Law. Subject to Article 16, Customer shall maintain its status as a rural electric cooperative that is subject to, and entitled to the powers provided by, Chapter 425 of the Florida Statutes.

11.4 Warranty Disclaimer. EXCEPT AS SET FORTH IN THIS ARTICLE 11, COMPANY MAKES NO WARRANTIES (EXPRESS OR IMPLIED) WITH REGARD TO FULL REQUIREMENTS ELECTRIC SERVICE, CAPACITY, ENERGY OR ANCILLARY SERVICES SOLD OR PROVIDED PURSUANT TO THIS AGREEMENT, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, AND ANY AND ALL WARRANTIES ARE DISCLAIMED.

## ARTICLE 12 ASSIGNMENT

12.1 General Prohibition Against Assignments. Except as provided in Section 12.2 and Section 12.3, below, no Party shall assign, pledge or otherwise transfer this Agreement or any right or obligation under this Agreement without first obtaining the other Party's prior written consent, which consent shall not be unreasonably withheld. Any assignment, pledge or transfer in contravention of the terms of this Article 12 shall be null and void.

12.2 Exceptions to Prohibition Against Assignments by Company. Company may, without Customer's prior written consent, (i) pledge, encumber or collaterally assign this Agreement or the accounts, revenues or proceeds hereof in connection with any financing or other financial arrangements; (ii) in the case of Company only, transfer or assign this agreement to an Affiliate of Company, where such Affiliate's creditworthiness is equal to

or higher than that of Company; or (iii) transfer or assign this Agreement to any Person or entity succeeding by merger, consolidation, a change in control of Company or by acquisition of all or substantially all of the assets of Company; provided, however, that in each such case, any such transferee or assignee shall agree in writing to be bound by the terms and conditions hereof.

12.3 Exceptions to Prohibition Against Assignments by Customer. Customer may, without Company's prior written consent, (i) transfer or assign this Agreement to any Person or entity succeeding by merger or by acquisition to all or substantially all of the assets of Customer provided that all of the conditions set forth in Section 16.1(a) through (e) are first satisfied; or (ii) collaterally assign, mortgage or pledge its interest in this Agreement as security to one or more of the Lenders or an indenture trustee under any indenture securing the obligations of Customer (the "Other Secured Party") if Customer is then a borrower of such Lender or Other Secured Party for any obligations secured by any indenture, mortgage or similar lien on its system assets; provided that Lender or Other Secured Party may not (a) have itself or its designee substituted for Customer under this Agreement, or (b) sell, assign, transfer or otherwise dispose of this Agreement to itself or a third party, in each case without the prior written approval of Company (such approval not to be unreasonably withheld, delayed or conditioned) and unless itself and such third party satisfies the requirements set forth in Section 16.1(a) through (e). Upon any assignment for security to the Lender or the Other Secured Party, Company and Customer agree to enter into a consent and agreement, substantially in the form attached hereto as Appendix I, with such Lender or Other Secured Party and Customer.

### ARTICLE 13 CONFIDENTIALITY

## 13.1 Treatment of Confidential Information.

(a) To the extent permitted by Law, all Confidential Information shall be held and treated by the Parties and their agents, counsel and consultants in confidence, used solely in connection with this Agreement, and shall not, except as hereinafter provided, be disclosed without the other Party's prior written consent.

(b) Notwithstanding the foregoing, Confidential Information may be disclosed (i) to a third party for the purpose of effectuating the supply, transmission and/or distribution of Full Requirements Electric Service to be delivered pursuant to this Agreement; (ii) to regulatory authorities of competent jurisdiction, or as otherwise required by applicable Law, including state sunshine, open meeting, freedom of information, securities Laws or similar Laws; (iii) as part of any required, periodic filing or disclosure with or to any regulatory authority of competent jurisdiction; (iv) to rating agencies; and (v) to third parties in connection with merger, acquisition/disposition and/or financing transactions, provided that, in the case of (v), above, any such third party shall have signed a confidentiality agreement with the Disclosure of the Confidential Information and that strictly limit the recipient's use of such information only for the purpose of the subject transaction and that provide for remedies for non-compliance. Disclosing Party shall make all reasonable efforts to ensure that Confidential Information remains confidential even if disclosed, including marking such information confidential and requesting confidential treatment of such information.

(c) In the event that a Party ("Disclosing Party") is requested or required to disclose any Confidential Information pursuant to subsection (b)(ii) or (iii) of this Article 13, the Disclosing Party shall provide the other Party with prompt written notice of any such request or requirement, so that the other Party may seek an appropriate protective order, other confidentiality arrangement or waive compliance with the provisions of this Agreement. If, failing the entry of a protective order, other confidentiality arrangement or the receipt of a waiver hereunder, the Disclosing Party, in the opinion of counsel, is compelled to disclose Confidential Information, the Disclosing Party may disclose that portion of the Confidential Information which the Disclosing Party's counsel advises that the Disclosing Party is compelled to disclose.

(d) The Parties shall be entitled to all remedies available at Law or in equity to enforce, or seek relief in connection with, this confidentiality obligation. In addition to the foregoing, the Disclosing Party shall indemnify, defend and hold harmless the other Party from and against any Claims, threatened or filed, and, subject to Articles 7 and 10 of this Agreement, any direct losses, damages, expenses, attorneys' fees or court costs incurred by such Party in connection with or arising directly or indirectly from or out of the Disclosing Party's disclosure of the Confidential Information to third parties except as permitted above.

(e) Notwithstanding the above provisions, Company and Customer shall be permitted to communicate to the Transmission Provider any necessary information, including Confidential Information, with regard to implementation of this Agreement, and will make all reasonable efforts to ensure that such Confidential Information remains confidential.

# **ARTICLE 14 REGULATORY AUTHORITIES**

## 14.1 Effect of Regulation.

(a) Each Party shall perform its obligations hereunder in accordance with applicable Law. Unless specifically provided otherwise in this Agreement, nothing in this Agreement affects, modifies or negates either Party's rights or obligations under the FPA and the regulations promulgated thereunder, or any other federal or state Law. Nothing contained herein shall be construed to constitute consent or acquiescence by either Party to any action of the other Party which violates the Laws of the United States or any applicable state Laws, as those Laws may be amended, supplemented or superseded, or which violates any other Law.

(b) The Parties acknowledge that this Agreement is an agreement subject to the jurisdiction of the FERC under the FPA, that this Agreement is being entered into by Customer for the purpose of serving its Retail Load, and that neither Party shall terminate this Agreement except as provided in this Agreement and in accordance with 18 CFR 35.15.
(c) Customer shall not oppose, directly or indirectly, Company or any of its positions with regard to Company's provision of Full Requirements Electric Service or the provision of electric service or capacity to Company's Customers, including but not limited to the determination of any cost included in the formula charges established herein or in Company's retail tariff, and Customer hereby waives its right to do so, in each case, before the FPSC.

### ARTICLE 15 DISPUTE RESOLUTION

#### 15.1 Negotiation by Officers of the Parties; Provisional Relief

Except as specifically provided in Sections 4.6 and 4.11 herein, if any controversy, dispute, claim, counterclaim or cause of action involving the parties and/or their respective representatives ("Dispute") arises out of or relates to this Agreement or the interpretation, breach, validity or termination thereof, the Parties shall first seek to resolve the Dispute through negotiation; provided that all Disputes relating to termination of this Agreement shall be resolved by petition to FERC directly without negotiation. Either Party shall provide the other Party with written notice setting forth the parameters of the Dispute and a proposed means for resolving the same, and the support for such position ("Notice"). If such Dispute cannot be resolved through negotiation within sixty (60) days of the receipt by a party of Notice (or such longer period as the Parties may agree to in writing), the Parties agree that any such Dispute shall be resolved pursuant to Section 15.2 and, if applicable, Section 15.4 of this Agreement. The procedures specified in this Article 15 shall be the sole and exclusive procedures for the resolution of Disputes; provided, however, either Party may, without prejudice to any negotiation, FERC or arbitration procedures commenced pursuant to this Article 15, proceed in a Florida state court of competent jurisdiction located in Orange County, Florida to obtain temporary provisional injunctive relief (excluding permanent injunctive relief, including declaratory actions) if such action is necessary to avoid imminent irreparable harm, to provide uninterrupted electrical and other services, or to preserve the status quo pending the conclusion of such negotiation, FERC proceeding or arbitration. Without being prejudiced or bound by such temporary provisional injunctive relief as may be available or granted under the exclusive jurisdiction of a Florida state court of competent jurisdiction located in Orange County, Florida, and subject to the limitations set forth in this Agreement, including without limitation those set forth in Articles 7, 8 and 10 hereof, either FERC or the arbitrators (as the case may be) shall have full authority to grant the remedies set forth in this Agreement (including without limitation those set forth in Articles 7 and 8) or order the Parties to request that a court modify or vacate any temporary or preliminary relief issued by a court, and the Parties shall continue to participate in the procedures specified in this Article 15.

#### **15.2 Procedures for Resolution of Disputes**

(a) Disputes within the primary or exclusive jurisdiction of FERC shall be resolved by petition to FERC, subject to the further provisions of this paragraph. Where FERC does not act within sixty (60) Days of filing a Dispute with FERC or issues an order declining to act upon such a Dispute, the Dispute shall be subject to binding arbitration in accordance with the procedures set forth in Section 15.3(a). In addition, Disputes within the concurrent jurisdiction of FERC (which are those disputes subject to both FERC and court jurisdiction) shall be subject to binding arbitration in accordance with the procedures set forth in Section 15.3(a) if: (i) the current body of valid FERC precedent, as reasonably determined by counsel for the Party requesting resolution of the Dispute, reflects FERC's practice or policy of allowing disputes of a similar nature to be resolved by alternative dispute resolution rather than by FERC; or (ii) the Parties agree that the Dispute shall be arbitrated (each, an "Arbitrable FERC Dispute"). Disputes within the current jurisdiction of FERC that are not Arbitrable FERC Disputes shall be resolved by petition to FERC.

All other Disputes that are not resolved by FERC pursuant to Section 15.2(a), (b) above, shall be resolved by arbitration pursuant to Section 15.4; provided, however, that claims for breach of the confidentiality obligations set forth in Article 13, or claims for personal injury or tangible personal property damage arising in connection with this Agreement may at the election of either Party be resolved by any court of competent jurisdiction in the State of Florida, provided that exclusive jurisdiction for such claims shall reside with the courts of Orange County, Florida, and each of the Parties hereby irrevocably consents to the exclusive jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such limited suit, action or proceeding involving such claims and irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in any such court has been brought in an inconvenient forum. THE PARTIES HEREBY KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVE ANY RIGHTS THEY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY DISPUTE SUBJECT TO COURT RESOLUTION HEREUNDER OR OTHERWISE.

#### 15.3 Changes Subject to FERC and Standard of Review

(a) Subject to the exceptions set out herein in Section 15.3(b) and in Appendix B, the Parties hereby waive any rights they may have to request that any changes be made to this Agreement pursuant to Sections 205 or 206 of the FPA, and pursuant to the rules and regulations promulgated thereunder, and Customer further agrees to waive its rights to seek or support: (i) an order from FERC finding that the rate formulas or rate(s), charges, classifications, terms or conditions agreed to by the Parties in this Agreement are unjust and unreasonable; or (ii) any refund with respect thereto.

(b) Notwithstanding Section 15.3(a), the Parties expressly reserve their rights as follows:

(i) Customer reserves its rights under Section 4.6 to challenge certain aspects of Company's calculations, but only to the extent expressly provided in Section 4.6;

(ii) Either party may file under Section 205 or 206 of the FPA, as applicable, to the extent permitted in Section 4.7(c) and (d);

(iii) In the event that there is a change in Law or the FERC changes or adds any classification, rule or regulation or issues any order that impacts or otherwise effects the cost-of-service formulas provided for in Article 4, Appendix B or Appendix C or the FERC Form No. 1 data, Company may make a filing pursuant to Section 205 of the FPA to conform this Agreement or the associated Appendices to such changes or additions;

(iv) In the event that the FPSC changes or adds any classification, rule or regulation or otherwise issues any order that results in or otherwise recognizes or permits Company's recovery of additional or new costs, expenses, charges, Taxes, fees and/or assessments through Company's retail rates, Company may make a filing pursuant to Section 205 of the FPA to revise the cost-of-service formulas provided for in Article 4, Appendix B and Appendix C to permit the recovery of such costs, expenses, charges, Taxes, fees and/or assessments under this Agreement;

(v) (A) In the event that there is any change in Law that results in, or in the event any ISO, RTO, ITC or other future transmission organization creates, any additional or new costs, expenses, charges, Taxes, fees and/or assessments that are attributable or related to the production and/or provision of capacity, Energy and/or Generation-Related Services by Company to Company's retail and/or wholesale customers (including Full Requirements Electric Service to Customer); or (B) to the extent any production-related costs, expenses, charges, taxes, fees and/or assessments are incurred by Company that are not already provided for in the cost-of-service formulas provided for in Article 4, Appendix B or Appendix C, or are not reflected in the FERC Form No. 1 data, Company may make a filing pursuant to Section 205 of the FPA to revise the cost-of-service formulas provided for in Article 4 or the associated Appendices and/or to include the costs not reflected in the FERC Form No. 1 data in order to permit the recovery of such costs, expenses, charges, Taxes, fees and/or assessments under this Agreement;

(vi) The page, column and line number references to the FERC Form No. 1 data identified under the heading "Reference" in Appendix B shall be automatically amended to reflect any changes made by FERC that cause changes in the pagination, columns and line numbers of such FERC Form 1 data and Company will make a filing pursuant to Section 205 of the FPA to revise such page, column and line number references in Appendix B to reflect such changes in the FERC Form No. 1 data;

(vii) Company may file, pursuant to Section 205 of the FPA, to implement any changes permitted under Section 4.12 of the Agreement in the event of the advent of Retail Competition in Florida; and

(viii) Company may file, pursuant to Section 205 of the FPA, to implement any changes to the Delivery Points pursuant to Section 2.4 or any changes to the Company Generation Resources as contemplated by the definition of such term.

(ix) In the event that a filing is made pursuant to this Section 15.3(b), Customer and Company reserve their right to oppose any such filing to the extent such filing is inconsistent with the provisions of this Agreement. Customer shall be limited in any opposition to opposing the change proposed by Company as being inconsistent with the provisions of this Agreement and shall not be entitled to argue that changes should be made to any other aspect of this Agreement in order to make the overall rate just and reasonable or otherwise, and the scope of any proceeding initiated as a result of such filing shall be limited to those matters contained in the filing. Except as provided in this subsection (ix), Company and Customer shall support filings made pursuant to this Section 15.3(b).

(c) It is the intent of this Section and the Parties after a knowing, voluntary and due inquiry, to the maximum extent permitted by Law, that the provisions of this Agreement, except as enumerated in Section 15.3(b), shall not be subject to change under Sections 205 or 206 of the FPA, and that absent the written agreement of the Parties to change any of the exceptions enumerated in Section 15.3(b), above, the standard of review for changes to any of those enumerated exceptions proposed by a Party, or a non-party, or the FERC, acting sua sponte, shall be the public interest standard of review set forth in United Gas Pipe Line Co. v. Mobile Gas Service Corp., 350 U.S. 332 (1956) and Federal Power Commission v. Sierra Pacific Power Co., 350 U.S. 348 (1956) (the "Mobile-Sierra Doctrine").

In furtherance of the foregoing, each Party knowingly, voluntarily, and, after due inquiry and to the maximum extent permitted by Law: (x) covenants and agrees not to seek unilaterally from FERC, or any other authority, relief of any kind changing the provisions set forth in this Agreement under the public interest standard enumerated in the Mobile-Sierra Doctrine, notwithstanding any subsequent changes in applicable Law or market conditions that may occur; (y) completely waives any rights under statute, regulation, state or federal constitution or common law to assert or to support any complaint, claim, suit or other challenge in any regulatory, judicial or other forum, including without limitation, the public utility or service commission of any state, FERC, or any state or federal court, concerning or related in any way to unilateral abrogation or modification of the above enumerated provisions; and (z) acknowledges and agrees that: (1) it is aware of the Ninth Circuit's holding in Public Utility No. 1 of Snohomish v. FERC, 2006 WL 3717533 (9th Cir., Dec. 19, 2006), (2) such holding concerns contracts entered into pursuant to a seller's market-based rate authorization, (3) this Agreement is cost-ofservice based, (4) it intends that such holding has no precedential effect with respect to the Parties' rights and obligations under this Section 15.3 or this Agreement, and (5) acceptance for filing by FERC of this Agreement shall be sufficient to constitute as between the Parties, FERC's approval of this Agreement.

#### 15.4 Arbitration Procedure

(a) In the event of a Dispute which is to be resolved by binding arbitration in accordance with this Article 15, such arbitration shall be held in accordance with the Rules for Non-Administered Arbitration of the CPR International Institute for Conflict Prevention and Resolution ("<u>CPR</u>") then in effect (the "Rules"), except as modified herein, before a panel of three (3) arbitrators. The arbitration shall be held and the award shall be rendered in Orlando, Florida.

The Party initiating arbitration shall nominate one (1) arbitrator at the same time it **(b)** initiates arbitration. This nominee shall be neutral and impartial, shall not be a current or former representative or agent of such Party, shall be a CPR panel member and shall be reasonably believed by such Party to possess the requisite experience, education and expertise in respect of the matters to which the claim relates to enable such person to perform arbitral duties competently. The other Party shall nominate one (1) arbitrator within twenty (20) calendar days of receiving the notice of arbitration. This nominee shall be neutral and impartial, shall not be a current or former representative or agent of such Party, shall be a CPR panel member and shall be reasonably believed by such Party to possess the requisite experience, education and expertise in respect of the matters to which the claim relates to enable such person to perform arbitral duties competently. The two arbitrators shall appoint a third, neutral and impartial arbitrator, who shall serve as the chair of the arbitral tribunal, which arbitrator shall be a CPR panel member. The third, neutral arbitrator shall be a competent and experienced arbitrator, with at least fifteen (15) years of United States electric industry experience as a practicing attorney, and shall be unaffiliated with and without prior financial alliances with any Party, or either of the other arbitrators.

(i) If the two arbitrators are unable to agree on a third arbitrator within twenty (20) days of the appointment of the second arbitrator, a third arbitrator shall be selected by CPR with due regard given to the selection criteria above and input from the Parties and other arbitrators. The Parties shall undertake to request CPR to complete selection of the third arbitrator if possible, no later than forty (40) calendar days after the appointment of the second arbitrator. The costs charged by CPR for this service shall be borne equally by Company and Customer.

(ii) If prior to the conclusion of the arbitration any arbitrator becomes incapacitated or otherwise unable to serve, then a replacement arbitrator shall be appointed in the manner described above and applicable to the original arbitrator being replaced.

(c) Discovery and other pre-hearing procedures shall be conducted as agreed by the Parties, or if they cannot agree, as determined by a majority of the arbitrators. The hearing shall be held, if practicable, thirty (30) calendar days after all prehearing discovery has been completed.

(d) The arbitrators' decision shall be made in accordance with the terms and conditions of this Agreement, and shall consider any relevant evidence and testimony, and the arbitrators shall, if practicable, render their decision within thirty (30) calendar days following close of the hearing. The decision and award rendered by a majority of the arbitrators, made in writing, shall be final and binding upon the Parties. Any such decision and award may be entered and enforced in any court of competent jurisdiction. The arbitrators shall have no authority to award special, exemplary, multiple, punitive or consequential damages, or any other damages or remedies that are not permitted or provided for under this Agreement.

(e) The expenses of arbitration shall be borne equally by the Parties, except that each Party shall bear the compensation and expenses of its nominated arbitrator, own counsel,

witnesses and employees; provided further, that any costs incurred by a Party in seeking judicial enforcement of any decision and award rendered by the arbitrators, or a majority of the arbitrators, shall be chargeable to and borne exclusively by the Party against whom such court order of enforcement is obtained.

## ARTICLE 16 CONSOLIDATION, MERGER, CONVEYANCE AND TRANSFER

16.1 Consolidation, Merger, Conveyance, or Transfer Only to Certain Person. Customer covenants and agrees that a Change in Control shall not occur with respect to Customer without the prior written consent of Company (such consent not to be unreasonably withheld or delayed) unless all of the following conditions are satisfied:

(a) The Person formed, succeeding, surviving or resulting from such Change in Control, including without limitation the Person resulting from any consolidation or into which or with the Customer merges, that acquires all of or substantially all of the assets of Customer (whether by asset transfer, liquidation, dissolution or otherwise), or that acquires a majority of the ownership in or control of Customer (collectively, the "Successor Person"), shall be a Person organized and existing under the laws of the United States of America or any state or the District of Columbia, shall be organized as a cooperative with membership sufficient to recover the costs incurred under this Agreement on an annual basis, and shall expressly assume this Agreement by instrument supplemental hereto executed and delivered to Company, which instrument shall be in a form satisfactory to Company, and shall provide for the performance and observance of every covenant and condition hereof on the part of Customer to be performed or observed and, in addition, shall provide Company with the representations and warranties set forth in Sections 11.1 and 11.2.

(b) The Successor Person shall satisfy, on a pro-forma consolidated basis (based upon a pro forma consolidation as of the date of its last audited financial statements) the requirements set forth in Section 6.3(d) and shall have equal or better creditworthiness to that of Customer.

(c) Immediately after giving effect to such transaction, no Event of Default hereunder shall have occurred and be continuing.

(d) The Successor Person shall be organized as a cooperative with membership equal to or greater than that of Customer prior to such Change in Control so that it will have sufficient capability to satisfy the payment obligations of Customer under this Agreement.

(e) For purposes of this Section 16.1, the term "Change in Control" shall mean a merger or consolidation of Customer with or into another Person, a direct or indirect transfer or conveyance of all or substantially all of Customer's assets or the liquidation or dissolution of Customer, or a direct or indirect transfer, in one or a series of related transactions, of a majority of the ownership in or control of Customer.

16.2 Successor Substituted. Upon any consolidation or merger, or any conveyance or transfer of the properties and assets of any Person substantially as an entirety in

accordance with Section 16.1, the Successor Person shall succeed to, and be substituted for, and, subject to Section 16.3, may exercise every right and power of Customer hereunder with the same effect as if such Successor Person had been named as the Customer herein.

Coordination. Customer promptly shall notify Company of any proposed 16.3 Change in Control. Company is not required, but shall have the option, to serve new Customer load that as of the Effective Date of this Agreement is being served by or is in the service territory of another electric provider that Customer proposes to serve: provided, however, to the extent that such new load is currently being served by another generator pursuant to a written power purchase agreement with such generator that was in effect prior to any proposed service by Customer, Company shall have no obligation to serve such new load. Company shall have one hundred eighty (180) days from the date of notification from Customer to determine whether to serve new Customer load under the terms and conditions of this Agreement. If Company does not elect to serve such new Customer load under the terms and conditions of this Agreement prior to the expiration of such one hundred eighty (180) days, Customer shall have the right to enter into other arrangements to serve such new load. The provisions of this Section 16.3 shall not apply to de minimis new Member load which Customer serves as a result of acquisition of Members from or adjustments of service territory with adjacent electric providers, and any such load shall be part of Customer's Retail Load.

Non-Solicitation. Company agrees that neither Company nor its affiliates shall, 16.4 without the prior written consent of Customer or the board of trustees of Customer (or similar governing body of Customer), (1) acquire, offer to acquire, or agree to seek to acquire, directly or indirectly, all or substantially all of the assets of Customer or all or substantially all of its members; (2) make any public announcement with respect to, or enter into or agree to, offer, propose, or seek to enter into, directly or indirectly, any acquisition, transaction or other business combination for all or substantially all of the assets of Customer or all or substantially all of its members; or (3) make, or in any way participate in, directly or indirectly, any solicitation of proxies, votes, or gathering of consents, however denominated, seeking the vote of or consent of, or seeking to influence any member in connection with any election or vote related to the acquisition of all or substantially all of the assets of Customer or all or substantially all of its members. In no event are the provisions of this Section 16.4 intended to limit those rights which Company may obtain in the event of the adoption of Retail Competition in the State of Florida and Customer acknowledges that Company shall have the right to sell capacity, energy and related services directly to existing and/or former Members of Customer in the event of the adoption of Retail Competition in the State of Florida.

#### 16.5 Customer's Loss of Members.

(a) During the Term, if Customer experiences a Loss of Members, as set forth in Section 16.5(e), below, Customer shall pay an amount calculated as set forth in Appendix L, and such amount shall be included as part of the Monthly Bill (the "Loss of Members Charge"). Customer shall receive a credit against the load represented by such Loss of Members used to calculate the Loss of Members Charge by reducing the annual kW demand subject to Loss of Members Charge in column (6), Table 2, of Appendix L to the extent that (i) the FPSC, in a proceeding, recognizes Company's generation that was used to serve the load represented by such Loss of Members as benefiting Company's Customers and permits Company to recover the costs (including the ROE) associated with such generation through the retail rates payable by Company's Customers, or (ii) Company sells the capacity and energy associated with Company's generation used to serve the load represented by such Loss of Members to another wholesale customer such that Company is able to recover the costs (including the ROE) associated with such generation from another wholesale customer. To the extent Company does not fully recover the costs of Company's generation (including the ROE) that was used to serve the load represented by such Loss of Members, Customer shall continue to pay an amount equal to the difference between the costs (including the ROE) of Company's generation that was used to serve the load represented by such Loss of Members, Customer shall continue to pay an amount equal to the difference between the costs (including the ROE) of Company's generation that was used to serve the load represented by such Loss of Members, Customer shall continue to pay an amount equal to the difference between the costs (including the ROE) of Company's generation that was used to serve the load represented by such Loss of Members, and the actual costs recovered by Company pursuant to (i) and (ii) above.

(b) Customer shall not directly or indirectly aid, encourage, participate in or otherwise facilitate any act undertaken to cause any Member of Customer to terminate its membership unless such Member has failed to comply with its obligations to Customer, and Customer shall use commercially reasonable efforts to renew and extend all franchise agreements, territorial agreements and similar arrangements for the Term of this Agreement.

(c) Company shall use commercially reasonable efforts to secure FPSC approval for Company to recover, through retail rates payable by Company's Customers, costs associated with Company's generation used to serve the load existing prior to the Loss of Members.

(d) To the extent related solely to and as necessary to indirectly serve the load represented by a Loss of Members, Customer shall be entitled to resell capacity and energy purchased from Company to any successor which obtains the same franchise formerly held by Customer or otherwise serves the same area associated with a Loss of Members; provided, however, Company also shall have the right to serve any Person that is subsumed under the definition of Loss of Members. To the extent any capacity and Energy is resold by Customer under the first sentence of this Section 16.5(d) or to the extent Company sells capacity and Energy to any former Members (in the same area such Members were formerly served by Customer Members that are part of the Loss of Members calculation, or to the extent any former Members that are part of the Loss of Members were formerly served by Customer again (in the same area such Members were formerly served by Customer shall not be subject to any Loss of Members Charge calculation based on such Loss of Members under Section 16.5(a).

(e) For purposes of this Section 16.5, the term "Loss of Members" means a cumulative decrease exceeding three percent (3%) in the number of Members' accounts, which cumulative percentage decrease is calculated for the three year period that includes the current Calendar Year and the two (2) prior Calendar Years, where the load represented by the accounts formerly served by Customer continues to be served at or about the same locations, but by an electric service provider other than Customer. The

Loss of Members calculation shall be performed each Calendar Year during the Delivery Period. The reasons for a Loss of Members could include, but are not limited to, Customer's failure to retain a franchise agreement granting Customer the right to provide retail electric service to Members within a particular geographic area, incorporation of a new municipality that acquires Customer's facilities and thereafter serves the Members within said municipality, and changes of service territorial boundaries. The cumulative percentage decrease in the number of Members' accounts in a Calendar Year shall be calculated by dividing the number of former Members' accounts taking retail electric service at their existing locations from an electric service provider other than Customer during that Calendar Year plus such losses from the prior two (2) Calendar Years, by the highest number of Members' accounts during that three (3) year period, which resulting amount shall be i) multiplied by 100 percent. If the result of this calculation is greater than three percent (3%), then the calculation set forth in Appendix L shall be used to determine the amount of annual Loss of Members Charge that is attributable to the Loss of Members' accounts, and for which Customer continues to be responsible. Customer shall certify to Company, by January 30th of each year during the Term, the total number of Members and Members' accounts for the prior Calendar Year along with the total kWhs consumed in the prior Calendar Year by the Members' accounts that have been lost. Customer represents and warrants that the schedule set forth in Appendix L represents the total number of Members' accounts that Customer has had for the three (3) Calendar Years of 2004, 2005 and 2006 set forth in such Appendix. The method for calculating the amount of Loss of Members Charge and examples of such calculation are set forth in Appendix L.

#### ARTICLE 17 GENERAL PROVISIONS

17.1 Third Party Beneficiaries. This Agreement is intended solely for the benefit of the Parties hereto, and nothing herein shall be construed to create any duty to, or standard of care with reference to, or any liability to, any Person not a Party hereto. For the avoidance of doubt, neither Members nor the Lenders are third party beneficiaries of this Agreement.

**17.2** No Dedication of Facilities. Any undertakings or commitments by one Party to the other under this Agreement shall not constitute the dedication of generation facilities or the transmission system or any portion thereof by either Party to the other Party.

17.3 Waivers. The failure of a Party to insist in any instance upon strict performance of any of the provisions of this Agreement or to take advantage of any of its rights under this Agreement shall not be construed as a general waiver of any such provision or the relinquishment of any such right, except to the extent such waiver is in writing and signed by an authorized representative of such Party.

17.4 Choice of Law. The interpretation and performance of this Agreement shall be in accordance with and controlled by the Laws of the State of Florida, without giving effect to its conflict of Laws provisions.

17.5 Severability. If any provision or provisions of this Agreement shall be held to be invalid, illegal, or unenforceable, the validity, legality, and enforceability of the remaining provisions shall in no way be affected or impaired thereby; and, subject to Section 2.5, the Parties hereby agree to effect such modifications to this Agreement as shall be reasonably necessary in order to give effect to the original intention of the Parties.

17.6 Modification. No modification to this Agreement will be binding on any Party unless it is in writing and signed by the Parties.

17.7 Counterparts. This Agreement may be executed in any number of counterparts, and each executed counterpart shall have the same force and effect as an original instrument. Execution of this Agreement by facsimile or PDF signature is deemed to be, and has the same effect as, execution by original signature.

**17.8 Headings.** Article and Section headings used throughout this Agreement are for the convenience of the Parties only and are not to be construed as part of this Agreement.

17.9 Records. The Parties shall keep (or as necessary cause to be kept by their respective agents) for a period of at least five (5) years such records as may be needed to afford a clear history of the Full Requirements Electric Service supplied pursuant to this Agreement. For any matters in dispute, the Parties shall keep the records related to such matters until the dispute is ended.

17.10 Survival. The provisions of Section 3.3, Section 3.4, Section 3.5 and Section 3.6, Article 7, Article 10, Article 11, Article 13, Article 15, Section 17.4, Section 17.9, Section 18.2 and any other Section of this Agreement that specifies by its terms that it survives termination and any other Section of this Agreement that pertains to the obligation to pay amounts due for service rendered prior to termination, shall survive the cancellation, termination or expiration of this Agreement.

17.11 Cooperation to Effectuate Agreement. Each Party shall cooperate to implement the provisions of and to administer this Agreement in accordance with the intent of the Parties.

17.12 No Public Announcement. The Parties agree that no press release or public announcement concerning the transactions contemplated by this Agreement will be made unless mutually agreed to by the Parties in writing; provided, however, such mutual agreement shall not be required if:

(a) the disclosing Party determines that disclosure is reasonably necessary to comply with applicable Laws of a governmental authority having jurisdiction; or

(b) disclosure is made to the FERC or the FPSC in connection with a hearing, proceeding or a submission, whether voluntary or compulsory.

17.13 Stranded Costs. If a Party or any of its Affiliates becomes entitled to receive compensation associated with stranded generation, transmission, distribution or other

assets or costs, the other Party shall have no claim or entitlement to any such compensation.

17.14 Further Assurances. If either Party determines in its reasonable discretion that any further instruments, assurances, or other things are necessary or desirable to carry out the terms of this Agreement, the other Party shall execute and deliver all such instruments or assurances, and do all things reasonably necessary or desirable to carry out the terms of this Agreement.

17.15 No Joint Venture. This Agreement is not intended, and shall not be construed, to create any association, joint venture, agency relationship or partnership between the Parties or to impose any such obligation or liability upon either Party. Except as set forth in the Limited Services Agreement, neither Party shall have any right, power or authority to enter into any agreement or undertaking for, or act on behalf of, or to act as or be an agent or representative of, or otherwise bind, the other Party.

17.16 Joint Preparation. This Agreement shall be considered for all purposes as having been prepared through the joint efforts of the Parties and shall not be construed against one Party or the other as a result of the preparation, substitution, submission or other event of the negotiation, drafting or execution of this Agreement.

17.17 Entire Agreement. This Agreement and the Short-Term Agreement contain the entire agreement between the Parties with respect to the subject matter hereto and supersede all previous oral and written negotiations, commitments, and understandings of the Parties (including any preliminary term sheet) with respect to the Parties' respective rights and obligations set forth herein. There exist no other understandings, terms or conditions, written or oral, related to the rights and obligations established by this Agreement, and neither Party has relied on any representation, express or implied, not contained herein.

#### ARTICLE 18 TAXES

18.1 General. Company and Customer shall each use reasonable efforts to minimize Taxes applicable to the transactions to be carried out under the terms of this Agreement. Either Party, upon written request of the other, shall provide a certificate of exemption or other reasonably satisfactory evidence of exemption including, but not limited to, an applicable affidavit approved by the Florida Department of Revenue, if such Party is exempt from Taxes, and shall use reasonable efforts to obtain and cooperate with obtaining any exemption from or reduction of Tax.

#### 18.2 Applicable Taxes.

(a) Customer shall be responsible for its share of any existing or new Taxes imposed or levied upon Company and Customer's share of such Taxes shall be included in the Monthly Bill as costs identified in and allocated pursuant to the cost-of-service formulas set forth in Article 4, Appendix B and Appendix C. (b) Customer shall be responsible for all existing and any new Taxes, or other similar tax, fee or assessment 1) imposed or levied upon Customer, 2) attributable or related to Full Requirements Electric Service not included in any of the other elements of the Monthly Bill and/or 3) relating to the sale, use or consumption of Energy or Full Requirements Electric Service pursuant to this Agreement. In the event of a conflict between the provisions set forth in this subsection (b) and any other provision of this Agreement, this subsection (b) controls.

(c) If Company is required to collect or remit any Tax on behalf of Customer as a result of the transactions contemplated in this Agreement, as agent or otherwise, Customer shall reimburse any Tax to Company through the Monthly Bill, with such reimbursement to be made on an After-Tax-Basis.

To the extent that Company incurs a Tax liability due to denied or lost Tax (d) benefits previously recognized or enjoyed by Customer during the term of this Agreement, Customer shall be responsible for all such Taxes. Additionally, in the event that Company incurs interest expense or penalties on a Tax deficiency related to denied or lost Tax benefit previously recognized or enjoyed by Customer during the Term of this Agreement, Customer shall be responsible for all such related penalties and interest expense. To the extent such Taxes and interest are not captured and reflected in the Monthly Bill pursuant to Section 4.9, such Taxes or interest expense shall be incorporated into the Monthly Bill of Customer. To the extent any Taxes or interest is due from Customer to Company after termination of this Agreement, Company shall be paid all outstanding amounts for which Customer is responsible, in twelve (12) equal monthly payments, beginning with the month immediately following the month in which Customer is notified that such amount is due to Company. Such payments shall be made by Customer to Company on or before the first day of each month. This Section 18.2 shall survive the termination or expiration of this Agreement.

#### ARTICLE 19 RULES OF CONSTRUCTION

19.1 Terms used in this Agreement but not listed in this Article, or defined herein or in Article 1, shall have meanings as commonly used in the English language and, where applicable, in Good Utility Practice.

19.2 Words not otherwise defined herein that have well known and generally accepted technical or trade meanings are used herein in accordance with such recognized meanings.

**19.3** The masculine shall include the feminine and neuter.

**19.4** The words "include," "includes" and "including" are deemed to be followed by the words "without limitation."

19.5 References to contracts, agreements and other documents and instruments shall be references to the same as amended, supplemented or otherwise modified from time to time.

19.6 The Appendices attached hereto are incorporated in and are intended to be a part of this Agreement; provided, that in the event of a conflict between the terms of any Appendices and the terms of this Agreement, the terms of this Agreement shall take precedence.

19.7 References to Laws and to terms defined in, and other provisions of, Laws shall be references to the same (or a successor to the same) as amended, supplemented or otherwise modified from time to time.

**19.8** References to a Person shall include its permitted successors and assigns, and any entity succeeding to the functions and capacities of that Person.

**19.9** References to "Articles," "Sections," or "Appendices" shall be to articles, sections, or appendices of this Agreement, or, as appropriate, to sections of the FPA or FERC's regulations.

19.10 Unless the context plainly indicates otherwise, words importing the singular number shall be deemed to include the plural number (and vice versa); terms such as "hereof," "herein," "hereunder" and other similar compounds of the word "here" shall mean and refer to the entire Agreement rather than any particular part of the same.

19.11 This Agreement was negotiated and prepared by both Parties with the advice and participation of counsel. The Parties have agreed to the wording of this Agreement and none of the provisions hereof shall be construed against one Party on the ground that such Party is the author of this Agreement or any part hereof.

19.12 The Parties acknowledge and agree that this Agreement, the transactions contemplated hereby, and any instruments that may be provided by either Party hereunder (including any Performance Assurance) shall each, and together, constitute one and the same "forward contract" within the meaning of the United State Bankruptcy Code, and Company shall constitute a "forward contract merchant" under the United State Bankruptcy Code.

IN WITNESS WHEREOF, Company and Customer have caused this Agreement to be executed in duplicate by their respective duly authorized officers as of the Effective Date.

# FLORIDA POWER & LIGHT COMPANY

Juk BY:\_ NAME: Sam A. Forrest

TITLE: Vice President, Energy Marketing & Trading

LEE COUNTY ELECTRIC COOPERATIVE, INC.

1f BY:

NAME: William D. Hamilton TITLE: Executive Vice President and Chief Executive Officer



Lee County Electric Cooperative, Inc. Post Office Box 3455 North Fort Myers, FL 33918-3455 (239) 995-2121 • FAX (239) 995-7904 www.losc.net

September 4, 2007

Mr. Michael A. Robison Florida Power & Light Company 700 Universe Boulevard Mail Stop CTR/JB Juno Beach, FL 33408

Dear Mr. Robison:

SUBJECT: Amended Power Supply Agreement - Long- and Short-Term

Enclosed are two sets of the following executed documents for your files. Please sign and return one and retain the other for your files.

Long-Term Amended Agreement – Pages 14 and 15 Signed letter dated August 31, 2007, re: agreed-upon changes on above-mentioned pages

Short-Term Amended Agreement – Pages 12 and 13 Signed letter dated August 31, 2007, re: agreed-upon changes on above-mentioned pages

Sincerely,

Jemi Hanath

Dennie Hamilton Executive Vice President & Chief Executive Officer

Enclosures

Florida Power & Light Company, 700 Universe Bivd, Juno Beach, FŁ 33408



August 31, 2007

Lee County Electric Cooperative, Inc. 4980 Bayline Drive North Fort Myers, Florida 33917-3910 Attention: Mr. Dennie Hamilton, Chief Executive Officer

Re: Amendment to Long-Term Agreement for Full Requirements Electric Service Dated as of August 21, 2007, By and Between Florida Power & Light Company and Lee County Electric Cooperative, Inc. (the "Agreement")

This letter is being signed pursuant to Section 17.6 of the Agreement and constitutes a modification and amendment of the Agreement pursuant to such Section. Capitalized terms not otherwise defined in this letter have the meaning set forth in the Agreement. In consideration of the mutual covenants and agreements contained in this letter, by executing this letter in the space provided below, the Parties hereby agree that the Agreement is modified and amended (as of the date of this letter) as follows:

- 1. The sentence beginning with the words "In the event any of the above conditions" in the first full paragraph following Section 2.5(f) is amended by deleting the word "or" between the words "are not satisfied" and "are not waived" and by adding in its place the words "and, to the extent not satisfied,".
- 2. Section 2.8(a) of the Agreement is amended by deleting the word "or" between the words "are not satisfied" and "are not waived" and by adding in its place the words "and, to the extent not satisfied,".

For contract administration purposes, the Parties agree to replace pages 14 and 15 currently set forth in the original copies of the Agreement retained by each Party with replacement pages 14 and 15 that are attached to this letter as Exhibit A. Such replacement pages incorporate the above amendments. The Agreement, except as amended by this letter, remains unchanged.

FLORIDA POWER & LIGHT COMPANY BY:

NAME: Sam A. Forres

MAA

TITLE: Vice President, Energy Marketing & Trading

LEE COUNTY ELECTRIC COOPERATIVE, INC.

BY: 6 NAME: William D. Hamilton

TITLE: Executive Vice President and Chief Executive Officer

an FPL Group company

## EXHIBIT A REPLACEMENT PAGES

\_

an FPL Group company

È.

.

....

limited to an aggregate total of 300 MW during any hour unless otherwise mutually agreed in writing.

2.7 Cooperation. If requested by Company, Customer shall undertake commercially reasonable efforts to cooperate with and assist Company in Company's efforts to secure acceptance of this Agreement by FERC and request FERC action on these filings and, upon Company's request, shall make a timely submittal at FERC affirmatively supporting the acceptance of this Agreement by FERC in each case without modification, suspension, investigation, or condition. To the extent that this Agreement is modified pursuant to Section 2.6, Customer shall promptly execute an amendment to this Agreement that implements such modifications.

#### 2.8 Termination for Failure of Condition Precedent.

(a) In the event any of the above conditions precedent set forth in Section 2.5 are not satisfied or are not waived in writing by the dates provided above, this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(b) Notwithstanding any provision of this Agreement to the contrary, in the event this Agreement terminates pursuant to this Section 2.8, subject to Section 17.10, the Parties shall be released and discharged from any and all obligations arising or accruing hereunder from and after such date and shall not incur any additional liability to each other as a result thereof, except to the extent of any claims arising with respect to Section 2.7, Article 10 or Article 13. With respect to any such claims, Articles 7, 10, 13, 15, 17 and 19 shall survive and continue to apply.

#### ARTICLE 3 SALE AND PURCHASE

## 3.1 Partial Requirements Electric Service.

(a) During the Delivery Period, except as permitted by Sections 3.3, 3.4, 3.5 and 3.7, Company shall sell and deliver to Customer and Customer shall receive and purchase from Company, Partial Requirements Electric Service at the Receipt Points sufficient to serve Customer's Retail Load at the Delivery Points, as such load may be verified by Company. As a provider of Partial Requirements Electric Service, Company is solely responsible for satisfying all requirements and paying all costs incurred or to be incurred to provide Partial Requirements Electric Service to the Receipt Points, and Customer shall pay for such Partial Requirements Electric Service as provided in this Agreement, including without limitation as provided in this Article 4. In no event, however, shall Company be obligated to supply Customer during any hour of this Agreement more than 300 MW as measured at the Belle Meade and Buckingham Delivery Points combined.

(b) Subject to Section 16.5, Company understands that Customer's Retail Load may change from time to time. Subject to Sections 3.3, 3.4, 3.5 and 3.7, except as specifically provided herein, at no time during the Delivery Period shall Customer use, directly or indirectly, including through any of its Affiliates (i) constructed or purchased generation

Agreement that are acceptable to Company pursuant to Section 2.6, Customer shall promptly execute an amendment to this Agreement that implements such modifications.

### 2.8 Termination for Failure of Condition Precedent.

(a) In the event any of the above conditions precedent set forth in Section 2.5 are not satisfied or are not waived in writing by the dates provided above, this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(b) In the event that FERC issues an order requiring modifications or conditions on its acceptance or approval of the Short-Term Agreement and Company does not, within the earlier of (i) December 31, 2009, or (ii) six (6) months after the later of (x) the date of such FERC order or (y) the satisfaction of the conditions set forth in Section 2.5(c), provide written notice to Customer that such modifications are acceptable to Company in its sole discretion as set forth in Section 2.5(d), then this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(c) In the event that FERC issues an order requiring modifications or conditions on its acceptance or approval of this Agreement and Company does not, by December 31, 2009, provide written notice to Customer that such modifications are acceptable to Company in its sole discretion as set forth in Section 2.5(e) and Section 2.6, then this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(d) Notwithstanding any provision of this Agreement to the contrary, in the event this Agreement terminates pursuant to this Section 2.8, subject to Section 17.10, the Parties shall be released and discharged from any and all obligations arising or accruing hereunder from and after such date and shall not incur any additional liability to each other as a result thereof, except to the extent of any claims arising with respect to Section 2.7, Article 10 or Article 13. With respect to any such claims, Articles 7, 10, 13, 15, 17 and 19 shall survive and continue to apply.

#### ARTICLE 3 SALE AND PURCHASE

#### 3.1 Full Requirements Electric Service.

(a) During the Delivery Period, except as permitted by Sections 3.3, 3.4, 3.5 and 3.7, Company shall sell and deliver to Customer and Customer shall receive and purchase from Company, Full Requirements Electric Service at the Receipt Points sufficient to serve Customer's Retail Load at the Delivery Points, as such load may be verified by Company. As a provider of Full Requirements Electric Service, Company is solely responsible for satisfying all requirements and paying all costs incurred or to be incurred not approved such modifications or conditions to this Agreement by December 31, 2009, Company may waive this condition precedent by providing written notice of waiver to Customer by such date; otherwise this Agreement shall be null and void.

In the event any of the above conditions precedent set forth in this Section 2.5 are not satisfied or are not waived in writing by the dates provided above, Company's obligations under this Agreement shall automatically terminate by the terms of this Agreement, and Customer expressly waives any and all rights to raise in any forum a claim that Company must provide service (including without limitation Full Requirements Electric Service) to Customer on any basis other than pursuant to the terms of the Short-Term Agreement, including without limitation:

(i) any terms or provisions of this Agreement;

(ii) any previous agreements, if any, between Customer and Company, including, but not limited to, any previous electric service agreements, any settlement agreements resolving FERC proceedings or any settlement agreements resolving any State or Federal court suits;

(iii) any federal Law;

(iv) any Florida Law; or

(v) on any other basis.

2.6 FERC Acceptance. With respect to the condition set forth in Section 2.5(e), Company agrees to accept any modifications or conditions imposed by FERC with respect to this Agreement that, in the judgment of Company exercised in its sole discretion, do not (i) adversely affect the commercial, economic, legal and other benefits reasonably anticipated to be derived by Company from this Agreement or (ii) otherwise have an adverse impact on Company's Customers, Company or its shareholders; provided, however, Company will be under no obligation to accept any modifications to this Agreement or conditions imposed by FERC in connection with the acceptance or approval of this Agreement, if in the judgment of Company, exercised in its sole discretion, such modifications to this Agreement or conditions imposed by FERC in connection with the acceptance or approval of this Agreement would be adverse to Company as set forth in subsections (i) and/or (ii), above.

2.7 Cooperation. If requested by Company, Customer shall undertake commercially reasonable efforts to cooperate with and assist Company in Company's efforts to secure acceptance of this Agreement by FERC and request FERC action on these filings and, upon Company's request, shall make a timely submittal at FERC affirmatively supporting the acceptance of this Agreement by FERC in each case without modification, suspension, investigation, or condition. In addition, if requested by Company, Customer shall submit letters to the FPSC and take such other actions as reasonably requested by Company in support of Company's effort to secure authorization or approval from the FPSC pursuant to Section 2.5(c). To the extent that FERC requires modifications to this

Agreement that are acceptable to Company pursuant to Section 2.6, Customer shall promptly execute an amendment to this Agreement that implements such modifications.

### 2.8 Termination for Failure of Condition Precedent.

(a) In the event any of the above conditions precedent set forth in Section 2.5 are not satisfied or are not waived in writing by the dates provided above, this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(b) In the event that FERC issues an order requiring modifications or conditions on its acceptance or approval of the Short-Term Agreement and Company does not, within the earlier of (i) December 31, 2009, or (ii) six (6) months after the later of (x) the date of such FERC order or (y) the satisfaction of the conditions set forth in Section 2.5(c), provide written notice to Customer that such modifications are acceptable to Company in its sole discretion as set forth in Section 2.5(d), then this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(c) In the event that FERC issues an order requiring modifications or conditions on its acceptance or approval of this Agreement and Company does not, by December 31, 2009, provide written notice to Customer that such modifications are acceptable to Company in its sole discretion as set forth in Section 2.5(e) and Section 2.6, then this Agreement, except for those provisions that expressly survive, shall terminate automatically under the terms of this Agreement without any need by either Party to declare an Early Termination Date or take any further action, and this Agreement shall have no further force and effect.

(d) Notwithstanding any provision of this Agreement to the contrary, in the event this Agreement terminates pursuant to this Section 2.8, subject to Section 17.10, the Parties shall be released and discharged from any and all obligations arising or accruing hereunder from and after such date and shall not incur any additional liability to each other as a result thereof, except to the extent of any claims arising with respect to Section 2.7, Article 10 or Article 13. With respect to any such claims, Articles 7, 10, 13, 15, 17 and 19 shall survive and continue to apply.

#### ARTICLE 3 SALE AND PURCHASE

#### 3.1 Full Requirements Electric Service.

(a) During the Delivery Period, except as permitted by Sections 3.3, 3.4, 3.5 and 3.7, Company shall sell and deliver to Customer and Customer shall receive and purchase from Company, Full Requirements Electric Service at the Receipt Points sufficient to serve Customer's Retail Load at the Delivery Points, as such load may be verified by Company. As a provider of Full Requirements Electric Service, Company is solely responsible for satisfying all requirements and paying all costs incurred or to be incurred



February 15, 2008

Lee County Electric Cooperative, Inc. 4980 Bayline Drive North Fort Myers, Florida 33917-3910 Attention: Mr. Dennie Hamilton, Chief Executive Officer

Re: Amendment to Long-Term Agreement for Full Requirements Electric Service Dated as of August 21, 2007, By and Between Florida Power & Light Company and Lee County Electric Cooperative, Inc. (the "Agreement")

This letter is being signed pursuant to Section 17.6 of the Agreement and constitutes a modification and amendment of the Agreement pursuant to such Section. Capitalized terms not otherwise defined in this letter have the meaning set forth in the Agreement. In consideration of the mutual covenants and agreements contained in this letter, by executing this letter in the space provided below, the Parties hereby agree that the Agreement is modified and amended (as of the date of this letter) as follows:

1. A new Section 13.1(f) is added to Article 13 as follows:

"Notwithstanding anything in this Section to the contrary, if the FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to this Agreement, the Party shall provide the requested information to the FERC or its staff. In providing the information to FERC or its staff, the Party may, consistent with 18 C.F.R. 388.112, request that the information be treated as confidential and non-public by the FERC and its staff and that the information be withheld from public disclosure. The Party shall notify the other Party to the Agreement when it is notified by FERC or its staff that a request for disclosure of, or decision to disclose, confidential information has been received, at which time either of the parties may respond before such information is made public, pursuant to 18 CFR 388.112."

For contract administration purposes, the Parties agree to replace page 55 currently set forth in the original copies of the Agreement retained by each Party with replacement pages 55 and 55A that are attached to this letter as Exhibit A. Such replacement pages incorporate the above amendment. The Agreement, except as amended by this letter, remains unchanged.

**FLORIDA POWER & LIGHT COMPANY** BY:

NAME: Sam A. Forrest TITLE: Vice President, Energy Marketing & Trading

MAR

LEE COUNTY ELECTRIC COOPERATIVE, INC.

BY:

NAME: William D. Hamilton TITLE: Executive Vice President and Chief Executive Officer

an FPL Group company

## EXHIBIT A REPLACEMENT PAGES

even if disclosed, including marking such information confidential and requesting confidential treatment of such information.

(c) In the event that a Party ("Disclosing Party") is requested or required to disclose any Confidential Information pursuant to subsection (b)(ii) or (iii) of this Article 13, the Disclosing Party shall provide the other Party with prompt written notice of any such request or requirement, so that the other Party may seek an appropriate protective order, other confidentiality arrangement or waive compliance with the provisions of this Agreement. If, failing the entry of a protective order, other confidentiality arrangement or the receipt of a waiver hereunder, the Disclosing Party, in the opinion of counsel, is compelled to disclose Confidential Information, the Disclosing Party may disclose that portion of the Confidential Information which the Disclosing Party's counsel advises that the Disclosing Party is compelled to disclose.

(d) The Parties shall be entitled to all remedies available at Law or in equity to enforce, or seek relief in connection with, this confidentiality obligation. In addition to the foregoing, the Disclosing Party shall indemnify, defend and hold harmless the other Party from and against any Claims, threatened or filed, and, subject to Articles 7 and 10 of this Agreement, any direct losses, damages, expenses, attorneys' fees or court costs incurred by such Party in connection with or arising directly or indirectly from or out of the Disclosing Party's disclosure of the Confidential Information to third parties except as permitted above.

(e) Notwithstanding the above provisions, Company and Customer shall be permitted to communicate to the Transmission Provider any necessary information, including Confidential Information, with regard to implementation of this Agreement, and will make all reasonable efforts to ensure that such Confidential Information remains confidential.

(f) Notwithstanding anything in this Section to the contrary, if the FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to this Agreement, the Party shall provide the requested information to the FERC or its staff. In providing the information to FERC or its staff, the Party may, consistent with 18 C.F.R. 388.112, request that the information be treated as confidential and non-public by the FERC and its staff and that the information be withheld from public disclosure. The Party shall notify the other Party to the Agreement when it is notified by FERC or its staff that a request for disclosure of, or decision to disclose, confidential information has been received, at which time either of the parties may respond before such information is made public, pursuant to 18 CFR 388.112.

#### ARTICLE 14 REGULATORY AUTHORITIES

#### 14.1 Effect of Regulation.

(a) Each Party shall perform its obligations hereunder in accordance with applicable Law. Unless specifically provided otherwise in this Agreement, nothing in this Agreement affects, modifies or negates either Party's rights or obligations under the FPA and the regulations promulgated thereunder, or any other federal or state Law. Nothing contained herein shall be construed to constitute consent or acquiescence by either Party to any action of the other Party which violates the Laws of the United States or any applicable state Laws, as those Laws may be amended, supplemented or superseded, or which violates any other Law.

(b) The Parties acknowledge that this Agreement is an agreement subject to the jurisdiction of the FERC under the FPA, that this Agreement is being entered into by Customer for the purpose of serving its Retail Load, and that neither Party shall terminate this Agreement except as provided in this Agreement and in accordance with 18 CFR 35.15.



March 18, 2008

Lee County Electric Cooperative, Inc. 4980 Bayline Drive North Fort Myers, Florida 33917-3910 Attention: Mr. Dennie Hamilton, Chief Executive Officer

Re: Amendment to Long-Term Agreement for Full Requirements Electric Service Dated as of August 21, 2007, By and Between Florida Power & Light Company and Lee County Electric Cooperative, Inc. (the "Agreement")

This letter is being signed pursuant to Section 17.6 of the Agreement and constitutes a modification and amendment of the Agreement pursuant to such Section. Capitalized terms not otherwise defined in this letter have the meaning set forth in the Agreement. In consideration of the mutual covenants and agreements contained in this letter, by executing this letter in the space provided below, the Parties hereby agree that the Agreement is modified and amended (as of the date of this letter) as follows:

 At the request of the FERC, a new sentence is added to the end of Section 13.1(c) as follows: "Notwithstanding the above language, this provision does not permit a Disclosing Party to provide the other Party notice of a request or requirement of the FERC or its staff to disclose Confidential Information to the FERC or its staff."

For contract administration purposes, the Parties agree to replace pages 55 and 55A that were added on February 15, 2008 to the original copies of the Agreement retained by each Party with new replacement pages 55 and 55A that are attached to this letter as Exhibit A. Such replacement pages incorporate the above amendment. The Agreement, except as amended by this letter, remains unchanged.

FLORIDA POWER & LIGHT COMPANY

BY: Sam A. F

NAME: Sam A. Forest TITLE: Vice President, Energy Marketing & Trading

MAR

LEE COUNTY ELECTRIC COOPERATIVE, INC.

BY:

NAME: William D. Hamilton TITLE: Executive Vice President and Chief Executive Officer

## EXHIBIT A REPLACEMENT PAGES

even if disclosed, including marking such information confidential and requesting confidential treatment of such information.

(c) In the event that a Party ("Disclosing Party") is requested or required to disclose any Confidential Information pursuant to subsection (b)(ii) or (iii) of this Article 13, the Disclosing Party shall provide the other Party with prompt written notice of any such request or requirement, so that the other Party may seek an appropriate protective order, other confidentiality arrangement or waive compliance with the provisions of this Agreement. If, failing the entry of a protective order, other confidentiality arrangement or the receipt of a waiver hereunder, the Disclosing Party, in the opinion of counsel, is compelled to disclose Confidential Information, the Disclosing Party may disclose that portion of the Confidential Information which the Disclosing Party's counsel advises that the Disclosing Party is compelled to disclose. Notwithstanding the above language, this provision does not permit a Disclosing Party to provide the other Party notice of a request or requirement of the FERC or its staff.

(d) The Parties shall be entitled to all remedies available at Law or in equity to enforce, or seek relief in connection with, this confidentiality obligation. In addition to the foregoing, the Disclosing Party shall indemnify, defend and hold harmless the other Party from and against any Claims, threatened or filed, and, subject to Articles 7 and 10 of this Agreement, any direct losses, damages, expenses, attorneys' fees or court costs incurred by such Party in connection with or arising directly or indirectly from or out of the Disclosing Party's disclosure of the Confidential Information to third parties except as permitted above.

(e) Notwithstanding the above provisions, Company and Customer shall be permitted to communicate to the Transmission Provider any necessary information, including Confidential Information, with regard to implementation of this Agreement, and will make all reasonable efforts to ensure that such Confidential Information remains confidential.

(f) Notwithstanding anything in this Section to the contrary, if the FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to this Agreement, the Party shall provide the requested information to the FERC or its staff. In providing the information to FERC or its staff, the Party may, consistent with 18 C.F.R. 388.112, request that the information be treated as confidential and non-public by the FERC and its staff and that the information be withheld from public disclosure. The Party shall notify the other Party to the Agreement when it is notified by FERC or its staff that a request for disclosure of, or decision to disclose, confidential information has been received, at which time either of the parties may respond before such information is made public, pursuant to 18 CFR 388.112.

#### **ARTICLE 14 REGULATORY AUTHORITIES**

#### 14.1 Effect of Regulation.

(a) Each Party shall perform its obligations hereunder in accordance with applicable Law. Unless specifically provided otherwise in this Agreement, nothing in this Agreement affects, modifies or negates either Party's rights or obligations under the FPA and the regulations promulgated thereunder, or any other federal or state Law. Nothing contained herein shall be construed to constitute consent or acquiescence by either Party to any action of the other Party which violates the Laws of the United States or any applicable state Laws, as those Laws may be amended, supplemented or superseded, or which violates any other Law.

(b) The Parties acknowledge that this Agreement is an agreement subject to the jurisdiction of the FERC under the FPA, that this Agreement is being entered into by Customer for the purpose of serving its Retail Load, and that neither Party shall terminate this Agreement except as provided in this Agreement and in accordance with 18 CFR 35.15.

# APPENDIX A

# LIST OF DELIVERY POINTS

# Belle Meade

Buckingham

# Calusa

Note: Each of the above Delivery Points is at the point recognized by the Transmission Provider.

### APPENDIX B

# GENERATION DEMAND CHARGE AND

# GENERATION ENERGY CHARGE FORMULA RATE

# **APPENDIX B**

# Generation Demand Charge and Generation Energy Charge Formula Rate

The "Generation Demand Charge Rate" and "Generation Energy Charge Rate" used to calculate the Generation Demand and Energy Charges shall be determined pursuant to the cost-of-service formulas set forth in this Appendix B and in accordance with the provisions of Article 4 of this Agreement.

The Generation Energy Charge Rate shall exclude costs recovered through the Fuel Charge and the Fuel Adjustment Charge, set forth in Section 4.5 of the Agreement.

The references to FERC FORM No. 1 pages, columns and line numbers set forth in this Appendix B were derived from FPL's 2006 FERC FORM No.1. Such references are expected to change during the term of this Agreement and the Parties agree to jointly support any FERC filings required to reflect any such changes. FPL Cost of Service Formulas A-1

Determination of Generation Demand Charge Rate December 31

Line	Description	Reference	Demand Related	
1	Return on investment	A-5, L19 Col (2)	#DIV/0!	
2	Operation & maintenance expense	A-14, L19 Col (2)	#DIV/0!	
3	Depreciation expense	A-15, L14 Col (2)	#DIV/0!	
4	Taxes other than income taxes	A-16, L14 Col (3)	#DIV/0!	
5	Income tax	A-17, L5 Col (2)	#DIV/01	
6	Subtotal	Sum L1 thru L5	#DIV/0!	•
7	Less: Share of gains on non-separated wholesale power sales excluding emergency sales	A-4, L5 Col (2)	•	
8	Less: Other revenue credits	A-4a, L5 Col (3)	#DIV/0!	
9	Annual production fixed cost	L6 - L7 - L8	#DIV/0!	
10	Total of 12 monthly peak firm MWs @ generator	FPL CP Demand Schedule	-	MW
11	Customer's adjusted monthly coincident peaks	Customer CP Demand Schedule	#DIV/0!	MW
12	Customer's share of annual production fixed cost	L11/L10	#DIV/0!	
13	Customer's annual production fixed cost	L9 X L12	#DIV/0!	
14	Customer's sum of monthly kW billing demand	Customer Billing Demand Schedule	#DIV/0!	kW
15	Generation Demand Charge Rate	L13/L14	#DIV/0!	\$/kW-mor

FPL Cost of Service Formulas A-2 Determination of Generation Energy Charge Rate December 31

Line	Description	Reference	Energy Related	
1	Total fuel	A-14, L21 Col (4) - L1 Col (4)		
2	Purchased power (555)	A-14, L1 Col (4)	-	
3	Other production expense	A-14, L6 & L15 Col (3)	•	
4	Total production cost	Sum L1 thru L3		
5	Administrative and general expense	A-10, L18 Col (5)	#DIV/0!	
6	Return on investment	A-5, L19 Col (3)	#DIV/0!	
7	Depreciation expense	A-15, L14 Col (3)	#DIV/0!	
8	Income tax	A-17, L5 Col (3)	#DIV/0!	
9	Annual production variable costs	Sum L4 thru L8	#DIV/0!	
10	Less: Fuel Costs	Sum L1 + L2	-	
11	Non-fuel costs	L9 - L10	#DIV/0!	
12	Net MWh generated and purchased, less MWh sold	FERC 1, p. 401b Col (b) - Col (c)	• MW	
13	Generation Energy Charge Rate	L11/L12/1,000	#DIV/0! \$/kV	

# FPL Cost of Service Formulas

## A-4

Share of gains on non-separated wholesale power sales excluding emergency sales December 31

			]	Production		
Line	Description	Reference	Total (1)	Demand (2)	Energy (3)	
1	Actual Gain on Non-Separated Wholesale Power Sales	a/			<u> </u>	
2	excluding emergency sales Incentive Threshold	a/		<u>-</u>		
3	Difference of Actual vs. Threshold	L1 - L2		-	-	
4	FPL Share of Gains	b/	-			
5	Deduction to Production O&M from the Gain on non- separated wholesale power sales excluding emergency sales	 L1 - L4	· · · •			

a/ As provided by Article 4.7(c) of the Agreement.

b/ Equal to 20% of the gains above the Incentive Threshold as provided by Article 4.7(c) of the Agreement.

## FPL Cost of Service Formulas A-4a Other Revenue Credits December 31

Line	Description	Reference	System (1)	Allocator (2)	Production (3)
1	Ancillary Service Revenues - Generation Related	s/		100%	-
2	Other Generation Related Revenue Credits	b/		100%	-
3	Rent From Electric Property - General Plant	c/		#DIV/0!	#DIV/0!
4	Other General Revenue Credits	d/		#DIV/0!	#DIV/0!
5	Total	Sum LI thru L4	-		#DIV/0!

a/ Ancillary Service Revenues - Generation Related means Generation - Related Services as defined in Article 1 of the Agreement and includes the revenues booked in accounts 447 and 456 for generation-related ancillary services. These revenues include amounts charged to transmission customers as well as amounts charged to FPL for the use of the FPL system in making off-system power sales. The ancillary services charged transmission customers include reactive, regulation, spinning operating reserves and supplemental operating reserves. Revenues from energy imbalance service are revenue credited through the fuel charge. Reactive charges imputed for FPL's use of the FPL system in making third-party sales are also included in Ancillary Services Revenues -Generation Related.

b/ Other Generation Related Revenue Credits includes generation-related revenues booked in accounts 447 and 456, excluding amounts for: 1) transactions included in the divisor of this formula rate and 2) revenues credited through other components of this formula.

c/ Rental From Electric Property includes revenues booked to account 454 for the rental of general plant, allocated to production based on gross plant.

d/ Other General Revenue Credits includes miscellaneous revenues booked to accounts 454 and 456 not associated with a specific function, allocated to production based on gross plant.
A-5

Return on production related investment December 31, Simple Average

			Production				
			Total	Demand	Energy		
Line	Description	Reference	(1)	(2)	(3)		
1	Electric plant:	-		·	· · · · · · · · · · · · · · · · · · ·		
2	Gross plant in service	A-6, L9	#DIV/0!	#DIV/0!	#DIV/0!		
3	Accumulated depreciation	A-6, L16	#DIV/0!	#DIV/01	#D <b>I</b> V/01		
4	Accumulated Deferred Taxes	A-6, L23	#DIV/0!	#DIV/0!	#D <b>r</b> V/0!		
5	Net plant in service	L2 + L3 + L4	#DIV/0!	#DIV/0!	#DIV/0!		
6	Plant held for future use	FERC-1 p. 214	-	-			
7	Construction work in progress	CWIP Schedule	-	-			
8	Adjustment for FERC CWIP treatment	CWIP Adjustment Schedule					
9	Subtotal - Electric Plant	Sum LSthru LB	#DIV/01	#DIV/0!	#DIV/0!		
10	Working capital:						
11	Materials & supplies						
12	Fuel	A-9, L2	-	-			
13	Non-fuel	A-9, L9	#DIV/0!	#DIV/0!	#DIV/0!		
14	Total Materials and Supplies	LI2 + L13	#DIV/0!	#DIV/0!	#DIV/0!		
15	Prepayments	e/	#DIV/0!	#DIV/0!	#DIV/0!		
16	Cash requirements	A-8, L2	#DIV/0!	#DIV/0!	#DIV/0!		
17	Total investment	L9 + L14 thru L16	#DIV/0!	#DIV/0!	#DIV/0!		
18	Composite cost of capital	A-11, L5 Col (4)	#DIV/01	#DIV/0!	#DIV/0!		
19	Return on investment	L17 x L18	#DIV/0!	#DIV/01	#DIV/01		

a/ Classified and functionalized using Gross Plant, A-6 L10 Col (2)

FERC-1 p. 110 L57	
#DIV/0!	
#DIV/0!	
	#DIV/01

\_\_\_\_\_

·..•

A-6 Production related electric plant in service December 31, Simple Average

				Production					
Line	Description	Reference	System	Production	Demand	Energy			
	· · · · · · · · · · · · · · · · · · ·		(1)	(2)	(3)	(4)			
1	Gross plant in service:								
2	Plant in service, excl G&I	FERC-1 p. 204 - 207	#DTV/01	-	-	-			
3	Acquisition Adjustment Scherer 4	<b>s/</b> .		-	•				
4	Less: ARO related plant	Currently o/a	·						
5	Restated plant in service	L2 + L3 - L4	#DIV/0	-	-	-			
6	GSUs included in system total above (353)	OSU Statement		-	-				
7	General, intangible & electric plant purchased	A-7, L19	#DTV/01	#DIV/0!	#DIV/0!	#DIV/0!			
8	Less: ARO related general and intangible plant	<b>ک</b>		#DIV/01	#DIV/0t	#DIV/01			
9	Total	L5 + L6 + L7 - L8	#DIV/01	#DIV/0t	#DIV/01	#DIV/01			
10	Gross Plant allocator		#DIV/01	#DTV/01	= #D[V/0]	#DIV/01			
11	Accumulated Depreciation:								
.12	Plant in service, excl G&l	Acom Dap Statement, Rows 2 + 4 time 9	-	-	-	-			
13	Acquisition Adjustment Scherer 4	Acam Dep Statement, Row 3	-	-	•				
14	GSUs included in system total above (353)	GSU Statement		-	-				
15	General, intangible & electric plant purchased	Acam Dep Statement, Rows J + 10		#DIV/01	#DIV/0	#DIV/0!			
16	Total	Sum Li2 thru Li3	-	#DIV/0!	#DIV/0	#DIV/01			
17	Accumulated Deferred Taxes:								
18	Accumulated Deferred Taxes - Account 190 (debit)	FERC-1 p. 111 Lt2	-	#DTV/01	#DTV/01	#DIV/0			
19	Accumulated Deferred Taxes - Account 282 (credit)	FERC-) p. 113 L63	•	#DIV/01	#DIV/01	#DIV/01			
20	Accumulated Deferred Taxes - Account 283 (credit)	FERC-1 p. 113 L64	-	#DIV/01	#DIV/01	#DIV/0!			
21	Other Reg Asset - Deferred Taxes FAS 109 (debit)	FERC-1 p. 232 L29	-	#DIV/01	#DTV/01	#DIV/0			
22	Other Reg Liability - Deferred Taxes FAS 109 (credit)	FERC-1 p. 278 L32	-	#DIV/0)	#DTV/0)	#DIV/0)			
23	Total	Sam Lill thru L22	-	#DIV/0	#DTV/01	#DIV/0!			
			Ending Balance	Beginning Balance	Simple Average				
B/	Acquisition Adjustment	FERC-1 p. 200 L.12		·····					
		Acct	Ending Balance	Beginning Balance	Simple Average				
ь/	Plant in Service-ARO Asset	101098							

A-7

1.1

ĥ.

Production related general plant allocation December 31, Simple Average

			General plant accounts 101 and 106									
			(8)	(b) Payroli		(c)		(d)			(c)	(f)
Line	Description	Reference	Total system	Allocator	_	Production					Demand	Energy
1	General plant											·
2	389 Land and land rights	FERC-1 p. 206 1.86	#DIV/0!									
3	390 Structures and improvements	FERC-1 p. 206 L87	#DIV/01									
4	391 Office furniture and equipment	FERC-1 p. 206 L88	#DIV/01					1				
5	392 Transportation equipment	FERC-1 p. 206 L89	#DIV/0!									
6	393 Stores equipment	FERC-1 p. 206 L90	#DIV/01									
7	394 Tools, shop and garage equipment	FERC-1 p. 206 L91	#DIV/01									
8	395 Laboratory equipment	PERC-1 p. 206 L92	#DIV/01									
9	396 Power operated equipment	FERC-1 p. 206 L93	#DIV/01						·			
10	397 Communication equipment	FERC-1 p. 206 L94	#DIV/01									
11	398 Miscellaneous equipment	FERC-1 p. 206 L95	#DIV/0									
12	102 Electric plant purchased	FERC-1 p. 206 L101	#D[V/0]									
13	Subtotal	Sum L2 Unu L12	#D1V/0!	#DIV/01	a/ "	#DIV/0	x	#DIV/0!	Ъ/	=	#DIV/01	#DIV/0!
14	Percent of subtotal	% of LI3 Col (s)				#DIV/0!				=	#DIV/0	#DIV/0!
15	399 Other tangible property	FERC-1 p. 206 L97	•									
16	Total general plant	Sum L13 + L15	#DIV/01		-	#DIV/0					#DIV/0!	#DIV/0!
17	Percent of total	% of L 16 Col (a)				#DIV/01				-	#DIV/01	#DIV/0!
18	Intangible plant	FERC-1 p. 204 L5	-			#DIV/01					#DIV/0	#DIV/01
19	General and intangible plant	Sum L16 + L18	#DIV/01		-	#DIV/01	• • •		•••••••	····.	#D[V/0]	#DIV/0

a/ FERC-1 p. 354: Production payroll / Total payroll excluding A&G payroll b/ A-14a L51 Col (a)

98

.

### DRAFT - SUBJECT TO MANAGEMENT APPROVALS

FPL Cost of Service Formulas A-8 Production related Cash Requirement December 31

				Production	·
Line	Description	Reference	Production	Demand	Non-Fuel Energy
1	Other O&M expenses	A-14, L19 - L1	#DIV/0!	#DIV/0!	#DIV/0!
2	Total cash requirements	LI X 1/8	#DIV/0!	#DIV/0!	#DIV/0!

FPL Cost of Service Formulas A-9 Production related Materials & Supplies December 31, Simple Average

Line	Description	Reference	System	Allocator	Production	Demand	Energy
1	Materials & supplies						
2	Fuel (151)	FERC-1 p. 110 L45		100.00%	-	-	-
3	Non-fue!						
4	Account (158-allowances)	FERC-1 p. 110 L52	•	100.00%	-	•	-
5	Plant materials (154 & 163) a/	FERC-1 p. 110 L48 + L54	-	#DIV/0!	#DIV/01	#DIV/0!	#DIV/01
6	Merchandise (155)	FERC-1 p. 110 L49	- ·	#DIV/0i	#DIV/0!	#DIV/01	#DIV/0!
7	Other M&S (156)	FERC-1 p. 110 L50	•	#DIV/01	#DIV/01	#DIV/01	#DIV/0!
8	Nuclear held for sale (157)	FERC-1 p. 110 L51	-	#DIV/01	#DIV/0!	#DIV/0!	#DIV/0!
9.	Total non-fuel	Sum L4 thru L6	•	-	#DIV/01	#DIV/0!	#DIV/0!
10	Total materials & supplies	Sum L2 + L9	-	-	#DIV/0!	#DIV/01	#DIV/0!

a/ functionalized to Production Plant based on Gross Plant allocator, A-6, L10

A-10

Production related Administrative & general expense allocation December 31

			System	Allocator	Production	Demand	Energy
Line	Description	Reference	(l)	(2)	(3)	(4)	(5)
1	Administrative & General Expense						
2	920 Administrative and general salaries	FERC-1 p. 323 L181	-				
3	921 Office supplies and expenses	FERC-1 p. 323 L1#2	-				
4	922 Administrative expenses transferred-Credit	FBRC-1 p. 323 L113	-				
5	923 Outside services employed	FERC-1 p. 323 L114	-				
6	925 Injuries and damages	FERC-1 p. 323 L186	-				
7	926 Employee pensions and benefits	FERC-I p. 323 LIR7	-				
8	927 Franchise requirements	FERC-I p. 323 LIRE	-				
9	929 Duplicate charges-Credit	FERC-1 p. 323 L190	-				
10	930 Miscellaneous general expenses	FERC-1 p. 323 L191 + L192	-				
11	931 Rents	FERC-1 p. 323 L193	-				
12	Subtotal a/	Sum 1.2 thru 1.1 (	-	#DIV/0!	#DIV/0!	#DIV/01	#DIV/0!
13	924 Property insurance	FERC-1 p. 323 L185	-	#DIV/0!	#DTV/0!	#DIV/0!	#DIV/0!
14	Production related storm expense not recorded in 924	ы			-		-
15	928 Regulatory commission expenses a/	FERC-I p. 323 L/89	-	#DIV/0!	#DIV/0!	#DIV/0!	#DIV/0!
16	Reg comm exp FERC annual assessment a/	FERC Assessment Fee		#DIV/0	#DIV/0!	#D <b>IV</b> /0!	#DIV/0!
17	935 Maintenance of general plant a/	FERC-1 p. 323 L196	-	#DIV/01	#DIV/0!	#DIV/0	#DIV/01
18	Total	Sum LI2 to LI7	-	-	#DIV/0!	#DIV/0!	#DIV/0!

a/ Payroll allocator, A-7, L13

b/ The amount of total company production-related storm damage expenses (if any) that for retail rate-making purposes is charged against an operating reserve, as authorized by the FPSC.

# FPL Cost of Service Formulas A-11 Composite cost of capital December 31, Simple Average

			l otal company							
Line	Description	Description Reference	Average ca	Average capitalization			Wtd cost			
			(1)	(2)	_	<u>(3)</u>	(4)			
1	Long term debt	A-12, L3	•	#DIV/0!		#DIV/0!	#DIV/0!			
2	Preferred stock	A-13, L3	-	#DIV/0!		#DIV/0!	#DIV/01			
3	Common stock	<b>a/</b>	•••	#DIV/0!	b/		#DIV/0!			
4	Total equity	Sum L2 + L3	-	#DIV/0!			#DIV/0!			
5	Total	Sum L1 + L3		#DIV/0!			#DIV/0!			

Ending

**T**... 1

Proprietary capital

\_

Average

Beginning

Total Proprietary Capital FERC-1 p. 112 L16

a/

b/ Return on Common Equity as defined in article 4.7(b) of the PSA

# FPL Cost of Service Formulas A-12 Average Long Term Debt December 31, Simple Average

		,			Debt b	alances		
Line	Description	Reference	Ending	<u>.                                    </u>	Beginning		Average	
1	Total long-term debt	FERC-1 p. 112 L24		-		·		-
2	(Less) Securitization Bonds	Currently n/s						
3	Adjusted Long-Term Debt	Sum L1 - L2		-		-		-
							Year End	Total
4	Interest on long-term debt (427)	FERC-1 p. 117 L62						-
5	Amort. of Debt Disc. and Expense (428)	FERC-1 p. 117 L63						-
6	Amortization of Loss on Required Debt (428.1)	FERC-1 p. 117 L64						-
7	Amort. of Premium on Debt-Credit (429)	FERC-1 p. 117 L65						-
8	Amort. of Gain on Required Debt-Credit (429.1)	FERC-1 p. 117 L66						-
9	Interest on Debt to Assoc. Companies (430)	FERC-1 p. 117 L67						-
10	(Less) Interest on Securitization Bonds	Currently n/a						
11	Total interest	Sum L4 thru L10						-
12	Cost of long-term debt	LIT/L3				;	#DIV/	0!

# FPL Cost of Service Formulas A-13 Average preferred stock December 31, Simple Average

			Preferred stock balances					
Line	Description	Reference	Ending	Beginning	Average			
1	Preferred stock issued (204)	FERC-1 p. 112 L3	-	-	-			
2	Premium on preferred stock	Currently n/a						
3	Total preferred stock	Sum L1 thru L2	-	-				
4	Preferred stock dividends	FERC-1 p. 121 L80		-	-			
5	Embedded costs	L4/L3		-	#DIV/0!			

# FPL Cost of Service Formulas A-14 Production O&M Expenses December 31

.

				•	Ener	87
			Total company	Demand	Non-fuel	Fuci
Line	Description	Reference	(1)	(2)	(3)	(4)
1	555 Purchased power d/	FERC-1 p.327		-	_	
2	556 System control and load dispatching	FERC-1 p.321 L77		-		
3	557 Other expenses	° ₩	-	-	-	-
4	565 Transmission of electricity by others	۵/	-	-	-	-
5	Other production expenses	A-14(a)	-	•	•	
. 6	Total production excluding fuel used in generation	Sure Lithra L5	-		•	
7	A&G expenses	A-IO, LIS	•	#DIV/01	#DIV/0	•
8	GSU related O&M	GSU Schedule	<b>-</b>	-		
9	Total O&M, excluding fuel	SUM Ló thra Li		#DIV/01	#DIV/0	-
10	501 Fuel	FERC-  p. 320 L5	-	i _	-	
11	518 Fuel	FERC-1 9. 320 L25	-	· -	-	-
12	Less: Fuel handling	Corrently a/a				
13	Less: Sale of fly ash	Currently a/e				•
14	547 Fuel	FERC-1 p. 321 L63	-	-	-	-
15	Subtotal fuel	SUM LIG thes L 4	-	-	-	_
16	Subtotal Production O&M and fuel costs	19+L15	•	#D[V/0]	#DIV/0	
17	Other costs and expenses	•/				
18	Gains/Losses on disposition of allowances (411.8 and 411.9)	al/				
19	Total Production O&M and fuel costs	LI6+L17+L11	•	#DIV/01	#DIV/0	-
20	Fuel Adjustments and Revenue Credits	<u>م</u>				-
21	Total Fuel Cost	Sway 1.19 + 1.20				
	es any other costs and expenses incurred for compliance with the RPS provided by Accounting	S Requirements provid	ed in Article 3.6 of t	he Agreement.		
<b>Б</b> /	Description	Acct	Total company	Demand	Non-fuel	Fuel
	Other Power Supply Expenses	557000				

۰.

	Other Power Supply Expenses	337000				
ď	Description	Acct	Total company	Demand	Non-fuel	Fuel
	Transmission of Electricity by Others	565000		•		
	Transmission of Electricity by Others - CCRC	565120		-		
	Transmission of Electricity by Others - CCR	565121		-		
	Transmission of Electricity by Others - FCRC	565130				-
	Transmission of Electricity by Others - Fuel Rec-L/T CNT-Fuel	565131				-
	Total Transmission of Electricity by Others		-		-	-
					<u> </u>	
ď	Description	Acct	Total company	Demand	Non-fuel	Fuel
	PURCH PWR-RECOVERABLE INTRCHG-LOC 54	555110	•			· · ·
	PUR PWR-RCV INTC-L/T CNTR-MIN PYMT-FUEL	555111	-			
	PURCH PWR-UNIT PWR PURCH-SOUTHERN CO-EGY	555140	•			
	PURCHASE POWER - PPA - ENERGY	555141	•			
	PURCH PWR-SJRPP ENERGY EXPENSE	555142	-			
	PUR PWR-PPA-L/T CONTR-MIN PYMT-FUEL	555143	-			
	PURCH PWR-RECOVERABLE EXP-QUALIFY FACIL	555160	-			
	PURCH PWR-NON-RECOVERABLE EXP-LOC 54	555210	-			
	COST OF EMT THIRD PARTY POWER SALES	555225	•.			
	CAP CHARGES-NOT CCR-FPSC-'90 RATE REDUCN	555250	-			
	UPS CAPACITY CHARGES-CCR	555410				
	QF CAPACITY CHARGES-CCR	555420	•			
	OTH DEF CR-SJRPP CAPACITY ACCEL RECOVRY	555429	-			
	SJRPP CAPACITY CHARGES-CCR	555430				
	CAP CHARGES-CCR-FPSC-90 RATE REDUCTION	555431				
	SJRPP DEFERRED INTEREST PAYMENTS-CCR	555432	ه			
	SHORT TERM CAPACITY PURCHASES-CCR	555440	-			
	PUR PWR-CAPAC PUR-L/T CNTR-MIN PYMT-CCR	555441	-			
	SSS Purchased power - Total				-	
e/	Description	Acct				
-	SALES FOR RESALE - RECOV INTERCHANGE POWER SALES	447110				-
	FUEL COSTS OF SALES TO FKEC AND CKW	N/A				_
	ENERGY IMBALANCE SERVICE	456225				_
	OTHER MINOR ITEMS - ROUNDING	N/A				-
					<u> </u>	

Fuel Adjustments and Revenue Credits

#### FPL Cost of Service Formulas A-14(a)

•

A-14(8) Classification of fixed and variable production expenses December 31

	Description	Raference	(#) Demand	(b) Energy		(c) 	(d) Tota
	500 Operation supervision and orginouring	FERCA P. SHILA	-				
	501 Fusi 502 Stram expenses	PERC-1 p. 130 (.) PERC-1 p. 130 (.)	-		•	•	<b>s/</b>
	503 Steam from other sources	FRUC-1 p. 320 L7	•				
	504 Stram transforred-Cruitt	FIRIC-1 p. 320 1.6					
	505 Electric expenses	PSRC-1 p. 329 L 9	-				
	S06 Missellaneous steam power expenses	PERC 1 p. 318 L IN	•				
	S07 Roats	PERC-I p. 339 LI	•				
	S09 Allowances	PERC-1 p. 338 L12	-				
	510 Maintenance supervision and orgineoring 511 Maintenance of structures	PERC-1 p. 338 L11 PERC-1 p. 338 L14			•		
	512 Maintenance of boiler plant	PROLES I p. 330 L11	-				
	513 Maintenance of electric plant	PERC-1 p. 320 L11					
	514 Majamanoc of miscellaneous steam plant	FERCI & SHOLIN	-				
	517 Operation supervision and engineering	FIRC-1 p. 120 L14	•				
	51# Fuel	FBRC-L p. 376 L25			b/		-
	519 Coolants and water	FERC-1 p. 120 Las		•			
	520 Stoam expenses	FRAC-1 p. 330 1.37	•				
	521 Steam from other sources	FBRC-1 p. 320 1.31		4			
	522 Steam transferredCredit 523 Electric imponent	FBRC-1 p. 336128	-				
	524 Miscellancus nuclear power expenses	FIRE-1 p. 320 1.31					
	525 Rents	P200C-1 p. 328 L32					
	528 Maintonance supervision and ongineering	FURC-1 p. 344 LDT					
4	529 Maintenance of structures	PERC-I p. 338-L36	•				
1	530 Maintenance of reactor plant equipment	FERC-1 p. 330 L37		-			
	531 Maintonance of cloetric plant	PURC-1 p. 330 (.31		•			
	532 Maintenance of miscellancous nucluar plant	PERC-1 p. 230 134	•				
	133 Operation supervision and engineering	FIRC-Ly, THE LAN					
	536 Water for power 137 Hydraulic expenses	7580-1 p. 338 Let 7580-1 p. 338 Let					
	537 Flyeraune expenses 538 Electric expenses	788C-L ± 326 L47					
	33 Mincellancous hydraulic power generation expenses	PRINC-1 p. 338 L48					
	40 Rents	FIRE-1 p. 338 1.49					
	541 Maintonance supervision and onginvering	PERC-1 p. 338 L43					
	142 Maintonance of structures	FERC-1 p. 329 2.14					
	43 Maintenance of reservoirs, dans and waterways	FRRC-1 p. 328 1.45					
	544 Maintenance of electric plant 545 Meintenance of miscollanges hydraulis plant	#5000-1 p. 330 L46 #6000-1 p. 330 L47					
	45 Metatopapeo or miscosianapus nyarautic plant. 446 Operation supervision and engineering.	PERC-I p. 328 L57 PERC-I p. 33   L61	-	• •			
	H7 Fusi	PERC-1 p. 31 ( 43			ď		
	Hill Generation expenses	PRICH J. 333 LAH	- 1		-		
5	49 Microilancous other power generation expenses	PRIC-4 p. 301 L61	•				
5	iso Ranu	7EBC-L p. 321 L44	. •				
	<ol> <li>Maintenance supervision and engineering</li> </ol>	P\$\$\$\$254 p. 321 L49	•				
	52 Maintenance of structures	FRRC-1 p. 321 L70	•				
	53 Maintenance of generating and electric plant 54 Maintenance of mixedlanaeux other power generation plant	1920(C+1 p. 32) 1.71 FXXC-1 p. 32) 1.71	-				
	134 Maintenance of mucchandous outer power generation plant	PRIC-1 p. 321 L71 Jun Li 1-1.41		-			
		•		-			
	ca Fuel Demand/Energy Allocator Rocator used in A-7, L13 Col (d)	L-19 Col (n) (h) - % of total Col (d)	#DIV/0	#DIV/0		: -	#DIV/0
<b>1</b> 11		50/110				-	
	UEL-RECOVERABLE FUEL OIL UEL OIL RECOVERABLE ADJUSTMENTS	501110 501111					
	CREMENTAL HEDGING COSTS	501115					
	UBL-RECOVERABLE FUEL GAS	501120					
	IL, RECOV TEMPERATURE & CALIBRATION ADJ	501130					
	ECOVERABLE FUEL COAL (GENERATION)						
		501140					
SJ	RPP/SCHERER COAL CARS DEPR EXPENSE	501141					
SJ N	(RPP/SCHERER COAL CARS DEPR EXPENSE ECOVERABLE FÜEL, PETCOKE (GENERATION)	301141 301142					·
S) N R	(RPP/SCHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE PUEL COAL ADJUSTMENTS	301141 301142 301143					·
SJ N N N	(RPP/SCHERER COAL CARS DEPR EXPENSE ECOVERABLE FUEL, PETCORE (GENERATION) ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTIVES	301141 301142 301143 301143					·
sj Ri Ri Ri	IRPP/SCHERER COAL CARS DEPR EXPENSE ECOVERABLE FUEL, FETCORE (GENERATION) ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADDITIVES ECOVERABLE FUEL, DISTILLATE(GENERATION)	S01141 S01142 S01143 S01144 S01144 S01160					·
SJ N N N N N	IRPRSCHERER COAL CARS DEPR EXFENSE ECOVERABLE FUEL, PETCOKE (GENERATION) ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUTYES ECOVERABLE FUEL, DISTILLATE(GENERATION) JEL-NONRECOVERABLE FUEL, DIL	S01141 S01142 S01143 S01144 S01160 S01210					
SI RI RI RI RI RI O	IRPRSCHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE PUEL COAL ADUSTMENTS ECOVERABLE FUEL COAL ADUTIVES ECOVERABLE FUEL, DISTILLATE(GENERATION) UELNONRECOVERABLE VIEL, OIL IL, NON-RECOVERABLE TERMINAL & TRSPT EXP	S01141 S01142 S01143 S01144 S01144 S01160					
SI N RI	IRPRSCHERER COAL CARS DEPR EXFENSE ECOVERABLE FUEL, PETCOKE (GENERATION) ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUTYES ECOVERABLE FUEL, DISTILLATE(GENERATION) JEL-NONRECOVERABLE FUEL, DIL	301141 301142 301143 501143 501144 501160 501210 501230					
SHARRINAN	IRPRSCHERER COAL CARS DEPR EXPENSE ECOVERABLE FUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADUSTMENTS ECOVERABLE FUEL COAL ADUSTNERTS ECOVERABLE FUEL, DISTILLATE(GENERATION) JUELNONRECOVERABLE FUEL, OIL UL, NON-RECOVERABLE TERMINAL & TRSPT EXP OR-RECOVERABLE FUEL, COAL (ADI) SH HANDLING EXPENSE SIL-NONRECOVERABLE-NON M&S EXPENSES	501141 501142 501143 501144 501240 501210 501250 501250 501250 501270					
SHARRINAN	IRPRISCHERER COAL CARS DEPR EXFENSE ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL, COAL ADUSTMENTS JEL-NONRECOVERABLE FUEL, OIL IL, NON-RECOVERABLE FUEL, OAL (AD) SH HANDLING EXFENSE EL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES	S01141 S01142 S01143 S01144 S01160 S01210 S01230 S01230 S01230 S01240					
SHARRINAN	IRPRSCHERER COAL CARS DEPR EXPENSE ECOVERABLE FUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADUSTMENTS ECOVERABLE FUEL COAL ADUSTNERTS ECOVERABLE FUEL, DISTILLATE(GENERATION) JUELNONRECOVERABLE FUEL, OIL UL, NON-RECOVERABLE TERMINAL & TRSPT EXP OR-RECOVERABLE FUEL, COAL (ADI) SH HANDLING EXPENSE SIL-NONRECOVERABLE-NON M&S EXPENSES	501141 501142 501143 501144 501240 501210 501250 501250 501250 501270				<u> </u>	
SINNERPONATIO	IRPRSCHERER COAL CARS DEPR EXPENSE ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL COAL ADUSTMENTS ECOVERABLE FUEL COAL ADUSTMENTS ECOVERABLE FUEL COAL ADUSTMENTS ECOVERABLE FUEL, COAL (ADUSTMENT) JEL-NONRECOVERABLE FUEL, OIL IL, NON-RECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (ADI) SH HANDLING EXPENSE JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S, GAS Total AC 301	501141 501142 501143 501144 501160 501210 501230 501230 501260 501270 501271	-				
SINNER PONALLY N	IRPRISCHERER COAL CARS DEPR EXFENSE ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL, COAL ADUSTMENTS ECOVERABLE FUEL, COAL ADUSTMENTS JEL-NONRECOVERABLE FUEL, OIL IL, NON-RECOVERABLE FUEL, OAL (AD) SH HANDLING EXFENSE EL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES	501141 501142 501143 501144 501240 501210 501250 501250 501250 501270	-				
SINNER ON ANY NN	IRPRICHERER COAL CARS DEPR EXFENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL, COAL ADUISTMENTS ECOVERABLE FUEL, COAL ADUITYES ECOVERABLE FUEL, OIL (ATE) JEL-NONRECOVERABLE FUEL, OIL (ADI) SH HANDLING EXPENSE JEL-NONRECOVERABLE HOEL, COAL (ADI) SH HANDLING EXPENSE JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S GAS TOWA AC 501 UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL	501141 501142 501143 501144 501160 501210 501210 501250 501250 501270 501271					-
SHARRFORATE NEWS	IRPROCHERER COAL CARS DEPS EXPENSE ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL, COAL (ADJ) JEL-NONRECOVERABLE FUEL, OIL (L, NON-RECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXTENSE EAL-NONRECOVERABLE HON M&S EXPENSES JEL-NON-RECOVERABLE HON M&S CAS Tosal AC 301 UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV OTH ADMIN FEES-LEASED UC FUEL EXP-RECOV OTH ADMIN FEES-LEASED UC FUEL EXP-RECOV THADAIN FEES-LEASED UC FUEL EXP-RECOV THADAIN FEES-LEASED UC FUEL EXP-RECOV THADAIN FEES-LEASED	501143 501142 501163 501260 501230 501230 501230 501230 501270 501271 518110 518130 518130 518151	-				
	IRPRICHERER COAL CARS DEPR EXFENSE ECOVERABLE FUEL, COAL ADDISTMENTS ECOVERABLE FUEL COAL ADDISTMENTS ECOVERABLE FUEL COAL ADDISTMENTS ECOVERABLE FUEL, COAL ADDISTMENTS ECOVERABLE FUEL, COAL (ADDISTMENTS) JEL-NONRECOVERABLE TENINGLA TRSPT EXP OR-RECOVERABLE TENINGLA TRSPT EXP OR-RECOVERABLE FUEL, COAL (AD) SH HANDLING EXFENSE ELL-NONRECOVERABLE-HON M&S CAS TOWARCOVERABLE-HON M&S CAS LEL-NON-RECOVERABLE-HON M&S CAS TOWARCOVERABLE-HON M&S CAS TOWARCOVERABLE-HON M&S CAS TOWARCOVERABLE-HON M&S CAS TOWARCOVERABLE-HON M&S CAS LEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV OTH ADMIN PEES-LEASED UC FUEL EXP-RECOV OTH ADMIN PEES-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1	501141 501142 501143 501140 501210 501220 501220 501220 501220 501270 501270 501270 501271 518110 518120 518130 518151 518152	-				-
	IRPRSCHERER COAL CARS DEPR EXFENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL, OCAL (ADUISTMENTS) ECOVERABLE FUEL, DILLATE(GENERATION) JEL-NONRECOVERABLE FUEL, OIL (I, NON-RECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE DEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES JEL-NONRECOVERABLE-HON M&S EXPENSES UC FUEL EXP-RECOV FINANCIMO CST-LEASED UC FUEL EXP-RECOV FINANCIMO CST-LEASED UC FUEL EXP-RECOV FINANCIMO CST-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXPOSPL CST-CURR-ST LUCIE #2 UC FUEL EXPOSPL CST-CURR-TURKEY FT #3	S01143 S01142 S01143 S01144 S01240 S01210 S01220 S01220 S01220 S01220 S01270 S01271 S18110 S18120 S18151 S18152 S18152 S18152	-				-
	IRPRICHERER COAL CARS DEPS EXFENSE ECOVERABLE FUEL, COAL ADDISTMENTS ECOVERABLE FUEL COAL ADDISTMENTS ECOVERABLE FUEL COAL ADDISTMENTS ECOVERABLE FUEL, COAL ADDISTMENTS ECOVERABLE FUEL, COAL (ADDISTMENTS) UEL-NONRECOVERABLE FUEL, OIL L, NON-RECOVERABLE TELMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (AD) SH MANDLING EXFENSE GLALNONRECOVERABLE-NON MAS EXPENSES JEL-NON-RECOVERABLE-NON MAS GAS Total ACS 301 UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #2 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1	S01141 S01142 S01143 S01144 S01240 S01230 S01230 S01230 S01270 S01270 S01271 S18110 S18120 S18151 S18151 S18153 S18153					- 
	IRPRICHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE PUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL, COAL (ADUISTMENTS) JEL-NONRECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (ADI) SH HANDLING EXPENSE JEL-NONRECOVERABLE-NON M&S EXPENSES JEL-NONRECOVERABLE-NON M&S EXPENSES JEL-NORRECOVERABLE-NON M&S EXPENSES UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE # UC	S01143 S01142 S01143 S01144 S01240 S01210 S01220 S01220 S01220 S01220 S01270 S01271 S18110 S18120 S18151 S18152 S18152 S18152	-				
	IRPRICHERER COAL CARS DEPS EXFENSE ECOVERABLE FUEL, COAL ADDISTMENTS ECOVERABLE FUEL COAL ADDISTMENTS ECOVERABLE FUEL COAL ADDISTMENTS ECOVERABLE FUEL, COAL ADDISTMENTS ECOVERABLE FUEL, COAL (ADDISTMENTS) UEL-NONRECOVERABLE FUEL, OIL L, NON-RECOVERABLE TELMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (AD) SH MANDLING EXFENSE GLALNONRECOVERABLE-NON MAS EXPENSES JEL-NON-RECOVERABLE-NON MAS GAS Total ACS 301 UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #2 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1	S01141 S01142 S01142 S01144 S01100 S01210 S01220 S01220 S01270 S01270 S01271 S18110 S18120 S18132 S18151 S18154 S18154	-				-
	IRPROCHERER COAL CARS DEPS EXPENSE ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL, COAL (ADJ) JEL-NONRECOVERABLE FUEL, OIL (L, NON-RECOVERABLE TERMINAL & TRSPT EXP CN-RECOVERABLE FUEL, COAL (AD) SH HANDLING EXPENSE SIGLNONRECOVERABLE FNON M&S EXPENSES JEL-NONRECOVERABLE NON M&S CAS TOGIAC SOI UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV TINANCING CST-LEASED UC FUEL FOR FUEL TO FUEL CST-CURR-TINANCING CST-LEASED UC FUEL EXP-RECOV TINANCING CST-LEASED UC FUEL EXP-RECOV TINANCING CST-LEASED UC FUEL FOR FUEL CST-CURR-TINANCING CST-LEASED UC FUEL FOR FUEL CST-CURR-TINANCING CST-LEASED UC FUEL EXP-DSPL CST-CURR-TURKEY FI FI	S01141 S01142 S01142 S01143 S01240 S01250 S01250 S01250 S01270 S01270 S18110 S18110 S18120 S18151 S18151 S18151 S18151 S18154 S18161 S18161					- 
	IRPRICHERER COAL CARS DEPS EXFENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE PUEL, COAL ADDITTMENTS ECOVERABLE FUEL, COAL ADDITTMENTS ECOVERABLE FUEL, COAL ADDITTMES ECOVERABLE FUEL, COAL ADDITTVES ECOVERABLE FUEL, COAL (ADD) UEL-NONRECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (AD) SH MANDLING EXFENSE ISL-NONRECOVERABLE FON M&S GAS Total AC 301 UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-SPL CST-CURR-ST LUCIE #1 UC FUEL EXP-SPL CST-CURR-ST LUCIE #1 UC FUEL EXP-SPL CST-CURR-TURKEY FT #1 CC FUEL EXP-SL M-DETT OF ENERGY-CRED UC FUEL EXP-FIL #0-DETT OF ENERGY-CRED	S01141 S01142 S01143 S01143 S01240 S01230 S01230 S01230 S01220 S01270 S01271 S18110 S18120 S18151 S18153 S18154 S18164 S18164 S18165					- 
	IRPRICHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADUITMENTS ECOVERABLE FUEL COAL ADUITMENTS ECOVERABLE FUEL COAL ADUITMENTS ECOVERABLE FUEL COAL ADUITMENTS ECOVERABLE FUEL, COAL (ADD) JEL-NONRECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (ADD) SH HANDLING EXPENSE JEL-NONRECOVERABLE-MON M&S EXPENSES JEL-NONRECOVERABLE-MON M&S EXPENSES JEL-NONRECOVERABLE-MON M&S EXPENSES UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV FUNANCING CST-LEASED UC FUEL EXP-RECOV FUNANCING CST-LEASED UC FUEL EXP-RECOV FUNANCING CST-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE # UC FUEL EXP-DSPL CST-CURR-ST LUCIE # UC FUEL EXP-DSPL CST-CURR-TURKEY FT # UC FUEL EXP-DSPL CST-CURR-TURKEY FT # UC FUEL EXP-DSPL OFT OF ENERGY-CRED UC FUEL EXP-PIS # J-DEFT OF ENERGY-CRED UC FUEL EXP-D & D FUND-FERC	501143 501142 501143 501143 501240 501220 501220 501220 501270 501270 501270 501271 518120 518120 518151 518152 518154 518163 518165 518166	•				-
	IRPRICHERER COAL CARS DEPS EXPENSE ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUSTMENTS ICOVERABLE FUEL, COAL (ADJ) BLI-NONRECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE ICL-NONRECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE ICL-NONRECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE ICL-NONRECOVERABLE FON M&S GAS TOGIAC SOI UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV THANCING CST-LEASED UC FUEL EXP-DS DE CST-CURR-THANCING CST-CRED UC FUEL EXP-DS DE CST-CURR-THANCING CST-CRED UC FUEL EXP-D & D FUND-FEXC UC FUEL EXP-D & D FUND-FEXC	S01141 S01142 S01142 S01143 S01240 S01250 S01250 S01250 S01270 S01270 S01271 S18110 S18120 S18151 S18151 S18151 S18154 S18164 S18166 S18166					- 
	IRPRICHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL, COAL (ADUISTMENTS) UEL-NONRECOVERABLE TERMINAL & TRSPT EXP OR-RECOVERABLE FUEL, COAL (AD) SH HANDLING EXPENSE SIL-NONRECOVERABLE-HON MAS EXPENSES JEL-NON-RECOVERABLE-HON MAS CAS Toul AC 301 UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV OTH ADMIN PEES-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #2 UC FUEL EXP-DSPL CST-CURR-TURKEY FT M UC FUEL EXP-PSL M-DEFT OF ENERGY-CRED UC FUEL EXP-FT M-DEFT OF ENERGY-CRED	501143 501142 501143 501143 501240 501220 501220 501220 501270 501270 501270 501271 518120 518120 518151 518152 518154 518163 518165 518166		•			-
	IRPRICHERER COAL CARS DEPS EXPENSE ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUSTMENTS ICOVERABLE FUEL, COAL (ADJ) BLI-NONRECOVERABLE TERMINAL & TRSPT EXP ON-RECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE ICL-NONRECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE ICL-NONRECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE ICL-NONRECOVERABLE FON M&S GAS TOGIAC SOI UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV THANCING CST-LEASED UC FUEL EXP-DS DE CST-CURR-THANCING CST-CRED UC FUEL EXP-DS DE CST-CURR-THANCING CST-CRED UC FUEL EXP-D & D FUND-FEXC UC FUEL EXP-D & D FUND-FEXC	S01141 S01142 S01142 S01143 S01240 S01250 S01250 S01250 S01270 S01270 S01271 S18110 S18120 S18151 S18151 S18151 S18154 S18164 S18166 S18166	- -				
	IRPRICHERER COAL CARS DEPS EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADDITTMENTS ECOVERABLE FUEL COAL ADDITTMENTS ECOVERABLE FUEL COAL ADDITTYES ECOVERABLE FUEL, COAL ADDITYES ECOVERABLE FUEL, COAL (AD) UEL-NONRECOVERABLE FUEL, OIL L, NON-RECOVERABLE FUEL, OIL & NON-RECOVERABLE FON MAS, GAS Total AC 301 UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXPRECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING FINANCING SS C FUEL EXP-RECOV FINANCING FINANCING SS C FUEL EXP-RECOV FINANCING FINANCING SS C FUEL EXP-RECOVERABLE ADUSTMENTS JCLEAR FLANTS RECOVERABLE ADUSTMENTS JCLEAR FLANTS RECOVERABLE ADUSTMENTS JCLEAR FUEL EXPENSE-LAST CORE TOULACC SIR	S01141 S01142 S01142 S01143 S01240 S01230 S01230 S01230 S01270 S01271 S18110 S18120 S18151 S18153 S18153 S18154 S18155 S18154 S18165 S18165 S18165 S18165 S18165 S18165 S18204					
	IRPRICHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL, COAL (ADU) JEL-NONRECOVERABLE TERMINAL & TRSPT EXP OR-RECOVERABLE FUEL, COAL (ADU) SH HANDLING EXTENSE JEL-NON-RECOVERABLE-HON M&S EXPENSES JEL-NON-RECOVERABLE-HON M&S EXPENSES JEL-NON-RECOVERABLE-HON M&S EXPENSES JEL-NON-RECOVERABLE-HON M&S EXPENSES UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV THA DAMIN PEES-LEASED UC FUEL EXP-RECOV THA DAMIN PEES-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-TTURKEY FT #1 UC FUEL EXP-PSL #1-DEPT OF ENERCY-CRED UC FUEL EXP-PSL #1-DEPT OF ENERCY-CRED UC FUEL EXP-PSL #1-DEPT OF ENERCY-CRED UC FUEL EXP-TF #1-DEPT OF ENERCY-CRED UC	S01143 S01142 S01143 S01143 S01140 S01210 S01220 S01220 S01220 S01270 S01270 S01270 S01271 S18120 S18120 S18120 S18153 S18154 S18164 S18165 S18166 S18166 S18166 S18166 S18166 S18166 S18166					
	IRPRICHERER COAL CARS DEPS EXPENSE ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUSTMENTS ECOVERABLE FUEL, COAL ADJUSTMENTS ICOVERABLE FUEL, COAL (ADJ) MELNONRECOVERABLE FUEL, OIL L, NON-RECOVERABLE FUEL, OIL L, NON-RECOVERABLE FUEL, OIL SEL-NONRECOVERABLE FUEL, OIL SEL-NONRECOVERABLE FUEL, OIL SEL-NONRECOVERABLE FUEL, OIL CON-RECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXPENSE SEL-NONRECOVERABLE-NON MAS GAS TOGIAC SOI UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL LUC FUEL EXP-RECOV FINANCING CST-LEASED FUEL EXPRESOV FINANCING CST-LEASED FUEL EXP-DSPL CST-CURR-ST LUCIE #1 FUEL EXP-DSPL CST-CURR-ST LUCIE #1 FUEL EXP-DSPL CST-CURR-TURKEY FT #3 FUE FUEL EXP-DSPL CST-CURR-TURKEY FT #3 FUE FUEL EXP-FIS IN-DEFT OF ENERGY-CRED FUEL EXP-FIS INFORMALE ADUSTMENTS FUEL EXP-FIS IN-DEFT OF ENERGY-CRED FUEL EXP-FIS IN-DEFT OF FUEL FUEL EXP-FIS IN-DEFT OF ENERGY-CRED FUEL EXP-FIS IN-DEFT OF FUEL FUEL FUEL FIS FIS FUEL FUEL FUEL FIS FIS FUEL FUEL FUEL FUEL FIS FIS FUEL FUEL FUEL FUEL FUEL FIS FUEL FIS FUEL FUEL FUEL FUEL FUEL FIS FUEL FIS FUEL FUEL FUEL FUEL FUEL FIS FUEL FUEL FUEL FUEL FUEL F	S01141 S01142 S01142 S01143 S01240 S01230 S01230 S01230 S01270 S01271 S18110 S18120 S18151 S18151 S18151 S18151 S18153 S18154 S18155 S18154 S18165 S18166 S18166 S18166 S18166 S18166 S18166 S18166 S18166 S18166 S18166 S18166	- - - -				
	IRPRICHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL COAL ADUISTMENTS ECOVERABLE FUEL, COAL (ADU) JEL-NONRECOVERABLE TERMINAL & TRSPT EXP OR-RECOVERABLE FUEL, COAL (ADU) SH HANDLING EXTENSE JEL-NON-RECOVERABLE-HON M&S EXPENSES JEL-NON-RECOVERABLE-HON M&S EXPENSES JEL-NON-RECOVERABLE-HON M&S EXPENSES JEL-NON-RECOVERABLE-HON M&S EXPENSES UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV BURNUT CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-RECOV THA DAMIN PEES-LEASED UC FUEL EXP-RECOV THA DAMIN PEES-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-TTURKEY FT #1 UC FUEL EXP-PSL #1-DEPT OF ENERCY-CRED UC FUEL EXP-PSL #1-DEPT OF ENERCY-CRED UC FUEL EXP-PSL #1-DEPT OF ENERCY-CRED UC FUEL EXP-TF #1-DEPT OF ENERCY-CRED UC	S01143 S01142 S01143 S01143 S01140 S01210 S01220 S01220 S01220 S01270 S01270 S01270 S01271 S18120 S18120 S18120 S18153 S18154 S18164 S18165 S18166 S18166 S18166 S18166 S18166 S18166 S18166	-				
	IRPRICHERER COAL CARS DEPR EXPENSE ECOVERABLE PUEL, PETCOKE (GENERATION) ECOVERABLE FUEL COAL ADDITTMENTS ECOVERABLE FUEL COAL ADDITTYES ECOVERABLE FUEL COAL ADDITYES ECOVERABLE FUEL COAL ADDITYES ECOVERABLE FUEL, COAL (ADD) UEL-NONRECOVERABLE FUEL, OIL L, NON-RECOVERABLE FUEL, OIL L, NON-RECOVERABLE FUEL, OIL SCH-RECOVERABLE FUEL, COAL (AD) SH HANDLING EXFENSE SIL-NON-RECOVERABLE FOR MAS, GAS TGAL NON-RECOVERABLE FOR MAS, GAS TGAL NON-RECOVERABLE-NON MAS, GAS TGAL NON-RECOVERABLE-NON MAS, GAS TGAL NON-RECOVERABLE-NON MAS, GAS UC FUEL EXP-RECOV FINANCING CST-LEASED UC FUEL EXP-DSPL CST-CURR-ST LUCIE #1 UC FUEL EXP-DSPL CST-CURR-ST LUCIE #2 CC FUEL EXP-DSPL CST-CURR-ST LUCIE S2 CC FUEL EXP-DSPL S3 CST CST S3 CST S3 CS	S01141 S01142 S01143 S01143 S01240 S01230 S01230 S01230 S01230 S01270 S01270 S1220 S01271 S12120 S1210 S12120 S12120 S12120 S12120 S12120 S12120 S12120 S12120 S12120 S12120 S12120 S12155	- -				
	IRPRISCHERER COAL CARS DEPS EXPENSE ECOVERABLE FUEL COAL ADJUSTMENTS ECOVERABLE FUEL, COAL (ADJ) BELNONRECOVERABLE FUEL, OLL IL, NON-RECOVERABLE FUEL, COAL (ADJ) SH HANDLING EXTENSE ELELNONRECOVERABLE HOEL, COAL (ADJ) SH HANDLING EXTENSE ELELNONRECOVERABLE HOEL, COAL (ADJ) SH HANDLING EXTENSE IEL-NONRECOVERABLE HON M&S EXPENSES IEL-NONRECOVERABLE HON M&S CAS Tosl AC 501 UC FUEL EXP-RECOV BURNUP CHG-LEASD FUEL UC FUEL EXP-RECOV FINANCING CST-LEASED DUC FUEL EXP-RECOV FINANCING CST-LEASED DUC FUEL EXP-RECOV OTH ADMIN PEES-LEASED UC FUEL EXP-RECOV OTH ADMIN PEES-LEASED DUC FUEL EXP-RECOV OTH ADMIN PEES-LEASED DUC FUEL EXP-DSPL CST-CURR-ST LUCIE # UC FUEL EXP-PSL # JDEPT OF ENERGY-CRED DUC FUEL EXP-PT # JDEPT OF ENERGY-CRED DUC FUEL EXP-RECOVER	S01143 S01143 S01143 S01144 S01140 S01210 S01220 S01220 S01270 S01270 S01271 S18120 S18120 S18151 S18153 S18154 S18154 S18154 S18155 S18154 S18165 S18175 S1	- - -				

R,

FPL Cost of Service Formulas A-15 Production related depreciation expense December 31

				Production		
			Payroll	Related	Demand	Energy
Line	Description	Reference	Allocator	(1)	(2)	(3)
1	Steam production	FERC-1 p. 336 L2 Col (f)		-	-	
2	Nuclear production	FERC-1 p. 336 L3 Col (f)		-	-	
3	Add: ARO decommissioning and dismantling credit	A/C 403 [71, 403333			-	
4	Hydro production					
5	Conventional	FERC-1 p. 336 L4 Cal (f)				
6	Pumped storage	FERC-1 p. 336 L5 Col (f)				
7	Other production	FERC-1 p. 336 L6 Col (f)		-	-	
8	Subtotal	Sum L1 - 1.7	-	-	-	-
9	Scherer Acquisition Adjustment	A/C 406000			-	
10	Production related G&I plant	FERC-1 p. 336 L1 & L10	#DIV/0!	#DIV/0!	#DIV/0!	#DIV/01
11	Less: ARO related plant	FERC-1 p. 336 L1 Col (c)	#DIV/0!	#DIV/0!	#DIV/0!	
12	GSU-related depreciation expense	GSU Schedule		-	-	
13	Amortization for FERC CWIP treatment	CWIP Adjustment Schedule				
14	Total production	Sum L8 thru L13	-	#DIV/0!	#DIV/0!	#DIV/0!

FPL Cost of Service Formulas A-16 Production related taxes other than income taxes December 31

			System	Allocator	Production
Line	Description	Reference	(1)	(2)	(3)
1	Unemployment	FERC-1 p. 263 L10,11, 12, 20, 21 Col (i)			· · · · · · · · · · · · · · · · · · ·
2	FICA	FERC-1 p. 263 L6, 7 Col (i)	-		
3	Total taxes related to wages & salaries	\$UM L1 + L2	-	#DIV/0!	#DIV/0!
4	Real and personal property taxes	FERC-1 p. 263.1 L17 to L18 Col (i)	-		
5	Use tax	FERC-1 p. 263 L32 Col (i)	-		
6	Total taxes related to property	SUM 14 + L5	-	#DIV/0!	#DIV/0!
7	Total taxes other than income taxes	SUM L3 + L6	-		
8	Franchise tax	FERC-1 p. 263.1 L3, 6, 7 Col (i)	-		
9	Gross receipts	FERC-1 p. 263 L23, 24 Col (i)	-		
10	FPSC Fee	FERC-I p. 263 L29, 30 Col (i)	-		
11	Intangible Tax	FERC-1 p. 263 L34 Col (i)	· -		
12	Occupational License	FERC-1 p. 263.1 L8 Col (i)	-	#DIV/0!	#DIV/0!
13	Other Taxes	a/	•		
14	Total Taxes Other than Income Taxes	Sum L7 - L13	- 5	Sum L3, L6, L12, L13	#DIV/01

(1) System - FERC-1 p. 263-263.1 Col (i)

(2) Allocators:

.

L3 and L12 - Payroll Allocator, A-7, L13

L6 - Gross Plant Allocator, A-6, L10

a/ Includes any other Taxes, as defined in Article 1 of the Agreement. Production allocator based on how the tax is attributable or related to the production and/or provision of capacity, Energy and Generation-Related Services by Company to Company's retail and wholesale customers and Full Requirements Electric Service to Customer.

FPL Cost of Service Formulas A-17 Production Related Income Tax December 31

Line	Description	Reference	Total (1)	Demand (2)	Energy (3)
1	Total Investment	A-5, L17	#DIV/0!	#DIV/0!	#DIV/0!
2	Weighted return on equity	A-11, L4 Col (4)	#DIV/0!	#DIV/0!	#DIV/0!
3	Equity earnings		#DIV/0!	#DIV/0!	#DIV/0!
4	Combined income tax factor	A-18, L13	#DIV/0!	#DIV/0!	#DIV/0!
5	Income tax	L3 X L4	#DIV/0!	#DIV/0!	#DIV/0!

FPL Cost of Service Formulas A-18 Computation of effective income tax rate December 31

Line	Description	Reference	FPL Amount
1	Net Utility Operating Income	FERC-1 p. 117 L27 Col (c)	-
2	Less: Net interest charges	FERC-1 p. 117 L70 Col (c)	-
3	Adjusted net income	L1 - L2	~
4	Income taxes:		
5	Federal	FERC-1 p. 114 L15 Col (c)	-
6	Other	FERC-1 p. 114 L16 Col (c)	-
7	Provision for deferred income taxes	FERC-1 p. 114 L17 Col (c)	-
8	Provision for deferred income taxes-Cr	FERC-1 p. 114 L18 Col (c)	-
9	ITC adjustment - net	FERC-1 p. 114 L19 Col (c)	-
10	Total income taxes	L5 thru L9	
11	Pretax earnings base	L3 + L10	-
12	Effective income tax rate	L10/L11	#DIV/0!
13	Combined income tax factor	L12/(1 - L12) ==	#DIV/0!

### FPL December 31 FERC Form No. 1 Comparative Balance Sheet Page 110, 111, 112, 113, 118 and 119

rage			1	a street to be an an and the start of
Sin - Sinnin				
ណូ រត្តខេត្ត				
1	UTILITY PLANT			
2	Utility Plant (101-106, 114)			
3	Construction Work in Progress (107)		:	
4	TOTAL Utility Plant (Enter Total of lines 2 and 3)		-	-
5	(Less) Accum. Prov. for Depr. Amort. Depl. (108, 110, 111, 115)			
6	Net Utility Plant (Enter Total of line 4 less 5)		-	-
7	Nuclear Fuel In Process of Ref (120.1)			· · · · · · · · · · · · · · · · · · ·
8	Nuclear Fuel Materials and Assemblies-Stock Acct (120.2)	n alasta an sanaratka marang sa siyang		· ····································
9	Nuclear Fuels Assemblies in Reactor (120.3)			-
10	Spent Nuclear Fuel (120.4)	***	***	
.11	Nuclear Fuel Under Capital Leases (120.6)			
12	(Less) Accum. Prov. For Amort. Of Nucl. Fuel Assemblies (120.5)			
13	Net Nuclear Fuel (Total of lines 7 - 11 less 12)			
14	Net Utility Plant (Enter Totel of lines 6 and 13)		-	
15	Utility Plant Adjustments (116)			-
16	Gas Stored Underground - Noncurrent (117)			
17	OTHER PROPERTY AND INVESTMENTS			
18	(Nonutility Property (121)	"- Indexe de la composition de la compo		
19	(Less) Accum. Prov. for Depr. and Amort. (122)			
20	Investments in Associated Companies (123)	<u>مەر مەرە</u> بەرەر		
21	Investment in Subsidiary Companies (123.1)	****	,	
22	(For Cost of Account 123.1)			
23	Noncurrent Portion Allowances			
24	Other Investments (124)		، الم	······································
25	Sinking Funds (125)	-{		
26	Depreciation Fund (126)	1971 - 1996 - 1993 - 1996 - 1997 - 19		
27	Amortization Fund - Federal (127)			
28	Other Special Funds (128)			-
29	Special Funds (Non Major Only) (129)	**************************************		-
30	Long-Term Portion of Derivative Assets (175)			
31	Long-Term Portion of Derivative Assets- Hedges (176)	······································		
32	TOTAL Other Property and Investments (Lines 18-21 and 23-31)			
33	CURRENT AND ACCRUED ASSETS			
34	Cash and Working Funds (Non-Major Only) (130)		araa ahaa maxaa ahaadaadaa	
35	(Cash (131)	* ************************************		
36	Special Deposits (132-134)			
37	Working Fund (135)			
38	Temporary Cash Investments (136)			
39	Notes Receivable (141)			-
40	Customer Accounts Receivable (142)			jj
41	Other Accounts Receivable (143)			
42	(Less) Accum. Prov. for Uncollectible AcctCredit (144)			······
43	Notes Receivable form Assoclated Companies (146)			
44	Accounts Receivable from Assoc. Companies (146)	annen market i nir Versensene er mensen en er I		الله ، « بر بر بر بر بر ۲۰ هم محمد المحمد المحمد الله :
45	Fuel Stock (151)			ן, איז ווייאייי-איגעער איז אייייאייייש און ן- ן
46	Fuel Stock Expenses Undistributed (152)			
47	Residuals (Elec) and Extracted Products (153)			•
48	Plant Materials and Operating Supplies (154)	1		-
49	Merchandise (155)			=
50	Other Materials and Supplies (156)		······································	-
	ali ka manana ka manana manana manana manana manana ka ka ka manana ka manana ka manana ka manana ka manana ka		an and an and the second s	survey and a survey provide the statement of

FPL December 31

### FERC Form No. 1 Comparative Balance Sheet

Combai	ativ	C Datation	Blicer	
Page 11	0.1	11, 112, 1	13, 118 and 1	19

52				
	Nuclear Materials Held For Sale (157)			
	Aliowances (158.1 and 158.2)			-
53	(Less) Noncurrent Portion of Allowances			-
54	Stores Expense Undistributed (163)			-
55	Gas Stored Underground - Current (164.1)		anna anggina mangka kanan mengenangkakan dalam belampangkanan	
56	Liquefied Natural Gas Stored and Held for Processing (164.2-164.3)			-
57	Prepayments (165)			-
58	Allowances for Gas (166-167)		-	
59	Interest and Dividends Receivable (171)	and a shareh out other has a management of a special of		14. La 19.0 1, 19
60	Rents Receivable (172)			
61	Accrued Utility Revenues (173)			-
62	Miscellaneous Current and Accrued Assets (174)			
63	Derivative Instrument Assets (175)		······································	•
64	(Less) Long-Term Portion of Derivative Instrument Assets (175)		· · · · · · · · · · · · · · · · · · ·	-
a company of the second second	Derivative Instrument Assets - Hedges (176)		,	
66	(Less) Long-Term Portion of Derivative Instrument Assets - Hedges (176)			-
67	Total Current and Accrued Assets (Lines 34 through 66)	-		_
68	DEFERRED DEBITS			
	Unamortized Debt Expenses (181)			
	Extraordinary Property Losses (182.1)		······	a a second a la sub la sun restance dans la desanara de de
71	Unrecovered Plant and Regulatory Study Costs (182.2)			
	Other Regulatory Assets (182.3)			
73	Prelim, Survey and Investigation Charges (Electric) (183)			-
	Preliminary Natural gas Survey and Investigation Charges (182.1)	1		
75	Other Preliminary Survey and Investigation Charges (183.2)			
76	Clearing Accounts (184)		To Medicine and March 2 a rate of Support programming	
77	Temporary Facilities (185)			
78	Miscellaneous Deferred Debits (186)			-
79	Def. Losses from Disposition of Utility Plant (187)	(		and the second
80	Research, Development and Demonstration Expend. (188)			-
81	Unamortized Loss on Required Debt (189)		]	ی. ایر
	Accumulated Deferred Income Taxes (190)			لي معد الله ( ۲۰۱۰ معد ۲۰۱۰ معد
82 83	Unrecovered Purchased Gas Costs (191)		······································	-
82 83				

### FPL December 31

FERC Form No. 1

# Comparative Balance Sheet Page 110, 111, 112, 113, 118 and 119

	10, 111, 112, 113, 118 and 119		
<u>Inumpe</u>			
1			
2	Common Stock issued (201)		
3	Preferred Stock Issued (204)		
4	Capital Stock Subscribed (202, 205)		
5	Stock Liability for Conversion (203, 206)		
6	Premium on Capital Stock (207)		
7	Other Pald-in-Capital (208-211)		
8	Installments Received on Capital Stock (212)		*
9	(Less) Discount on Capital Stock (213)		
10	(Less) Capital Stock Expense (214)		
11	Retained Earnings (215, 215.1, 216)		
12	Unappropriated Undistributed Subsidiary Earnings (216.1)		
13	(Less) Required Capital Stock (217)		الم
14	No corporate Proprietorship (Non-Major only) (218)		
15	Accumulated Other Comprehensive Income (219)		
16	Total Proprietary Capital (lines 2 through 15)	-	
17	LONG TERM DEBT		合于16世纪的中国19世纪
18	Bonds (221)		
19	(Less) Required Bonds (222)		annan kanan kana kana kana kana sa ang sa
20	Advances from Associated Companies (223)		
21	Other Long-Term Debt (224)		
22	Unamortized Premium on Long Term debt (225)		
23	(Less) Unamortized Discount on Long-Term Debt-Debit (226)	D 1979 (heliochina de celor C bhadaiste annes que sur l'ann Annair al Cais e a d'agus	
24	Total Long-Term Debt (lines 18 through 23)		
25	OTHER NONCURRENT LIABILITIES		
26	Obligations Under Capital Leases - Noncurrent (227)		
27	Accumulated Provision for Property Insurance (228.1)	· · · · · · · · · · · · · · · · · · ·	
28	Accumulated Provision for Injuries and Damages (228.2)		
29			
	Accumulated Provision for Pensions and Benefits (228.3)		
30	Accumulated Provision for Pensions and Benefits (228.3) Accumulated Miscellaneous Operating Provisions (228.4)		
30 31		· · · · · · · · · · · · · · · · · · ·	
	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229)		
31	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities		
31 32	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229)		
31 32 33	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Refirement Obligations (230)		
31 32 33 34	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges		
31 32 33 34 35	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES		
31 32 33 34 35 36	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34)		
31 32 33 34 35 36 37	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231)		
31 32 33 34 35 36 37 38	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234)		
31 32 33 34 35 36 37 38 39	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232)		
31 32 33 34 35 36 37 38 39 40 41	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Refirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235)		
31 32 33 34 35 36 37 38 39 40 41 42	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236)		
31 32 33 34 35 36 37 38 39 40 41 42 43	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Refirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237)		
31 32 33 34 35 36 37 38 39 40 41 42 43 44	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Refirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237) Dividends Declared (238)		
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237) Dividends Declared (238) Matured Long Term Debt		
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237) Dividends Declared (238) Matured Long Term Debt Matured Interest		
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237) Dividends Declared (238) Matured Long Term Debt Matured Interest Tax Collections Payable (241)		
31           32           33           34           35           36           37           38           39           40           41           42           43           44           45           46           47           48	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Refirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237) Dividends Declared (238) Matured Long Term Debt Matured Interest Tax Collections Payable (241) Miscellaneous Current and Accrued Liabilities (242)		
31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49	Accumulated Miscellaneous Operating Provisions (228.4) Accumulated Provisions for Rate Refunds (229) Long-Term Portion of Derivative Instrument Liabilities Long-Term Portion of Derivative Instrument Liabilities - Hedges Asset Retirement Obligations (230) Total Other Noncurrent Liabilities (lines 26 through 34) CURRENT AND ACCRUED LIABILITIES Notes Payable (231) Accounts Payable (232) Notes Payable to Associated Companies (234) Accounts Payable to Associated Companies (234) Customer Deposits (235) Taxes Accrued (236) Interest Accrued (237) Dividends Declared (238) Matured Long Term Debt Matured Interest Tax Collections Payable (241)		

		· 		
	2		· ·	
	December 31			
· · ·	C Form No. 1		·	
	parative Balance Sheet	e fa se a se	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	
	110, 111, 112, 113, 118 and 119			
51	(Less) Long-Term Portion of Derivative Instrument Liabilities			
52	Derivative Instrument Liabilities - Hedges (245)			
53	(Less) Long-Term Portion of Derivative Instrument Liabilities-Hedges			<u> </u>
54	Total Current and Accrued Liabilities (lines 37 through 53)			-
55	DEFERRED CREDITS			
56	Customer Advances for Construction (252)			
57	Accumulated Deferred Investment Tax Credits (255)	· · · · · · · · · · · · · · · · · · ·		
58	Deferred Gains from Disposition of Utility Plant (256)	·		
59	Other Deferred Credits (253)			
60	Other Regulatory Liabilities (254)	1		
61	Unamortized Gain on Required Debt (257)			
62	Accum, Deferred income Taxes-Accel, Amort, (281)	1		
63	Accum, Deferred Income Taxes-Other Property (282)			
64	Accum. Deferred Income Taxes-Other (283)			
65	Total Deferred Credits (lines 56 through 64)		-	-
.66	TOTAL LIAB AND STOCKHOLDER EQUITY (lines 16, 24, 35, 54 and 65	)	-	
<u>19009</u>	UNAPPROPRIATED RETAINED EARNINGS (216) Balance-Beginning of Period			
2	Changes			
3	Adjustments to Retained Earnings	······································	1	
4	207			
9	Total Credits to Retained Earnings			
10	214		· +	
11	242			<u></u>
15	Total Debits to Retained Earnings			<u>.</u>
16	Balance Transferred from Income (Account 433 less Account 418.1)		· {	
17	APPROPRIATIONS OF RETAINED EARNINGS (436)			
18	242			
22	Total Appropriations of Retained Earnings (436)		l	i
23	Dividends Declared-Preferred Stock (437)			
24	125			
25	238	1		
29	Total Dividends Declared-Preferred Stock (437)		i	
30	Dividends Declared-Common Stock (438)	TRANSFER OF THE AVE		e : linda da de la como a como de la como de
30	238			
36	TOTAL Dividends Declared-Common Stock (Acct. 438)		<u></u>	· · · · · · · · · · · · · · · · · · ·
37 38	Transfers from Acct. 216.1 Unapprop. Undistrib. Subsidiary Earnings Balance - End of Period (Total 1,9,15,16,22,29,36,37)	<u></u>		
48		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	
40	TOTAL Retained Earnings (Acct. 215, 215.1, 216) (Total 38, 47) (216.1)	- <u>-</u>	-	

# FPL December 31 FERC Form No. 1 Statement of Income Page 114, 115 and 117

- 3

						11.3
1	Utility Operating Income	Particular and the second second				
	Operating Revenues (400) Operating Expenses					
4	Operation Expenses (401)					
5	Maintenance Expenses (402)					
8	Depreciation Expense (403)					ļ.,
7	Deprediction Expense for Asset Retirement Costs (403.1)				!	
	Amort, & Depi, of Utility Plant (404-405) Amort, of Utility Plant Acq, Adj. (406)		┈┶╌╁╾┥╌╾╍╌╴╸┥┈		┉	
	Amort Property Losses (407)	, fan 1. a an an an an 1. 1. fan analar anna in sind			┈┼╍┤	
In a sum of sums and a	Amort of Conversion Expenses (407)	The property of second data in the second data and the second data				****
	Regulatory Debits (407.3)	and the west was marked at the second of the second s				
13	(Less) Regulatory Credits (407.4)					ļ
	Taxes Other Than Income Taxes (408.1) Income Taxes - Federal (409.1)		╶┼╌┥╍╋╴┉╍╍┉╺┉╸╍┥╴		÷	Ļ.,.
16	- Other (409.1)		╶╄╾╌┾╴┍╉┉┉╌╌╴╌╌┉╖╌╌┉╉┉		-+	
	Provision for Deferred Income Taxes (410.1)				÷	
	(Less) Provision for Deferred Income Taxes-Cr. (411.1)				$\pm 1$	
	Investment Tax Credit Adj Net (411.4)					
	(Less) Gains from Disp. of Utility Plant (411.6) Lossea from Disp. of Utility Plant (411.7)		. <u>↓</u> ↓,	··· • • • • • • • • • • • • • • • • • •	44	
	(Less) Gains from Disposition of Allowances (411.8)			·····		
	Losaes from Disposition of Allowances (411.9)	· · · · · · · · · · · · · · · · · · ·	·/		┿	
24	Accretion Expense (411.10)				11	hii
	TOTAL Utility Operating Expenses (Enter Total of lines 4 thru 24)	J	-		Tİ	
and the second s	Net Util Oper Inc (Enter Tol line 2 less 25) Carry to Pg117,line 27		1.1.1		1	
	Other Income and Deductions					
	Other Income Nanutility Operating Income					÷
	Revenues From Merchandising, Jobbing and Contract Work (415)		N.			
A PROPERTY AND ADDRESS OF	(Less) Costs and Exp. of Merchandising, Job. & Contract Work (416)		-			
	Revenues From Norufilky Operations (417)					
	(Less) Expenses of Nonutliky Operations (417,1)		]			
	Nonopenating Rental Income (416) Equity in Earnings of Subaidiary Companies (418.1)		-1			
	Equity in Earnings of Schedury Companies (410.1)		-			
	Allowance for Other Funds Used During Construction (419.1)	· · · · · · · · · · · · · · · · · · ·	-1			
	Miscelleneous Nonoperating income (421)					
	Gein on Disposition of Property (421.1)					
	TOTAL Other Income (Enter Total of lines 31 thru 40) Other Income Deductions					
	Loss on Disposition of Property (421.2)	的分子的一个分化。如此这些问题的问题是是	iii			
	Miscellaneous Amortization (425)		1			
45	Donations (426.1)	an an ann ann an ann an ann an ann ann	1			
	Life insurance (426.2)	• • •				
47 48	Penatias (426,3) Exp. for Certain Civic, Political & Related Activities (426,4)					
49	Other Deductions (426.5)			· .		
50	TOTAL Other Income Deductions (Total of lines 43 Ibru 49)		4			
	Texes Applic, To Other Income and Deductions					
	Texes Other Then Income Taxes (408.2)		i J			
	Income Taxes-Federal (409.2)		ł			
	Income Taxes-Other (409.2) Provision for Deferred Inc. Taxes (410.2)		1	-		
presenters a stre	Less) Provision for Deferred Income Taxes - Cr (411.2)		1			
	nvestment Tax Credit AdJ-Nei (411.5)					
	Lees) Investment Tax Credits (420)					
	IOTAL Taxes on Other income and Deductions (Total of lines 52-58)	······································				
	Vet Other Income and Deductions (Total of lines 41, 50, 59) Interest Changes		·			
	nterest on Long-Term Debt (427)					
	Amort, of Debi Disc. and Expense (428)					
	Amortization of Loss on Required Debi (428.1)					
. <u> </u>	Less) Amort. of Premium on Debt-Credit (429)		1			
65 !(	Land Americation of Only on Description October October 19					
65 (( 66 ((	Less) Amortization of Gain on Required Debt-Credit (429.1)		- '			
65 (1 66 (1 67 1	nteresi on Debi to Assoc. Companies (430)	ant provide the mean of the second				
65 (1 66 (1 67 ) 68   69	nterest on Debi to Assoc. Companies (430) Diher Interest Expense (431) Less) Allowance for Borrowed Funds Used During Construction-Cr. (432)		· .			
65 (( 66 (( 67 ) 68 C 69 (( 70 )	nterest on Debi to Assoc. Companies (430) Dither Interest Expense (431) Less) Allowance for Borrowed Funds Used During Construction-Cr. (432) Het Interest Charges (Total of Ines 62 thru 69)					
65 (1 66 (1 67  r 68  C 69 (1 70  N 71  r	nterest on Debi to Assoc. Companies (430) Ther Interest Expense (431) Less) Allowance for Borrowed Funds Used During Construction-Cr. (432) let Interest Charges (Total of Inse 62 thru 69) ncome Before Extraordinary liams (Total of Inse 27, 60 and 70)					
65 (1 66 (1 67 ir 68 C 59 (1 70 N 71 ir 72 E	nterest on Debi to Assoc. Companies (430) Ther Interest Expense (431) Less ) Allowance for Borrowed Funds Used During Construction-Cr. (432) He Interest Charges (Total of Intel 62 thru 63) ncome Before Extraordinary flams (Total of Intel 27, 60 and 70) Extraordinary Items					
65 (1 66 (1 67 1 68 C 69 (1 70 N 71 1 72 E 73 E	nterest on Debi to Assoc. Companies (430) Ther Interest Expense (431) Less ) Allowance for Borowed Funds Used During Construction-Cr. (432) Net Interest Charges (Total of Ines 62 Intu 69) nonne Before Extraordinary (Iems (Total of Ines 27, 60 and 70) ixtnaordinary Isens ixtraordinary Income (434)					
65 (( 66 (( 67 )) 68 C 69 (( 70 )) 71 ) 71 ) 72 E 73 E 73 E 74 ((L 75 ))	nterest on Debi to Assoc. Companies (430) 2ther Interest Expense (431) Less) Allowance for Borrowed Funds Used During Construction-Cr. (432) 4et Interest Charges (Total of Inte 62 thru 69) ncome Before Extraordinary Items (Total of Intes 27, 60 and 70) Extraordinary Income (434) Less) Extraordinary Items (435) 4et Extraordinary Items (Total of Inte 73 less Inte 74)					
65 (( 66 (( 67 ) ) ) 68 C 69 (( 70 ) N 71 ) ) 72 E 73 : E 73 : E 74 (( 175 ) N 76 )   n	nterest on Debi to Assoc. Companies (430) Ther Interest Expense (431) Less) Allowance for Borrowed Funds Used During Construction-Cr. (432) Net Interest Charges (Total of lines 52 hhu 89) Income Before Extraordinary Items (Total of lines 27, 60 and 70) Extraordinary Income (434) Less) Extraordinary Deductions (435)					

# FPL December 31

# FERC Form No. 1

# Statement of Cash Flows

# Page 121

	e de la companya de la contra de la companya de la La companya de la comp	nir, <u>auten (Hur), Helev</u> e III. III. energen (Hur), (Hyver)	un (extensi penter Deferonen en Vicen In Thistophysics
46	Loans Made or Purchased		
47	Collections on Loans		
48		·	
49	Net (Increase) Decrease in Receivables		
50	Net (Increase) Decrease in Inventory		
51	Net (Increase) Decrease in Allowances Held for Speculation		
52	Net Increase (Decrease) in Payables and Accrued Expenses		
53	Other (provide details in footnote):		
54			
55			Autors and a second second second second
56	Net Cash Provided by (Used in) Investing Activities		
.57	Total of lines 34 thru 55		
58			
59	Cash Flows from Financing Activities		
	n and the second s	-	
	Not Increase (Decrease) is Short-Term Daht ( a)		
		· · · · · · · · · · · · · · · · · · ·	
			j
-	Cash Provided by Outside Sources (Total 61 thru 69)		-
71			
72	Payments for Retirement of:		
	Long-Term Debt (b)	A STREAMENT COMPANY OF THE REAL OF THE	
74	Preferred Stock		
75	Common Stock		
76	Other (provide details in footnote):		
77			
j manuscription of	Net Decrease in Short-Term Debt ( c)		
79			
1 1			
harris and a second sec			
	( I OTAI OF IINES / U TN/U 81)		- 
for the second s	Not Increase (Decrease) in Cash and Cash Faulyalente		
à			
1	Cash and Cash Equivalents at Beginning of Period		
89			
t	Cash and Cash Equivalents at End of Period	996-216 EU 79-11-246-21699-5599-569-569-569-569-5 	
62           63           64           65           66           67           68           69           70           71           72           73           74           75           76           77           78           79           80           81           82           83           84           85           86           87           88           89	Process from Issuance of: Long-Term Debt (b) Preferred Stock Common Stock Other (provide details in footnote): Net Increase (Decrease) in Short-Term Debt ( c) Other (provide details in footnote): Cash Provided by Outside Sources (Total 61 thru 69) Payments for Retirement of: Long-Term Debt (b) Preferred Stock Common Stock Other (provide details in footnote): Net Decrease in Short-Term Debt ( c) Dividends on Preferred Stock Dividends on Preferred Stock Net Cash Provided by (Used in) Financing Activities (Total of lines 70 thru 81) Net Increase (Decrease) in Cash and Cash Equivalents (Total of lines 22, 57 and 83) Cash and Cash Equivalents at Beginning of Period		

FPL December 31 FERC Form No. 1 Electric Plant In Service Page 204, 205, 206 and 207

rage 2	04, 205, 206 and 207				
្រីវេណ្ឌីស្ទ					
1	1. Intangible Plant				
2	(301) Organization			301	#DIV/0!
3	(302) Franchises and Consents				#DIV/0!
4	(303) Miscellaneous Intangible Plant			303	#DIV/0!
5	TOTAL Intangible Plant (lines 2, 3, and 4)			- OTA	
6	2. Production Plant				
7	A. Steam Production Plant				
8	(310) Land and Land Rights	:	·	310	#DIV/0!
9	(311) Structures and Improvements	and a state of the		311	#DIV/0!
10	(312) Boiler Plant Equipment			312	#DIV/0!
11	(313) Engines and Engine Driven Generators				#DIV/0!
12	(314) Turbogenerator Units			314	#DIV/0!
13	(315) Accessory Electric Equipment	ar angele ar physics a parameter an angeletiker ar dar strangent is		315	#DIV/0!
14	(316) Misc. Power Plant Equipment	-		316	#DIV/0!
15	(317) Asset Retirement Costs for Steam Production			317	#DIV/0!
16	TOTAL Steam Production Plant (lines 8 thru 15)		-	- OTA	#DIV/0!
17	B. Nuclear Production Plant		日本市法をお生物		
18	(320) Land and Land Rights				#DIV/0!
19	(321) Structures and Improvements				#DIV/0!
20	(322) Reactor Plant Equipment				#DIV/0!
21	(323) Turbogenerator Units				#DIV/0!
22	(324) Accessory Electric Equipment			1	#DIV/0!
23	(325) Misc. Power Plant Equipment			7	#DIV/0!
24	(326) ARO for Nuclear Production				#DIV/0!
25	Total Nuclear Production Plant (lines 18 - 24)		-	-	#DIV/0!
131212			<b>使用的时候的</b> 的第三人称单数		
36	D. Other Production Plant				
37	(340) Land and Land Rights	ner of Landston and BEUCHSLIE Albert to Antoine and Evelo	2 ( 1273) <b>(1773) (1773) - 1773)</b> 	340	#DIV/0!
38	(341) Structures and Improvements			341	#DIV/0!
39	(342) Fuel Holders, Products, and Accessories		englanantatik ara ura 2 minu kituri altinin kanadik 6 milik int	342	#DIV/0!
40	(343) Prime Movers		// pro-reading to a sublishing to all and such handling a limit i	343	#DIV/0!
41	(344) Generators	and and the second s		344	#DIV/0!
42	(345) Accessory Electric Equipment			345	#DIV/0!
43	(346) Misc. Power Plant Equipment			346	#DIV/0!
44	(347) ARO for Other Production			347	#DIV/0!
45	TOTAL Other Prod. Plant (lines 37 thru 44)		-	OTA 🗌	#DIV/0!
46	TOTAL Prod. Plant (lines 16, 25, 35, and 45)			- OTA -	
47	3. TRANSMISSION PLANT				
48	(350) Land and Land Rights	In the second state of the		350	#DIV/0!
49	(352) Structures and improvements			352	#DIV/0!
50	(353) Station Equipment			353	#DIV/0!
51	(354) Towers and Fixtures			354	#DIV/0!
52	(355) Poles and Fixtures	1		355	#DIV/0!
53	(356) Overhead Conductors and Devices		· · · · · · · · · · · · · · · · · · ·	356	#DIV/0!

 $\tilde{\mathcal{V}}$ 

# FPL December 31

FERC Form No. 1

Electric Plant In Service Page 204, 205, 206 and 207

Page 2	204, 205, 206 and 207		TITLESS STATISTICS		
		A Constant of the second s Second second seco			
54	(357) Underground Conduit		·	357	#DIV/0!
55	(358) Underground Conductors and Devices			358	#DIV/0!
56	(359) Roads and Trails		-	359	#DIV/0!
57	(359.1) ARO for Transmission Plant			359	#DIV/0!
58	TOTAL Transmission Plant (lines 48 thru 57)		-	-  OTA -	#DIV/0!
59	4. DISTRIBUTION PLANT	位 建水油 建固定 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)			
60	(360) Land and Land Rights			360	#DIV/0!
61	(361) Structures and Improvements			361	#DIV/0!
62	(362) Station Equipment		1	362	#DIV/0!
63	(363) Storage Battery Equipment			363	#DIV/0!
64	(364) Poles, Towers, and Fixtures		1	364	#DIV/0!
65	(365) Overhead Conductors and Devices	an a the second se		365	#DIV/0!
66	(366) Underground Conduit			366	#DIV/0!
67	(367) Underground Conductors and Devices		""""""""""""""""""""""""""""""""""""""	367	#DIV/0!
68	(368) Line Transformers	هه - برای مانو از مانو 		368	#DIV/0!
69	(369) Services	, 1, 1, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2,	and a find distance with a strategy data and a superson of the strategy data and a	369	#DIV/0!
70	(370) Meters			370	#DIV/0!
71	(371) Installations on Customer Premises	annen er eine seine s		371	#DIV/0!
72	(372) Leased Property on Customer Premises			372	#DIV/0!
73	(373) Street Lighting and Signal Systems		**** = ** - *** ***********************	373	#DIV/0!
74	(374) Asset Retirement Costs for Distribution Plant		***	374	#DIV/0!
75	TOTAL Distribution Plant (lines 60 thru 74)		+	OTA	#DIV/0!
76	5. GENERAL PLANT				
77	(389) Land and Land Rights			389	#DIV/0!
78	(390) Structures and Improvements		814 8 1 40 1	390	#DIV/0!
79	(391) Office Furniture and Equipment			391	#DIV/0!
80	(392) Transportation Equipment			392	#DIV/0!
81	(393) Stores Equipment			393	#DIV/0!
82	(394) Tools, Shop and Garage Equipment	· · · · · · · · · · · · · · · · · · ·		394	#DIV/0!
83	(395) Laboratory Equipment			395	#DIV/0!
84	(396) Power Operated Equipment	1,		396	#DIV/0!
85	(397) Communication Equipment		The second state of the second state proper sound an external second state of the seco	397	#DIV/0!
86	(398) Miscellaneous Equipment			398	#DIV/0!
87	SUBTOTAL (Enter Total of lines 77 thru 86)			UBT	#DIV/0!
88	(399) Other Tangible Property			399	H WATE VE
89	(399.1) ARO for General Plant	······································		399	
90	TOTAL General Plant (lines 87, 88 and 89)		•	OTA	· _
91	TOTAL (Accounts 101 and 106)			OTA	
92	(102) Electric Plant Purchased			Les	- #DIV/0!
93	(Less) (102) Electric Plant Sold	<b>New Production and Annual States and Annual S</b>		103	₩ <b>₩</b> ₩1, <b>₩</b> 1, <b>₩</b> 1,₩1,₩1,₩1,₩1,₩1,₩1,₩1,₩1,₩1,₩1,₩1,₩1,₩1
94	(103) Experimental Plant Unclassified			105	
95	TOTAL Electric Plant in Service (lines 91 thru 94)		-	OTA -	
····	Add Acquisition Adjustment Scherer 4		3		#DIV/0!
96 97	TOTAL		<u> </u>		
			·		#DIV/0!

FPL December 31 FERC Form No. 1 Electric Plant Held for Future Use Page 214 L 47

			Other	Total			_	
Ledger Month	Steam	Nuclear	Production	Production	Transmission	Distribution	General	Grand Total
Beginning Balance Ending Balance				-				-
Total	-		-	_	-	-	-	
Simple Average	-		-		-	-		

FPL December 31
Accumulated Provision for Depreciation of Electric Utility Plant
Page 219

		Simple Av	erage		
Intangible				Row #1	
Steam Production				Row #2	-
Acquisition Adjustmer	nt Scherer 4 A/	C 115000		Row #3	-
Nuclear Production				Row #4	· -
Hydraulic Production -	- Conventional		1. A.	Row #5	. –
Hydraulic Production-I	Pumped Storag	ge		Row #6	
Other Production				Row #7	-
Transmission				Row #8	-
Distribution				Row #9	
General				Row #10	
Total				Row #11	
······		Beginning B	alance		· · · ·
		Less	Less	Less	
	Total	ARO Accounts	Decommissionin	FPL-NED	Adjusted Totals
<u>Intangible:</u> ARO					_
Unallocated					
\$25 million/yr					
Other					
Subtotal Intangible		-	•	-	-
Production:					
Steam					-
Nuclear					-
Other					-
Scherer Acquisition Subtotal Production	-				•
Fransmission					
					-
Distribution					-
General Fotal		······	· · ·		<u> </u>
. 0(2)	-	-	**	*	

		Ending Bai	lance		
	Total	Less ARO Accounts	Less Decommissionin	Less FPL-NED	Adjusted totals
Intangible: ARO Unallocated				۰.	-
\$25 million/yr Other Subtotal Intangible					• • •
Production; Steam		÷.			•
Nuclear Other Scherer Acquisition					
Subtotal Production Transmission Distribution General	-	-	-	•	•
Total _	•		•	·	

FPL December 31 FERC Form No. 1 Other Regulatory Assets Page 232

		F Sign hurder		ing stepped
្រុំស្រាល់ ទោះ		ાર્ગ નું પદ્મ-દુધોવા પિદ્ધારિક [1] ું ઉદ્યદ્ધવિદ્યાલય નવે છે.		
	Special Assessment for Decontamination and			
3	Decommissioning Fund			<u>i</u>
4				· · · · · · · · · · · · · · · · · · ·
5	Underrecovered Fuel Clause Costs - FPSC			
6				<u> </u>
7	Underrecovered Fuel Clause Costs - FERC			
8				 
9				
10	Deferred Loss on Sale of Land			
11				
12				[ 
13	Tax Audit Settlements			
14	and a standard from the standard and manage and manage and management and part of the standard	<u> </u>		
15			ļ	
16	Standard Offer "QF" Contract Settlement			
17				
18		<u> </u>		
19	Mark to Market Adjustments	[		
20				
21	Underrecovered Franchise Fees	1 ] j**		
22				
23	Underrecovered Capacity Costs			-
24		} }		
25		]		
26	Costs Associated with NRC Design Basis Threat Order for Security Systems at Nuclear Sites	1		_
20	Soconty Systems at Hudical Siles			
27	Deferred Income Taxes			
28		1	,, _,, _	
			· · · · · · · · · · · · · · · · · · ·	
30 31		<b> </b>		
44	Total			
44				-

FPL December 31 FERC Form No. 1 Taxes Accrued, Prepaid and Charged During Year Page 262, 263,262.1 and 263.1

\*unemployment\* \*FICA\* \*worker\* \*franchise\* \*use\* \*use\* \*property\* \*receipts\* \*income\* Motor Vehicles FPSC Fee Intangible Tax Occupational License TOTAL Test (must be zero)

				5			Test (must be	zero)		
a da Salanda)						医红色的现在形态的				
いた時間	带着 法法律 网络马马马									
1.1.1	·利益的增加的产品。	國際的感到			PERDONA	A. 12 12 13	2. 17. 17 - 1	L. Set Markin	n and the second	
1	FEDERAL:		j		L				Γ	1
2				) d			[			L
- 3	INCOME TAXES			L	i		ι		1	
4					1	1	1			
5	FICA:					1	L			
8	YEAR		·	1				1		
7	IYEAR		a di sa							L
8	1			1	L					
9	UNEMPLOYMENT:	1. · • • • • • • • • • • • • • • • •		<u>.</u>		1) 1	י היאס האסי איי איי איי איי איי איי איי איי איי		}	
10	YEAR		· · · · · · · · · · · · · · · · · · ·				<u> </u>			1
11	YEAR				·		!	<u> </u>		
. 12	SUBTOTAL - FEDERAL				_; 		Commune de commune dans se		-	
13 -					[ 	1	······································			
14	STATE			  -	1 					
15			E					· · · · · · · · · · · · · · · · · · ·		·
18	INCOME TAXES	, . 		i	: 			ļ	1 	
17			) ,	<u> </u>	Ļ					
18	UNEMPLOYMENT:	1			արեւ առաջություններություն					
19	YEAR		د د محمد المحمد ( 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 199	; • • • • • • • • • • • • • • • • • • •	1		, 	<u> </u>	<u> </u>	+
	YEAR	بيرهموه والدرسار 1990	1		<u> </u>		{	<u> </u>	<u> </u>	Į
21	) 				ļ. <u>.</u>					ļ
	GROSS RECEIPTS:	÷								
	YEAR			<u> </u>	{					
	YEAR						···· · ···· · ·····			
25	MOTOR VEHICLES:	• • • • • • • • • • • • • • • • • • • •								
		F								
	YEAR				<u> </u>		······································	{		
29	TCAR	ta mananana i	امد ،							
	FPSC FEE:	1 1 1 1		· · · · · · · · · · · · · · · · · · ·				<u>}</u>	j	} í
	YEAR						····· •			
32	YEAR		len 4.7 á ásainn a sánar le allánaithn á súlliadhlin		<u></u>	+				<u> </u>
33	TEAN	· · · · · · · · · · · · · · · · · · ·	۱۰۰۰ - ۱۰۰۰ - ۱۰۰۰ می سامد و اف		· ·					<u>-</u>
34	SALES TAX - A/C 408					·				
34 35		····· ····	р	L						
	INTANGIBLE TAX	******	· · · · · · · · · · · · · · · · · · ·							
37		···· ··· ···	· · · · · · · · · · · · ·		*· · · · · · ·					
38	SUBTOTAL STATE		ļ			t				
的感觉	1970年1月1日,1月1日日日 1月1日日 - 1月1日日 - 1月1日日 1月1日日 - 1月1日日 - 1月			<b>和你们的</b> 有不可以			<b>这一些出来的</b> 的		Carl Martha	W CALL ST LEAD
1	LOCAL	andream ad the date of the second second	1 1	000007.004092;0049712100	PLANNAR ACTUE OF NEW		CHARLES PROVIDE A CONTRACT OF STATES	TELEVISION OF BUILDING		
2		!'			1999 - Constanting and a sub-		······································			j
	FRANCHISE PREPAID	t								1
. 4									 	J
	FRANCHISE ACCRUED:									]
	YEAR		L				······			
	YEAR		• • • • • • • • • • • • • • • • • • •	·						
8		L	L							
	OCCUPATIONAL LICENSES	<u>.</u>	<u>وماريح معمد معمد محمد الم</u>	· · · ·	·					
10			,		·····					
	REAL AND PERSONAL		ښد مو د مور در د							
	PROPERTY TAX:									
And an owner surgery of	YEAR									
	YEAR YEAR		·							
	YEAR						···			
	YEAR					······				
17	YEAR		المستقدين ، 1	إلى دىنە مارىيە			·	··· -··	··	
	YEAR									
20	an and the full second and the second s	an a ann anns an a'	·········			····				
	SUBTOTAL LOCAL					<u> </u>				
										1
41	TOTAL		······			<u>/</u>				
<b></b>					`	<u></u>			<sup>}</sup>	

FPL December 31 FERC Form No. 1 Other Regulatory Liabilities Page 278

		R PEINING	REAL CONTRACT	
្រុះរាជាតាតាត			A Carter of the second second	的现在分词建筑建立
1	Deferred Interest Income - Tax Refunds			· •
2		. ] 		
3				
4	Deferred Gains on Sale of Land			-
5		·		
6				Ś
7				-
· 8	······································	) ] 14]		
9	Derivatives			-
10			]	<u> </u>
11	Nuclear Amortization		-	
12				
13				
14				
15	Overrecovered Energy Conservation			-
16				
17	Deferred Gain on Sale of Emission Allowances			
18				
19	Asset Retirement Obligation	 		
20				
21	Big Bend Transmission Line Sale			-
22		] 		
23				
24	Overrecovered ECRC Revenues			
25				
26	Deferred Regulatory Assessment Fee			_
27				
28	Overrecovered Fuel Clause Revenues - FERC			_
29				
30	Deferred Income Taxes	[		-
31			<b>N</b>	
41	Total	•		-

Power Production Expenses         2       Steam Power Generation         3       Operation         4       (600) Operation Supervision and Engineering         5       (601) Fuel         5       (602) Steam Expenses         7       (503) Steam Transforter-Cr.         9       (600) Expension Supervision and Engineering         9       (600) Genetic Expenses         9       (600) Genetic Expenses         10       (600) Miscelaneous Steam Power Expenses         113       10771A, Operation (Enter Total of Lines 4 thru 12)         14       Maintenance         15       (151) Maintenance of Expenses         16       (151) Maintenance of Expenses         17       (151) Maintenance of Expenses/Steam Plant         18       (151) Maintenance of Expenses/Steam Plant         19       (151) Maintenance of Expenses/Steam Plant         19       (151) Maintenance of Expenses/Steam Plant         19       (151) Operation Supervision and Engineering         10       (151) Color Plant         10       (151) Maintenance of Expenses/Steam Plant         10       (151) Color Plant         10       (151) Color Plant         11       (151) Color Plant <td< th=""><th>LICCI</th><th>ric Operation and Maintenance Expense</th><th></th><th></th><th></th></td<>	LICCI	ric Operation and Maintenance Expense			
1       Power Forduction Expenses         2       Sitem Power Generation         3       Operation         4       (600) Operation Supervision and Engineering         6       (652) Sitem Expenses         7       (503) Sitem Transforred-Or.         9       (605) Excinc Expenses         1000 Sitem Transforred-Or.       9         9       (605) Excinc Expenses         11       (607) Fents         12       (607) Fents         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Maintenance         15       (151) Maintenance of Structures         16       (151) Maintenance of Structures         17       (151) Maintenance of Enciric Pent         18       (151) Maintenance of Enciric Pent         19       (151) Operation Supervision and Engineering         21       TOTAL Maintenance of Enciric Pent         22       Nuclear Expenses         23       (1510) Operation Supervision and Engineering         24       (1512) Operation Supervision and Engineering         <	'age	320 - 323		:	
1       Power Forduction Expenses         2       Sitem Power Generation         3       Operation         4       (600) Operation Supervision and Engineering         6       (652) Sitem Expenses         7       (503) Sitem Transforred-Or.         9       (605) Excinc Expenses         1000 Sitem Transforred-Or.       9         9       (605) Excinc Expenses         11       (607) Fents         12       (607) Fents         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Maintenance         15       (151) Maintenance of Structures         16       (151) Maintenance of Structures         17       (151) Maintenance of Enciric Pent         18       (151) Maintenance of Enciric Pent         19       (151) Operation Supervision and Engineering         21       TOTAL Maintenance of Enciric Pent         22       Nuclear Expenses         23       (1510) Operation Supervision and Engineering         24       (1512) Operation Supervision and Engineering         <	0			NUTRE CONTROL	
2       Sham Pover Generation         3       Operation         4       (500) Operation Supervision and Engineering         5       (601) Fuel         5       (602) Steam Transforred-Cr.         9       (602) Gotesi Expanses         9       (602) Gotesi Expanses         10       (602) Miscelianeuros Steam Power Expanses         10       (602) Miscelianeuros Steam Power Expanses         110       (602) Miscelianeuros Steam Power Expanses         12       (603) Alovences         13       TOTAL, Operation (Enter Total of Lines 4 thru 12)         4       Maintenance         15       (611) Maintenance of Exciture Plant         16       (613) Maintenance of Exciture Plant         17       (613) Maintenance of Exciture Plant         18       (614) Maintenance of Exciture Plant         19       (613) Maintenance of Micrutures         17       TOTAL, Maintenance (Enter Total of Lines 15 thru 16)         20       TOTAL Maintenance (Enter Total of Lines 15 thru 16)         21       TOTAL Maintenance (Enter Total of Lines 15 thru 16)         22       Stam Expanses         23       Coperation         24       (517) Operation Supervision and Engineering         25					
3       Operation         4       (600) Operation Supervision and Engineering         6       (652) Steam Expenses         7       (503) Steam Transforred-Cr.         9       (1605) Feedine Expenses         1000 (300) Beam Transforred-Cr.       9         9       (1605) Feedine Expenses         10       (607) Feedine Expenses         10       (600) Feedine Expenses         11       (607) Feedine Expenses         12       (609) Allowneas         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Maintenance         15       (110) Maintenance of Expenses         16       (121) Maintenance of Expenses         17       (121) Maintenance of Expenses         18       (111) Maintenance of Total Plant         19       (121) Maintenance of Expenses-Steam Plant         10       (121) Maintenance of Macellaneous Steam Plant         21       TOTAL Power Production Expenses         22		strand and a state of the state		· 동안 · · · · · · · · · · · · · · · · · ·	and the second se
4       (500) Operation Supervision and Engineering       4         5       (602) Steam Expenses       6         7       (503) Steam Expenses       6         8       (4.648) Steam Transford-Cr.       6         9       (602) Steam Form Other Sources       6         10       (602) Steam Transford-Cr.       6         10       (602) Rearts       7         11       (607) Fraits       7         12       (609) Allovances       7         13       TOTAL Operation (Enter Total of Lines 4 thru 12)       1         14       Maintenance       1         15(10) Maintenance of Supervision and Engineering       1         16(11) Maintenance of Ender Plant       1         17       (512) Maintenance of Supervision and Engineering       1         16(12) Maintenance of Ender Plant       1         17       107AL Maintenance (Enter Total of Lines 15 thru 19)       1         21       TOTAL Maintenance (Enter Total of Lines 15 thru 19)       1         22       5. Nuclear Power Cotal Cine Expenses-Staam Power (Tot lines 13 4.20)       1         23       Operation       1       1         24       (617) Operation Supervision and Engineering       1         25		history was a second and a second and a second at the seco	·····································		
6       (501) Fuel         7       (503) Steam from Other Sources         8       (Less) (504) Steam Transferre-Cr.         9       (605) Electic Expenses         10       (506) Miscelianeous Steam Power Expenses         11       (607) Florits         12       (507) Rents         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Mahitenance Supervision and Engineering         15       (611) Maintenance of Structures         16       (612) Maintenance of Structures         17       (712) Maintenance of Checkric Parent         18       (613) Maintenance of Checkric Parent         19       (614) Maintenance of Miscellaneous Steam Plant         20       TOTAL. Power Production Expenses-Steam Power (Tot lines 13 & 20)         21       TOTAL. Power Production Expenses-Steam Power (Tot lines 13 & 20)         23       Operation         24       (619) Cooleris and Waisr         25       Nuclear Power Generation         26       (519) Cooleris and Waisr         27       (522) Steam Transfered - Cr.         26       (519) Cooleris and Waisr         27       (522) Steam Transfered - Cr.         28       (Less) (522) Steam Transfered - Cr.         <				승규는 아이들 것이 같아요.	
6       (502) Steam From Other Sources         7       (503) Steam from Other Sources         9       (604) Steam Transferreu-Cr.         9       (505) Maccianeous Steam Power Expanses         10       (607) Rents         11       (607) Rents         12       (609) Allowances         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Mathemance         15       (611) Mathemance Supervision and Engineering         16       (611) Mathemance of Electric Plant         16       (612) Mathemance of Electric Plant         16       (613) Mathemance of Indicalianous Steam Plant         17       (614) Mathemance (Enter Total of Lines 15 thru 19)         20       TOTAL Dover Production Expenses-Steam Power (Tot lines 13 & 20)         21       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         22       5. Nuclear Power Sources         23       Operation         24       (517) Operation Supervision and Engineering         25       (518) Colarits and Watar         26       (519) Colarits and Watar         27       (520) Steam Transfered - Cr.         26       (521) Steam Expenses         26       (522) Steam Transfered - Cr.	-				
7       (503) Steam from Other Sources         9       (Lees) (504) Steam Transformet-Or.         9       (1000) Miscellanous Steam Power Expanses         10       (000) Miscellanous Steam Power Expanses         11       (507) Rents         12       (509) Allowances         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Maintenance of Structures         15       (511) Maintenance of Structures         16       (513) Meintenance of Stear Plant         17       (514) Maintenance of Structures         16       (513) Meintenance of Calcin Plant         17       (514) Maintenance of Expanses-Steam Plant         10       (514) Maintenance of Expanses-Steam Power (Tot lines 13 & 20)         17       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         18       (517) Operation Supervision and Engineering         19       (517) Operation Supervision and Engineering         19       (517) Operation Supervision and Engineering         19       (517) Operation Supervision and Engineering         10       (513) Meintenance Of Succes         10       (524) Miscellanous Nuclear Power Expenses         10       (523) Steam Transforred - Or.         10       (522) Steam Transforred - Or.<	÷				
8       (Lear) (504) Steam Transformat-Cr.         9       (1050) Electine Expenses         11       (607) Rents         12       (509) Allowances         13       TOTAL. Operation (Enter Total of Lines 4 thru 12)         14       Maintenance         15       (511) Maintenance of Structures         16       (513) Maintenance of Structures         17       (742) Maintenance of Structures         16       (513) Maintenance of Electric Plant         16       (514) Maintenance of Miscellaneous Steam Plant         17       (712) Maintenance of Miscellaneous Steam Plant         18       (513) Maintenance of Miscellaneous Steam Plant         19       (514) Maintenance of Miscellaneous Steam Plant         10       TOTAL. Maintenance of Miscellaneous Steam Plant         20       TOTAL Maintenance (Enter Total of Lines 15 thru 19)         21       TOTAL Maintenance Science Steam Power (Tot lines 13 & 20)         22       B. Nuclear Power Generation         23       Operation         24       (617) Operation Supervision and Engineering         25       (513) Fuel         26       (512) Rent Expenses         31       (523) Electric Expenses         32       (523) Maintenance Greet Power Expense		erente feine und eine eine eine eine eine eine eine ei			
9       (305) Electric Expenses         10       (600) Miccolaneous Steam Power Expenses         11       (607) Ronts         12       (609) Allowances         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Maintenance Supervision and Engineering         15       (611) Maintenance of Solier Plant         16       (612) Maintenance of Electric Plant         17       (613) Maintenance of Electric Plant         19       (614) Maintenance of Clecking Plant         10       (615) Maintenance of Miccolanahous Steam Plant         20       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         21       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         22       Operation         23       Operation         24       (517) Operation Supervision and Engineering         25       (518) Fuel         26       (202) Steam Expenses         27       (202) Steam Expenses         28       (202) Steam Expenses         29       (Less) (522) Steam Transferred - Cr.         20       (323) Electric Expenses         21       (202) Steam Expenses         22       (223) Electric Expenses         21       (523) M			, , , , , , , , , , , , , , , , , , ,		
10       (500) Macalianeous Steam Power Expenses	8	(Less) (504) Steam Transferred-Cr.	i .		
11       607) Fionts         12       (509) Allowances         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         14       Maintenance         15       (611) Maintenance of Structures         17       (612) Maintenance of Structures         17       (613) Maintenance of Structures         17       (614) Maintenance of Structures         16       (613) Maintenance of Structures         17       (614) Maintenance (Enter Total of Lines 15 fm 19)         17       TOTAL Maintenance (Enter Total of Lines 15 fm 19)         17       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         18       Cyperation         19       (514) Maintenance (Enter Total of Lines 15 fm 19)         11       TOTAL Maintenance (Enter Total of Lines 15 fm 19)         12       TOTAL Power Production and Engineering         19       (517) Operation Supervision and Engineering         15       (518) Fuel         16       (519) Colorists and Waiter         17       (522) Steam Transfered - Cr.         10       (522) Steam Transfered - Cr.         10       (522) Maintenance of Structures         17       Total Operation Expenses         16       (524) Maintenance of Electric Plant				······································	
12       (509) Allowances         13       TOTAL Operation (Enter Total of Lines 4 thru 12)         Maintenance       (510) Maintenance of Structures         15       (511) Maintenance of Other Plant         16       (513) Maintenance of Other Plant         17       (514) Maintenance of Other Plant         18       (513) Maintenance of Other Plant         19       (614) Maintenance of Miscellaneous Steam Plant         20       TOTAL, Power Production Expenses-Steam Power (Tot lines 13 & 20)         21       TOTAL, Power Production Expenses-Steam Power (Tot lines 13 & 20)         22       B. Nuclear Power Generation         23       Operation         24       (517) Operation Supervision and Engineering         25       (518) Fuel         26       (519) Coolerits and Water         7       (520) Steam Expenses         26       (512) Steam Transfered - Cr.         9       (142) Miscellaneous Nuclear Power Expenses         2       (523) Steam Transfered - Cr.         9       (524) Miscellaneous Nuclear Power Expenses         2       (524) Miscellaneous Nuclear Power Expenses         2       (522) Maintenance of Structures         3       Total Operation (Enter Total of lines 24 thru 32)         4	10	(506) Miscellaneous Steam Power Expenses			
13       TOTAL Operation (Enter Total of Lines 4 thru 12)				······································	
14       Maintenance       Maintenance         15       (610) Maintenance of Structures       (611) Maintenance of Structures         17       (612) Maintenance of Electric Plant       (613) Maintenance of Electric Plant         18       (613) Maintenance of Electric Plant       (614) Maintenance of Electric Plant         19       (613) Maintenance of Kinscellaneous Steam Plant       (614) Maintenance (Enter Total of Unes 15 thru 19)         11       TOTAL. Maintenance (Enter Total of Unes 15 thru 19)       (7074)         12       TOTAL. Power Production Expenses-Steam Power (Tot lines 13 & 20)       (7074)         13       Operation       (7074)         14       (617) Operation Supervision and Engineering       (7074)         15       (516) Fuel       (7074)         15       (512) Electric Expenses       (7074)         16       (512) Operation Supervision and Engineering       (7074)         16       (512) Steam Expenses       (7074)         16       (522) Steam Transfered - Cr.       (7074)         17       (522) Steam Transfered - Cr.       (7074)         15       (522) Steam Expenses       (7074)         15       (523) Electric Expenses       (7074)         15       (524) Rentanance of Fowed Creater Plant Equipment       (707	12				
15       (11) Maintenance Supervision and Engineering       1         16       (611) Maintenance of Bolter Plant       1         17       (12) Maintenance of Bolter Plant       1         18       (613) Maintenance of Electric Plant       1         19       (514) Maintenance of Electric Plant       1         19       (514) Maintenance (Enter Total of Lines 15 thru 19)       1         21       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)       1         22       B. Nuclear Power Generation       1         30       Operation       1         4       (517) Operation Supervision and Engineering       1         5       (518) Fuel       1         6       (19) Coolents and Water       1         7       (202) Steam Transfered - Cr.       1         0       (522) Steam Transfered - Cr.       1         0       (522) Steam Transfered - Cr.       1         0       (523) Maintenance Supervision and Engineering       1         1       (524) Miscelianeous Nuclear Power Expenses       1         1       (524) Miscelianeous Nuclear Power Expenses       1         1       (524) Maintenance of Reactor Part Equipment       1         1       (524) Maintenance	13	TOTAL Operation (Enter Total of Lines 4 thru 12)		4	
15       (11) Maintenance Supervision and Engineering       1         16       (611) Maintenance of Bolter Plant       1         17       (12) Maintenance of Bolter Plant       1         18       (613) Maintenance of Electric Plant       1         19       (514) Maintenance of Electric Plant       1         19       (514) Maintenance (Enter Total of Lines 15 thru 19)       1         21       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)       1         22       B. Nuclear Power Generation       1         30       Operation       1         4       (517) Operation Supervision and Engineering       1         5       (518) Fuel       1         6       (19) Coolents and Water       1         7       (202) Steam Transfered - Cr.       1         0       (522) Steam Transfered - Cr.       1         0       (522) Steam Transfered - Cr.       1         0       (523) Maintenance Supervision and Engineering       1         1       (524) Miscelianeous Nuclear Power Expenses       1         1       (524) Miscelianeous Nuclear Power Expenses       1         1       (524) Maintenance of Reactor Part Equipment       1         1       (524) Maintenance	14				
7       (512) Maintenance of Boler Plant         18       (513) Maintenance of Electric Plant         19       (514) Maintenance (Enter Total of Lines 15 thru 19)         10       TQTAL Maintenance (Enter Total of Lines 15 thru 19)         11       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         2       B. Nuclear Power Generation         30       Operation         4       (517) Operation Supervision and Engineering         5       (618) Fuel         6       (512) Steam Transferred - Cr.         0       (522) Steam Transferred - Cr.         0       (523) Stearts         1       (524) Maintenance Supervision and Engineering         5       (525) Rents         1       (526) Maintenance Generation Muclear Plant         1       (527) Maintenance of Mactor Plant Equipment         1       (528) Maintenance of Mactor Plant Equipment         1       (529) Maintenance Genineeus Nuclear Plant					
13       [513] Meintenance of Miscellaneous Steam Plant         19       [514] Maintenance of Miscellaneous Steam Plant         10       TOTAL Power Production Expense-Steam Plant         21       TOTAL Power Production Expense-Steam Power (Tot lines 13 & 20)         22       B. Nuclear Power Generation         30       Operation         31       Operation Supervision and Engineering         35       [519] Coolants and Water         7       (520) Steam Expenses         30       (521) Operation Supervision and Engineering         35       (521) Steam Transferred - Cr.         36       (522) Steam Transferred - Cr.         37       (522) Steam Transferred - Cr.         38       (522) Steam Transferred - Cr.         39       (Less) (522) Steam Transferred - Cr.         31       (524) Steam Transferred - Cr.         31       (524) Steam Transferred - Cr.         32       (525) Rents         33       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance Supervision and Engineering         5       (525) Maintenance of Structures         5       (522) Maintenance of Maccellaneous Nuclear Plant         33       Total Operation Supervision and Engineering         (532) Mainten	16				
99       (514) Maintenance of Miscellaneous Steam Plant         00       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         11       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)         12       B. Nuclear Power Generation         30       Operation         31       Operation Supervision and Engineering         32       (517) Operation Supervision and Engineering         33       (517) Operation Supervision and Engineering         34       (517) Operation Supervision and Engineering         35       (518) Fuel         36       (521) Steam Transferred - Cr.         37       (522) Electric Expenses         38       (522) Electric Expenses         39       (523) Electric Expenses         30       (524) Macellaneous Nuclear Power Expenses         31       Total Operation (Enter Total of lines 24 thru 32)         41       Maintenance Supervision and Engineering         (522) Maintenance of Structures       (533) Maintenance of Structures         31       (543) Maintenance of Structures         32       (533) Maintenance of Miscellaneous Nuclear Plant         33       (533) Maintenance of Miscellaneous Nuclear Plant         34       (533) Maintenance of Miscellaneous Nuclear Plant         35<	17		1		
20       TQTAL Maintenance (Enter Total of Lines 15 thru 10)				neren erende ennere anderen. F	
21       TOTAL Power Production Expenses-Steam Power (Tot lines 13 & 20)       -         22       B. Nuclear Power Generation       -         23       Operation       -         24       (Bi1) Operation Supervision and Engineering       -         25       (518) Fuel       -         26       (B17) Operation Supervision and Engineering       -         27       (S20) Steam Expenses       -         28       (S21) Steam Transferred - Cr.       -         29       (Less) (S22) Steam Transferred - Cr.       -         29       (S22) Steam Transferred - Cr.       -         29       (S22) Steam Transferred - Cr.       -         20       (S22) Reantian       -         31       Total Operation (Enter Total of lines 24 thru 32)       -         32       Total Operation and Engineering       -         33       Total Operation and Engineering       -         35(252) Maintenance of Structures       -       -         33       Total Operation Genetar Plant       -         34(533) Maintenance of Structures       -       -         35(32) Maintenance of Structures       -       -         35(32) Maintenance of Miscelianeous Nuclear Plant       -         <				·····	
22       9. Nuclear Power Generation         33       Operation         44       (617) Operation Supervision and Engineering         55       (618) Fuel         56       (619) Coolents and Water         76       (620) Steam Expenses         8       (521) Steam Transferred - Cr.         9       (Less) (622) Steam Transferred - Cr.         9       (Less) (622) Steam Transferred - Cr.         9       (1624) Miscellaneous Nuclear Power Expenses         10       (524) Miscellaneous Nuclear Power Expenses         11       (524) Miscellaneous Nuclear Power Expenses         12       (522) Rentis         31       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance of Structures         15       (523) Maintenance of Structures         15       (523) Maintenance of Electric Plant         15       (531) Maintenance of Bloctic Plant         16       (532) Maintenance of Miscellaneous Nuclear Plant         17       Total Meintenance (Enter Total of lines 35 thru 39)         17       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         17       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         17       Total Power Production Expenses         1640) Op	20				
22       9. Nuclear Power Generation         33       Operation         44       (617) Operation Supervision and Engineering         55       (618) Fuel         56       (619) Coolents and Water         76       (620) Steam Expenses         8       (521) Steam Transferred - Cr.         9       (Less) (622) Steam Transferred - Cr.         9       (Less) (622) Steam Transferred - Cr.         9       (1624) Miscellaneous Nuclear Power Expenses         10       (524) Miscellaneous Nuclear Power Expenses         11       (524) Miscellaneous Nuclear Power Expenses         12       (522) Rentis         31       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance of Structures         15       (523) Maintenance of Structures         15       (523) Maintenance of Electric Plant         15       (531) Maintenance of Bloctic Plant         16       (532) Maintenance of Miscellaneous Nuclear Plant         17       Total Meintenance (Enter Total of lines 35 thru 39)         17       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         17       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         17       Total Power Production Expenses         1640) Op	21	TOTAL Power Production Expenses-Steam Power (Tot lines 13 &	1 20)		
4       (517) Operation Supervision and Engineering       5         55       (518) Fuel       5         66       (519) Coolents and Water       5         7       (520) Steam Expenses       5         8       (521) Steam Transferred - Cr.       5         0       (522) Steam Transferred - Cr.       5         0       (523) Electric Expenses       5         1       (524) Miscelianeous Nuclear Power Expenses       5         2       (625) Remits       5         3       Total Operation (Enter Total of lines 24 thru 32)       5         4       Maintenance Supervision and Engineering       5         5       (528) Maintenance of Structures       7         7       (530) Maintenance of Reactor Plant Equipment       5         5       (521) Maintenance of Electric Plant       5         6       (532) Maintenance of Electric Plant       5         7       Total Maintenance of Electric Plant       5         6       (532) Maintenance of Electric Plant       5         7       (532) Maintenance of Electric Plant       5         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40)       5         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40	22			B TR AN A CLEAN	
5       (513) Fuel         6       (519) Coolents and Water         7       (520) Steam Expenses         8       (321) Steam Transferred - Cr.         9       (Less) (522) Steam Transferred - Cr.         0       (1523) Electric Expenses         1       (524) Miscelianeous Nuclear Power Expenses         2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (522) Maintenance of Structures         7       (523) Maintenance of Structures         7       (523) Maintenance of Electric Plant         2       (523) Maintenance of Electric Plant         5       (523) Maintenance of Electric Plant         5       (531) Maintenance of Electric Plant         5       (532) Maintenance of Electric Plant         6       (532) Maintenance of Electric Plant         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         7       (540) Operation Supervision and Engineering         (544) Operation	3	Operation			
6       (519) Coolants and Water         7       (520) Steam Expenses         8       (521) Steam Transferred - Cr.         0       (523) Electric Expenses         1       (524) Miscellaneous Nuclear Power Expenses         2       (625) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (522) Maintenance Supervision and Engineering         6       (530) Maintenance of Structures         7       (530) Maintenance of Electric Plant         5       (532) Maintenance of Electric Plant         5       (531) Maintenance of Electric Plant         5       (532) Maintenance of Electric Plant         5       (532) Maintenance of Electric Plant         5       (532) Maintenance of Electric Plant         7       Total Power Production Expenses - Nuo. Power (bt 33 & 40)         7       Total Power Production Expenses - Nuo. Power (bt 33 & 40)         7       Total Power Production Expenses - Nuo. Power (bt 33 & 40)         7       (540) Operation         (547) Fuel       (547) Fuel         (548) Generation Expenses       (548) Generation Expenses         (549) Miscelianeous Other Power Generation Expenses       (540) Miscelianeous Other Power Generation Expenses <td>4</td> <td>(517) Operation Supervision and Engineering</td> <td>an an a</td> <td></td> <td></td>	4	(517) Operation Supervision and Engineering	an a		
7       (520) Steam Expenses         8       (521) Steam Transferred - Cr.         0       (523) Electric Expenses         1       (524) Miscellaneous Nuclear Power Expenses         2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (528) Maintenance Supervision and Engineering         6       (529) Maintenance of Structures         7       (530) Maintenance of Electric Plant         5       (529) Maintenance of Electric Plant         6       (529) Maintenance of Electric Plant         7       (530) Maintenance of Miscellaneous Nuclear Plant         7       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         7       (540) Operation Supervision and Engineering         (544) Operation Supervision and Engineering       (545) Generation Expenses         (547) Fuel       (548) Generation Expenses         (549) Maintenance Other Power Generation Expenses       (549) Miscelaneous Other Power Generation Expenses	5	(518) Fuel			5
8       (521) Steam from Other Sources         9       (Less) (522) Steam Transferred - Cr.         0       (523) Electric Expenses         1       (524) Miscellaneous Nuclear Power Expenses         2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (528) Maintenance of Structures         7       (530) Meintenance of Reactor Plant Equipment         3       (531) Maintenance of Reactor Plant Equipment         3       (532) Maintenance of Miscellaneous Nuclear Plant         9       (522) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         1       Total Power Production Expenses         (546) Operation       (547) Fuel         (547) Fuel       (548) Generation Expenses         (548) Generation Expenses       (549) Miscellaneous Other Power Generation Expenses         (549) Miscellaneous Other Power Generation Expenses <td>6</td> <td>(519) Coolants and Water</td> <td>· ····</td> <td></td> <td>5</td>	6	(519) Coolants and Water	· ····		5
9       (Less) (522) Steam Transferred - Cr.         0       (523) Electric Expenses         1       (524) Miscellaneous Nuclear Power Expenses         2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (528) Maintenance Supervision and Engineering         6       (529) Maintenance of Structures         7       (530) Maintenance of Reactor Plant Equipment         5       (531) Maintenance of Electric Plant         6       (532) Maintenance of Miscellaneous Nuclear Plant         7       Total Maintenance of Miscellaneous Nuclear Plant         7       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuo. Power (tot 33 & 40)         7       Value         (540) Operation       (547) Fuel         (547) Fuel       (547) Fuel         (547) Fuel       (547) Fuel         (548) Generation Expenses       (549) Miscellaneous Other Power Generation Expenses         (549) Miscellaneous Other Power Generation Expenses       (549) Miscellaneous Other Power Generation Expenses	7	(520) Steam Expenses			5
0       (523) Electric Expenses         1       (524) Miscellaneous Nuclear Power Expenses         2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (528) Maintenance Supervision and Engineering         6       (529) Maintenance of Structures         7       (530) Maintenance of Reactor Plant Equipment         3       (531) Maintenance of Electric Plant         3       (532) Maintenance of Miscellaneous Nuclear Plant         3       (532) Maintenance of Miscellaneous Nuclear Plant         4       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         4       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         4       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         4       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         4       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         4       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         5       (540) Operation Supervision and Engineering         (547) Fuel       (548) Generation Expenses         (547) Fuel       (547) Fuel         (547) Fuel       (548) Generation Expenses         (547) Bends       Senis	8		· .		5
1       (524) Miscellaneous Nuclear Power Expenses         2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (529) Maintenance Supervision and Engineering         6       (529) Maintenance of Structures         7       (530) Maintenance of Reactor Plant Equipment         8       (531) Maintenance of Electric Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Inscellaneous Nuclear Plant         9       (532) Maintenance (Enter Total of lines 35 thru 39)         1       Total Maintenance (Enter Total of lines 35 thru 39)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Vision and Engineering         (540) Operation Supervision and Engineering         (547) Fuel         (547) Fuel         (547) Fuel         (547) Fuel         (547) Rents         (549) Miscellaneous Other Power Generation Expenses         (540) Miscellaneous Other Power Generation Expenses	9				
2       (525) Rents         3       Total Operation (Enter Total of lines 24 thru 32)         4       Maintenance         5       (528) Maintenance Supervision and Engineering         6       (529) Maintenance of Structures         7       (530) Maintenance of Reactor Plant Equipment         8       (531) Maintenance of Electric Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Clientic Plant         9       (532) Maintenance of Cliencic Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance (Enter Total of lines 35 thru 39)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses         1       (540) Operation Supervision and Engineering         1       (543) Generation Expenses         1       (544) Generation Expenses         1       (545) Generation Expenses	0				5
3       Total Operation (Enter Total of lines 24 thru 32)					5
4       Maintenance         5       (528) Maintenance Supervision and Engineering         6       (529) Maintenance of Structures         7       (530) Maintenance of Reactor Plant Equipment         8       (531) Maintenance of Reactor Plant Equipment         9       (532) Maintenance of Electric Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         0       Total Maintenance (Enter Total of lines 35 thru 39)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         2       Figure 1         3       Figure 2         4       Figure 2         4       Figure 2         5       Figure 2         5       Figure 2         5       Figure 2         6       Figure 2         6       Figure 2         6       Figure 2         6       Figure 2	2		Ľ	· ·	5
5       (528) Maintenance Supervision and Engineering       Provide and the second sec				-	- 10
6       (529) Maintenance of Structures         7       (530) Maintenance of Reactor Plant Equipment         8       (531) Maintenance of Electric Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         9       Total Maintenance (Enter Total of lines 35 thru 39)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         7       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         9       Coperation         1       Coperation         1       Coperation Supervision and Engineering         1       C48) Generation Expenses         1       C48) Generation Expenses         1       C540) Miscellaneous Other Power Generation Expenses			20.40 M		
7       (530) Maintenance of Reactor Plant Equipment         8       (531) Maintenance of Electric Plant         9       (532) Maintenance of Miscellaneous Nuclear Plant         0       Total Maintenance (Enter Total of lines 35 thru 39)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Total Power Production Expenses         0       Total Power Production Expenses         0       Total Power Production Expenses         0       Total Power Power Generation Expenses         (549) Miscellaneous Other Power Generation Expenses       (549) Miscellaneous Other Power Generation Expenses		The second s		1	
B       (531) Maintenance of Electric Plant         I       (532) Maintenance of Miscellaneous Nuclear Plant         I       Total Maintenance (Enter Total of lines 35 thru 39)         I       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         I       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         Operation       Status         Operation       Status         (540) Operation Supervision and Engineering         (547) Fuel       Status         (549) Miscellaneous Other Power Generation Expenses         (540) Renis					
9       (532) Maintenance of Miscellaneous Nuclear Plant         0       Total Maintenance (Enter Total of lines 35 thru 39)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         1       Total Power Production Expenses - Nuc. Power (tot 33 & 40)         0       Operation         0       Government of the power of the power (tot 33 & 40)         0       Operation         0       Government of the power o	-				
Total Maintenance (Enter Total of lines 35 thru 39)       -         Total Power Production Expenses - Nuo. Power (tot 33 & 40)       -         Total Power Production Expenses - Nuo. Power (tot 33 & 40)       -         Operation       -         (546) Operation Supervision and Engineering       -         (547) Fuel       -         (548) Generation Expenses       -         (549) Miscellaneous Other Power Generation Expenses       -	_		}		
Total Power Production Expenses - Nuc. Power (tot 33 & 40)         Operation         (546) Operation Supervision and Engineering         (547) Fuel         (548) Generation Expenses         (549) Miscellaneous Other Power Generation Expenses					
Operation       (546) Operation Supervision and Engineering       (547) Fuel       (548) Generation Expenses       (549) Miscellaneous Other Power Generation Expenses	)			•	
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses		Total Power Production Expenses - Nuc. Power (tot 33 & 40)			-
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses		可以已经没有保障的问题。			
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses	1				
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses			化成本法律		本の構成では、
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses					
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses					
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses					50 - 50 - A 14
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses					
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses	i li Li				
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses	نې ( ۱۹۹۱ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ -				
(546) Operation Supervision and Engineering (547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses				it is an a set of the	
(547) Fuel (548) Generation Expenses (549) Miscellaneous Other Power Generation Expenses					
(548) Generation Expenses (549) Miscellanaous Other Power Generation Expenses				<u> </u>	
(549) Miscellaneous Other Power Generation Expenses				n 1994 ( 1994 -	54
(550) Repts					54
		and the second	s as a la se se s	1	54

		(PTAL Maintensuce (Enter Total of line 211 and 2101)	9
		(598) Maintenance of Miscellaneous Distribution Plant	*
		(597) Maintenance of Missellageour Distribution Blass	5
<b>N</b>		(596) Maintenance of Street Lighting and Signal Systems	z
6	الم المحمد الم	(595) Maintenance of Line Transformers	
		(304) Maintenance of Underground Lines	- 0
<b>.</b>		(593) Maintenance of Overhead Lines	6
		(592) Maintence of Station Equipment	8
<b>.</b>		(591) Maintenance of Structures	۳ ۲
		And a second s	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		(690) Maintenance Supervision and Engineering	9
			<u> </u>
-		TOTAL Operation (Enter Total of lines 103 thru 113)	4
94 1		sineA (683)	3
		(568) Miscellaneous Expenses	5
		(587) Customer Installations Expenses	۱
2		(586) Meter Expenses	Ø
		Seenest Lighting and Signal System Expenses	6
		(584) Underground Line Expenses	8
		(583) Overhead Line Expenses	2
		(582) Station Expanses	
		(1683) Load Dispatching	9
		3. Distribution Expenses (continued)	2 7
		(560) Operation Supervision and Engineering	
		and a second	3
		Орегацои	S
		3. Distribution Expenses	
-		(99 bris 18 senii to IstoTitefi Expenses (Enter Totsi of lines 81 and 99)	0
<del>ال</del>	F	TOTAL Maintenance (Enter Total of lines 93 thru 96)	e
		(573) Maintenance of Miscellaneous Transmission Plant	8
6		(572) Maintenance of Underground Lines	2
	1	6511) Maintenance of Overhead Lines	g
	······································	final memory of Station Equipment	g
		(569) Maintenance of Structures	\$
		(568) Maintenance Supervision and Engineering	ŝ
		Waintenance	7
		TOTAL Operation (Enter Total of lines 63 thru 90)	์ เ
ä			0
M	- magandana sa na	(566) Miscellaneous Transmission Expenses	6
<b></b>		(265) Transmission of Electricity by Others	.8
		(564) Underground Lines Expenses	4
		(263) Overhead Lines Expenses	9
		(562) Station Expenses	S
<u>N</u>		Buindteqai Dispatching	4
		(560) Operation Supervision and Engineering	3
		Operation	2
		2. Transmission Expenses	
-	- (6/ 8 4	TOTAL Power Production Expenses (Total of lines 21, 41, 59, 7	0
	and the second		
		COTAL Other Power Supply Exp (Enter Total of lines 76 thru 76	6
			8
		(556) System Control and Load Dispatching	
Contraction of the second second	BCM and the Destination of the State of the	(222) Furchesed Power	9,
下的相談		E. Other Power Supply	ç,
-	-	(52)	۴.
1-	[-: i	TOTAL Maintenance (Enter Total of lines 69 thru 72)	3
	Ttroi	(554) Maintenance of Miscalianeous Other Power Generation P	<u>ت</u>
		(553) Maintenance of Generating and Electric Plant	يع را
		**************************************	<u> </u>
		(662) Maintenance of Structures	0
		Construction Supervision and Engineering	6
			8 1
		920 - 323	
		: Form No. 1 ic Operation and Maintenance Expense	

FPL December 31	
FERC Form No. 1	

Electric Operation and Maintenance Expense

		•	
Done	220	_	222

H ton b	····································				
126	TOTAL Distribution Exp (Enter Total of lines 114 and 125)			OTA	
127	4. Customer Accounts Expenses	<b>北京市内市</b> 市		. C	
128	Operation	<b>建设的建筑的</b> 的		per	
129	(901) Supervision			901	
130	(902) Meter Reading Expenses		·····	902	
131	(903) Customer Records and Collection Expenses		·	<b>903</b>	
132	(904) Uncollectible Accounts			904	
133	(905) Miscellaneous Customer Accounts Expenses			905	
134	TOTAL Customer Accounts Expenses (Total of lines 129 thru 133)	-	-8	OTA	
135	5. Customer Service and Informational Expenses	Sec. C. Chiller		.C	
136	Operation	<b>美国新港市</b> 市		рег	
137	(907) Supervision			907	
138	(908) Customer Assistance Expenses	1		908	
139	(909) Informational and Instructional Expenses	1		909	
140	(910) Miscellaneous Customer Service and Information Expenses		ļ	910	
141	TOTAL Cust. Service and Information. Exp. (Total lines 137 thru 140)	3	í -	OTA	
142	6. Sales Expenses			. S	
143	Operation			per	
144	(911) Supervision			911	
145	(912) Demonstrating and Selling Expenses			912	
146	(913) Advertising Expenses		1	913	
147	(916) Miscellaneous Sales Expenses			916	
148	TOTAL Sales Expenses (Enter Total of lines 144 thru 147)		-4	OTA	•
149	7. Administrative and General Expenses			, A	FPLNEI
150	Operation	Al Chabrel of Statistics		per	ADJ
151	(920) Administrative and General Salaries			920	
152	(921) Office Supplies and Expenses		44	921	
153	(Less) (922) Administrative Expenses Transferred-Credit	{		922	
	7. Administrative and General Expenses (continued)	J			
- 155	(923) Outside Services Employed			923	
156	(924) Property Insurance			924	
157	(925) Injuries and Damages			925	
and the second second second	(926) Employee Pensions and Benefits			926	
159	(927) Franchise Requirements			927	
160	(928) Regulatory Commission Expenses	андаран — — — — — — — — — — — — — — — — — — —		928	
	(929) (Less) Dupilcate Charges-Cr.			929	
Instrument of the second second	(930.1) General Advertising Expenses	1		930	
i	(930.2) Miscellaneous General Expenses			930	
	(931) Rents			931	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -
165	TOTAL Operation (Enter Total of lines 151 thru 164)	-	-	ΟΤΑ	
	Maintenance			ain	
	(935) Maintenance of General Plant			935	
	TOTAL Admin & General Expenses (Total of lines 165 thru 167)	2 I	-	ΟΤΑ	
1. · · · · · · · · · · · · · · · · · · ·	TOTAL Elec Op and Maint Expn (Tot 80, 100, 126, 134, 141, 148, 168)		-103	OTA	



	ecember 31										
	Form No. 1										
	sed Power 27, 327.1, 327.2, 32	, 73377	A								
				肌心		N 4 3 .			用於分裂的	1	$15^{-2}$
			n D			LE PURC			CHARLES (THE		<b>秋秋</b>
<u>백리의보망</u> 1	Southern Co	LF	NG SALES	I Berni	anaette T					80 80 BB	315. 1
2	Southern Co	AD	4			<del> </del> -	·····		<u> </u>		
3	FL Municipal	τu		1							
4	FL Municipal	AD				<u>i</u>		***		1	
5	Orlando Utilities	<u> </u>	: yme				<b></b> ,,				
6 7	Orlando Utilities	AD - LU	<u>.</u>	+							
<u>6</u>	JEA	AD			<u>+</u>		•••••••				
9	Bio-Energy	LU		- in		an an garan an g					
10	Bio-Energy	AD		••••••••••••••••••••••••••••••••••••••		•					
11	Broward Resource	LU		1	· - · · ·	· · ·			<b>7</b>		
12	Broward Resource	AD						<b>.</b>			
13 14	Broward Resource	RQ							ļ		
14 1	Broward Resource	ĂĎ LU			ŀ						
2	Cedar Bay	AD									
3	Ellott	ĽŪ	• • • • • • • • • • • • • • • • • • •	T	4. <b>.</b>				j		
4	Fi Crushed Stone	LU		<u>.</u>					****** *******************************		
5	FI Crushed Stone	AD		T					· · · · · · · · · · · · · · · · · · ·		
	Georgia Pacific	LU							L		
7 8	Georgia Pacific Indiantown Cogen	AD LU	al Glado — a - Mashanna Abrahi a Manan	÷	Ļ						
	Indiantown Cogen	AD	• ··· ·•· ·•·		<b>1</b>				۰۰۰۰۰ ۰۰۰۰۰۰ ۱		
10	MM Tomoka Farms	10			<u> - : :</u>						
11	MM Tomoka Farms	AD				* ****** * *, #*****				-*j	
12	Okeelanta Power	LU		••••••••	1					**************************************	
13	Okeelanta Powar	AD									
14	Solid Waste Auth	LU		 							
1 2	Tropicana	AD LU									
3	Tropicana	AD		r				· ·			
4	US Sugar	LU		L			·				
5	US Sugar	AD I		r						··• • • • • • • • • • • • • • • • • • •	•••••
6	Calpine	os								1	
7	Calpine	<u>IU</u>		·							
8	Calpine Cargil	οs	است	۱	. بى					-	
10	Cincinnati Gas	OS :	······		<u>+</u>						
11	Cobb	os i			· <u>-</u>			•		ļ	\
12	Conoco	OS .			<u>بة</u>			·	**		
13	Energy Authority	OS :	· ••••••••••••••••••••••••••••••••••••								* *** ***
14	FI Power Corp	OS									
	Ft Power Corp	08.	 							4	
	Hardee Power Homestead	OS OS	• ••••	·· -			·			<u>_</u>	
	Merrill Lynch	-os-+		+	t.e. ,		_ <del> </del>	<del></del> +		 	
	Morgan Stanley	08	···· • • • • • • • • • • • •	• •	<b></b> †-				• }		
6	Oglethorpe	ÖŠ .							·		
	Orlando Utilities	ŌS			· • • • •					1	
	Progress Ventures :	OS	· · · ·			•••• • ••• •• •••		 السبب د			
9 10	Progress Ventures   Progress Ventures !							·			
	Progress Ventures		····								
12	Reliant Energy	OS		-	n nado S			·· ·····			
	Reliant Energy	ΪU i			· · · ·	·····				1	
	Reliant Energy	ÎU .		Į		- 1900 Million Plane and 1				1	
Long a straight	Reliant Energy	IU OR		· · +		a manadar analis bi shaya adaga					
	Seminole Southern Co	OS OS			1 				• • • • • • • • • • • • • • • • • • •		
	Southern Co	10		•···- <u> </u>			<u>+</u>		······································	• • •	
	Southern Co	ΪU .		!	••••		•••••••••••••••••••••••••••••••••••••••	·••••••	. عبير وجير بيون شور		-
	Southern Co	IU 🗍								•	AT 2000-0-1
	Tempa Electric	os	••••••••••••••••••••••••••••••••••••••	i				·		jaas⊷en, as as ¦ hu san '	• • • • • • • • • • • • • • • • • • •
8 🗄	Tenaska	OS .		:				1			
	Villams Power	ÖS		*** *******						ն հարդապանությունը է են	· · · · · · · · · · · · ·

# FPL December 31

FERC Form No. 1

# Depreciation and Amortization of Electric Plant

Page 336

<u> </u>					
			CPALE 1555 1555 1555 1555 1555 1555 1555 15	n (f. Standard) 17. Juni (f. Standard) 19. Juni (f. Standard) 19. Juni (f. Standard) 19. Juni (f. Standard)	(Alian Charles and Charles and Charles Angle (Charles and Charles) (Charles and Charles) (Charles) (Charles and Charles) (Charles)
1	Intangible Plant			· · · ·	
2	Steam Production Plant				
3	Nuclear Production Plant				-
4	Hydraulic Production Plant				
5	Hydraulic Production Plant-Pumped Storage				
6	Other Production Plant				
7	Transmission Plant				
8	Distribution Plant				-
9	General Plant				
10	Common Plant - Electric	antisense conserve are make any all the second			
11	TOTAL	-	-	-	-
	é éterne an annu 18 ar 18 ar é raisean anna 18 an anna an an an anna an an an an an an		the second s		

FPL December 31 FERC Form No. 1 Distribution of Salaries and Wages Page 354 and 355

Production payroll	-
A&G Payroll	-
Total payroll	-
Payroll excluding A&G	
Production/Total excl. A&G	#DIV/0!
1	

1	Electric		
2	Operation		
3	Production		
4	Transmission		
5	Distribution	- 1	
6	Customer Accounts		
7	Customer Service and Informational		
B	Sales	· · · · · · · · · · · · · · · · · · ·	
Ð	Administrative and General		
10	TOTAL Operation (Enter Total of lines 3 thru 9)		
11	Maintenance		
12	Production		
13	Transmission		
14	Distribution		
15	Administrative and General	······	
16	TOTAL Maint. (Total of lines 12 thru 15)		
17	Total Operation and Maintenance		
18	Production (Enter Total of lines 3 and 12)		
19	Transmission (Enter Total of lines 4 and 13)		
20	Distribution (Enter Total of lines 5 and 14)		
21	Customer Accounts (Transcribe from line 6)	-	
22	Customer Service and Informational (Transcribe from line 7)		
23	Sales (Transcribe from line 8)	•	
24	Administrative and General (Enter Total of lines 9 and 15)		
25	TOTAL Oper. and Maint, (Total of lines 18 thru 24)		
62	TOTAL All Utility Dept. (Total of lines 25, 59, and 61)		
63	Utility Plant		
64	Construction ( By Utility Departments)		和哈特的基本的自己的特
65	Electric Plant		
66	Gas Plant		
67	Other		
68	TOTAL Construction (Total of lines 65 thru 67)	-	
69	Plant Removal (by Utility Departments)		
70	Electric Plant		
71	Gas Plant		
72	Pother		······
73	TOTAL Plant Removal (Total of lines 70 thru 72)	-	
74	Other Accounts		***************************************
75	ο επιτηροτορισμοτρικό το μεται τη ποροχεία με τη απόσο ματική το το το μοτορισμοτο ματική το		ار میں زیر در دون سامند و دور در میں میں ہو و میں میں اور
76	Accounts Receivable from Associated Companies (146)		
77	Misc. Current and Accrued Assets - Jobbing Accounts (174.1)	ar ann anna marris 201 Pombleona, or -1	
78	Temporary Facilities (185)		8 Mh > 869644444 holds/Halleddiddiddiaeniau ny 9 ywraiddady a fer byr glygarif braw ywr arwyn y
79	Misc Deferred Debits - Primary Storm Maintenance (186)		
80	Accumulated Provision for Pensions and Benefits (228.3)		en fair stradell (hait is a skiller to a bibling formation and any second a fair and any of a
81	Other Electric Revenues (456)		all Manuf Mar feilid and antill (d. Galgerstang) of a linear a lar constant - a second stargets of a space of a
82	Various	· · · · · · · · · · · · · · · · · · ·	979-1989 dati dati berta data da ana persona persona da ana ana ana ana ana ana ana ana an
	TOTAL Other Accounts	-	-
96	TOTAL SALARIES AND WAGES	-	
-			

FPL December 31 Monthly Peaks and Output Page 401b

(1)	(2)	(3)	(4)	(5)	(6)	(7)
			a distant a distant distant a distant a distant a distant a distant a distan		activated load management (Includes LCEC) MWV	adjusted peak demand MW (5) + (6)
29	January					-
30	February	1				-
31	March					-
32	April					-
33	Мау					•
34	June					-
35	July		[		-	_
36	August				-	-
37	September				-	~
38	October					-
39	November					-
40	December					
41	TOTAL	-	-	-	-	~

Winter Summer

5

### Less: FPL Load Management Capability:

Residential Load Management b/

C/I Load Management b/

Total FPL Load Management Capability b/

Number of months

FPL Load Management Capability

### Less: 7 X LCEC Summer Adjusted Load Management Capability :

Summer Adjusted Load Management Capability = The lesser of: a) SLM Ratio times Customer's actual peak load coincident with the peak hour of Company's Summer Months with the effect of any Customer Load Management activated removed

b) LCEC's actual load management during the Summer Months resulting from a verification test or pursuant to a request by Company

### Less: 5 X LCEC Winter Load Management Capability:

LCEC Winter Adjusted Load Management Capability = The lesser of:

a) WLM Ratio times Customer's actual peak load coincident with the peak hour of Company's Winter Months with the effect of any Customer Load Management activated removed or

b) LCEC's actual load management during the Winter Months resulting from a verification test or pursuant to a request by the Company

Total of 12 monthly peak firm MWs @ generator

a/ per article 3.7 of the PSA b/ per Ten Year Site Plan FPL December 31

Construction Work in Progress Statement

LINE <u>NO.</u>	Period	Gross Steam Production	Nuclear Production	Other Production	<u>n</u>	Total Production	Non-Production CWIP	Total CWIP FF 1, p.110 L3
.1	Beginning Balance				=	0		0
2 3 4	Ending Balance				=	0		0
5		0	0		0 =	0	0	0
`6 7	100% Pollution Con 50% All other Produ		d Fuel Conversion	facilities		0	· .	
8	Total CWIP for St				· •	0		


	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
	Input	Input	(Prior Bal)+(1)+(2)	(2) * 50%	(Prior Bal)+(1)+(4)	(4) - (2)	Note 1	Cumulative Sum (6)	(7) * Depreciation Rate	Cumulative Sum (9)	(8) - (10)
Year	Expenditures	AFUDC FPSC	CWIP BALANCE - FPSC	AFUDC LCEC	CWIP BALANCE -	AFUDC DIFF	Plant In Service Diff · Balance		Amort of LCEC Plant Adjustment	Accumulated Amort of LCEC Plant Adjustment	LCEC CWIP Adjustment Balance After Arnort

132

Note 1 - Plant in Service difference balance due to FPSC and LCEC AFUDC method difference.

## FPL December 31 GSU Statement

Year	Plant In Service	Reserve	Annual Depr Expense	GSU O&M	·	
Average	0	0			•	

#### Florida Power & Light Company ACTUAL LOSSES Loss Study - Development of Loss Expansion Factors For The Year Ended December 31

		(†) ENER	(2) (2)		(3) (4) DEMAND (2)		
Line No D	Description	Flow MWH	Loss Expansio Factor	n	Flow MW	Loss Expansio Factor	n
1 2	Net Energy / Demand to Transmission	(	A)		(C	3)	
34	Step-up Losses	(	B)		()	ł)	
5 6	Flow to Transmission	(	C)		(	)	
7 8	Transmission Losses	. (	D) .		(.	))	
9 10	Flow on Transmission	(	E)		()	0	
11	Transmission Energy / Demand Losses		#D1V/01	(F)		#DIV/0!	(

Notes:

(1) Energy Loss Expansion Factor is calculated based on information provided by Power Supply.

(A) Power Supply (NEL plus net power wheeled for others)

(B) Power Supply

(C) = (A) + (B)

(D) Power Supply

(E) = (C) + (D)

(F) = 1/(E/A)

(2) Demand Loss Expansion Factor is calculated based on the Westinghouse Formula.

(G) Represents system demand including power wheeled for others.

(H) Based on Westinghouse Formula --> Demand Loss = Flow to Transmission (line 1) \* Energy Loss Factor / (.3+(.7\*Load Factor)) (I) = (G) + (H)

(J) Based on Westinghouse Formula --> Demand Loss = Flow to Transmission (line 5) \* Energy Loss Factor / (.3+(.7\*Load Factor)) (K) = (I) + (J)

(L) = 1/(K/G)

# FPL December 31 Calculation of Customer's CP Demand

135

10/AIQ#	10/A10#	#DIA\0i	10/A10#	#DIA\0i	10/AICI#	i0/AIC#	i0/AlQ#	#DIA\0i	10/AIC#	10/AIQ#	₩DIA\0	10/AIQ#	Customer's adjusted monthly coincident peaks (L 17 / L 2, col 1)
													91 7 '£1 7 0} 7 '£ 7 ums
	-											_	St oni i the it is greater than ine 15 then take line 14 minute line 15
				_	-				· ·	,		e/u	Maximum Allowed KW Credit For Distributed Generation Purchases
												e/u	minute Interval for Measured CP Demand
													Distributed Generation Purchases Behlind the Delivery Point(s) During 60
						·		·····				<u>.</u>	Stenil sunim ftenilexistnerti Steni insrbretseng si fteril 11 mu2
												B/L	session Allowed (W Credit For Renewable Energy Purchases
												8/U	Renewable Energy Purchasses Behind the Delivery Point(s) During 60 Menute Internation Measured CP Demand
												-	6 anii auriim 6 anii axisi nadi 6 anii madi najaang si 6 anii 11 mu2
												¥/U	Maximum Allowed kW Credit For Qualitying Facility Purchases
												в/ш	Interval for Measured CP Demand
										1		-	Qualifying Facility Purchases Behind the Delivery Point(s) During 50 minute
	-			-		-	-		-	-		·	- LCEC Adjusted CP Demand (Line 3 + Line 4 - Line 5 - Line 5)
										-	-		b) LCEC actual load management for the winter months D.1.0.2.0.3.
												-	Load Management activated removed or peak hour of Company's Winter Months with the effect of any Customer
													<ul> <li>MLM Ratio times Customer's actual peak load coincident with the series of Company Minter Method Mathematics</li> </ul>
													LCEC Winter Adjusted Load Management Capability = The lower of:
						-							ess. LOEC 5 Winter Months Load Management Capability.
													<ul> <li>b) LCEC actual load management for the summer months</li> </ul>
												· •	peak hour of Company's Summer Months with the effect of any Customer Load Management activated removed or
													a) SLM Ratio times Customer's actual peak load colnoident with the
													LCEC Summer Adjusted Load Management Capability = The lower of:
													stis: Adjustment for any activated Load Management .ess: LCEC 7. Summer Months Adjusted Load Magagement Cagability :
												-	Sum of 60 Minute KW CP Demand Measured At the Delivery Point(s)
12/11 D.54	B/U B/U	ълı D	12/13 12/13	8/11	ълі Влі	12/U	12/U	<b>E/</b> 0	B'U	ели	B/U		Annual Loss Factor (1-Line 1)
8/0	e/u	8/0	B/U	- <del> </del>	E/U	B/U	ели 		ъла 11.00	e/u	B/0		futual Loss % (per Demand and Energy Loss schedule)
bruemoG	Demand	Demand	Demand		Demand			• -	Demand		Densard	(Sum Cols 2 - 13)	Description
Cb	Cb	Cb	Cb	Cb	Cb	Cb	Ch tree	Cb '	Cb	Cb (The second s	Cb	Demand	
• •		(11)	September (10)	isuguA (9)	մյոք (8)	Jane (7)	(a) May	(S) (J)	March	February (2)	វីរាមមាន (=)	Sum of Monthly CP	
(EI)	(21)	(II)	(UI)	ŝ	(8)	L)	(9)	(S)	(†)	(£)	(z)	(1)	(a) A set of the se

.

FPL December 31 Calculation of Customer's Billing Demand

(I) Sum of Mo	Billing Dernand Crann Cale 7 = 13	schedule)	Highest 60 Minute kW Demand Measured At the Delivery Point(s)	Qualifying Facility Purchases Behind the Delivery Point(s) During 60 minute n/a Interval for Measured Demand Marcimum Allowed kW Credit For Qualifying Facility Purchases n/a Sum: If line 4 is greater than line 5 then take line 4 minus line 5	Renewahle Energy Purchases Behind fae Delivery Point(s) During 60 minute Interval for Measured Demand Maximum Allowed kW Credit For Renewable Emergy Purchases Sum: Ef line 7 is greater than line 8 (hen take line 7 minus line 8	Distributed Generation Purchases Behind the Delivery Point(s) During 60 n/t minute Interval for Measured Demand Maximum Allowed kW Credit For Distributed Generation Purchases n/a. Sum: If fine 10 is greater than fine 11 then take line 10 minus time 11		Sum of Customer's mouthly kW billing demand (L 13 / L 2) #D(V/0)
athly (2)	January I Billing	,	ı	 	•	•		
(1)	ry February is Billing	i						IV/NIC# IQ
€	y March Billing	1						iovald# io
છ	April Billing	n/a n/a	•					i0/AIG#
(9)	May Billing	1						iQ/AIC#
E	Billing	4	·		·			#DIV/0
6)	July Billing	:						#DIV/0
6)	August S Billing	1						i0/AIC#
( <u>6</u> )	September Billing	n/a 1/a						#DIV/0
(11)	October Billing	n/a B/a			• .	·		i0//IC#
(71)	November Billing	D'a D'a			•			#DIV/0
(EI) -	December Billing	e/u						i0/A10#

. . .

ž

136

#### APPENDIX C

### Fuel Charge Factor Formula and Fuel Adjustment Charge Factor Formula

The Fuel Charge Factors.

1.

(c)

(a) The amounts included in the estimated and actual total expense of system fuel and purchased economic power shall be consistent with 18 CFR 35.14 and shall include without limitation fees for disposal of spent nuclear fuel and/or high-level radioactive waste as specified in the Contract for Disposal of Spent Nuclear Fuel And/or High-Level Radioactive Waste between the United States of America represented by the US Department of Energy and Florida Power and Light Company dated June 1983.

- (b) The total expense of estimated and actual system fuel and purchased economic power included in the Fuel Charge Factors and the Fuel Adjustment Charge Factors shall be the cost of:
  - (i) fuel consumed in Company's own plants, and Company's share of fuel consumed in jointly owned or leased plants;
  - (ii) the actual identifiable fuel costs associated with energy purchased for reasons other than identified in section 1(b)(iii) of this Appendix;
  - (iii) the total cost of the purchase of economic power as defined in section 1(d) of this Appendix, if the reserve capacity of Company is adequate independent of all other purchases where nonfuel charges are included;
  - (iv) generation energy charges for any purchase if the total amount of generation energy charges is less than Company's total avoided variable costs;
  - (v) less the cost of fuel recovered through all intersystem sales;
  - (vi) plus any Taxes on the energy cost of fuel or, electric energy generated, where such Taxes are not included elsewhere.
  - The cost of fuel included in the estimated and actual system fuel and purchased economic power expenses shall include no items other than those listed in the account 151 of the FERC Uniform System of Accounts For Public Utilities and Licensees. The cost of nuclear fuel shall be that as shown in account 518, except that if account 518 also contains any expenses for fossil fuel that has already been included in the cost of fossil fuel, it shall be deducted from this account.

- (d) For the purpose of section 1 (b) (iii) and (iv), the following definitions apply:
  - Economic power is economic power or energy purchased over a period of 12 months or less where the total cost of the purchase is less than Company's avoided variable cost;
  - (ii) Total cost of the purchase is all charges incurred in buying economic power and having such power delivered to Company system. The total cost includes, but is not limited to, capacity reservation charges, generation energy charges, adders, and any transmission or wheeling charges associated with the purchase.
  - (iii) Total avoided variable costs is all identified and documented variable costs that would have been incurred by Company had a particular purchase not been made. Such costs include, but are not limited to, those associated with fuel, startup, shutdown or any purchases that would have been made in lieu of the purchase made.
- (e) For the purpose of section 1 (b) (iii), the system reserved capacity criteria used by Company's system operators is demand and energy purchased for a period of less than a year and shall be deemed as being for reliability purposes if Company expects that the purchase is required in order to maintain operating reserves in accordance with Good Utility Practice.
- (f) Total system net generation and purchased economic power costs included in the Fuel Charge Factors and the Fuel Adjustment Charge Factors shall be the sum of:
  - (i) generation,
  - (ii) purchases,
  - (iii) exchange received, less
  - (iv) energy associated with pumped storage operations, less
  - (v) intersystem sales referred to in section 1 (b) (v) of this Appendix C, less
  - (vi) total system losses (losses shall be deemed to be zero because Customer takes Full Requirements Electric Service at the Receipt Points).
- (g) Calculation of estimated On-peak Fuel Charge Factor:

((Estimated total fuel costs and net power transactions defined in 1(b) and (c)<sup>1</sup> \* On-peak cost ratio<sup>2</sup>) / (estimated total net generation defined in  $1(f)^1$  \* On-peak load ratio<sup>2</sup>))

<sup>&</sup>lt;sup>1</sup> Total fuel costs and net power transactions and total MWh generation estimates approved by the FPSC for the current period. Includes applicable FERC adjustments.

(h) Calculation of estimated Off-peak Fuel Charge Factor:

((Estimated total fuel costs and net power transactions defined in 1(b) and (c)<sup>1</sup> \* Off-peak cost ratio<sup>2</sup>) / (estimated total net generation defined in 1(f)<sup>1</sup> \* Off-peak load ratio<sup>2</sup>))

- (i) The attached Schedule 1 illustrates the calculation of the On-peak and Off-peak Fuel Charge Factors.
- (j) "On-peak" and "Off-peak" shall have the meanings attributed to such terms in Company's retail tariff GSLDT-3.
- 2. The Fuel Adjustment Charge Factors.
  - (a) Calculation of On-peak Fuel Adjustment Charge Factor:

Actual On-peak Fuel Charge Factor minus estimated On-peak Fuel Charge Factor

(b) Calculation of Off-peak Fuel Adjustment Charge Factor:

Actual Off-peak Fuel Charge Factor minus estimated Off-peak Fuel Charge Factor

The actual On-peak and Off-peak Fuel Charge Factors shall be calculated by applying actual costs and expenses to the formulas used to calculate the estimated On-peak and Off-peak Fuel Charge Factors in part 1 of this Appendix C.

<sup>2</sup> On-peak and off-peak cost and load ratios calculated using cost and load data from the Production Costing Model POWRSYM.

## APPENDIX C

Schedule 1

## Calculation of On-Peak and Off-Peak Fuel Charge Factors

Line	Description	Amount
	Total Fuel Costs and Net Power Transactions	-
ī	Cost of Fuel Consumed - Section 1(b)(i)	-
2	Fuel Costs for Energy Purchased - Section 1(b)(ii)	-
3	Total Cost of Purchased Economic Power - Section 1(b)(iii)	-
4	Generation energy charges - Section 1(b)(iv)	-
5	Cost of Fuel Recovered Through all Intersystem Sales - Section 1(b)(v)	_
6	Taxes on Energy Cost of Fuel or Electric Energy Generated - Section 1(b)(vi)	
7	Total Fuel Costs and Net Power Transactions - Section 1(g) (Sum Lines1- 4, less line 5, plus line 6)	····
	Total Net Generation	
8	Generation - Section 1(f)(i)	
9	Purchases - Section 1(f)(ii)	
10	Exchanged Received 1(f)(iii)	
11	Energy Associated With Pumped Storage Operations - Section 1(f)(iv)	
12	Intersystem Sales Included in Line 5 - Section 1(f)(v)	·
13	Total Net Generation - Section 1(f) (Sum lines 8 - 10, less lines 11 - 12)	
14	On-peak Cost Ratio - Section 1(g) (POWRSYM)	
15	Total On-peak Fuel Costs and Net Power Transactions - Section 1(g) (Line 7 x Line 14)	
16	On-peak Load Ratio - Section 1(g) (POWRSYM)	
17	On-peak Net Generation - Section 1(g) (Line 13 x Line 16)	
18	ON-PEAK FUEL FACTOR - Section 1(g)(Line 15 / Line 17)	·
19	Off-peak Cost Ratio - Section 1(h) (POWRSYM)	;
20	Total Off-peak Fuel Costs and Net Power Transactions - Section 1(h) (Line 7 x Line 19)	. <u></u>
21	Off-peak Load Ratio - Section 1(h) (POWRSYM)	
22	Off-peak Net Generation - Section 1(h) (Line 13 x Line 21)	
23	OFF-PEAK FUEL FACTOR - Section 1(h) (Line 20 / Line 22)	

ł

## APPENDIX D

### FINANCIAL COVENANT COMPLIANCE WORKSHEET

Deb Service Coverage Debt Service Coverage = Margins + Interest Expense + Depreciati / Principal and Interest on Long Term Debt	
	bon and Amontization
Margins:*	
Interest Expense:	
Premium on Discounted RUS Debt	
Early Payment Penalty	
Depreciation and Amortization:	
1) Total	
Long Term Debt Principal Payments Bank A:	
Long Term Debt Principal Payments Bank 8; Interest Expense:	
2) Total	
Debt Service Coverage Ratio: (1/2)	Not less than 1.75 to 1.0
imes Interest Earned Ratio:	1001000 (Mart 1.7010 1.0
Net Margins + Interest Expense / Interest Expense	
Margins:* Interest Expense:	
1) Total	
Interest Expense:	
2) Total	
TIER: (1/2)	Not less than 1.5
Equity Ratio	11011000 11211110
otal Margine & Equitles / Total Assets 1) Total Margins and Equities	
2) Total Assets	
Equity Percent; (1/2)	
otal Debt to EBITDA Earnings Before Interest Taxes Depreciation otal Debt (all borrowed money, notes, bonds, debentures & capital /Operating Revenues Operating Expenses + Dep	leases)
Long Term Debt & LOC Notes Payable Capital Leases	
Japina Lebsco	
1) Total	
Operating Margins:	
Premium on Discounted RUS Debt	
Early Payment Penalty	,
Interest Expense	
Depreciation and Amontization: 2) Total	
Debt to EBITDA	· · · · · · · · · · · · · · · · · · ·
	Not greater than 8 to 1
ther investments:	1
otal of Other Investments do not exceed 15% of "Net Utility Plant"	and the second
otal of Other Investments do not exceed 15% of "Net Utility Plant" Other Investments	
otal of Other Investments do not exceed 15% of "Net Utility Plant" Other Investments Net Utility Plant Other Investments/Net Utility Plant	
Net Uility Plant Other Investments/Net Utility Plant iole: *= Net Margins	
otal of Other Investments do not exceed 15% of "Net Utility Plant" Other Investments Net Utility Plant Other Investments/Net Utility Plant ote: *= Net Margins + Cash received from Bank A	
otal of Other Investments do not exceed 15% of "Net Utility Plant" Other Investments Net Utility Plant Other Investments/Net Utility Plant ote: *= Net Margins + Cash received from Bank A + Cash received from Company X	· · · · · · · · · · · · · · · · · · ·
otal of Other Investments do not exceed 15% of "Net Utility Plant" Other Investments Net Utility Plant Other Investments/Net Utility Plant ote: *= Net Margins + Cash received from Bank A + Cash received from Company X + Cash received from Y	· · · · · · · · · · · · · · · · · · ·
tal of Other Investments do not exceed 15% of "Net Utility Plant" Other Investments Net Utility Plant Other Investments/Net Utility Plant te: *= Net Margins + Cash received from Bank A + Cash received from Company X	

141

Company X capital credits
 Return of CTC-Bank A

+ capital oredits + Cash received from Z

Marcins:

Debt Service Coverage Ratio. For purposes hereof, the term "Debt Service Coverage Ratio" (A) shall mean the ratio of: (1) net income (after taxes and after eliminating any gain or loss on sale of assets or other extraordinary gain or loss), plus depreciation expense, amortization expense, and interest expense, minus non-cash patronage, and non-cash income from subsidiaries and/or joint ventures; to (2) all principal payments due within the period on all "Long-Term Debt" (as defined below) plus total interest expense (all as calculated on a consolidated basis for the applicable period in accordance with GAAP consistently applied or the appropriate standards of the regulatory agency having jurisdiction over Customer). For purposes hereof, "Long-Term Debt" shall mean, for Customer, on a consolidated basis. the sum of (a) all indebtedness for borrowed money, (b) obligations which are evidenced by notes, bonds, debentures or similar instruments, and (c) that portion of obligations with respect to capital leases or other capitalized agreements that are properly classified as a liability on the balance sheet in conformity with GAAP or which are treated as operating leases under regulations applicable to them but which otherwise would be required to be capitalized under GAAP, in each case having a maturity of more than one year from the date of its creation or having a maturity date within one year from such date, but that is renewable or extendible, at Customer's option, to a date more than one year from such date or that arises under a revolving credit or similar agreement that obligates the lenders to extend credit during a period of more than one year from such date, including all current maturities in respect of such indebtedness whether or not required to be paid within one year from the date of its creation.

TIER DSC EBITDA

(B) Times Interest Earned Ratio ("TIER"): "Times Interest Earned Ratio" shall mean the ratio of (1) net income (after taxes and after eliminating any gain or loss on sale of assets or other extraordinary gain or loss), plus total interest expense (all as calculated on a consolidated basis for the applicable period in accordance with GAAP consistently applied or the appropriate standards of the regulatory agency having jurisdiction over Customer); to (2) total interest expense (all as calculated on a consolidated basis for the appropriate standards of the regulatory agency having jurisdiction over Customer). Said TIER ratio shall be determined by averaging the two highest annual ratios during the most recent three calendar years.

(C) Equity Ratio. "Equity Ratio" shall mean the ratio of consolidated total margins and equities to consolidated total assets (both as determined in accordance with GAAP consistently applied or the appropriate standards of the regulatory agency having jurisdiction over Customer).

(D) Total Debt to EBITDA. "Total Debt to EBITDA" shall mean the ratio of Total Debt to EBITDA. For the purposes hereof: (1) "Total Debt" shall mean, for Customer, on a consolidated basis, the sum of (a) all indebtedness for borrowed money, (b) obligations which are evidenced by notes, bonds, debentures, or similar instruments, and (c) that portion of obligations with respect to capital leases or other capitalized agreements that are properly classified as a liability on the balance sheet in conformity with GAAP or which are treated as operating leases under regulations applicable to them but which otherwise would be required to be capitalized under GAAP; and (2) "EBITDA" shall mean, for Customer, on a consolidated basis, operating revenues minus operating expenses, plus depreciation and amortization expenses for such fiscal year (all as calculated on a consolidated basis in accordance with GAAP consistently applied or the appropriate standards of the regulatory agency having jurisdiction over Customer).

#### **APPENDIX E**

## **FPL Generating Resources by Location**

Jocation/ Map Key	Piant Name	Number of Units	Summer NW
A	Turkey Point	4	2,174
6	St, Lucie *	2	1,553
¢	Manatee	3	2,742
Ð	Fort Myons	2	1,764
ε	Cutier	2	205
F	Leuderdale	2	872
3	Port Everglades	4	1,219
ĸ	Riviera	2	585
F	Martin	5	3,738
J	Cape Canaveral	2 2 2 4 2 5 2 3	792
ĸ	Senford	3	2,044
L	Puinani	2	498
м	SJRPP **	2 2	250
	Scherer ***	1	646
	Gas Turbines	45	1,908
	Internal Combustion Turbines	5	12



\* Represents FPL's ownership share: St Lucie nuclear, 100% unit 1, 65% unit 2, St. Johns River, 20% of two units.

\*\* SJRPP = St. John's River Power Park

\*\* The Scherer unit is localed in Georgia and is not shown on this map.

Figure I.A.1: Capacity Resources by Location (as of December 31, 2006)

#### APPENDIX F

#### LOAD MANAGEMENT

#### 1.0 Load Management Verification Testing Methodology.

- 1.1 If Company believes Customer's monthly Load Management capability is not representative of Customer's actual Load Management capability, a verification test ("Test") may be requested as set forth in Section 3.7(g) of this Agreement to determine Customer's Adjusted Load Management Capability.
- 1.2 A Test shall entail the establishment of the Customer's Reference Average Hourly Load Without Load Management, as set forth in Section 1.2.2, below, and the establishment of the Customer's Reference Average Hourly Load With Load Management for the Test hour during the Test day, as set forth in Section 1.2.3, below.
  - 1.2.1 Tests shall be performed for the Summer Months from June through August on a day when rain or significant cloud cover is not expected. If rain occurs in the Customer's service territory on a Test day between 13:00 and 18:00, Customer or Company shall have the option of postponing the Test until the next Business Day. Tests shall be performed in the Winter Months from December through February on a day when the forecasted lows are predicted to be below 38 degrees in N. Ft. Myers, Florida. If by February 15<sup>th</sup> of each year, no Test has been conducted during the Winter Months, Company may select a day between February 15<sup>th</sup> and 28<sup>th</sup> to conduct a Test. All tests shall be performed on a Business Day.
  - 1.2.2 Customer's Reference Average Hourly Load Without Load Management shall be determined as follows: a comparison day load shape for days Load Management is not used will be established by calculating the hourly average load for each hour of the day over a four day period consisting of the two days before the Test and the two days after the Test. The average hourly load for the hour of the comparison day that is the same hour as the designated Test hour shall be the Customer's Reference Average Hourly Load Without Load Management.
  - 1.2.3 Customer's Reference Average Hourly Load With Load Management shall be determined as follows: the Test day load shape will be determined by measurement of the hourly average loads for each hour of the day during the Test day. On the Test day, the Customer shall initiate full Load Management capabilities such that all capabilities are utilized for the Test hour. The average hourly load during the Test hour shall be the Customer's Reference Average Hourly Load With Load Management. The Test Load Management capability shall be determined by subtracting Customer's Reference Average Hourly Load With Load Management from Customer's Reference Average Hourly Load With Load Management from Customer's Reference Average Hourly Load With Load Management.

- 1.2.4 Scheduling of the testing shall be arranged in accordance with the provisions of Section 3.7 (g) of this Agreement or as otherwise mutually agreed. Measurements shall be made using the Delivery Point meters. Upon request by Customer, Company shall provide assistance to Customer in arranging with the Transmission Provider the right to obtain dial up, "read only" access (not real time data) to the output of the Delivery Point meters; provided, however, should the Transmission Provider not grant such access rights, then Company shall have no obligation to provide such assistance to Customer.
- **1.2.5** Test frequency shall not exceed one test during the Summer Months and one test during the Winter Months consistent with Section 1.2.1 for each Party, unless the Parties mutually agree otherwise.
- **1.2.6** Results of a Test shall establish a new Load Management Capability for the entire season of Summer Months or Winter Months, as applicable, of the applicable Calendar Year.
- 2.0 Customer Adjusted Load Management Capability.
  - 2.1 Customer Adjusted Load Management Capability shall equal the amount specified in Section 3.7(a) of the Agreement.
  - 2.2 A Test may be used by Company to determine Customer Adjusted Load Management Capability, provided the following Test conditions are satisfied:
    - 2.2.1 Customer is notified by Company to implement 100% of its Load Management capability.
    - 2.2.2 Customer's hourly load shape for the day of the Test, excluding the Test hour, is substantially the same as the load shape for the two days prior to the Test and the two days after the Test.
    - 2.2.3 Customer is not prevented from initiating Load Management because of Force Majeure events.
  - 2.3 As provided in Section 3.7(a) of this Agreement, Company may also determine Customer Adjusted Load Management Capability by requesting that Customer dispatch 100% of Customer's Load Management capability.
    - 2.3.1 Upon Company's request, Customer shall provide all operating data available to demonstrate Load Management was dispatched as requested.
    - 2.3.2 In the event Company determines that 100% of Customer's Load Management capability was not dispatched as requested and, upon request from Company, Customer fails to demonstrate that it used reasonable efforts to dispatch 100% of its Load Management, Company may establish Customer's Adjusted Load Management Capability for the applicable Summer or Winter Months for the current Calendar Year and the following two Calendar Years at the amount of Customer's Load Management actually dispatched.
    - 2.3.3 Absent a Force Majeure event, should Customer have a complete failure to respond to Company's request for dispatch of Load Management and/or for operating data, Company may establish Customer's Adjusted Load Management Capability at zero for the applicable Summer or Winter

Months for the current Calendar Year and the following two Calendar Years.

### 3.0 Notification to Implement Load Management

- 3.1 Company shall notify Customer to implement Load Management consistent with Company's dispatch of its Load Management in the Ft. Myers area of Company's service territory.
- 3.2 Company may request interruption of the following devices:
  - 3.2.1 Water Heaters
    - 3.2.2 Pool Pumps
    - 3.2.3 Air Conditioners/ Heaters
  - 3.2.4 Interruptible (All other devices)
- 3.3 Company agrees to provide Customer with an alarm and printer for receipt of Load Management requests from Company.
- 3.4 Proper notice shall consist of a request by Company and a response by Customer acknowledging the request.
- 3.5 Notice shall consist of an e-mail to Customer at <u>SYSOPS@LCEC.NET</u>. Once received, Customer shall initiate a reply to <u>XXX@FPL.COM</u> (e-mail address to be provided by Company).
  - 3.5.1 Should Company not receive a response in the time Company deems appropriate, Company shall then contact Customer by telephone at the telephone numbers listed below:
    - 3.5.1.1 (239) 995-5892
    - 3.5.1.2 (239) 656-2161
    - 3.5.1.3 (239) 656-2405
  - **3.5.2** Upon request, Company shall provide to Customer e-mail and telephone records detailing the day, time, and communication methods utilized to request dispatch of Customer's Load Management.
  - 3.5.3 Customer shall notify Company of any changes in e-mail addresses and telephone numbers in writing.
  - 3.5.4 Customer and Company may change communication methods from time to time upon mutual written consent.
  - 3.5.5 All notices of changes shall be made in accordance with Section 9.1 of this Agreement.
- 4.0 Company shall comply with any reporting requirements of FRCC or any other jurisdictional entity with respect to any information required for the use of Customer's Load Management equipment.

#### APPENDIX G

#### DEDICATED LINE OF CREDIT

The following are some required general attributes of the Dedicated Line of Credit:

- The Dedicated Line of Credit shall be a committed revolving credit facility provided by a financial institution reasonably acceptable to Company of at least 364 days in duration, with an automatic evergreen renewal provision similar to a Letter of Credit.
- Customer's rights under the Dedicated Line of Credit shall be pledged to Company and a first priority security interest therein properly perfected in favor of Company.
- The Lender shall enter into a conventional consent and agreement in form and substance satisfactory to Company, whereby it shall, among other things:
  - o acknowledge the pledge of the credit agreement to Company and agree that, to the extent of any contradictory instructions from Customer and Company, it will act at the instruction of Company,
  - agree to provide Company notice and independent opportunity to cure defaults of Customer under the credit agreement prior to having the right to terminate or cease funding,
  - agree that payments will be made only to Company pursuant to draws made only by Company (the credit agreement will incorporate the Customer's authorization to the Lender to fund draws made only by Company and agree that all such draws shall be paid directly to Company), and
  - o agree that the credit agreement will not be terminated or amended without the prior written consent of Company.
- The credit agreement shall not contain extensive representations, warranties, covenants or defaults and the right of the Lender to terminate the commitment to fund or refuse to fund draws presented by Company shall be limited to the following:
  - a payment default which remains uncured or unwaived for a period of five (5) Business Days after the Lender gives notice to Company and Customer, and
  - certain events of bankruptcy in respect of Customer.

Lender may retain its right to pursue remedies against the Customer, yet not refuse to make the funds available to Company due to the breach of any representation, warranty, covenant or event of default (other than the limited exceptions agreed to by Company).

- Company's right to draw shall include:
  - the default by Customer in the observance or performance of any term of the Short-Term Agreement or this Agreement,
  - o the failure of Customer to renew or replace the Dedicated Line of Credit with other agreeable security at least thirty (30) calendar days prior to the current expiration of the Dedicated Line of Credit;
  - the failure of Customer to replace the Dedicated Line of Credit with an agreed Letter of Credit within five (5) Business Days following the failure to remain in compliance with the financial covenant; and
  - o any other rights to draw set forth in the Agreement.

#### APPENDIX H

#### FORM OF LETTER OF CREDIT

[ISSUING BANK NAME] IRREVOCABLE NONTRANSFERABLE STANDBY LETTER OF CREDIT NO. { }

DATE:

BENEFICIARY:

APPLICANT:

L	
1	•
[	

	]
[	]
Ĺ	]
L	1

INITIAL AMOUNT: USD \$\_\_\_\_\_ DATE OF EXPIRY: On the Expiration Date (as hereinafter defined) PLACE OF EXPIRY: At our Counters

We hereby issue in your favor our Irrevocable Nontransferable Standby Letter of Credit No. { } (this "Letter of Credit") for the account of Lee County Electric Cooperative, Inc. (the "Applicant"), in the aggregate stated amount of not to exceed AND \_) (as the same may be reduced from time to /100 US DOLLARS (US\$ time as a result of draws made pursuant to the provisions of this Letter of Credit, the "Available Amount"), effective immediately and expiring at 5:00 p.m., New York, New York, time, on the Expiration hereinafter defined) Date (as at our counters at

This Letter of Credit shall be of no further force or effect upon the close of business on \_\_\_\_\_\_\_\_\_ (or, if such day is not a Business Day (as hereinafter defined), on the next preceding Business Day); provided, however, that this Letter of Credit will be automatically extended without amendment for successive one (1) year periods from the present or any future expiration date hereof, unless we provide you with written notice of our election not to renew this Letter of Credit at least forty-five (45) days prior to any such expiration date (the present or any future expiration date as aforesaid is referred to herein as the "Expiration Date"). For the purposes hereof, "Business Day" shall mean any day on which commercial banks are not authorized or required to close in Miami, Florida.

Subject to the terms and conditions herein, funds under this Letter of Credit are available to you by presentation in strict compliance on or prior to 5:00 p.m., New York, New York, time, on or prior to the Expiration Date at our counters of:

(1) the original of this Letter of Credit and all amendments; and

(2) your sight draft drawn on us; and

#### (3) either:

(i) Beneficiary's Certificate issued in the form of Annex I attached hereto and which forms an integral part hereof, duly completed and purportedly bearing the original signature of an officer of the Beneficiary; or

(ii) Beneficiary's Certificate issued in the form of Annex II attached hereto and which forms an integral part hereof, duly completed and purportedly bearing the original signature of an officer of the Beneficiary.

Drafts drawn under this Letter of Credit must contain the clause: "Drawn under [Issuing Bank Name] Irrevocable Nontransferable Standby Letter of Credit No. { }, dated

Multiple draws are permitted under this Letter of Credit; provided that the Available Amount of this Letter of Credit shall be permanently reduced by the amount of each such draw.

This Letter of Credit may not be transferred or any of the rights hereunder assigned. Any purported transfer or assignment shall be void and of no force or effect.

This Letter of Credit sets forth in full our undertaking and such undertaking shall not in any way be modified, amended, amplified or limited by reference to any documents, instruments or agreements referred to herein, except only the annexes referred to herein; and any such reference shall not be deemed to incorporate by reference any document, instrument or agreement except for such annexes.

We engage with you that your drafts drawn under and in strict compliance with the terms of this Letter of Credit will be duly honored if presented to us on or before the Expiration Date.

This Letter of Credit shall be governed by the Uniform Customs and Practice for Documentary Credits, 1993 Revision, International Chamber of Commerce Publication No. 500 as in effect on the date of issuance thereof (the "UCP"), except to the extent that the terms hereof are inconsistent with the provisions of the UCP, including but not limited to Articles 13(b) and 17 of the UCP, in which case the terms of this Letter of Credit shall govern. As to matters not covered by the UCP, this Letter of Credit shall be governed by and construed in accordance with the laws of the State of New York, without regard to the principles of conflicts of laws thereunder.

[ISSUING BANK NAME]

By:\_\_\_\_

Authorized Signature

Address:

[	
[	]

#### ANNEX I TO [Issuing Bank Name] IRREVOCABLE NONTRANSFERABLE STANDBY LETTER OF CREDIT NO. {

}

[Issuing Bank Name]	
[]	
	ĺ

Date:\_\_\_\_\_

Ladies and Gentlemen:

The undersigned \_\_\_\_\_\_, the duly elected and acting \_\_\_\_\_\_\_\_\_of Florida Power & Light Company (the "Beneficiary"), hereby certifies to [Issuing Bank Name] (the "Bank") and Lee County Electric Cooperative, Inc. (the "Applicant"), with reference to Irrevocable Nontransferable Standby Letter of Credit No. { }, dated \_\_\_\_\_\_\_, \_\_\_\_\_\_(the "Letter of Credit"), issued by the Bank in favor of the Beneficiary, as follows as of the date hereof:

- 1. The Beneficiary is a party to that certain Long-Term Agreement for Full Requirements Electric Service dated as of August 21, 2007, by and between Beneficiary and Applicant (as amended from time to time, the "Long-Term Agreement"). In addition, the Beneficiary is a party to that certain Short-Term Agreement for Partial Requirements Electric Service dated as of August 21, 2007, by and between Beneficiary and Applicant (as amended from time to time the "Short-Term Agreement").
- Based upon the foregoing, the Beneficiary hereby makes demand under the Letter of Credit for payment of U.S. DOLLARS \_\_\_\_\_\_ AND \_\_\_\_/100ths (U.S.\$\_\_\_\_\_), which amount does not exceed the lesser of (i) the amount set forth in paragraph 2, above, and (ii) the Available Amount under the Letter of Credit as of the date hereof.
- 4. The undersigned has concurrently presented to you its sight draft drawn in the amount specified in Paragraph 3 above. The date of the sight draft is the date hereof, which is not later than the Expiration Date.

Funds paid pursuant to the provisions of the Letter of Credit shall be wire transferred to the Beneficiary in accordance with the following instructions:

Unless otherwise provided herein, capitalized terms which are used and not defined herein shall have the meaning given each such term in the Letter of Credit.

IN WITNESS WHEREOF, this Certificate has been duly executed and delivered on behalf of the Beneficiary by its duly elected and acting \_\_\_\_\_\_ as of this day of \_\_\_\_\_\_.

Beneficiary:

[\_\_\_\_]

By: Name: Title:

5.

#### ANNEX II TO [Issuing Bank Name] IRREVOCABLE NONTRANSFERABLE STANDBY LETTER OF CREDIT NO. {

[Issuing Bank Name] [\_\_\_\_\_]

Date:\_\_\_\_\_

}

Ladies and Gentlemen:

The undersigned \_\_\_\_\_\_, the duly elected and acting \_\_\_\_\_\_ of Florida Power & Light Company (the "Beneficiary"), hereby certifies to [Issuing Bank Name] (the "Bank"), and Lee County Electric Cooperative, Inc. (the "Applicant"), with reference to Irrevocable Nontransferable Standby Letter of Credit No. { }, dated \_\_\_\_\_\_ (the "Letter of Credit"), issued by the Bank in favor of the Beneficiary, as follows as of the date hereof:

- 1. The Beneficiary is a party to that certain Long-Term Agreement for Full Requirements Electric Service dated as of August 21, 2007, by and between Beneficiary and Applicant (as amended from time to time, the "Agreement").
- 2. The Bank has heretofore provided written notice to the Beneficiary of the Bank's intent not to renew the Letter of Credit prior to the present Expiration Date thereof and the Applicant has failed to provide the Beneficiary with a substitute letter of credit substantially in the same form as the Letter of Credit referred to in this paragraph.

or

- 3. The Beneficiary has provided at least thirty (30) calendar days' prior written notice to the Applicant of the Bank's intent not to renew the Letter of Credit prior to the present Expiration Date thereof and the Applicant has failed to provide the Beneficiary with a substitute letter of credit substantially in the same form as the Letter of Credit within the thirty (30) calendar day period referred to in this paragraph.
- 4. Based upon either 2 or 3, above, the Beneficiary hereby makes demand under the Letter of Credit for payment of U.S. DOLLARS \_\_\_\_\_\_& /100ths (U.S. \$\_\_\_\_\_\_), which amount does not exceed the lesser of (i) the amount Beneficiary is entitled to draw under the terms of the Agreement and (ii) the Available Amount under the Letter of Credit as of the date hereof.
- 5. The undersigned has concurrently presented to you its sight draft drawn in the amount specified in Paragraph 4, above, which amount does not exceed the lesser of (a) the amount the Beneficiary is entitled to draw pursuant to the provisions of the Agreement, and (b) the Available Amount as of the date hereof. The date of the sight draft is the date of this Certificate, which is not later than the Expiration Date.
- 6. Funds paid pursuant to the provisions of the Letter of Credit shall be wire transferred to the Beneficiary in accordance with the following instructions:

SSD\_ADMIN/106297,1

153

Unless otherwise provided herein, capitalized terms which are used and not defined herein shall have the meaning given each such term in the Letter of Credit.

IN WITNESS WHEREOF, this Certificate has been duly executed and delivered on behalf of the Beneficiary by its duly elected and acting \_\_\_\_\_\_ as of this day of \_\_\_\_\_\_.

Beneficiary:

[\_\_\_\_\_]

By: Name: Title:

SSD\_ADMIN/106297.1

154

#### APPENDIX I

#### FORM OF CONSENT AND AGREEMENT

This CONSENT AND AGREEMENT (this "<u>Consent and Agreement</u>") dated as of \_\_\_\_\_\_, is executed by and among FLORIDA POWER & LIGHT COMPANY, a corporation organized under the laws of the State of Florida ("<u>FPL</u>"), LEE COUNTY ELECTRIC COOPERATIVE, INC., a rural electric cooperative ("<u>LCEC</u>"), and [LENDER NAME, a \_\_\_\_\_] (the "<u>Lender</u>").

1. FPL and LCEC have entered into the agreement specified in Schedule I hereto (as amended, restated, supplemented or otherwise modified from time to time in accordance with the terms thereof and hereof, the "Assigned Agreement");

2. The Assigned Agreement expressly contemplates that LCEC may, subject to the provisions of the Assigned Agreement, collaterally assign, transfer, mortgage or pledge its interest in the Assigned Agreement to one or more of the Lenders (as defined in the Assigned Agreement) or an indenture trustee under any indenture securing the obligations of LCEC (the "Other Secured Party") if LCEC is then a borrower of such Lender or Other Secured Party for any obligations secured by any indenture, mortgage or similar lien on its system assets;

3. LCEC has obtained financing from the Lender pursuant to [Describe Financing Agreement], dated as of \_\_\_\_\_\_ (the "<u>Financing Agreement</u>"), and FPL has consented to the collateral assignment of the Assigned Agreement to the extent provided in Section 12.2(i) of the Assigned Agreement, which terms are supplemented and modified by the terms of this Consent and Agreement; and

NOW, THEREFORE, the parties hereto hereby agree as follows:

1. <u>Consent to Assignment</u>. Subject to the terms and conditions of this Consent and Agreement, FPL hereby acknowledges and consents solely to the pledge and collateral assignment of all right, title and interest of LCEC in, to and under (but not its obligations, liabilities or duties with respect to) the Assigned Agreement by LCEC to the Lender.

2. <u>Right to Cure.</u>

(a) Subject to Section 2(c), below, from and after the date hereof and unless and until FPL shall have received written notice from the Lender that the lien of the Security Agreement has been released in full, the Lender shall have the right, but not the obligation, following an "event of default" or "default" (or any other similar event however defined) by LCEC under an Assigned Agreement, to pay all sums due under such Assigned Agreement by LCEC and, to the extent it may lawfully do so, to perform any other act, duty or obligation required of LCEC thereunder as described in Section

SSD\_ADMIN/106297.1

2(c) below; <u>provided</u>, that no such payment or performance shall alone be construed as an assumption by the Lender of any covenants, agreements or obligations of LCEC under or in respect of the Assigned Agreement.

(b) FPL agrees that it will not terminate the Assigned Agreement without first giving the Lender notice and opportunity to cure as provided in Section 2(c), below.

(c) If a Termination Event shall occur, and FPL shall then be entitled to and shall desire to terminate the Assigned Agreement, FPL shall, prior to exercising such remedies or taking any other action with respect to such Termination Event, give written notice to the Lender of such Termination Event. If the Lender elects to exercise its right to cure as herein provided, it shall have a period of two (2) business days after receipt by it of notice from FPL referred to in the preceding sentence in which to cure the Termination Event specified in such notice if such Termination Event consists of a payment default, or if such Termination Event is an event other than a failure to pay amounts due and owing by LCEC (a "<u>Non-monetary Event</u>"), the Lender shall have ten (10) business days after receipt by it of notice from FPL referred to in the preceding sentence in which to cure the Termination Event specified in such notice of such Termination Event. If Lender fails to cure within the aforementioned cure period, Lender's right to cure shall be extinguished and FPL may exercise all its available rights and remedies.

(d) Any curing of or attempt to cure any Termination Event shall not be construed as an assumption by the Lender of any covenants, agreements or obligations of LCEC under or in respect of the Assigned Agreement.

(e) Notwithstanding any provision herein to the contrary, nothing herein shall be deemed to affect, impair or diminish any other rights or remedies FPL may have against LCEC in respect of such Termination Event, including, without limitation, the right to suspend performance under the Assigned Agreement.

Substitute Owner. Upon an event of default by LCEC under the Security 3. Agreement, the Lender may not, without the prior written consent of FPL (a) have itself or its designee substituted for LCEC under the Assigned Agreement, or (b) sell, assign, transfer or otherwise dispose of the Assigned Agreement, and any such substitution, sale, assignment, transfer or other disposition in violation of this Section 3 shall be void and of no force and effect. In addition, if FPL provides such prior written consent, with respect to (a) and (b), Lender and any third party assignce or transferee must first satisfy the requirements set forth in Section 16.1(a) through (e) of the Assigned Agreement. If FPL provides such prior written consent and such conditions are satisfied, the Lender may (but shall not be obligated to) assume, or cause any purchaser at any foreclosure sale or any assignee or transferee under any instrument of assignment or transfer in lieu of foreclosure to assume, all of the interests, rights and obligations of LCEC thereafter arising under the Assigned Agreement, subject, however, to the provisions of Section 12.3 and 16.1 of the Assigned Agreement. If the interest of LCEC in the Assigned Agreement shall be assumed, sold, assigned, transferred or otherwise disposed as provided above and FPL shall have given its prior written consent thereto, the assuming party shall agree in writing to be bound by and to assume the terms and conditions of the Assigned Agreement and

SSD\_ADMIN/106297.1

any and all obligations to FPL arising or accruing thereunder from and after the date of such assumption, and FPL shall continue to perform its obligations under the Assigned Agreement in favor of the assuming party as if such party had thereafter been named as LCEC under the Assigned Agreement.

4. <u>Notices</u>. Notice to any Party hereto shall be in writing and shall be deemed to be delivered on the earlier of: (a) the date of personal delivery, (b) if deposited in a United States Postal Service depository, postage prepaid, registered or certified mail, return receipt requested, or sent by express courier, in each case addressed to such party at the address indicated below (or at such other address as such party may have theretofore specified by written notice delivered in accordance herewith), upon delivery or refusal to accept delivery, or (c) if transmitted by facsimile, the date when sent and facsimile confirmation is received; provided that any facsimile communication shall be followed promptly by a hard copy original thereof by express courier;

The Lender:

LCEC:

Attn: \_\_\_\_\_\_ Telephone: (\_\_\_\_\_\_\_ Facsimile: (\_\_\_\_\_\_

Lee County Electric Cooperative, Inc. 4980 Bayline Drive North Fort Myers, FL 33917-3910 Attn: Chief Executive Officer Telephone: (\_\_\_\_\_\_\_\_\_ Facsimile: (\_\_\_\_\_\_\_\_

With a copy to:

Florida Power & Light Company 700 Universe Boulevard Mail Stop EMT/JB Juno Beach, FL 33408-0428 Attention: Vice President Telephone: ()

FPL:

SSD\_ADMIN/106297.1

157

Facsimile: (\_\_\_\_)

With a copy to:

5. <u>Successors and Assigns</u>. This Consent shall be binding upon and shall inure to the benefit of the permitted successors and assigns of FPL, and shall inure to the benefit of the Lender, LCEC and their respective permitted successors, transferees and assigns.

6. <u>Counterparts</u>. This Consent maybe executed in one or more counterparts with the same effect as if the signatures thereto and hereto were upon the same instrument.

7. <u>Governing Law</u>. This Consent shall be governed by and construed in accordance with the laws of the State of Florida (regardless of Florida's or any other jurisdiction's choice of law rules).

IN WITNESS WHEREOF, the parties hereto have caused their duly authorized officers to execute and deliver this Consent and Agreement as of the date first written above.

FLORIDA POWER & LIGHT COMPANY

By:	 
Name:	 
Title:	

SSD\_ADMIN/106297.1

## [LENDER NAME], as Lender

.

By:	
Name:	
Title:	 

Acknowledged and Agreed:

## LEE COUNTY ELECTRIC COOPERATIVE, INC.

Ву:	
Name:	
Title:	

### Assigned Agreement

Long-Term Agreement for Full Requirements Electric Service, dated as of August 21, 2007, between Florida Power & Light Company and Lee County Electric Cooperative, Inc

SSD\_ADMIN/106297.1

.

APPENDIX J

NOT USED

## APPENDIX K

#### **BILLING AND CP EXAMPLES**

		Example 1		Example 2		Example 3	
1	Loss %		3.20%		3.20%		3.20%
2	Loss Factor (1-Line 1)	0.96800		0.96800		0.96800	
3	Highest 60 Minute kW Demand Measured At the Delivery Points	980,000		972,700		<del>9</del> 22,000	
4	kW Qualifying Facility Purchases Behind the Delivery Points During 60 minute Interval for Measured Demand		-	4,000		50,000	
5	Maximum Allowed kW Credit For Qualifying Facility Purchases	5,000		5,000		5,000	
6	If line 4 is greater than line 5 then take line 4 minus line 5		-		-	45,000	
7	kW Renewable Energy Purchases Behind the Delivery Points During 60 minute Interval for Measured Demand		-	2,200		3,000	
8	Maximum Allowed kW Credit For Renewable Energy Purchases	2,000		2,000		2,000	
9	If line 7 is greater than line 8 then take line 7 minus line 8		-	200		1,000	
10	kW Distributed Generation Purchases Behind the Delivery Points During 60 minute Interval for Measured Demand		-	800		5,000	
11	Maximum Allowed kW Credit For Distributed Generation Purchases	2,000		2,000		2,000	
12	If line 10 is greater than line 11 then take line 10 minus line 11		-		-	3,000	

13	Monthly Billing KW Demand ( Sum Of Lines 3, 6, 9 and 12 )	980,000	972,900	971,000
14	Monthly Billing kW Demand Adjusted For Losses (Line 13/ Line 2)	1,012,397	1,005,062	1,003,099

		Exan	nple 1	Example	2 <u>Example</u>	3
1	Loss %		3.20%	3.2	20% 3.	20%
2	Loss Factor (1-Line 1)	0.96800	)	0.96800	0.96800	
3	60 Minute kW CP Demand Measured At the Delivery Points	900,000		892,700	842,000	
4	Adjustment for kW Load Management Activated Behind the Delivery Points During 60 minute Interval for Measured CP Demand	37,000		30,000		-
5	60 Minute kW Summer CP Demand Measured At the Delivery Points	950,000		950,000	950,000	
6	SLM Ratio		4.933%	4.93	3% 4.93	33%
7	Maximum Allowed Load Management kW (Line 5 x Line 6)	46,864		46,864	46,864	
8	Actual Verified Load Management kW For Summer Months	37,000		52,000	45,000	
9	Allowed Load Management kW Adjustment (Lower of line 7 or line 8)		37,000	46,8	64 45,0	100
10	kW Qualifying Facility Purchases Behind the Delivery Points During 60 minute Interval for Measured CP Demand		-	4,000	50,000	
11	Maximum Allowed kW Credit For Qualifying Facility Purchases	5,000		5,000	5,000	
12	If line 10 is greater than line 11 then take line 10minus line11		-		- 45,000	
13	kW Renewable Energy Purchases Behind the Delivery Points During 60 minute Interval for Measured CP Demand		-	2,200	3,000	
14	Maximum Allowed kW Credit For Renewable Energy Purchases	2,000		2,000	2,000	

15	If line 13 is greater than line 14 then take line 13 minus line 14	·	- 200	1,000
16	kW Distributed Generation Purchases Behind the Delivery Points During 60 minute Interval for Measured CP Demand		- 800	5,000
17	Maximum Allowed kW Credit For Distributed Generation Purchases	2,000	2,000	2,000
18	If line 16 is greater than line 17 then take line 16 minus line 17	-		- 3,000
19	Monthly CP kW Demand (Lines 3+4+12+15+18 minus Line 9)	900,000	876,036	846,000
20	Monthly CP kW Demand Adjusted For Losses (Line 19/ Line 2)	929,752	904,996	873,967

#### APPENDIX L

#### LOSS OF MEMBERS

#### Page 1 of 3

#### LOSS OF MEMBERS CHARGE CALCULATION (TABLE 1 - Reference Columns are on Page 2 of 3)

- (A) Applicable Calendar Year Column (1)
- (B) Customer provides the total number of Member Accounts as of December 31 for the current Calendar Year Column (2)
- (C) Subtract the number of Member Accounts in the immediately prior Calendar Year from the amount in the current Calendar Year to determine the amount of Member Accounts lost Column (3)
- (D) Customer provides the total kWh consumption of Member Accounts lost based on the consumption during the 12-month period immediately prior for each Member Account lost Column (4)
- (E) Add Member Accounts lost for the current Calendar Year and the two prior Calendar Years Column (5)
- (F) Identify the highest number of total Member Accounts in the current Calendar Year and the two prior Calendar Years Column (6)
- (G) Calculate the percentage of Member Accounts lost by dividing the result in step (E) by the result in step (F) Column (7)
- (H) If the percentage calculated in step (G) is greater than 3%, the threshold has been exceeded Column (8)
- (I) If step (H) results in the threshold being exceeded, calculate the average Calendar Year consumption per Member Account lost based on the current Calendar Year and the prior 2 Calendar Years Column (9)
- (J) Multiply the result of step (I) by the result of step (C) in order to determine the Calendar Year kWh Contribution to be included in the calculation of the Loss of Members Charge Column (10)
- (K) Determine the final Cumulative kWh Contribution to be included in the calculation of the Loss of Members Charge by summing the current year and all prior year values obtained in step (J) - Column (11)

#### LOSS OF MEMBERS CHARGE CALCULATION (TABLE 2 - Reference Columns are on Page 3 of 3)

- (A) Applicable Calendar Year Column (1)
- (B) Cumulative kWh Consumption lost used in the calculation of the Loss of Members Charge from Column (11) in Table 1 Column (2)
- (C) Actual Calendar Year Average Monthly kWh Consumption data that will be used in the calculation of the Average Monthly Load Factor Column (3)
- (D) Actual Calendar Year Average Monthly kW Demand data that will be used in the calculation of the Average Monthly Load Factor Column (4)
- (E) Divide the applicable Calendar Year data from step (C) by the product of the applicable Calendar Year data from step (D) and 730 hours Column (5)
- (F) Divide the Cumulative kWh Contribution from step (B) by product of the result obtained in step (E) and 730 hours to determine the Calendar Year kW demand applicable to the Loss of Members Charge Column (6)
- (G) The annual Loss of Members Charge equals the kW amount obtained in step (F) multiplied by the actual Yearty Generation Demand Charge Rate for the Calendar Year as trued-up.
### Page 2 of 3

# LOSS OF MEMBER'S ACCOUNTS EXAMPLE (TABLE 1)

### Historical Actual Total Member Accounts Are Set Forth For the Calendar Years 2004, 2005 and 2006.

(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
	Customer Supplied Data	(2) Current Year minus (2) Prior Year	Customer Supplied Data	SUM (3) - Current Year plus Prior 2 Years	MAX (2) - Current Year through Prior 2 Years	(5)/(0)	(7) > 3%, YES (7) < 3%, NO	SUM (4) / SUM (3) - Current Year through Prior 2 Years	(3) * (9)	SUM (10) - Current Year through all Prior Years
Calendar Year	Celenciar Year Total Member Accounts	Calendar Year Total Member Accounts Lost	Total Calendar, Year kWh Consumption of Mamber Accounts Lost	Rolling 3 Year Cumulative Member Accounts Lost	Rolling 3 Year Highest Number of Member Accounts	Percentage of Member Accounts Lost	3% Threshold Exceeded?	Total Average Calendar Year Consumption per Member Account Lost	Calendar Year kWh Contribution to Loss of Members Charge Calculation	Cumulative kWh Contribution to Loss of Members Charge Calculation
2004	171,000	<u> </u>			171,000	0.00%	NO	•		-
2005	183,000				183,000	0.00%	NO	<u> </u>		-
2006	195,000			_	195,000	0.00%	NO	-		-
_2007	193,000	2,000	32,400,000	2,000	195,000	1.03%	NO	-	-	-
2008	189,000	4,000		6,000	195,000	3.08%	YES	16,200	64,800,000	64,800,000
2009	188,000	1,000	15,200,000	7,000	193,000	3.63%	YES	16,200	16,200,000	81,000,000
2010	190,000			5,000	190,000	2.63%	NO	-	-	81,000,000
	195,000	-		1,000	195,000	0.51%	NO	-	-	81,000,000
2012	200,000			-	200,000	0.00%	NO	-	-	81,000,000
2013	198,000	2,000	36,000,000	2,000	200,000	1.00%	NO	•	-	81,000,000
2014	198,000	-		2,000	200,000	1.00%	NO	-		81,000,000
2015	200,000			2,000	200,000	1.00%	NO	-	-	81,000,000
2016	150,000	50,000	1,185,000,000	50,000	200,000	25.00%	YES	23,700	1,185,000,000	1,266,000,000
2017	149,500	500	11,850,000	50,500	200,000	25.25%	YES	23,700	11,850,000	1,277,850,000
2018	150,000			50,500	150,000	<u>33.</u> 67%	YES	23,700	-	1,277,850,000
2019	150,000	·		500	150,000	0.33%	NO	-	-	1,277,850,000

#### Page 3 of 3

# LOSS OF MEMBER'S ACCOUNTS EXAMPLE (TABLE 2)

(9),	(ç)	(7)	(2)	(z)	(j)
(2) / (2)	(3) / ((4),730)	Actual Calendar:Year Data	Actual Calendar Year Data	(11)-1 əldisT	
Annual KW Demand Subject to Loss of Members Charge	yintinoM egsievA LossT bso.1	Average Monthly Wy Demand	ylnîno₩egaravA notiqmuzno⊃ nWA	Cumulative KWh Contribution to Loss of Members Charge Calculation	alendar Year
-	%81'09	000'£21	000'000'92	-	2004
-	%07'09	000'921	000'009'22		5002
	%22.09	000,081	000,002,67		9007
	%51.09	000,1481	000,008,08		1002
599'ZÞL	%11.09	000,881	85,500,000	64,800,000	8002
184,703	%20.09	000,201	84'500'000	000,000,18	5009
184'602	%11.09	000'961	600,000,88	000,000,18	2010
157,481	%20.09	200'000	000,007,78	81,000,000	1102
184,832	60.03%	204°000	600,000,68	000'000'18	2012
757,481	%90'09	508'000	000,005,16	000,000,18	2013
297,481	%90.09	000,108	000'000'566	000,000,18	5014
185,600	%87.63	000'826	405,000,000	000,000,18	5102
2,878,643	<b>%97.09</b>	000'996	000'000'077	1'566,000,000	2016
2,918,253	%86.65	982,000	430'000'000	1,277,850,000	2102
2,917,083	%10.09	000'600'L	442,000,000	1,277,850,000	8102
5/9/6'5	%0.03	- 000'9€0'1	000,000,484	000,028,775,1	6102

\* Column Referenced in Section 16.5 (a) and 16.5 (d)

# APPENDIX I

Retail Impact Analysis (July 2007)

EXHIBIT TWG-2 DOCKET NO. \_\_\_\_\_ PAGES 1-10 November 7, 2008

					Ŀ	Retail Impact	teat		Fuel		ē	Demand and Energy	y .	
	Base Case - Naciear / IGUC	[Lee Case - Nuclear / IGUC resource]	Base - Reserve			-				ł	000	a a	Date	
	resource plan	plan	Margin	Lee - Reserve Margin Y	Year				ASPO	CCM		0000	Vene	
Year					2	Millions 5	Millions 5	Millions \$	Millions \$	Millions 5	Millions \$	Millions \$	Miltons \$	Check
2010	WCEC 2	WCEC 2	22.6%		010	E	6	5,290	5,256	(34)	3,079	3,106	27	(o)
2011	104 MW Nuclear Upgrade	WCEC 3, 104MW Nuclear Upgrade	20,4%		2011	35	29	5,033	5, 135	101	3,200	3,225	(65)	0
2012	WCEC 3. 304MW Additional	304MW Additional Nuclear Upgrade	24.4%	23.3% 2	2012	(22)	1	5,471	5,476	~	3,620	3,594	(27)	(0)
2013			22,5%	-	013	-	14	5,896	5,858	1351	3,854	3,699	45	0
2014		CC CC	20.6%		014	28	43	8,271	6,236	(36)	4,022	4,085	\$	9
2015	8	S	23.0%		015	(36)	4	6,890	6,840	(00)	4,373	4,384	1	•
2016	8	y	20.0%		910	18	21	7,920	7,902	(81)	4,745	4,781	35	0
2017	20	8	22.2%		110	2	53	8,719	8,563	(26)	5,183	5,241	58	(0)
2018	Nuclear I	Nuckar I	23.9%		018	12	8	9,279	8,186	63	6,148	6,251	<u>1</u> 05	(Q)
2019			21.4%	21.2%	610	32	69	8,842	9,851	(25)	6,424	6,548	124	0
2020	Nuclear 2	Nuckar 2	22.6%		020	42	110	10,515	10,391	(123)	7,310	2,478	186	(0)
2021	1 1-600MW IGCC	1-600MW JGCC	22.2%		2021	50	160	10,929	10,795	(134)	7,618	7,603	184	(0)
2022	Nuclear 3	Nuclear 3	23.1%		022	59	218	11,221	11,045	(176)	8,714	8,849	234	6
2023	1	S	20.2%		023	(27)	192	11,678	11,535	(0+1)	8,841	8,067	116	Ð
2024	Nuclear 4	Nuclear 4	21.0%		1024	(141)	51	12, 155	11,939	(215)	9,774	9,B49	74	0
2025	1-600MW JGCC, CC	1-600MW IGCC	23.3%	22.5%	520	(33)	18	12,739	12,463	(276)	10,162	10,405	243	0
2026	00	g	23.4%		026	110	127	13,563	13,409	(254)	10,568	10,932	364	0
2027	ł	22	20.1%		120	(23)	104	14,682	14,423	(256)	10,945	11,100	236	(Q)
2028	00		21.1%		820	Ē	101	15.707	15,398	(906)	11,202	11,507	305	(0)
2029	00	cc	22,1%		2029	156	259	16,505	16,322	(273)	11,474	11,804	430	0
2030	Nuclear 5	Nuclear 5	1.9.77		0200	104	E++3	17.044	18.742	(302)	12,067	13,152	981	0
2031		20	50.0%		1031	(18)	425	17,682	17,328	(354)	12,787	13,123	90E	6)
2032	Nuclear 6	Nuclear 6	20.5%		202	(175)	250	18,234	17,789	(445)	13,838	14,108	270	0
2033	CC		21.4%		6033	26	276	18,796	18,370	(428)	13,685	14,137	452	( <u>)</u>
2034	8	1	X0'77			276		272,350	268,352	(3,996)	194,321	194,596	127	6
2035	<u>cc</u>	cc cc	22.74	22.7%	ļ		•							
Jul CC units added:	1ded: 12	12												
New CC Capacity 2010-2035 (NW)		14,628		Cumulative Negative (Altions \$)	( <b>§</b> suo	(487)								
				Comparison Buckless of State	10	787								

Veltow highlighted entries denote changes from basecase expansion plan.

Cumulative Negative (Antitions \$) Cumulative Postitive (Millions \$) NPV (Millions \$)

16 5) (487) 15 5) 763 16 5) 95

	<u>و</u>		7		1	Ť	Ţ	1	η	Τ	٦	-	η	Ϊ	Т	-	ļ	1	-1	Ţ	Ţ	Ī	1		Τ	Τ	٦	-1	η		Π	
	FPL Fim 12 CP (MN)					19,450	19,048	20,184	20,546	20,903	21,363	21,860	22,360	22,654	23,347	23,961	24,393	24,892	25,597	26,208	26,835	27,454	28,076	28.705	28,351	29,896	30,642	31,295	31,961			602,106
	FPL & Other Total 12 CP (MW)					21,613	Z2,109	22,546	22,969	767.62	24,032	24,611	25,209	25,803	26,395	27,008	27,609	28,206	28,613	29,424	30,051	30,670	31,284	31,924	32,567	33,212	33,859	34,511	35,177	_		673,052
	LOEC 12 CP (MW)																															•
	FPL Load Management (MW)					2.1.2.008	2.05	0000	ELLC	12.2424	965.5	1103 2 541	2,40	2.839	866,25	7.7 3,037.	201 (E) +32	3)06	AC - 3 106	·901(11)(10)	301°C 3.106	301121106	301 CF	901'E	30(T.	2	301 Ed. 32	301.0	1.05.106			60,171
	LCEC Load Management - Ukodified Firel 4 Years (MW)																															0
	NPV Cumulative Capital Costs 2010- 2035 (Millions)					256	534	963	1,519	2,064	2,569	3,364	4,106	5,232	6,240	1,394	8,546	0,850	11,005	12.213	13,399	14,570	15,685	16,699	17,663	18,714	19,826	20,541	21,354	22,094	22,760	
	NPV Annual Capital P Courte 2010-2035 Co	•	142	127	174	256	211	450	535	545	804	718	104	1,044	1,008	1,154	1,153	206,1	1,155	1,208	1,186	1,171	1,095	1,034	186	1.031	913	914	613	740	e75	22,708
(Asseming all Customers are Retail)	8					15.86	95,44	96.90	102,29	104.19	106.38	116.62	122.76	130.39	133.96	140,06	143.06	146,38	149.26	153,36	156.94	162.33	165.96	189.25	173,13	176.23	177.50	180,87	180.78	163,45	186.05	
<	FPL Orey Arrual Energy Salars (GWH)					120,195	124.040	127,860	130,872	134,348	138,011	141,944	145,289	148,977	152,649	156,889	160,352	164,189	168,075	172,443	176,127	180,161	184,215	188,747	192,545	196,789	201,033	205,821	209,732	214,267	218,895	
	Tink Ore	5,749	5,787	5,879	5,934	6,058	6,116	6,248	6,379	6,503	6,626	6,754	6,883	7,010	7,148	7,281	7,544	7,688	7,838	7,992	8,148	8,309	8.474	8,644	8,818	8,999	9,189	9,386	9.588	9,797	10,020	
118	NPV Cumdates Annual Control		5,037	668,6	14,231	18,355	22,105	25,957	20,835	33,647	37,540	41,748	46,071	50,573	55,006	905'85	928'69	962'89	72,329	76,359	80,271	M,133	87,879	01,506	95,017	96.446	101,651	104.743	107,634	110,402	113,085	
101	2 M H	•	5,037	4,902	4,292	4,124	3,750	3,852	3,676	3,811	3,694	4,209	4, 322	4,502	4,433	4,500	4,369	4,255	4,099	4,030	3,812	3,851	3.746	3,629	3,509	3,429	3,205	3,052	2.691	2,765	2,663	
ĕ	Total Annual Costs (Manual	0	5,402	5,805	5,530	5,783	5,722	6,307	7,006	7.495	555,8	9,800	10,952	12,415	13,301	14 (82)	15,525	16,838	17.249	18,454	19,483	20,837	22,103	23,302	24,517	28,075	26,515	27,BM1	ZE 327	28,510	30,903	113,065
8	Fitand Gau Trensportation (Millions)	0	112	317	379	392	392	441	473	473	551	687	524	882	882	883	282	882	882	883	982	1,157	1,230	965,1	1,527	1,606	i.606	1,607	1,729	1,943	2,163	5,937
E	Emissions Conta (Millions)	•	(28)	(46)	(J0)	(62)	61	96	216	293	399	563	671	767	889	1,039	1,226	1,393	1,542	1.681	1,844	2,085	2,310	2,478	2,578	2,579	2,650	2,710	2,756	2,897	3,068	6,968
æ	System Fuel (Milliona)	-	4,983	5,289	4,B96	4,986	1771	4,986	5,220	5,523	5,947	6,715	7,236	7,609	8,156	8,550	B,772	8,857	9,201	9,469	9,733	10,269	10,989	11,690	12,330	12,673	13.191	13,594	14,010	14,845	15,829	72,107
5	n Filmed Mens)	0	12	12	20	¥	40	48	55	56	65	80	95	189	257	356	455	582	672	797	932	966	1,034	1,077	1,129	1,299	1,435	1,617	1,781	1,860	1,942	3,074
Ĩ	₩ Ø ¥	0	84	82	88	84	78	77	7	11	78	88	89	8	92	94	92	89	91	93	8	82	80	62	61	78	3	80	34	7	72	587
ص تو	<u>.</u>	0	154	150	224	359	423	747	967	1,072	1,293	1,667	2,038	2,878	3,024	3,770	4,098	5,035	4,860	5,532	5,911	6,348	6,460	6,639	6,875	7,840	7,550	8,233	7,967	7,890		23,211
0.0682	Armust Discount Factor	1.000	0.919	0.844	0,776	0.713	0.655	0.602	0.553	0.509	0,467	0.429	0.395	0,363	0,333	0.308	0.201	0.259	0.238	0.218	0.201	0.184	0,168	0,158	0.143	0,132	0.121	0.111	0.102	Nei O	0.066	Total NPV =
⊔ ₽	Year	1900Z	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021	2022	2023	2024	\$202	2026	202	2026	502	2030	1602	202	2033	2034	2035	

Columns A through P (Resource Planning)

Generation Capital Includes Generation Capital, and PPA Capacity payments FOM includes Field OBM, Capital Replacements Charges, and Sold Fuel Capital Adda .

	Ī	FPL (\$AAWH)	Τ	T		ľ	40.59	8.8	39.88	21.43	80				2 2 2	10.76	8	81.24	62.25	62 20	63.46	50.12	65.43	60.62	72.40	75.44	78,39	78.66	79.71	79.93	B1.00			63.30
		Fuel Costs (Millions) (		-	+									ſ	┢													_		-				
	-	Cither Tolal L Fuel Costs F (Millions)					ŝ	47	4	j.			5	3	8	21	= :	8	8	87	8	*	52	102	8	8	5	116	£1 <del>8</del>	122	125			2,032
		FPL Total Fuel Costs (Millions)					5,256	5,135	5,476	5.858	6.236	6 240	1 000			B01.9	100'2	162.01		11,045	11,535	11,939	12,463	13,409	14,423	15,398	16,322	16,742	17,328	17,789	18,370			268,352
	ł	Refer		╎	+	╞	-	[			┝			┢	$\frac{1}{1}$			╀	╎	+	┥	┦	-		1	-						┤	_	
		FPL Nuclear	┢	┢		-	100.000%	100,000%	100.000%	100.000%	100.000%	100.000		Too oot	100000		K.000 00	Kono'oot	1000	X.000.001	100.00%	100.00%	100.00%	100.000%	100.000%	100.000%	100.000%	100.000%	100.000%	100,000%	100.000%			
	ľ	Fossi F	l	ſ					ſ		t		ŀ	T	1		ļ		t			-+						1						-
		Come Found Radio	ĺ	ſ	ſ		0.956%	%166'0	0.913%	0.691%	0.672%	0.641%	A BIJC	7844	7074	0 70542		CONC.		2,808.0	0.803%	508.0	0.801%	0.778%	0.751%	0.725%	0.706%	0.707%	0.702%	0.703%	0.702%			-
	-	FPL Food		ļ			99.042%	\$89068%	¥180.68	⊢	t.	⊢	┢	╀	╀	┢	t	╈		t	+	┿	+	┽	+	+	-	╉	1	+	¥962-66			
ſ	+	. <u> </u>			┝							-					t	╈	t	T		┢	+		╈	┥	1	1	+		+	+		•
	(RAP)	Totel Fuel Coel (Millions)					5,305	5,182	5,525	5,000	692-9	8,036	7 045	N2.4	970	BCO D	10 T			251,11	11,525	12031	12,560	115,51	14,529	15,507	18,435	16,858	17,447	17,911	16,485			Z70.384
		Food Fuel Coel (Millions)					5,178	5,049	5,378	5,756	6,125	6.731	7 107	A 557	0.047	0 506	100.01	10,001	202.01	10,01	11,257	17011	12,120	13,056	14,070	SED'CI	15,955	16,333	10,534	17,295	12,834			282,373
	(-M-)	Nuclear Fuel Cost (Millions)					121	(3)	147	153	ŧ	165	16A	174	SUS.	Pic Pic	5	17		656			140	20		89	420	52	293	619	660			8,D11
-		LCEC Nuclear Fuel Energy (GWH) Cost (Netrons)																										ĺ						
		FPL Nuclear 1 Energy (OWH) E					22,582	23,765	25,774	26.361	27,751	27,353	55E /Z	27.751	105 02	76,060	12	2011	10 000	200.94	100			07,000	62.491	DAC 20	62,059	67,842	71,180	76,558	80,540			1 138 289
	1	Nuclear Energy (GWH) E					22,582	23,785	25,774	26,361	27,751	27,353	27.355	27.751	32.504	36.060	41 790	44 922	10 005	Cient and		800'AC	42/414	800'70	145.20	0.007.0	680720	67,542	1186	78,556	80,540			1,130,230
L		Total NEEL (MWH)				-	130,521	134,782	139,043	142,422	148,307	150,319	154,570	158,164	162,168	166.136	170.748	174 400	174 GAG		104,040		20218	C00'081	BNZ'002	201 ON7	192,902	Z13,833	218,415	223,592	227,518			4 204 248
							1,034	1,034	1,034	1,034	1,034	1,034	1.034	12	1,034	1.034	1024	101	201	-		100	ten'		10.04		HCD'L	1.034	153	1,034	1,034			010'52
		LCEC NEL (MWH) OTHER (MWH)							-																							T		
		FPL NEL (MWH)					129,467	85/ 521	138,009	141,308	145,273	149-285	153,536	157,150	161,134	165.102	169.684	173.426	177 572	101 277	101,172	11.101	100 101	170'001	CITRAL		CIZ BOZ	212,789	211-204	222,558	226,754		1 222 222	714 5074
		FPL & Other NEL (MWH)		i			125,061	134,792	139,043	142,422	146,307	150,319	154,570	156,184	162,165	166,126	170.718	174.460	17A GIM	182 8/16	102 695	101 506	A DE REE	100,141	The Line	200.047	242 202	213,053	210,410	205 622	227,518			101 210
		Other 12 CP (MW)					55	Ð	155	<u>5</u>	10	110	011	91	110	110	2	110	<b>B</b>	2	-	-	2			2				110	0			5113
		LCEC FJm 12 CP (MW)																															4	>
Ξ	Ľ	Year	8 X	1002	2006	8 R	2010	ş	2012	107	2014	2015	2016	2017	2010	\$0 20	2020	Ř	2022		-	2025			acuc.	0.01				2002			ŝ	

\_....

4

UCED         And Number         Examp Yound Muchan         Command Muchan <thcommand muchan<="" th="">         Command Muchan<th>(RAP) In Demand</th><th>Total Existing homomorulal homomorulal homological forest homological</th><th></th><th></th><th>384.751 0</th><th></th><th>397,128 0</th><th>403,584 D</th><th>409,661 0</th><th>416,095 0</th><th>422,546 0</th><th>428,096 0</th><th></th><th>442,447 0</th><th>448,447 0</th><th>458,592 0</th><th>463,696 0</th><th>471,443 0</th><th>479,196 D</th><th>487,096 0</th><th>495,550 0</th><th>503,369 D</th><th>511,741 0</th><th>520,275 0</th><th></th><th></th><th></th><th></th><th></th><th></th></thcommand>	(RAP) In Demand	Total Existing homomorulal homomorulal homological forest homological			384.751 0		397,128 0	403,584 D	409,661 0	416,095 0	422,546 0	428,096 0		442,447 0	448,447 0	458,592 0	463,696 0	471,443 0	479,196 D	487,096 0	495,550 0	503,369 D	511,741 0	520,275 0						
All (500001)         Contraction	(HAP)				139,229	141,161	143,185	145,242	147,250	149.243	151,310	153,394	155,458	157,863	159,508	162,159	184,478	106,065	169.315	171,810	174,351	175,936	_		185,164	154,130	191,155	194,238		-
A (50411)         Exercision Conversion Conversion         Conversion Conversion         Conversion         Conversion <thconversion< th=""> <thconversion< th="">         C</thconversion<></thconversion<>	_									_					-	_		_									╞			_
A (50001)         Extern (5001)         Total Extern (5001)         Total Extern (5001)         Total Extern (5001)         Connection (5001) <th< td=""><td></td><td></td><td>+</td><td></td><td></td><td></td><td></td><td>-</td><td></td><td></td><td>_</td><td></td><td></td><td></td><td>-</td><td></td><td>-</td><td></td><td>-</td><td></td><td></td><td></td><td>-</td><td></td><td></td><td> </td><td></td><td></td><td></td><td></td></th<>			+					-			_				-		-		-				-							
At (50000)         Extension frozen         Total Extension         Total Extension         Total Extension         Commental Found         Commental Foun					1,817,001	1,850,735	1,928,85	1,005,050	1,984,315	2,078,77	2,182,44	2,350,749	2,404,400	2,376,94	2,353,960	2,484,78	2,687,931	2,648,05	2,510,58	3,051,55	3,445,21	3,472,70	3,554,86	3,748,60	3,701,25	3,707,6	3,506,99	3,871,06		
A (50000)         Extension Notices         Extension Notices         Convention Notices         Convices         Convices         Co					\$10'126	1,001,618	1,256,789	1,513,034	1,704,956	1,903,403	2,190,179	2,475,918	3,427,055	3,741,661	4,667,724	4,877,032	5,814,162	5,653,850	6,775,573	6,884,967	2,011,843	7,224,108	7,450,279	7,855,240	B.862,540	E,696,439	966,961	9,926,068		
At (54000)         Exating functions         Exating functions         Communication (000)         Communication (000)           Communication         Communication (000)         Communication (000)         Communication (000)         Communication (000)           Communication         Communication (000)         Communication (000)         Communication (000)         Communication (000)           2014         Program         1,283,345         2,331,255         S1,142         S1,142           2014         Program         1,283,345         2,331,255         S1,142         S1,142           2014         Program         1,283,345         2,341,255         S1,142         S1,142           2014         Program         1,283,345         2,341,355         S1,142         S1,142           2014         Program         1,283,345         2,341,355         S1,122         S1,122           2114         Program         2,341,355         S1,122         S1,122         S1,122           2114         Program         1,280,346         2,244,365         S1,122         S1,122           2112         Program         1,280,346         2,244,365         S1,126         S1,122           2112         Program         1,280,467         2,244,167					477,40M	540,527	872,173	1,000,669	1,205,411	1,435,098	1,634,930	2,221,393	3.156,737	3,373,416	4,219,747	4,845,406	5,705,795	5,623,632	6,421,532	6,932,511	7,426,319	7,574,359	759,937	6,062,605	9,216,966	9,067,344	9,930,042	9,832,318		
All (SAMVI)         Existing Nuclear         Existing Nuclear         Existing Nuclear         Existing Nuclear         Commution           All (SAMVI)         Ownerston         Commution         Commution         Commution         Commution           All (SAMVI)         Ownerston         Commution         Commution         Commution         Commution         COM           All (SAMVI)         Deminand (DOD)         Deminand (DOD)         Deminand (DOD)         Deminand (DOD)         (COD)           All (SAMVI)         Eligitii         1,380,116         1,380,116         2,320,212         21,400           All (SAMVI)         Eligitii         1,380,116         1,380,116         2,340,120         21,400           All (SAMVI)         Eligitii         1,380,116         1,380,122         24,112,223         24,112           All (SAMVI)         Eligitii         1,380,116         1,380,120         2,240,120         2,240,120           All (SAMVI)         Eligitii         1,380,120         1,380,120         2,240,120         2,240,120           All (SAMVI)         1,380,120         1,380,120         2,240,120         2,240,120         2,240,120           All (SAMVI)         1,380,120         1,380,120         2,240,120         2,241,123         2,24					423,262	456,726	559,939	619,764	600,776	687,905	858,948	1,019,663	1,065,774	1,032,362	858,459	1,304,624	1,409,677	1,452,278	1,405,206	1,806,480	2,335,930	2,357,153	2,430,044	2,614,575	2,636,967	2,552,907	2,424,015	2,496,320		
Al. (SAMVII)         Existing Nuclear         Existing Nuclear         Existing Nuclear           Al. (SAMVII)         Convertion         Convertion         Convertion           Convertion         Convertion         Convertion         Convertion           Stati         1916 (11)         1,382,145         1,382,145           Stati         24,44         1,916 (11)         1,382,145           Stati         24,46         1,382,145         1,387,155           Stati         1,100,2100         1,387,155         1,387,155           Stati         1,100,2100         1,382,145         1,387,155           Stati         1,100,2100         1,386,166         1,387,155           Stati         1,100,2100         1,386,166         1,387,155           Stati         1,100,2100         1,386,166         1,387,155           Stati         1,100,2101         1,386,166         1,106,766           Stati         1,100,201         1,286,260         1,101,765           Stati         1,286,450         1,117,860         1,286,460           Stati         1,286,450         1,117,860         1,286,460           Stati         1,392,460         1,117,860         1,286,450           Stati		Promental Nucle Onne dan Danse (000)			54,142	61,801	312,234	473,905	604,835	747,193	975,961	1,201,730	2,000,963	2,341,055	3,220,366	3,340,582	4,208,118	4,171,358	5,016,324	5,048,031	5,090,369	5,217,205	5,364,893	5,486,227	5,579,961	6,514,830	7,506,027	7,335,898		
AL (\$AAVVI) Plake (builder AL (\$AAVVI) Consertion Co		Total Existing Generation Demand (000)			2,262,621	2,311,526	075.EMC.2	Z411,223	2,483,862	2,547,080	2,537,095	2,805,274	2,674,719	2,747,191	2,621,946	2,716,410	2,786,305	2,679,073	2,964,727	3,004,109	3,030,737	3,122,455	3,220,204	3,321,243	3,426,053	3,536,035	3.048.912	3,765,613		
A (54000) 46.000 46.000 46.					1,383,745	1, 382,008	1,308,715	1,373,185	1,383,542	1,380,669	1,323,407	1,331,067	1.338,675	1,345,585	1,354,610	1,179,960	1,186,262	1,190,770	1,205,478	1,185,173	1,109,283	1,115,462	1,124,817	1,134,230	1,144,295	1,154,533	1,184,948	1.175,544		
					-	-	<u>874,555</u>	┥	+	+	+	-	+	┿	╋	+	+	+	-		+	-+	+	+	+		_	+		
					40.65	38.44	39.74	41.48	42.86	45.68	51.53	88	21.58	59,76	61.35	82.35	62.33	63.59	64.17	\$5.59	66.85	72.56	75.59	78.54	78.64	79.61	11.00	B1.15		
		Year Other Base (\$AMVH)	2005	2008							1					2021 51.70					1				_			2033 121.09	1007	

 4.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65
 64.65

5

.

	Ļ	2034	2033		2030	2028	2028	2028	2025	2024	£202	2022	100	2019	2018	2017	2010		2013	2012	2011	2010	2008	2007	2008	Base	
	7,196,549		364 432	350,948	344,418	336,019	332,168	320,805	315,269	309,881	304,578	298.420	100 402	284,784	280,054	275,672	271 226	262,611	259.341	253,843	249,627	245.533				Nuclear Energy (000)	
	3,949,459		194,239	184,130	185,164	182,256	179 573	174,351	171,810	169,315	100,005	184.476	108,009	157,683	155,456	153,394	0151210	147,250	145,242	143,185	141.161	139.229				Fassi Energy (1999)	Yotal Non-Fuel
	11,146,104		558,671	539,078	529,580	520,275	511.741	485,158	487,099	479,196	471,443	453 506	148,447	442,447	435,511	429,066	ans cor	409,861	403,564	397,128	390,769	314 7/11				Energy (000)	Total New-Frank
	199,015,460		14,127,728	13,142,957	13,173,401	11,824,323	11.520.001	10,932,212	19,423,720	9,865,455	8,974,148	7,010,407	7,491,140	6,563,054	6,286,965	5,255,733	4,398,273	4,098,135	3,913,476	3,612,571	3,243,139	1 174 788				ord Energy Demand (000)	Yolal Generation
	504,361		31,405	30,752	30,108	29,461	20,100	27_964	26,845	26,318	25,707	× 13	23,971	23,457	22,964	22.470	21,493	21,013	20,658	20,339	400.0C					(MM)	Total 43 /rB
	145,836		10,012	5,922	8,922	7,832	200 L 44-	7,832	7,632	7,632	6.742	2,652	5,852	4,562	4,582	1472	3,472	3,472	3,472	3,472	3,004					Capacity (NW)	
	145,839		10,012	0,922	8,922	7,832	7.032	7,832	7,832	7,832	0.742	2,852	5,852	4,562	4,562	2172	3472	3,472	3,472	3,472	3,084					(MAN)	
	•		• •	•	0				0	•	• •		C	9	•	-	, c	•	0	-						Nucley LCEC Fossi Capacity 12CP Ratio (MW)	
			0.514%	0.504	0.519	0.509	0.540	0.557%	0.576	0.595	0.500	0.584	0,600	0.582	0.590	CACO	01910	0.627	0.840	13-2	0.937%			+	-	Ratio 120P Ratio	
		11	× 9914301×									_	_					1		+	¥ 99,063%	+		+		alio 12CP Ralio	
																Ì										Admin. Cost (900)	LEC EC
	0			0	•	•	0			•	, •	0	0	¢	•	, .	-	9	•		•			ŀ		Energy Demand Share (009)	
	419,780	- 1416-	19,798	20,003	20.944	19,800 50,810	20,094	20,562	18,940	18,597	17,426	15,794	15,417	15,103	15,517	14,208	13,844	13,728	14.071	16,01	18,361					and Energy mend Share (000)	Other Generation
· · ·	199,595,880	075/05/	14,107,930	13,122,955	13.152.457	11,006,900	11,180,088	10,831,950	10,404,760	9.54/ A43	6,948,570	7,802,815	1,475,723	6,547,952	3,240,803	4,780,982	4,384,379	4,085,407	3,899,406	CD/ 2776	3,106,426					Energy Demand Share (000)	EPI Remealing and
																									(SMWH)	Demand Share (	LOEC
		Evite	19.15	19.34		19,23	19.43	19.89		16.16	18,85	15.27	14,91	14,61	14.34	13.74	13.49	13.28		17.75	17.76				(SUMWH)	Demand Share	Oltree
		670	63.39	7E 08	57.17	56,36	50,12	<b>56</b> ,11	1070	49.28	50.39	44.89	44.98	99.6E	33.35	31.14	29,37	26.12	10.01	24.11	23.98				(annual)	and Energy Demand Shake	FPL Generation
	469,399,340	50C1C97E	32,038,834	30,589,747	5/0/846'87	27,034,378	25,728,585	24,462,750	27 084 021	20,509,557	20,097,985	10,090,248	17,864,034	16,490,753	13,996,730	12,760,410	11,284,758	10,386,377	2,121,022	8,425,032	8,429,900					Total Cost W/O LDEC (000)	
	466,946,053	32,506,505		30,451,156	20,225,802	26,905,352	25,803,180	24,340,649	17,767,903	20,492,414	19,993,733	16,597,975	17,855,803						+	8,359,854						Al-In Cost W/O	

11.841.11	651'616'E	186,871,68	30,056,330
	_		<u>                                      </u>
19 800	662 961		
		818'997'8	HIS'S21'1
	551'161	3,648,912	896,961,1
	061.681	SE0'965'6	SEC ISI I
	192,154	£28,054,C	\$62'991'1
	182,256	3'351'543	062,461,1
	E25'821	3,220,204	218'921'1
	926'941	SCA.SZ1.E	Z55'511'1
	ISE'#21	767,060,6	1,109,263
	OIB'IZI	501 100'E	EZ1'591'1
	SIE.881	127,488,5	871,205,1
	266,881	2,079,073	8/// 961 1
	927'791	906'962'2	292,881,1
	861,281	014,017,5	096'621'1
	699'65L	5'851'846	018'995'1
	E99/251	161'471'2	595'976'1
15,552	122'428	2117,478,2	SZ0'9EC'1
158 06	186'551	\$12,808,2	190'166'1
	012 151	2,537,695	161 EZE'I
	842,841	2,547,080	698'065'1
	062 731	2,463,662	ZN2 686.1
	142'545	2411,223	561'9/2'1
	S81'E11	072,646,5	\$12'89£'1
	141'191	126,116,2	1'385'009
92'#8E	138,229	129'292'2	SPL DOE I
	_		
6aug	(sucies) (standard)	(snothi)	(succession)
	Line Lote	Generation	CONDUCTOR
	ON DUGGET		

.

0000       00000	20 12-241 1-144(246 2)(44(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(24 2)))))))))))))))))))))))))))))))))))		L						
Open         Open <t< td=""><td>20 12-241 1-144(246 2)(44(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(24 2)))))))))))))))))))))))))))))))))))</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>5032</td></t<>	20 12-241 1-144(246 2)(44(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(246 2)(46(24 2)))))))))))))))))))))))))))))))))))								5032
0000       20100       2010       2010	1/141(321(321)) 1/141(348) 3/646(31(32) 161(122) 246(345))	Cool Traci							95036
Markadia							32,460,564	119,910,55	1033
Windowsky Wardsong         Machinany (Wardsong	940'665 061'691 SE0'965'E 665'651'1 (625'41) BS							S#2 866 CC	ZEOZ
302     302 </td <td>085'825 1961'521 197'553'556'582'92'5 192'92'5 192'92'5</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>1602</td>	085'825 1961'521 197'553'556'582'92'5 192'92'5 192'92'5								1602
3023     3231/43     2521/43     2521/43     1020     1127611     11276111     11276111     11276111	(a) (2)*(2) (2)*(2)0 (2)*(2)0 (2)*(2)0 (2)*(2)0 (2)*(2)0 (2)*(2)0 (2)*(2)*(2)*(2)*(2)*(2)*(2)*(2)*(2)*(2)*								OCOZ
	(A)(1) 29								8202
123     25.591/260     666241     5.6180     1.561900     1.56190	3013 2C) 6E								2058
302     202,01,02     202,02     120,01     120,02     120,01     120,02     120,01     120,02     120,01     120,02     120,01     120,02     120,01     120,02     120,01     120,02     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,01     120,00 <td>cardion cardion cardion (Cardion )</td> <td>124.37 134.5</td> <td>124.80</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	cardion cardion cardion (Cardion )	124.37 134.5	124.80						
Market       Str. Start	AND THE DESCRIPTION OF THE OWNER	120.23 134							
325       25/18/200	chart in and and a second and a second	1123 05711	16.711	E9'911					
3011       3712/328       10/12/32       11/12/30       11/	192 (22 (24 )	SZI 69711	11334	115.74					
11/2016       11/2016	28/965 661 701 01/01/0 CR 86 8 9 9 93 02		115.60	85 211	115.53				
250     11,202,000	/##/S## ###/9/11 044/17/17 998/05 E8		107.24	101.24	81 201	966'929			_
Construction         Inclusion         Construction         Statution         Construction         C	John State			102'58	£2.201	192 399			
Clear (equ)         Clear (ord)         FPL Aborded Mith (cond)         FPL Aborded Mith (					92'66	918'06S	956'996'91		
Cost	200°875 400°661 400°661 53				62 98	961,852	12,424,700		
Cerr     FPL Aborded Mith (Fer     CPEC Aborded Mith (Fer Total Code)     CPE Tota	GPC772 DICICL continet? PS					012'567	516,100,61	095'629'91	1102
Cerrer (et al. (2))     FPL Aborded Math.     CEC Aborded	CROBIT PATCAL ODD' LATT CONTACT						12,866,357	13,193,520	5018
Construction     FPR_Monored     Main Construction     FPR_Monored     Main Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction     Construction     Construction     Construction     Construction       Construction     Construction <td< td=""><td>100'50t 007'101 T00'000-7</td><td></td><td></td><td>-</td><td></td><td></td><td></td><td>920'202'11</td><td>5102</td></td<>	100'50t 007'101 T00'000-7			-				920'202'11	5102
Cerr     FPL Aborded ML     CEC Aborded ML     Cert Aborded	1.376, 7,2066 1,376, 105 2,411,223 145,242 403,564								¥102
Rear     FPL Absorded Math (ear     CEC Absorded Math (Ear Total Code)     FPL Absorded (FPL Absorded Math (EarWork)     FPL Absorded (FPL Absorded Math (FPL Absorded Math (FPL Absorded Math)     FPL Absorded FPL Absorded Math)     FPL Absorded Math)     FPL Absorded Math)     FPL Absorded	221.7265 281,EM 072,EM5,2 217,686,1 3162,153 265								6102
Control         Control <t< td=""><td>692/196 194'191 920'116'Z 900'28C'1 621'92 92</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>2102</td></t<>	692/196 194'191 920'116'Z 900'28C'1 621'92 92								2102
Construction     FPL Abounded Mut (rector)     FPL Abounded (rector)     FPL	192729 1921 1387292 142 128729 192129 192129 192129 192129 192129 192129 192129 192129 192129 192129 192129 1921								1102
Control     FPL Aborated ML     CEC Aborated ML     FPL Abo						69702	950'696'8	995,702,0	
Control		· · · · · · · · · · · · · · · · · · ·			·				
2008     Coal W1LCEC     FPL. Abcred A14;     Coal W1LCEC     FPL. Abcred A14;     Coal W1CEC     FAA10 FA     FPL. Abcred A14;     FAA10 FA									
few     [FPL Aboration MutriceC     [FPL Aboration									
	MAH) pubmerty therman provided to the provided	10000 VIIV	COEC	MVO TCEC		in Cast (000)	(acc) Com MITCEC	(000)	
	avasi and see a	Peterowy 193	""	PORSONY 7.63	Total Cod		ri-IA beteraded J93	194 Cost With CEC	
			L				L, J		$L_{m}$

Fuel Cost M\$	BASE	BASE	LCEC	LCEC	BASE	BASE	LCEC	LCEC
Year	Existing Nuclear	New Nuclear						
2010	127	-	127	-	22,582	-	22,582	-
2011	133	-	133	-	23,765	-	23,765	-
2012	147	-	147	_	25,774	_	25,774	-
2013	153	-	153	-	26,361	-	26,361	-
2014	164	-	164	-	27,751	-	27,751	-
2015	165	-	165	-	27,353	-	27,353	-
2016	168	-	168	-	27,355	-	27,355	-
2017	174	-	174	-	27,751	-	27,751	-
2018	175	31	175	31	27,353	5,150	27,353	5,150
2019	178	53	178	53	27,276	8,785	27,276	8,785
2020	186	86	186	86	27,831	13,959	27,831	13,959
2021	186	110	186	110	27,353	17,569	27,353	17,569
2022	189	146	189	146	27,276	22,720	27,276	22,720
2023	196	172	196	172	27,751	26,354	27,751	26,354
2024	198	211	198	211	27,433	31,576	27,433	31,576
2025	201	239	201	239	27,276	35,138	27,276	35,138
2026	208	244	208	244	27,751	35,138	27,751	35,138
2027	209	249	209	249	27,353	35,138	27,353	35,138
2028	214	254	214	254	27,355	35,235	27,355	35,235
2029	221	259	221	259	27,751	35,138	27,751	35,138
2030	222	303	222	303	27,353	40,289	27,353	40,289
2031	226	337	226	337	27,276	43,923	27,276	43,923
2032	231	384	231	384	27,364	49,194	27,364	49,194
2033	240	420	240	420	27,833	52,707	27,833	52,707
2034	248	429	248	429	28,217	52,707	28,217	52,707
2035	249	437	249	437	27,742	52,707	27,742	52,707

Nuclear Summary Variable TAB (RAP)

	ummary Fixed TAE			Capital Revenue Requiremen	nt		
BASE							
	Uprates (\$000)	Nuclear 2018 (\$000)	Nuclear 2020 (\$000)	Nuclear 2022 (\$000)	Nuclear 2024 (\$000)	Nuclear 2030 (\$000)	Nuclear 2032 (\$000)
Year	-		-		-		-
2006	-	-	-	-	-	-	-
2007	-	26,848		-	-	-	-
2008	-	26,848	-	-	-	-	-
2009	-	2,282	29,039	-	-	-	-
2010	-	25,103	29,039	-	-	-	-
2011	-	47,924	2,468	31,408	-	-	-
2012	169,236	84,437	27,151	31,408	-	-	-
2013	246,658	143,771	51,834	2,670	33,971	-	-
2014	237,735	212,234	91,327	29,367	33,971	-	-
2015	229,221	303,518	155,503	56,064	2,888	-	-
2016	221,085	394,801	229,552	98,780	31,763	-	-
2017	213,300	431,314	328,285	168,192	60,639	-	-
2018	205,839	1,016,766	427,017	248,284	106,840	-	-
2019	198,677	943,391	466,510	355,073	181,917	42,985	-
2020	191,683	906,434	1,099,734	461,861	268,544	42,985	-
2021	184,712	871,753	1,020,372	504,577	384,047	3,654	46,492
2022	177,742	839,120	980,399	1,189,472	499,549	40,191	46,492
2023	170,772	808,346	942,888	1,103,634	545,750	76,728	3,952
2024	163,801	778,873	907,592	1,060,400	1,286,533	135,187	43,470
2025	156,831	749,812	874,307	1,019,827	1,193,691	230,183	82,989
2026	149,861	720,745	842,429	981,652	1,146,928	339,794	146,218
2027	142,891	691,680	810,997	945,651	1,103,045	485,941	248,966
2028	135,920	662,614	779,558	911,171	1,061,755	632,089	367,521
2029	128,950	633,549	748,121	877,175	1,022,816	690,548	525,594
2030	121,980	604,497	716,683	843,170	985,522	1,627,874	683,668
2031	115,009	575,466	685,246	809,168	948,752	1,510,400	746,897
2032	108,915	546,446	653,824	775,165	911,973	1,451,230	1,760,709
2033	104,573	521,146	622,424	741,163	875,196	1,395,704	1,633,648
2034	101,108	503,311	591,036	707,176	838,418	1,343,459	1,569,651
2035	97,642	489,204	563,671	673,214	801,641	1,294,188	1,509,594

Nuclear Summary Fixed TAB (RAP)

Nuclear :						-						
			Fixed	D&M					Capital Rep	placement		
BASE	1	1	<b>.</b>			1						
	2018	2020	2022	2024	2030	2032	2018	2020	2022	2024	2030	2032
	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)	(\$000)
Year	-	-					-	-				· · ·
2006	-	-					-	-				
2007	-	-					-	-				
2008	-	-					-	-				
2009	-	-	-	- 1	-	-	-	-	-	-	-	- 1
2010	-	-	-	-	-	- 1	-	-	-	-	-	-
2011	-	-	-	-	-	-	-	-	-	-	-	-
2012	-	-	-	-	-	- 1	-	-	-	-	-	<b>-</b> '
2013	***	-	-	-	-	-	-	-	-	-	-	-
2014	-	-	-	-	-	-	-	-	-		-	-
2015	-	-	-	-	-	- 1	-	-	-	-	-	-
2016	-	-	-	-	-	-	-	-	-	-	-	-
2017		-	-	-	-	-	-	-	-	-	· +	-
2018	74,886	-	-	-	-	- ]	11,333	-	-	-	-	~
2019	132,868	-	-	-	-	-	19,635	-	-	-	-	-
2020	137,549	80,237	-	-	-	-	19,808	11,554	-	-	-	-
2021	142,504	142,504	-	-	-	- 1	19,984	19,984	-	-	-	-
2022	147,502	147,502	86,043	-	-	-	20,170	20,170	11,766	-	-	-
2023	152,716	152,716	152,716	-	-	-	20,379	20,379	20,379	-	-	-
2024	158,137	158,137	158,137	92,247	-	~ 1	20,598	20,598	20,598	12,016	-	-
2025	163,776	163,776	163,776	163,776	-	-	20,821	20,821	20,821	20,821	-	-
2026	169,633	169,633	169,633	169,633	-	-	21,058	21,058	21,058	21,058	-	-
2027	175,719	175,719	175,719	175,719	-	-	21,290	21,290	21,290	21,290	-	-
2028	182,051	182,051	182,051	182,051	-	-	21,516	21,516	21,516	21,516	-	-
2029	188,618	188,618	188,618	188,618	+	-	21,750	21,750	21,750	21,750	-	-
2030	195,429	195,429	195,429	195,429	114,000	-	22,008	22,008	22,008	22,008	12,838	-
2031	202,500 209,842	202,500	202,500	202,500	202,500	-	22,300	22,300	22,300	22,300	22,300	-
2032	209,842 217,468	209,842	209,842	209,842	209,842	122,408	22,593	22,593	22,593	22,593	22,593	13,179
2033	217,408	217,468 225,382	217,468 225,382	217,468	217,468	217,468	22,889	22,889	22,889	22,889	22,889	22,889
2034 2035	223,382	225,382 233,586	225,382 233,586	225,382 233,586	225,382 233,586	225,382	23,207	23,207	23,207	23,207	23,207	23,207
2035	200,000	200,000	233,000	233,360	233,300	233,586	23,542	23,542	23,542	23,542	23,542	23,542

Nuclear :

.

វោ	[2]	[3]	[4]	[5]	(6)	171	[8]	[9]	[10]	[11]			(Assuming all Customers are Retail)
Year - LCEC	Annual Discount Factor	Generation Capital* (Millions)	Generation Variable O&M (Millions)	Generation Fixed O&M** (Millions)		Emissions Costs (Millions)	Fixed Gas Transportation (Miltions)	Total Annual Costs (Millions)	NPV Total Annual Cost (Millions)	NPV Cumulative Annual Costs (Millions)	Est. Other Fixed Costs (Millions)	FPL with LCEC Annual Energy Sales (GWH)	Average Monthly Bill (\$/1000 kwh)
2006	1.000	0	0	0	0	0	0	0	0	0	5,749		
2007	0.919	154	84	12	4,983	(28)	277	5,482	5,037	5,037	5,787		
2008	0.844	150	82	12	5,289	(46)	317	5,805	4,902	9,939	5,879		
2009	0.776	224	88	20	4,896	(76)	379	5,530	4,292	14,231	5,934		
2010	0.713	359	85	34	5,058	(68)	392	5,860	4,179	18,410	6,058	121,151	98.37
2011	0.655	510	80	47	4,686	(7)	440	5,756	3,772	22,182	6,116	125,015	94.97
2012	0.602	802	77	54	4,995	93	474	6,495	3,911	26,093	6,248	128,856	98.89
2013	0.553	957	77	55	5,294	226	473	7,082	3,919	30,012	6,379	131,885	102.06
2014	0.509	1,158	79	63	5,690	292	549	7,830	3,982	33,994	6,503	138,762	103.29
2015	0.467	1,441	78	78	6,099	392	680	8,769	4,098	38,092	6,626	142,530	108.01
2016	0.429	1,810	87	93	6,876	552	816	10,233	4,395	42,487	6,754	146,608	115.87
2017	0.395	2,174	89	109	7,448	672	953	11,446	4,517	47,004	6,883	150,071	122.13
2018	0.363	3,009	88	203	7,865	780	1,012	12,958	4,699	51,703	7,010	153,891	129.75
2019	0.333	3,150	90	272	8,432	904	1,012	13,860	4,619	56,322	7,148	157,695	133.22
2020	0.306	3,890	93	371	8,867	1,065	1,013	15,299	4,685	61,007	7,281	162,082	139.31
2021	0.281	4,214	92	470	9,107	1,257	1,012	16,152	4,548	65,553	7,544	165,660	143.04
2022	0.259	5,145	90	597	9,231	1,437	1,012	17,512	4,529	70,082	7,688	169,630	148,56
2023	0,238	5,089	90	696	9,501	1,548	1,107	18,031	4,285	74,367	7,838	173,649	148.98
2024	0.218	5,837	90	829	9,782	1,693	1,176	19,408	4,239	78,606	7,992	178,164	153.79
2025	0.201	6,072	87	955	10,170	1,914	1,175	20,373	4,089	82,695	8,148	181,962	156.74
2026	0.184	6,410	82	1,012	10,766	2,174	1,278	21,722	4,006	86,701	8,309	186,129	161.34
2027	0.169	6,657	80	1,061	11,445	2,371	1,457	23,072	3,910	90,611	8,474	190,318	165,75
2028	0.156	6,781	78	1,101	12,228	2,576	1,533	24,297	3,784	94,395	8,644	194,996	168.93
2029	0.143	6,909	79	1,146	12,918	2,695	1,643	25,390	3,634	98,029	8,818	198,907	171.98
2030	0.132	7,870	77	1,317	13,299	2,701	1,722	26,986	3,549	101,578	8,999	203,284	177.02
2031	0.121	7,733	73	1,464	13,785	2,760	1,839	27,654	3,342	104,920	9,189	207,660	177.42
2032	0.111	8,517	73	1,655	14,217	2,817	1,923	29,202	3,243	108,163	9,386	212,599	181.50
2033	0.102	8,071	77	1,809	14,723	2,894	1,922	29,495	3,010	111,174	9,588	216,621	180.43
2034	0.094	7,699	70	1,868	14,832	2,892	1,922	29,283	2,746	113,920	9,797	214,267	182.39
2035	0.086	7,524	72	1,942	15,829	3,068	2,052	30,485	2,627	116,548	10,020	218,895	185.04
·	Total NPV =	23,960	864	3,172	73,790	7.018	7,744	116,548	1		- <b>L</b>	· · · · · · · · · · · · · · · · · · ·	<b>I</b>

-----

هما و و د د اسي

	[1]		[	0.25		_							
	Year ^ LCEC	NPV Annual Capital Costs 2010-2035 (Millions)	NPV Cumulative Capital Costs 2010- 2035 (Millions)	LCEC Load Management - Modified First 4 Years (MW)	FPL Load Management (MW)	LCEC 12 CP (MW)	FPL & Other Total 12 CP (MW)	FPL Firm 12 CP (MW)	LCEC Firm 12 CP (MW)	Other 12 CP (MW)	FPL & Other NEL (MWH)	FPL NEL (MWH)	LCEC NEL (MWH)
	2006	0	·										
	2007	142											
	2008	127											
	2009	174						]			·····		· · · · ·
	2010	256	256	10	2,008	192	21,613	19,450	182	155	130,521	129,487	1,033
	2011	334	590	10	2,105	196	22,109	19,849	185	155	134,792	133,758	1,054
	2012	483	1,073	11	2,207	199	22,546	20,184	189	155	139,043	138,009	1,076
	2013	530	1,603	11	2,313	203	22,969	20,546	192	110	142,422	141,388	1,095
	2014	589	2,192	44	2,424	881	23,437	20,903	837	110	146,307	145,273	4,769
	2015	673	2,865	45	2,539	908	24,032	21,383	862	110	150,319	149,285	4,884
	2016	777	3,642	46	2,641	934	24,611	21,860	888	110	154,570	153,536	5,039
	2017	858	4,500		2,740	961	25,209	22,360	914	110	158,184	157,150	5,167
	2018	1,091	5,592	49	2,839	989	25,803	22,854	940	110	162,168	161,134	5,310
	2019	1,050	6,641	50	2,938	1017	26,395	23,347	967	110	166,136	165,102	5,453
	2020	1,191	7,833	51	3,037	1045	27,009	23,861	993	110	170,718	169,684	5,611
21	2021	1,186	9,019	52	3,106	1073	27,609	24,393	1,021	110	174,460	173,426	5,736
$\sim$	2022	1,331	10,349	54	3,106	1102	28,208	24,992	1,049	110	178,606	177,572	5,879
	2023	1,209	11,559	55	3,106	1132		25,597	1,077	110	182,806	181,772	6,022
	2024	1,275	12,834	57	3,106	1162	29,424	26,208	1,105	110	187,525	186,491	6,182
	2025	1,219	14,063	58	3,106	1191	30,051	26,835	1,134	110	191,506	190,472	6,305
	2026	1,182	15,235	59	3,106	1221	30,670	27,454	1,162	110	195,865	194,831	6,449
	2027	1,128	16,363	61	3,106	1251	31,294	28,078	1,190	110	200,249	199,215	6,592
	2028	1,056	17,419	62	3,106	1281	31,924	28,708	1,219	110	205,143	204,109	6,753
	2029	969	18,408	64	3,108	1311		29,351	1,247	110	209,247	208,213	6,874
	2030	1,035	19,443	66	3,106	1340		29,996	1,275	110	213,833	212,799	7,018
	2031	935	20,378	67	3,106	1370	33,859	30,642	1,303	110	218,418	217,384	7,161
	2032	946	21,323	69	3,108	1400	34,511	31,295	1,331	110	223,592	222,558	7,324
	2033	824	22,147	71	3,106	1430	35,177	31,961	1,360	110	227,818	226,784	7,444
	2034	722	22,869										
	2035	648	23,518	· · · · ·									
		23,518	J	1,169	68,171	23,790	673,052	602,106	22,621	2,775	4,264,248	4,239,432	126,230

.

[1]			(RAP)	l		(RAP)		(RAP)						
Year - LCEC	Olher NEL (MWH)	Total NEL (MWH)	Nuclear Energy (GWH)		LCEC Nuclear Energy (GWH)		Fossil Fuel Cost (Millions)	Total Fuel Cost (Millions)	FPL Fossi Ratio	Other Fossil Ratio	LCEC Fossil Ratio	FPL Nuclear Ratio	LCEC Nuclear Ratio	FPL Total Fuel Costs (Millions)
2006														
2007						1		·			_			<b></b>
2008										[				
2009														
2010	1,034	131,553	22,582	22,404	179	127	5,254	5,382	98.267%	0.949%	0.784%	99.209%	0.791%	5,290
2011	1,034	135,846	23,765	23,580	186	133	4,986	5,119	98.303%	0.923%	0.774%	99.218%	0.782%	5,033
2012	1,034	140,119	25,774	25,575	199	147	5,416	5,562	98.329%	0.904%	0.767%	99,226%	0.774%	5,471
2013	1,034	143,517	26,361	26,158	203	153	5,840	5,993	98.356%	0.883%	0.762%	99.232%	0.768%	5,896
2014	1,034	151,076	27,751	26,869	882	164	6,366	6,531	96.010%	0.838%	3.152%	96.822%	3.178%	6,271
2015	1,034	155,203	27,353	26,487	867	165	7,006	7,172	95.049%	0.809%	3.142%	96.832%	3,168%	6,890
2016	1,034	159,609	27,355	26,486	869	168	8,075	8,243	96.065%	0.782%	3.153%	96.822%	3.178%	7,920
2017	1,034	163,351	27,751	26,867	883	174	8,899	9,073	96.078%	0.763%	3.159%	96.817%	3.183%	8,719
2018	1,034	167,479	32,504	31,467	1,037	206	9,451	9,657	96.068%	0.766%	3.166%	96.810%	3.190%	9,279
2019	1,034	171,589	36,060	34,907	1,153	231	10,117	10,348	96.064%	0.763%	3.173%	96,803%	3.197%	9,942
2020	1,034	176,329	41,790	40,452	1,338	272	10,673	10,944	96.055%	0.769%	3.176%	96.799%	3.201%	10,515
2021	1,034	180,195	44,922	43,484	1,438	298	11,079	11,375	96.059%	0.764%	3.177%	96,799%	3.201%	10,929
2022	1,034	184,486	49,995	48,393	1,602	335	11,345	11,680	96.051%	0.769%	3.180%	96,795%	3.205%	11,221
2023	1,034	188,828	54,104	52,369	1,735	369	11,787	12,156	96.050%	0.767%	3.182%	96.793%	3.207%	11,678
2024	1,034	193,707	59,009	57,116	1,893	409	12,243	12,651	96.048%	0.768%	3.184%	96.791%	3.209%	12,155
2025	1,034	197,811	62,414	60,414	2,000	440	12,819	13,259	96.057%	0.764%	3.180%	96.796%	3.204%	12,739
2026	1,034	202,313	62,689	60,874	2,015	452	13,766	14,218	96.078%	0.742%	3.180%	96.796%	3.204%	13,663
2027	1,034	206,840	62,491	60,490	2,002	458	14,815	15,273	96.104%	0,716%	3.180%	96.797%	3.203%	14,682
2028	1,034	211,896	62,590	80,586	2,004	468	15,868	16,336	96.127%	0.693%	3.180%	96.797%	3.203%	15,707
2029	1,034	216,121	62,889	60,879	2,010	480	16,776	17,256	96.151%	0.675%	3.174%	96.804%	3.196%	16,595
2030	1,034	220,851	67,642	65,482	2,160	525	17,197	17,722	96.154%	0.675%	3.171%	96.807%	3,193%	17,044
2031	1,034	225,579	71,198	68,928	2,271	563	17,821	18,384	96.162%	0.670%	3.168%	96.811%	3.189%	17,682
2032	1,034	230,916	76,558	74,119	2,439	616	18,342	18,957	96.165%	0.670%	3.165%	96.814%	3.186%	18,234
2033	1,034	235,262	80,540	77,981	2,560	660	18,878	19,539	96.175%	0.668%	3.157%	96.822%	3.178%	18,796
2034		ļ		<u> </u>							L	<b> </b>		
2035		1.000							<b> </b>		L		I	
	24,816	4,390,477	1,136,289	1,102,366	33,924	8,011	274,820	282,831	]					272,350

[1]			· · · · · · · · · · · · · · · · · · ·					(RAP)	(RAP)	(RAP)		(RAP)	
Year - LCEC	Other Total Fuel Costs (Millions)	LCEC Total Fuel Costs (Millions)	FPL (\$/MWH)	Other (\$/MWH)	LCEC (\$/MWH)	All (\$/MWH)	Existing Nuclear Generation Demand (000)	Existing Fossil Generation Demand (000)	Total Existing Generation Demand (000)	Incremental Nuclear Generation Demand (000)	Incremental Fossii Generation Demand (000)	Incremental Total Generation Demand (000)	Total Nuclear Generation Demand (000)
2006					<b></b>	<u> </u>		]		·	<u> </u>	<u> </u>	<u> </u>
2007			······································			<u> </u>					<u> </u>		··
2008						<u>+</u>	• • • • •				t	┟━┉╸━╧╾┯╸━┥	
2009							- 1		}	<u> </u>		┝━━╾───┤	
2010	50 .	42	40.85	48.22	40.85	40.91	868,876	1,393,745	2,262,621	54,142	424,222	478,364	923,018
2011	46	40	37.63	44.49	37.63	37.68	919,818	1,392,006	2.311.824	81,801	554,928	636,728	1,001,618
2012	49	43	39.64	47.36	39.84	39.70	974,555	1,368,715	2,343,270	312,234	620,284	932,517	1,286,789
2013	52	46	41.70	49.85	41.70	41.76	1,035,028	1,376,195	2,411,223	478,905	609,866	1,088,771	1,513,934
2014	53	206	43.17	51.62	43.17	43.23	1,100,320	1,383,542	2,483,862	604,635	695,156	1,299,791	1,704,955
2015	57	225	46.15	54.80	46.15	46.21	1,156,210	1,390,869	2,547,080	747,193	850,152	1,597,345	1,903,403
2016	63	260	51.59	61.06	51.59	51.65	1,214,198	1,323,497	2,537,695	975,981	1,013,922	1,989,903	2,190,179
2017	68	287	55.48	65.63	55.48	55.55	1,274,188	1,331,087	2,605,274	1,201,730	1,170,412	2,372,142	2,475,918
2018	72	306	57.58	70,02	57.58	57.66	1,336,092	1,338,626	2,674,719	2,090,963	1,209,696	3,300,659	3,427,055
2019	77	328	60.22	74,64	60.22	60.31	1,400,606	1,346,585	2,747,191	2,341,055	1,170,972	3,512,026	3,741,661
2020	82	348	61.97	79.33	61.97	62.07	1,467,336	1,354,610	2,821,946	3,220,388	1,134,028	4,354,416	4,687,724
2021	85	361	63.02	81,90	63.02	63.13	1,536,450	1,179,960	2,716,410	3,340,582	1,435,533	4,776,115	4,877,032
2022	87	372	63.19	84,35	63,19	63,31	1,608,044	1,188,262	2,796,306	4,206,118	1,626,707	5,832,825	5,814,162
2023	90	387	64.25	87,49	64.25	64.37	1,682,295	1,196,778	2,879,073	4,171,356	1,704,401	5,875,757	5,853,650
2024	94	403	65.17	90,89	65.17	65.31	1,759,249	1,205,478	2,964,727	5,016,324	1,739,754	6,756,077	6,775,573
2025	98	422	66.88	94,68	66.88	67,03	1,838,936	1,165,173	3,004,109	5,046,031	2,068,514	7,114,545	6,884,967
2028	102	452	70.13	98.73	70.13	70.28	1,921,454	1,109,283	3,030,737	5,090,389	2,414,002	7,504,392	7,011,843
2027	106	486	73.70	102.63	73.70	73.84	2,006,903	1,115,552	3,122,455	5,217,206	2,581,093	7,798,299	7,224,108
2028	110	520	76.95	106.28	76,95	77.10	2,095,386	1,124,817	3,220,204	5,364,893	2,595,386	7,960,279	7,460,279
2029	113	548	79.70	109,48	79.70	79.64	2,187,013	1,134,230	3,321,243	5,468,227	2,666,032	8,134,259	7,655,240
2030	116	562	80.09	112.25	80,09	80.25	2,282,559	1,144,295	3,426,853	6,579,981	2,683,676	9,263,657	8,862,540
2031	119	582	81.34	115.44	81,34	81.50	2,381,502	1,154,533	3,536,035	6,514,938	2,755,328	9,270,266	8,896,439
2032	123	600	81.93	118.83	81.93	82.10	2,483,964	1,164,948	3,648,912	7,506,027	2,738,163	10,244,190	9,989,991
2033	126	617	82.88	122.01	82.88	83.05	2,590,069	1,175,544	3,765,613	7,335,998	2,620,691	9,956,690	9,926,068
2034			<u> </u>	L	<u> </u>			l	ļ		<u> </u>		
2035	<u> </u>						I	L	Į	L			
	2,039	8,442	64.24	82.17	66.87	64.42	39,121,051	30,058,330	69,179,381	82,967,095	39,082,917	122,050,012	122,088,146

	1										
	н. С										
[1]				(RAP)	(RAP)		In Demand				
Year - LCEC	Total Fossil Generation Demand (000)	Total Generation Demand (000)	Existing Non-Fuel Nucleary Energy (000)	Existing Non-Fuel Fossil Energy (000)	Total Existing Non-Fuel Energy (000)	Incremental Nuclear Non-Fuel Energy (000)	Incremental Fossil Non-Fuel Energy (000)	Total Incremental Non-Fuel Energy (000)	Total Non-Fuel Nuclear Energy (900)	Total Non-Fuel Fossil Energy (000)	Total Non-Fuel Energy (000)
2006					· · · · · · · · · · · · · · · · · · ·	<b></b>					<u> </u>
2007				1				· · · -			
2008										1	
2009				1							
2010	1,817,967	2,740,985	245,533	139,229	384,761	0	0	0	245,533	139,229	384,761
2011	1,946,934	2,948,552	249,627	141,161	390,789	0	0	0	249,627	141,161	390,789
2012	1,988,998	3,275,787	253,943	143,185	397,128	0	0	0	253,943	143,185	397,128
2013	1,986,061	3,499,994	258,341	145,242	403,584	0	0	0	258,341	145,242	403,584
2014	2,078,698	3,783,653	262,611	147,250	409,861	0	0	0	262,611	147,250	409,861
2015	2,241,021	4,144,424	266,847	149,248	416,095	0	0	0	266,847	149,248	416,095
2016	2,337,418	4,527,598	271,236	151,310	422,546	0	0	0	271,236	151,310	422,546
2017	2,501,499	4,977,416	275,672	153,394	429,066	0	0	0	275,672	153,394	429,066
2018	2,548,323	5,975,378	280,054	155,456	435,511	0	0	0	280,054	155,456	435,511
2019	2,517,557	6,259,218	284,784	157,663	442,447	0	0	0	284,784	157,663	442,447
2020	2,488,638	7,176,362	289,557	159,889	449,447	0	0	0	289,557	159,889	449,447
2021	2,615,493	7,492,525	294,433	162,159	456,592	0	0	0	294,433	162,159	456,592
2022	2,814,969	8,629,131	299,420	164,476	463,896	0	0	0	299,420	164,476	463,896
2023	2,901,179	8,754,830	304,578	166,865	471,443	0	0	0	304,578	166,865	471,443
2024	2,945,231	9,720,804	309,881	169,315	479,196	0	0	0	309,881	169,315	479,196
2025	3,233,687	10,118,654	315,289	171,810	487,099	0	0	0	315,289	171,810	487,099
2026		10,535,128	320,805	174,351	495,156	0	0	0	320,805	174,351	495,156
2027		10,920,753	326,431	176,938	503,369	0	0	0	326,431	176,938	503,369
2028		11,180,482	332,168	179,573	511,741	0	0	0	332,168	179,573	511,741
2029		11,455,502	338,019	182,256	520,275	0	0	0	338,019	182,256	520,275
2030		12,690,510	344,416	185,164	529,580	0	0	0	344,416	185,164	529,580
2031		12,806,300	350,948	188,130	539,078	0	0	0	350,948	188,130	539,078
2032		13,893,101	357,619	191,155	548,774	0	0	0	357,619	191,155	548,774
2033		13,722,303	364,432	194,239	558,671	0	0	0	364,432	194,239	558,671
2034											1
2035		· · · · ·									
	69,141,247	191,229,393	7,196,646	3,949,459	11,146,104	0	0	0	7,196,646	3,949,459	11,146,104

rear - LCEC	Total Generation and Energy Demand (000)	Total 12 CP (MW)	Nuclear Capacity (MW)	FPL Nuclear Capacity (MW)	LCEC Nuclear Capacity (MW)	LCEC Fossil 12CP Ratio	Other Fossil 12CP Ratio	FPL Fossi 12CP Ratio	LCEC Admin. Cost (000)	LCEC Generation and Energy Demand Share (000)	Other Generation and Energy Demand Share (000)	FPL Generation and Energy Demand Share (000)	LCEC Generation and Energy Demand Share (\$/MWH)
2006												· · · ·	
2007												- <b></b>	
2008											Î		
2009											İ		
2010	3,125,746	19,787	3,064	3,036	28	0.918%	0.927%	98,155%		28,276	18,172	3,079,299	27.38
2011	3,339,341	20,189	3,168	3,139	29	0.917%	0,911%	98.172%	1	30,168	19,032	3,290,140	28.63
2012	3,672,915	20,527	3,472	3,440	32	0.918%	0,909%	98.173%		33,244	19,371	3,620,301	30.89
2013	3,903,578	20,848	3,472	3,440	32	0.921%	0.633%	98.446%		35,413	13,855	3,854,310	32,35
2014	4,193,514	21,850	3,472	3,338	134	3.827%	0.599%	95.574%		158,198	13,676	4.021,640	33,17
2015	4,560,519	22,355	3,472	3,337	135	3.854%	0.583%	95.564%	1	173,287	14,262	4,372,970	35.48
2016	4,950,143	22,859	3,472	3,336	136	3.882%	0.567%	95.551%	1	189,628	14,446	4,746,069	37.63
2017	5,406,482	23,383	3,472	3,338	136	3.905%	0.552%	95.543%	1	208,525	14,989	5,182,968	40,35
2018	6,410,889	23,904	4,562	4,382	180	3.928%	0.569%	95.503%		249,344	15.684	6,145,860	46.95
2019	6,701,664	24,424	4,562	4,381	181	3.954%	0,554%	95.492%		262,441	15,146	6,424,077	48.13
2020	7,625,809	24,965	5,652	5,426	226	3.974%	0.570%	95,457%		300,592	15,403	7,309,813	53.57
2021	7,949,117	25,524	5,652	5,425	227	3.994%	0.554%	95.452%		314,937	15,717	7,618,463	54.91
2022	9,093,027	26,151	6,742	6,471	271	4.004%	0.567%	95,430%		361,639	17,218	8,714,169	61.51
2023	9,226,273	26,784	6,742	6,470	272	4.014%	0,549%	95,437%		367,817	17,204	8,841,252	61.08
2024	10,200,000	27,423	7,832	7,515	317	4.024%	0.561%	95,415%		407,992	17,836	9,774,172	66.00
2025	10,605,754	28,078	7,832	7,515	317	4.031%	0,543%	95.426%	1	424,957	18,881	10,161,916	67,40
2026	11,030,284	28,726	7,832	7,514	318	4.039%	0.526%	95,435%		442.822	19,842	10,567,621	68,67
2027	11,424,122	29,378	7,832	7,514	318	4.046%	0,511%	95,444%	1	459,403	20,140	10,944,579	69.69
2028	11,692,223	30,036	7,832	7,513	319	4.052%	0,495%	95.453%	1	470,841	19,674	11,201,708	69.72
2029	11,975,778	30,708	7,832	7,513	319	4.055%	0.481%	95.464%		482,621	19,504	11,473,653	70,21
2030	13,220,090	31,381	8,922	8,558	364	4.057%	0.490%	95.453%	1	533,505	19,998	12,666,587	76.02
2031	13,345,378	32,056	8,922	8,558	364	4.060%	0.476%	95,465%		538,790	19,851	12,786,737	75.24
2032	14,441,875	32,738	10,012	9,603	409	4.061%	0.484%	95.455%		583,586	20,174	13,838,115	79.68
2033	14,280,974	33,431	10,012	9,604	408	4.061%	0.470%	95.469%	1	576,868	19,129	13,684,977	77.50
2034			1	1			1				101,20	10,000,017	
2035			T	1			t					L	
	202,375,497	627,502	145,836	140,362	5,474	+	<u> </u>			7,634,894	419,205	194.321.398	+

	FPL Generation and Energy Demand Share (\$AMVH)					23.78	24.60	26.23	27.26	27.68	29.29	30.91	32.98	38.14	36.91	43.08	43.83	49.07	48.64	52.41	53.35	54.24	54,94	54.88	55.11	59.52	58.82	62.18	60.34			
	Other Generation and Energy Demand Share (\$MWH)					17.57	18.41	18.73	13.40	13.23	13.79	13.97	14.50	15.17	14.65	14.90	15.20	16.65	16.64	17.25	18.26	19.19	19.48	19.03	18.86	19.34	19.20	19.51	18.50			
E	Year - LCEC	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	

	2002		NC34	2033	2032	2031	2030	204.4		2028	2027	2026	2025	7024	2023				200	2019	2018	1017	2018	200	1014	2012		i a	0102	2009	00	200	2005	Year			LCEC	_
	21,042	Cr 20	101.108	104,573	108,915	600101	121,980	120,000	138 050	135,820	142,891	149,861	156,831	163,601	110,112	761,141	104.112	1011000	191.6R3	1948 677	205,839	213,300	221,085	177'677	237,733	000,012	007,601	-			,	,	1	,	(note)	Uprates	_	
	469,204		517 514	521,146	546,446	575,466	604,497	031,549		662 614	691,680	720,745	749,812	778,873	808,346	639,120	6/1,/53	900, 90 <del>0</del>	000 101		1.016.766	431,314	394,801	303,518	212,234	143,771	84,437	47,924	25,103	2,282	26,848	26,848	•	٩	(2000)	Nuclear 2015		
	563,671	den'i an		627.424	653,824	685,246	716,683	748,121		770 559	810.997	842,429	874,307	907,59Z	942,668	990,399	1,020,372	1,099,734	ALC DOI:		427 N17	328,285	229,552	155,503	91,327	51,834	27,151	2,468	29.039	29,039			,	,	(\$000)	Nuclear 2010		
:	673,214	871,707	202 120	741 163	775.165	809,168	843,170	877,175	811,111		247 271	981.652	1.019.827	1,060,400	1,103,634	1,189,472	504,577	461,861	350,073	240,204		168.192	99,780	56,054	29,367	2,670	31,408	31,408		•			•	,	(\$000)	Nudear 2022		Copilal Revenue Regularment
	801.641	838,418	0/0,190		911 973	948.752	985,522	1,022,816	1,061,/55			1,146,928	1 197 601	1,286,533	545,750	499,549	384,047	268,544	181,917	106,840		BC8 08	31.763	2.888	33,971	33,971	•		•	•			•		(\$000)	1 Nuclear 2024		*
	1,294,188	1,343,450	1,395,704		1 451 330	1 510 400	1.627.874	690,548	632,089	1 46 CO4	105 014	130, 100	181 050	135 187	76,728	40,191	3,854	42,985	42,985	,			•		•	•	r	,	•	•		ŀ	•	·	(\$000)	Nuclear 2030		
	1,509,594	1,569,651	1,633,648	1,700,708	4 750 700	744 807	GRJ RRA	525,594	367,521	246,966		06,20		17 470	3.952	46.492	46.492	,	1		•				•			•	ł	•	•		•	-	(\$000)	Nuclear 2032		
												103,170						137 549	132,868	74,886		,	,					•				. ,		-	1000	2018		
	233.596	225 382	217,468	208,842	202,000	202,722	105 400	199 61A	182.051	175,719	109,033	163,776	1-10, 1-1	150 137	153 748	147 500	147 504	RD 237	•			,	,			•				,				1000	(\$000)	7070		ļ
	233 596	225.382	217,468	209,842	006,202	201 000		199.618	182 051	175,719	109,633	163,776	100,107	120,110		BS DA1		•			,			ļ		•								local	Soon		Fbed	,
2001000	222,002	225 192	217,468	209,842	202,202	140,428	105,010	199,001	182.051	175,719	169,633	163,776	92,241	· ·	,			I	ı	1	+	,		1			ų		,					10001	(1000)	1	d OWN	1
200,000	200,024	222 183	217.468	208,842	202,500	114,000	,		•	,	,	,	,	,	,				•	•	•	•	,	,		1	,	,	,					(0000)	1000	1020		
000 007	700,027		217,468	122,408	•		,			,	•				,		,		,	•	,	•	,	,	,	,	,		,					loonel	1002			
750.07	23,207	3	22,889	22,593	22,300	800,22	10,17	1.010	21.75	21,290	21,058	20,821	366,02	675,02	20,170	-06 B1	000	10,000	19 635	11 333					,	,	,	,	,	,	,	,	,	(MAR)	1019			
23,342	23,207		77 HAQ	22,593	22,300	22,008	21, 20	21,010		21,290	21,058	20,821	20,598	20,379	20,170	*06'61	11,004	1			•		,	•	,	,		,	ı	•	,		,	(0004)	1020	-		
23,542	23,207			22.593	22,300	22,008	21,750	21,010		21,290	21,058	20,821	20,598	20,379	11,766	1	ı	,			•	•	ı	•	1	,	ł	,	,					(0006)	1011		Capital Replacement	
23,542	23,207	22,000	33 880	22,593	22,300	22,008	21,750	21,516		21,220	21.058	20,821	12,016	•	•		•				,	•	,	•		,	·		•					(9000)	2024		lacement	
	23,207				22,300	12,638	ł	•		•	•	•	•	•	•	•	•	,			•	,	•	•	•	•	·	,	•					(5000)	2030	-		
23,542	23,207	22,008		13 179	•	,	•	•	•		•	•	•	•	'	•	,	•	•				•	•	,	•	•	,	'					(\$000)	2032			

[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]	[9]
Year - LCEC	Annual Discount Factor	Generation Capital* (Millions)	Generation Variable O&M (Millions)	Generation Fixed O&M** (Millions)	System Fuel (Millions)	Emissions Costs (Millions)	Fixed Gas Transportation (Millions)	Total Annual Costs (Millions)
2006	· · · ·	0	0	0	0	0	0	0
2007		0	0	0	0	0	0	0
2008	and the states of	0	0	0	0	0	0	0
2009	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	0	0	· 0	0	0	0	0
2010		0	1	0	72	4	0	78
2011		87	2	7	(84)	(26)	47	33
2012		55	0	5	9	(5)	34	98
2013		(10)	0	0	75	9	0	74
2014		85	2	8	167	(1)	76	336
2015		148	0	14	152	(7)	129	437
2016		142	(1)	14	160	(12)	129	433
2017		137	(0)	14	212	1	129	493
2018		131	(2)	14	256	13	129	543
2019		126	(2)	14	276	15	129	559
2020		121	(1)	15	316	26	129	606
2021	a myang tang tang tang tang tang tang tang t	116	0	15	335	31	129	626
2022		110	1	15	374	45	129	675
2023		229	(1)	25	300	5	225	782
2024	n des son tom The son	306	(3)	32	314	12	293	953
2025		162	(3)	23	437	70	192	881
2026		62	(0)	17	497	89	121	785
2027	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	197	0	27	457	62	226	969
2028		142	(1)	24	537	98	193	994
2029		33	0	18	588	117	116	873
2030	···	30	(1)	18	626	122	116	911
2031		183	(10)	29	594	110	233	1,139
2032		283	(7)	38	623	107	317	1,360
2033		104	(7)	27	713	138	194	1,168
2034		(191)	(4)	8	(14)	(5)	(21)	(227)
2035		(307)	0	0	0	0	(111)	(418)
	Total NPV ≈	749	(3)	97	1,683	149	807	3,482

	(1) Assumed	(2)	(3) = (1) +(2)		(1) Assumed	(2)	(3) = (1) +(2)	
	Additional		Total		Additional	Summer	Total	
	DSM	DSM	January		DSM	DSM	August	
	2016-2020	Capability	DSM		2016-2020	Capability	DSM	antistation from the two property and and and
	(MW)	(MW)	(MW)		(MW)	(MW)	(MW)	DSM 12 CP
2007		1,555	1,555	2007		1,768	1,768	1,679
2008		1,649	1,649	2008		1,908	1,908	1,801
2009		1,750	1,750	2009		2,034	2,034	1,916
2010		1,814	1,814	2010		2,146	2,146	2,008
2011		1,883	1,883	2011		2,264	2,264	2,105
2012		1,954	1,954	2012		2,388	2,388	2,207
2013		2,028	2,028	2013		2,516	2,516	2,313
2014		2,106	2,106	2014		2,651	2,651	2.424
2015		2,188	2,188	2015		2,790	2,790	2,539
2016	0	2,264	2,264	2016	52	2,858	2,910	2,641
2017	70	2,264	2,334	2017	172	2,858	3,030	<b>2,740</b>
2018	140	2,264	2,404	2018	292	2,858	3,150	2,839
2019	210	2,264	2,474	2019	412	2,858	3,270	2,938
2020	280	2,264	2,544	2020	532	2,858	3,390	3,037
2021	350	2,264	2,614	2021	600	2,858	3,458	3,106
2022	350	2,264	2,614	2022	600	2,858	3,458	3,106
2023	350	2,264	2,614	2023	600	2,858	3,458	3,106
2024	350	2,264	2,614	2024	600	2,858	3,458	3,106
2025	350	2,264	2,614	2025	600	2,858	3,458	3,106
2026	350	2,264	2,614	2026	600	2,858	3,458	3,106
2027	350	2,264	2,614	2027	600	2,858	3,458	3,106
2028	350	2,264	2,614	2028	600	2,858	3,458	3,106
2029	350	2,264	2,614	2029	600	2,858	3,458	3,106
2030	350	2,264	2,614	2030	600	2,858	3,458	3,106
2031	350	2,264	2,614	2031	600	2,858	3,458	3,106
2032	350	2,264	2,614	2032	600	2,858	3,458	3,106
2033	350	2,264	2,614	2033	600	2,858	3,458	3,106
2034	350	2,264	2,614	2034	600	2,858	3,458	3106
2035	350	2,264	2,614	2035	600	2,858	3,458	3,106
2036	350	2,264	2,614	2036	600	2,858	3,458	<b>3.106</b>
2037	350	2,264	2,614	2037	600	2,858	3,458	<b>31106</b>
2038	350	2,264	2,614	2038	600	2,858	3,458	3,106
2039	350	2,264	2,614	2039	600	2,858	3,458	3,106
2040	350	2,264	2,614	2040	600	2,858	3,458	3,106-

# MID BAND : Non Coincident Calendar Monthly Peaks Lee County (MW)

٠

Source: RAP

.

	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	Total	NCP
2010	235	194	183	163	181	189	192	196	191	177	199	203	2,303	192
2011	240	197	186	167	185	192	196	200	194	181	202	206	2,348	196
2012	244	201	190	170	188	196	200	204	198	184	206	210	2,391	199
2013	249	205	193	173	192	200	203	208	202	188	210	214	2,436	203
2014	1,078	889	838	750	832	866	883	901	875	814	915	934	10,575	881
2015	1,110	915	863	773	857	892	909	928	901	838	942	962	10,890	908
2016	1,144	943	890	795	882	918	935	955	927	862	971	991	11,214	934
2017	1,178	971	916	817	907	944	962	982	953	887	999	1,020	11,537	961
2018	1,213	1,000	943	840	932	970	988	1,009	979	911	1,029	1,050	11,865	989
2019	1,250	1,030	972	862	957	996	1,015	1,036	1,006	936	1,060	1,082	12,201	1017
2020	1,287	1,061	1,001	885	982	1,022	1,041	1,063	1,032	960	1,089	1,112	12,535	1045
2021	1,326	1,093	1,031	907	1,007	1,048	1,068	1,090	1,058	984	1,122	1,145	12,879	1073
2022	1,366	1,126	1,062	930	1,032	1,074	1,094	1,117	1,084	1,009	1,155	1,179	13,228	1102
2023	1,407	1,160	1,094	952	1,057	1,100	1,121	1,144	1,110	1,033	1,189	1,214	13,582	1132
2024	1,449	1,194	1,127	975	1,081	1,125	1,147	1,171	1,137	1,058	1,225	1,251	13,940	1162
2025	1,491	1,229	1,160	997	1,106	1,151	1,174	1,198	1,163	1,082	1,260	1,286	14,298	1191
2026	1,533	1,264	1,192	1,020	1,131	1,177	1,200	1,225	1,189	1,106	1,295	1,322	14,655	1221
2027	1,575	1,298	1,225	1,042	1,156	1,203	1,226	1,252	1,215	1,131	1,330	1,358	15,013	1251
2028	1,617	1,333	1,258	1,065	1,181	1,229	1,253	1,279	1,241	1,155	1,365	1,394	15,371	1 <b>281</b>
2029	1,659	1,368	1,290	1,087	1,206	1,255	1,279	1,306	1,268	1,180	1,401	1,430	15,729	1311
2030	1,701	1,402	1,323	1,110	1,231	1,281	1,306	1,333	1,294	1,204	1,436	1,466	16,086	1340
2031	1,743	1,437	1,356	1,132	1,256	1,307	1,332	1,360	1,320	1,228	1,471	1,501	16,444	1370
2032	1,785	1,471	1,388	1,155	1,281	1,333	1,359	1,387	1,346	1,253	1,506	1,538	16,802	1400
2033	1,827	1,506	1,421	1,177	1,306	1,359	1,385	1,414	1,373	1,277	1,542	1,574	17,161	1430

#### Florida Power & Light Projected 12 CP Source: RAP

Year	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	Total	12CP
	WPK						SEP0506	SPK						
			. <u></u>											
2006	14,800	19,683	16,946	18,975	20,143	20,963	21,364	21,810	21,170	19,698	18,432	18,817	232,801	
2007	22,247	18,338	17,303	18,531	20,558	21,395	21,805	22,259	21,607	20,104	18,748	19,139	242,032	
2008	22,627	18,652	17,599	18,956	21,030	21,886	22,305	22,770	22,103	20,565	19,152	19,552	247 197	
2009	23,115	19,054	17,978	19,509	21,643	22,524	22,956	23,435	22,747	21,165	19,543	19,951	253,620	
2010	23,587	19,444	18,346	19,982	22,168	23,070	23,512	24,003	23,299	21,678	19,924	20,340	259,351	21,613
2011	24,047	19,823	18,703	20,489	22,730	23,656	24,109	24,612	23,890	22,228	20,298	20,721	265,305	22,109
2012	24,498	20,194	19,054	20,908	23,195	24,139	24,602	25,115	24,378	22,683	20.674	21,106	270,547	22,546
2013	24,952	20,569	19,407	21,304	23,634	24,596	25,067	25,590	24,839	23,111	21,059	21,498	275,627	22,969
2014	25,416	20,951	19,768	21,728	24,105	25,086	25,567	26,100	25,334	23,572	21,582	22,032	281,242	23,437
2015	26,048	21,472	20,259	22,287	24,726	25,732	26,225	26,772	25.987	24,179	22,116	22,577	288,380	24,032
2016	26,692	22,003	20,761	22,818	25,314	26,345	26,850	27,410	26,606	24,755	22,654	23,126	295,333	24,611
2017	27,342	22,538	21,266	23,375	25,933	26,988	27,505	28,079	27,255	25,359	23,194	23,678	302,512	25,209
2018	27,994	23,076	21,773	23,923	26,540	27,621	28,150	28,737	27,894	25,954	23,737	24,232	309,632	25,803
2019	28,649	23,616	22,282	24,468	27,145	28,250	28,791	29,391	28,529	26,545	24,283	24,790	316,740	26,395
2020	29,308	24,159	22,795	25,051	27,791	28,922	29,476	30,091	29,209	27,177	24,803	25.321	324,104	27,009
2021	29,936	24,677	23,283	25,624	28,427	29,584	30,151	30,780	29,877	27,799	25,322	25,850	331,311	27,609
2022	30,562	25,193	23,770	26,195	29,061	30,244	30,823	31,466	30,543	28,419	25.843	26,382	338,502	28,208
2023	31,191	25,711	24,259	26,773	29,702	30,911	31,503	32,160	31,217	29,045	26,369	26,920	345,760	28,813
2024	31,826	26,235	24,754	27,355	30,347	31,582	32,187	32,859	31,895	29,676	26,907	27,468	353.092	29,424
2025	32,475	26,770	25,258	27,956	31,014	32.277	32,895	33,581	32,596	30,329	27,444	28,016	360,610	30,051
2026	33,123	27,304	25,762	28,547	31,669	32,958	33,590	34,290	33,285	30,969	27,981	28,565	368,044	30,670
2027	33,772	27,839	26,267	29,143	32,331	33,647	34,292	35,007	33,980	31,616	28,520	29,116	375,531	31,294
2028	34,422	28,375	26,773	29,746	33,000	34,343	35,001	35,731	34,683	32,270	29,068	29,675	383,086	31,924
2029	35,084	28,921	27,287	30,365	33,686	35,057	35,729	36,474	35,405	32,942	29,620	30,239	390,809	32,567
2030	35,750	29,470	27,805	30,984	34,374	35,773	36,458	37,219	36,127	33,614	30,172	30,802	398,548	33,212
2031	36,416	30,018	28,323	31,605	35,063	36,490	37,189	37,964	36.851	34,287	30,728	31,369	406,303	
2032	37,086	30,571	28,845	32,231	35,756	37,212	37,925	38,716	37,580	34,966	31,297	31,309		33,859
2033	37,773	31,137	29,379	32,867	36,462	37,946	38,673	39,480	38,322	35,656	31,882		414,133	34,511
2034	38,480	31,720	29,928	33,532	37,200	38,714	39,456	40.279	39.097	36,377	31,002	<u>32,547</u> 33,161	422,127	35,177
2035	39,205	32,318	30,493	34,202	37,944	39,488	40,245	41,084	39,879	37,105	33,095		430,427	35,869
2036	39,943	32,926	31,067	34,889	38,705	40,281	41,052	41,909	40.680	37,850	33,697	33,785	438,843	36,570
2037	40,670	33,525	31.632	35,564	39,455	41,061	41,847	42,720	40,000	38,582	34,337	34,400	447,398	37,283
2038	41,443	34,162	32,233	36,287	40,256	41,895	41,647	43,588	41,407	39,366	34,337	35,054	455,915	37,993
2039	42,235	34,815	32,849	37,028	41,079	42,751	43,570	44,478	42,310	40,171	34,993	35,724	464,954	38,746
2040	43,047	35,485	33,481	37,789	41,923	43,629	44,465	45,393	44,061	40,996		36,411	474,227	39,519
	,-			<u>.,</u>					44,001	40,990	36,210	36,966	483,447	40,287

. . .

		Lee Coun		
		at 60% Load I		
	Summer Peak MW	Lee Energy GWh	NEL GWh	Lee Sales GWh
2006-2009	0	0	-	0
2010	196	1,033	131,553	121,151
2011	200	1,053	135,846	125,015
2012	204	1,076	140,119	128,856
2013	208	1,094	143,517	131,885
2014	901	4,740	151,076	138,762
2015	928	4,883	155,203	142,530
2016	955	5,039	159,609	146,608
2017	982	5,167	163,351	150,071
2018	1,009	5,309	167,479	153,891
2019	1,036	5,451	171,589	157,695
2020	1,063	5,609	176,329	162,082
2021	1,090	5,736	180,195	165,660
2022	1,117	5,878	184,486	169,630
2023	1,144	6,020	188,828	173,649
2024	1,171	6,180	193,707	178,164
2025	1,198	6,305	197,811	181,962
2026	1,225	6,447	202,313	186,129
2027	1,252	6,590	206,840	190,318
2028	1,279	6,750	211,896	194,996
202 <del>9</del>	1,306	6,874	216,121	198,907
2030	1,333	7,017	220,851	203,284
2031	1,360	7,159	225,579	207,660
2032	1,387	7,321	230,916	212,599
2033	1,414	7,444 126,177	235,262	216,621

#### LEE COOP LOAD (At Generator)\*

## FPL LOAD\*\* (At Generator)

	NEL ENERGY	Winter PEAK		NEL ENERGY	- Winter PEAK
<u>YEAR</u>	<u>(GWH)</u>	MW 1	YEAR	<u>(GWH)</u>	<u>. MW</u>
2010	1,033	235	2010	130,521	24,003
2011	1,054	<b>*</b> . 6, 240 of 1	2011	134,792	+ 24,612
2012	1,076	244	2012	139,043	
2013	1,095	249	2013	142,422	25,590**1
2014	4,769	1078	2014	146,307	26,100
2015	4,884	1,110	2015	150,319	26,772
2016	5,039	1 1 44	2016	154,570	27,410
2017	5,167	1,178	2017	158,184	25-28,079
2018	5,310	1,213	2018	162,168	28,737
2019	5,453	1,250 2	2019	166,136	29,391
2020	5,611	1,287	2020	170,718	i <b>30,091</b> ≩ ∏
2021	5,736	1326日月	2021	174,460	i - 30,780 i -
2022	5,879	1366114	2022	178,606	1 31,466
2023	6,022	<b>1407</b> .5.5	2023	182,806	32,160
2024	6,182	a a 1449 a a	2024	187,525	32,859
2025	6,305	1.491 1	2025	191,506	33,581
2026	6,449	· 1533 · ·	2026	195,865	34,290
2027	6,592	1575	2027	200,249	35,007
2028	6,753	1 5 1 5 17	2028	205,143	35,731
2029	6,874	1,659,41	2029	209,247	36,474
2030	7,018	I IVOLA	2030	213,833	4.37,219
2031	7,161	5 1743	2031	218,418	37,964
2032	7,324	1785	2032	223,592	1 38,716
2033	7,444	<u>4 18271 I</u>	2033	227,818	39,480

\* The NEL forecasts and the peak load forecasts are assumed to be at the generator.

\*\* The FPL forecast in this spreadsheet does not include the Lee Coop load.

Prefixes         Power         Mutchar         Nuclear         Nuclear <th< th=""><th>of Total Justalled</th><th>tai lestalled</th><th></th><th>÷ ٿ</th><th>apachy</th><th>Change in Perchane</th><th>Cummulative Change in</th><th>Cummulative Channe in</th><th>Incremental</th><th>2:</th><th>testre</th><th>ja  </th><th>Existing</th><th>Ĩ</th><th>nsremental</th><th></th><th>Load</th></th<>	of Total Justalled	tai lestalled		÷ ٿ	apachy	Change in Perchane	Cummulative Change in	Cummulative Channe in	Incremental	2:	testre	ja 	Existing	Ĩ	nsremental		Load
Black         With Let Load         Base         Pict	ther Capacity* Purchases Fower Load	Purchases Power Load (MW)	Fower Load	Fower Load	Power Load	Power 1	ž	Purchase Power Land %	Power	M Harmus (B)	rr Peak Load)	z %	uclear .	zž	uclear Sources	C (C) (C) (C)	ontrol er Ptak Load)
R.91%         S.99,960,217         S.06         J.064         J.064 <thj.075< th="">         J.064         J.064</thj.075<>	Bate with Lee Load Base with Lee Load	d Base with Lee Load	with Lee Load						*		with Let Load		MW) Twith Lee Lond		MW)	- I	(MIM)
# 01%         54/376/61         19.6%         10.6% <th10.6%< th="">         10.6%         10.6%</th10.6%<>	20646 3103	101	LOIE														MIIN 146 1030
-3.8%         11.701.097         25.6%         21.6%         30.64         0         0         1.432           -3.8%         [11.701.097]         2.05%         2.05%         3.054         0         0         0         1.432           -3.8%         [11.701.097]         2.05%         2.05%         3.054         0         0         1.432           -3.45%         [11.701.097]         2.05%         2.05%         3.054         0         0         1.432           -3.45%         [12.500.073]         2.05%         2.05%         3.054         0         0         1.1792           -345.9%         [12.500.073]         2.01%         2.06%         3.064         3.064         1.064         1.07         1.1792           -345.9%         [12.200.073]         2.01%         2.064         3.064         3.064         3.064         3.064         3.05         1.92         3.05         1.92         3.05         1.92         3.05         1.92         3.05         1.92         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05         3.05	777 080.6 03.380	777 080.6 03.380		177	1		Ę	2010	609,950,227								
-5.8%         (11,701,097)         22.6%         22.6%         20.64         30.64         0         0         1757           -13.6%         (11,701,097)         22.9%         20.9%         30.64         0         0         1757           -13.6%         (11,701,097)         22.9%         20.9%         30.64         30.64         0         0         1775           -33.6%         (116,568,013)         20.9%         20.9%         30.64         30.64         0         0         1775           -33.6%         (116,568,013)         20.9%         20.66         30.64         30.64         10.64         0         0         1775           -33.6%         (116,506,013)         20.9%         30.64         30.64         30.64         30.64         30.64         1975           -33.9%         (235,900,073)         21.7%         21.9%         30.64	22.123 2.993 7 001 Ann	2.993 7 901	100 2			• :		1120	34"3/10'0984	19.6%	19.6%	3,064	3,064	¢		1435	STAT
-15.00%         (11/701.007)         20.5%         70.5%	22,150 2,993 7,961	2.993 7.904	1001	(195)		2:	5	× 90.1	(700,107,12)	22.6%	22.6%	3,064	3,064	•	. 0	1617	2
-13.06%         (19.6496.403)         20.9%	23.370 2.612 2.62	2 511 2 511				ΞI	5	×90.5	(21,701,097)	20.5%	20.5%	3,064	3.064	•		1 705	305
J.J.17         (195,4779,23)         22116         2106         306         306         306         316         0         0         1773           313.666         (197,75,200,173)         (213,590,073)				(784)		6	2)	-19.08%	(116,368,203)	20.9%	20.9%	3.064	3.064	• •	• c	634.1	<u>s</u>
31.56%         (714,715,477)         (19.3%)         23.66%         (104         (101<	24.589 2.067 2.101 (405)		(Sne) (Sne)	(500)		L66)	ć .	-32,11%	(195,879,923)	22.1%	21.0%	1064	3.064	• e			) (r) -
33.8.9%         (735,90,073)         20,1%         20,9%         3,0%         5,1%         1,9%         2,0%         2,1%         1,9%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,1%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0%         2,0% <td>75 RDK   SONC   1000</td> <td></td> <td></td> <td>(24)</td> <td>_</td> <td>(1,042</td> <td>-</td> <td>×95-65</td> <td>(204,725,479)</td> <td>19.3%</td> <td>23.6%</td> <td>3,064</td> <td>3.064</td> <td>104</td> <td>10</td> <td></td> <td>1,72</td>	75 RDK   SONC   1000			(24)	_	(1,042	-	×95-65	(204,725,479)	19.3%	23.6%	3,064	3.064	104	10		1,72
	140cf) 26,758 1,906 1 306 1 306 1 306 1	(acr) 906 1 906 1				26111)	-	38.59%	(ET0,095,252)	20,1%	20.9%	190	3,064	518	315	5781	970 <sup>4</sup> 1
-34.99         (735,990,073)         21.34         21.94         3064         3064         513         513         1927           -50.17%         (735,990,073)         21.34%         19.9%         3064         3064         518         513         1927           -50.17%         (712,212,212,12)         20.9%         19.9%         3064         3064         518         513         1927           -50.17%         (712,212,447)         20.9%         21.0%         10.64         518         513         513         1927           -50.17%         (712,212,447)         20.9%         20.64         3064         3064         3064         21.9%         2023           -50.17%         (711,212,447)         21.5%         21.3%         3064         20.64         21.9%         2023           -50.17%         (711,212,447)         21.5%         21.3%         3064         21.66         21.9%         21.3%         2023           -60.18%         (611,212,447)         21.5%         20.64         3064         21.60         21.1%         21.2%           -60.47         (711,122,447)         21.5%         20.64         3064         21.69         21.8%         21.9%           -60.	27.768 1.906 1.906			-		(1,195 22 100	Ξ.	-38.59%	(ET0,095,255)	19.1%	23.2%	3,064	3,064	518	315	200.1	
-5.0.07%         (313,239,0473)         21.7%         (19,9%         3064         3064         3064         318         513         513         513           -5.0.17%         (310,231,447)         21.0%         (19,9%         3064         3064         3064         3064         3063         3064         3063         3064         3063         2022         2022           -5.0.17%         (310,231,447)         24.2%         21.064         3064         3064         3064         3064         2022         2022         2022           -50.17%         (310,231,447)         24.2%         21.064         3064         3064         3064         21.09         21.190         2.022           -50.17%         (310,231,447)         21.2%         21.064         3.064         3.064         3.064         3.064         2.180         2.1180         2.213           -50.110,231,447         21.3%         21.064         3.064         3.064         3.064         2.180         2.180         2.213           -80.18%         (431,000,41)         19.9%         19.9%         3.064         3.064         3.066         2.066         2.066         2.066         2.066         2.066         2.066         2.066         <	28.321 1.906 1 suc	1.906	-			96T(1)	Ξ.	7.65'8F-	(ET0,096,262)	21.3%	21.9%	3,064	3,064	\$1\$	513	1 947	2.5
Subtry         (1)2.22.447)         (1)9.64         (1)9.94         (1)9.64         (1)9.74	28.374 1.525	1.525	1 2 2 2			(1,198	ά.	-38.59%	(ET0,09E,2ES)	23.7%	19.9%	190,5	3,064	513	515	66	
- 30.417%         (10.2.412.447)         2.0.9%         21.0%         1.064         3.064         2.180         2.1180         2.123         2.025         2.125         3.064         3.066 </td <td>29.980 1.522 1.522</td> <td>(rec) 345 1 555 1</td> <td>(rec) 563 1</td> <td>freeh</td> <td></td> <td></td> <td></td> <td>612 nc</td> <td>(310,282,447)</td> <td>19.6%</td> <td>19.9%</td> <td>3,064</td> <td>3,064</td> <td>518</td> <td>518</td> <td>101</td> <td></td>	29.980 1.522 1.522	(rec) 345 1 555 1	(rec) 563 1	freeh				612 nc	(310,282,447)	19.6%	19.9%	3,064	3,064	518	518	101	
	31,070 1.525 1.525	1.525 1.525	525				۰.	ALL IN	(7)0,282,447)	20.9%	21.0%	3,064	3,064	315	\$15	2.002	70012
Substration         21.27         24.0%         1,064         3,064         2,180         2,180         2,212           Substration         (10.2.82,4.4.4.7)         21.25%         21.13%         3,064         3,064         2,180         2,180         2,212           Substration         (692,000,604)         (19.5%         21.13%         3,064         3,064         3,064         3,064         2,180         2,180         2,212           Substration         (692,000,604)         (19.5%         21.3%         3,064         3,064         3,064         3,064         2,180         2,180         2,306           Substration         (697,090,604)         (19.5%         20.6%         3,064         3,064         3,064         3,064         2,180         2,306	32,160 1,525 1,325	1.525 1.325	1.325				ς,	%/ P.nc-	(310,282,447)	22.6%	22.5%	3.064	3,064	1,090	0601	151	1212
	32,160 1,525 1,525	1,525 1,525	1,525				_		(319,282,447)	24.2%	24.0%	3,064	3,064	2,180	2,180	2.2.12	2.217
-Bill and Bill a	595 595 (930)	595 595 (930)	595 (930)	(930)		(2.404)		A0.84%	(197 000 507)	21.5%	21.3%	3,064	3,064	2,180	2,180	2,272	2.272
Bit Link         (#33,0%10%1)         (19.5%)         (10.6%1)	34,372 595 595	595 595	595			(2.509		-60.84%	(*D0'040'54*)	19.0%	%L 61	8	3,064	2,180	2,180	2,306	2,306
Bit Link         (197)         20.55         3.064         3.064         3.064         3.064         4.360         2.306         2.306           99.53%         (43.3.37.3.37.3.21         20.3%         3.064         3.064         3.064         3.064         3.064         3.064         3.064         3.066         2.306         2.306           99.53%         (647.999.9972)         20.3%         3.064         3.064         3.064         3.064         3.064         2.306         2.306           100.00%         (667.999.99727)         21.3%         20.5%         3.064         3.064         3.064         4.360         2.306           100.00%         (609.960,277)         21.3%         20.5%         3.064         3.064         3.064         3.066         5.400         5.400         2.306           100.00%         (609.956,277)         11.7%         20.5%         3.064         3.064         5.400         5.400         2.306           100.00%         (609.956,277)         19.7%         20.6%         3.064         3.064         5.400         5.400         2.306           100.00%         (609.956,277)         19.7%         20.6%         3.064         5.400         5.400         2.306 </td <td>35,478 595 595</td> <td>595 595</td> <td>595</td> <td></td> <td>- (2.50</td> <td>(2.50</td> <td>16</td> <td>86. A.V.</td> <td>(*00'0.20'0.20)</td> <td>22.61</td> <td>19.5%</td> <td>190</td> <td>3,064</td> <td>3,270</td> <td>3,270</td> <td>2,306</td> <td>2,306</td>	35,478 595 595	595 595	595		- (2.50	(2.50	16	86. A.V.	(*00'0.20'0.20)	22.61	19.5%	190	3,064	3,270	3,270	2,306	2,306
-98.30%         (542,332,364)         3,064         3,066         1,066         3,066	36,031 595 595	295 295	595		12.50	05.5	5	-80.44%	(#00°0.00°0.00)	71.747	Z0.3%	190	3.064	4,360	4,360	2,306	2,306
99.53%         (67),099.921         20.1%         3,064         3,064         4,160         4,360         2,306           -100.00%         (667),099.921271         21.2%         19.7%         3,064         3,064         3,064         3,064         2,306         2,306           -100.00%         (669).950,2277         21.2%         20.3%         3,064         3,064         3,064         3,064         2,306         2,306           -100.00%         (669).950,2771         21.1%         20.5%         3,064         3,064         5,400         5,400         2,306           -100.00%         (669).950,2771         19,7%         20.6%         3,064         3,064         5,400         5,400         2,306           -100.00%         (669).950,2777         19,7%         20.6%         3,064         3,064         5,400         5,400         2,306           -100.00%         (669).950,2777         19,7%         20.6%         3,064         3,064         5,400         5,400         2,306           -100.00%         (669).950,2777         19,7%         20.5%         3,064         3,064         5,400         5,400         2,306           -100.00%         (669).950,2777         19,7%         20.5%	37,137 345	345 345 (250)	345 (250)	(250)		07.75	6	-RR On V		1.5.14	ZU.07	<b>190</b>	3,064	4,360	4,360	2,306	2,306
-100.00%         (609)950,227         20.55         3,064         5,450         2,306         2,306         2,306         2,306         5,490         2,306         2,306         2,306         2,306         5,400         2,306         2,306         2,306         6,540         5,400         2,306         2,306         1,006         6,540         5,400         2,306         1,006         2,306         1,006         6,540         5,400         2,306         2,306         1,006         6,540         5,400         2,306         2,306         1,006         6,540         5,306         2,306         2,306         1,006         6,540         5,306         2,306         2,306         1,006         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306         2,306	38,796 15 15 (330)	15 1 15 (330)	(050)	(055)			6	%US 66-	(785'757'746)	20.8%	20.1%	90	3,064	1,160	4,360	2,306	2,306
-100.00%         (60) 950,277         21.2%         20.3%         3,064         3,450         5,450         5,450         2,306           -100.00%         (60) 950,277         21.7%         20.5%         3,064         3,064         5,40         2,306           -100.00%         (60) 950,277         21.7%         20.5%         3,064         3,064         5,40         2,306           -100.00%         (60) 950,277         19.0%         19.6%         3,064         3,064         5,40         2,306           -100.00%         (60) 950,277         19.0%         19.6%         3,064         3,064         5,40         2,306           -100.00%         (60) 950,2777         19.7%         20.1%         3,064         3,064         5,40         5,40         2,306           -100.00%         (60) 950,2777         19.7%         20.1%         3,064         3,064         5,40         5,40         2,306           -100.00%         (60) 950,2777         19.7%         20.5%         3,064         5,40         5,40         2,306           -100.00%         (60) 950,2777         19.7%         3,054         5,40         5,40         2,306           -100.00%         (600) 950,2777         19.7%	39,349 0 0 0 (15)	0 0 (15)				10	ŝ	-100 00%		24C.UX	19. 1%	3,064	3,064	4,360	4,360	2,306	2,306
-100,00% (609,950,277) 2.1.3% 20.5% 3.064 3.064 6.540 6.540 2.306 -100,00% (609,950,277) 2.1.3% 2.0.6% 3.064 3.064 6.540 6.540 2.306 -100,00% (609,950,277) 19.7% 2.0.1% 3.064 3.064 6.540 6.540 2.306 -100,00% (609,950,277) 19.7% 2.0.1% 3.064 3.064 6.540 6.540 2.306 -100,00% (609,950,277) 19.7% 2.1.5% 3.064 1.064 6.540 6.540 2.306 -100,00% (609,950,277) 19.7% 2.1.5% 3.064 1.064 6.540 2.306 -100,00% (609,950,277) 19.7% 2.1.5% 3.064 1.064 6.540 2.306 -100,00% (609,950,277) 19.7% 2.1.5% 3.064 1.064 6.540 5.540 2.306 -100,00% (609,950,277) 19.7% 2.1.5% 3.064 1.064 6.540 6.540 2.306	+0,455 0 0	0					1	180 004	(177'056'600)	71.2%	20.3%	3,064	3,064	5,450	5,450	2,306	2,306
100,00%         (609,594,277)         21.7%         20.6%         3,064         5,540         6,540         2,306           100,00%         (609,594,277)         19,7%         20,6%         3,064         3,064         5,40         2,306           100,00%         (609,594,277)         19,7%         20,6%         3,064         3,064         5,40         2,306           100,00%         (609,594,277)         19,7%         20,5%         3,064         3,064         5,40         2,306           100,00%         (609,594,277)         19,7%         20,5%         3,064         3,064         5,40         2,306           100,00%         (609,594,277)         19,7%         20,5%         3,064         3,064         5,40         2,306           100,00%         (609,594,277)         19,7%         21,5%         3,064         3,064         5,40         5,40         2,306           100,00%         (609,594,277)         19,7%         21,5%         3,064         3,064         5,40         5,40         2,306           100,00%         (609,994,277)         19,7%         21,5%         3,064         3,064         5,60         5,40         2,306           100,00%         (609,994,277)						1101		100.001	(177'ncc' (ng)	21.5%	Z0.5%	3,064	3,064	6,540	6,540	2,306	2 306
100.00% (669.594,277) 19.0% 19.6% 3.064 3.064 6.540 6.540 2.306 100.00% (609.594,277) 19.7% 20.1% 3.064 3.064 6.540 6.540 2.306 100.00% (609.594,277) 20.3% 3.064 3.064 6.540 6.540 2.306 100.00% (609.594,277) 19.7% 19.5% 3.064 3.064 6.540 6.540 2.306 100.00% (609.594,277) 19.7% 21.5% 3.064 3.064 6.540 6.540 2.306 100.00% (609.594,277) 19.7% 21.5% 3.064 3.064 6.540 6.540 2.306 100.00% (609.594,277) 19.7% 21.5% 3.064 3.064 6.540 6.540 2.306							2		(1.77'INC6' 600)	21.7%	20.6%	3,064	3.064	6.540	6.540	2100	
-100.00% (609.550,227) 19.7% 20.1% 3.064 3.064 5.40 6.540 2.306 -100.00% (609.550,227) 20.3% 2.0.5% 3.064 3.064 6.540 6.540 2.306 -100.00% (609.550,227) 19.2% 19.5% 3.064 3.064 6.540 6.540 2.306 -100.00% (609.550,227) 19.7% 2.11.5% 3.064 1.064 6.540 6.540 2.306 -100.00% (609.550,227) 20.7% 2.11.5% 3.064 1.064 6.540 5.540 2.306	43.220 0 0				(FOI(F))	(501°F)	-	-100.00%	(609,950,217)	19.0%	19.6%	3,064	3,064	6.540	6 540	2 106	
-100.00% (609.950,227) 20.3% 20.5% 3,064 3,064 6,540 6,540 2,306 -100.00% (609.950,227) 19.2% 19.5% 3,064 3,064 6,540 6,540 2,306 -100.00% (609.950,227) 19.7% 21.5% 3,064 3,064 6,540 6,540 2,306 -100.00% (609.950,227) 20.5% 3,064 3,064 2,06 6,540 2,306 -2,306 -100.00% (609.950,227) 20.5% 3,064 3,064 3,064 2,00 5,40 2,306 -2,306 -1,00.00% (609.950,227) 20.5% 3,064 3,064 3,064 -1,060 -1			(3,103	(3)103	- (3,103	(3,103	-	-100.00%	(609,950,227)	19.7%	Z0, I %	3.064	3,064	6 540	2 240		
-100.00% (669.954,227) 19.2% 19.5% 3,664 3,064 3,064 5,400 6,540 2,300 -100.00% (669.954,227) 19.7% 21.5% 3,064 3,064 6,540 6,540 2,306 -100.00% (669.954,0277) 20.0% 2,315% 3,064 3,064 6,540 5,540 2,306 -100.0%		, , ,	(3,10)	(3,10)	- (3,10	(3,10	ŝ	-100.00%	(609,950,227)	20.3%	20.5%	3.064	1.064	6 540	1		00017
-100.00% (609.950,227) 19.7% 21.5% 3,064 3,064 3,064 5,540 5,540 2,306 -100.00% (609.956,227) 20.0% 20.3% 3,064 3,064 3,064 5,540 5,540 2,306	44.879	·			( <u>1</u> )	1	(2,103)	-100.00%	(609,950,227)	19.2%	19.5%	3,064	3.064	6 540	2 4 U		
-100.00% (609.950,2277) 20.0% 20.3% 3.064 1.064 2.500						6	(e) (e)	-100.00%	(172,029,950,2277)	19.7%	21.5%	3,064	3,064	6.540	6 540	225	80.74 2017
					- 1 (3,H	(3,1	5	-100.00%	(T12,059,908)	20.0%	20.3 V	3.064	3.064	6.540	2 C C C C C C C C C C C C C C C C C C C	2017	2017

"FPL existing and incremental generation capacity (includes nuclear generation and DOES NOT INCLUDE PURCHASES).

											1								
-			ļ						BASE	BASE WITH LEE COUNTY	VINY								
			Conitol D	Protection Date															]
	Nuclear	Nurfear	Number							Fixed O&M	05M								ļ
	Turning	2018			NHOCH	Nuclear	Nuclear	Nuctor	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear	Ι.	Cerpital Kep	Boement		1
Year	(2000)	(2000)	10000		1000	1000	1602	2018	2024	2022	2024	2030	2032	TIDI		The Nuclear	Nuchar	Nuclear	Nuclear
2006				(mane)	(annual	(nnat)	(nnot)	(1000)	(2000)	(\$000)	(2000)	(\$000)	(2000)	(2000)	(2000)	(20nm)	renno.	20.50	1602
2007	•	26.848				,	,	'										(anne)	moe
2004	•	26.648				•	•	•	•					,	,				
6002	•	2 282	20,030	, ,	•	•	•	•	•					•					
20102		25 107	20.030		•	•	•	,	•	'	•	•	•	•					
1	,		ACD 47	•	•	•				,	1			•	•	•	,	•	•
		78.1	2,468	31,408	•	•	1				,	•		•	•	•	•	•	
2012		84,437	27,151	31.408	•					•	•	•	•	•	•	,			
		143,771	51.834	2,670	33.971			•	'	•	•	•	,	,	,	,			
2014		212 234	01 227	796 06		•	,	•	•	•	•	•	•	•	,			•	
2015		A23 FOL	100 001			•		•	•	•	•	,						,	
			CUC CEL	10.00	2,536	•	•	•	,	•	,				•	,	,	,	,
		109,445	229,552	98,780	31,763	•	•	,		,	•	•		,	•	•	•	•	•
2017		431,314	328,285	166,192	60,639				•	•		•	•	,					
2011		1.018.766	427.017	248 284	106 840				•	•	•	•	'						
2019		102.530	466.510	246.073				14,868	•	•	•	,	•	11 333			I		•
0000		ONB 11				47,880	•	132,666	•	•			,	10 8.15		•	•	•	•
		121 121	101'ean'i		10 997	42,965	•		80,237	•				0000		•	•		•
1707		01110	275,020,1	LLC 100	364 047	0,654	48,482		142 504	,				000'81	HCC'LL	•	•	•	•
1207		839,120	66C'096	1,109.472	499,549	40.191	45.497		147 600		•	•		19,984	19,964	,	,	,	
1202		806,346	942.365	1.103.634	545,756	76.728	1000		700 1	20,043	•	•	,	Z0,170	20.170	11.766			
2024		C78,677	907.592	1.060.400	1.266 533	135 147	024 54	_	01,721	41/7cl	•	•	•	20,379	20,379	20.379		,	
2025		749,612	874.307	1 019 027	1 107 501	101 000			160,861	151,961	92,247	•		20,596	20.596	20.596	12 048	,	
2026	149,661	720,745	542,429	901 652	146 921	120,004	ADA'70	94.4,201	163,776	163,776	163,776	•		20,821	20,821	20.821	20.421		
2027		691,680	510 B97	GASS BIS 1	1 101 045		0170-070		222,841	169,633	169,633	•		21.058	21.058	21.054	21 050		
2026		562.614	779 554	811.174	1 061 755		006'017		175,719	175,719	175,719	•		21.290	21 290	21200	1000		, .
2029		633.549	101 072	177 17E			170,100		182,051	182,051	182,051	•	,	21516	21416	21 545		•	
2010		For ADY			91 9'77n't	8hc,040	180,526		155,615	188,618	158,618		,	31 760				•	
1000		164 400	10,003		222,000	1,627,874	683,668	-	195,429	195.429	195.420	114 000	, ,		ne/"17	06/17	21,750	,	•
			017 CD9	891 609	946,752	1,510,400	746,897		202.500	202 500	202 600		'	2001/22	900,22	222,000	22,008	12,036	•
7607		944 940	853,824	775,165	911,973	1,451,230	1,760,709		200 842	200 042				005-77	222,300	22,300	22,300	22,300	•
1002		521,146	622,424	741,163	875,196	1,385,704	1,633,640		217.468	217 468	200'AN7	240,802	BUA.221	22,583	22,593	22,503	22,593	22,583	13,179
2034		503,311	591,036	707,176	636,416	1 343 450	1,569,651		Call Sec	COC 410	50V/JLZ	217,465	217,468	22,889	222,069	22,889	22,889	22,689	22,089
1 5602	- 1	489,204	563,671	673,214	801,641	1.294.188	1.509 504				796'077	235,352	225,382	23,207	23,207	23,207	23,207	23.207	23,207
								1	140,000	235,300	233,546	233 586	233,506	23,542	23,542	23,542	23.542	23542	23 647

													ļ						
			Capital R	ovenue Reg.	urtement					i o ma			ľ						
	Indrac.	Ruchar	Necteur	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear	Worlease	Nuclear N	No.		ļ			Capital Repl	acement		ſ
,	Uprate	2107	2020	2022	2024	2039	2832	2014	7870	103-1	Nuclear .	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear	Nuclear
È		(0005)	(2000)	(\$000) (\$000) (\$000)	(000\$)	(\$000)	(2000)	(2005)	(000)	(TOOO)	10001	10001	2832	2018	1020	2022 2024	1024	2038	1601
9007	•	•	•		. 						- Anna	(nnnt)	Innoc	(0004)	(2000)	(2000)	(\$000)	(\$000)	(2000)
2007	'	26,848	•	,	•		,	•	•								i		
2008		28.640	,				•	•	•										
2009	'	2382	70 010			•	,	,	•					,					-
2010	•	75 403		•	,	•	,	•			•	,							-
101			870'97	•	•	•	•	•	,	•				•		•	•	•	•
		47,824	2,468	31,408	•		,				•	•		•	•	•	•	•	,
2012		161,437	27, 151	31,408	•					,	•	,	,	•	•		,		
2013		143.771	51.834	2.670	33.971			•	,		•	•					,		
\$102		212.234	1.227	790 287	12 071	,		•	•	•		,	•	•	•				
2015		303.516	165 E.			•		,		•		,	,					•	•
2016		104 105			2,000	•		•	•		,	,				•	•	,	
2017			700'577	00,.00	29/°LE			,	•	،	,		1		•		•	•	
			320,265	160,192	60,639	•	•							•	•	•	,	•	,
		1,016,766	427,017	24B,264	106,040	•		74 286	,		•	•	•	•	i		•	•	
2019		100,0391	486,510	355,073	181,917	42.985		112 ARA	•	•	,	•	•	11,333	,	,		,	
2020		906,434	1.099.734	481.861	268,544	47 985				•	1	•	•	19,635		•			
1202		671,753	1.020.372	504.577	TAG ANT	1.20.5	to Long		167'00	•		•	,	19,805	11.554			•	
2702		839,120	980.399	1.169.472	490 540		261.04	105,241	142,504		•	•	,	19,964	19.984	,			,
202		806,346	942 BUB	1 103 634	ANG TRO	1 81 104	7670	200,14	147,502	<b>BB,043</b>		•		20,170	20.170	11 766			,
2024		C/8,8/7	907 592	060 400	1 206 111	071.01		912,201	15Z,716	152,718		•	•	20.379	20.379	20.374			, ,
2025		749,812	102,910	1.019.827	1.193.691	TO THE			106,137	158,137	92,247			20,596	20,598	20,596	12.016		
9202		720,745	642.428	981,652	1 146 925			0///201	8//1991	163,776	163,776	•	'	20,821	20.021	20.821	20.621	,	
1202		691,680	610,997	945,651	1,103,045	105 944	748 065	250,901	108,633	169,633	169,633	•		Z1,058	21,058	21.056	21.058		
2028		662,614	779,558	911,171	1.061755	612 040	267 634	AL CL	RL/C/L	1/5,719	175,719	•	,	21,290	21290	21,290	21290	,	,
2029		633,549	748.121	877 175	1 027 816				100,201	150,581	102,051	•	•	21,515	21.518	21515	21516		,
2030		B04,497	716,683	843 170	DAK 577	1 2 2 2 2 4		10001	155,618	188,618	188,616	•	•	21,750	21.750	21 750	1250		
1602		575.466	645 246	SING 1KM	197 840		900,000	624,021	195,429	195,429	195,429	114,000		22,008	22,008	27 008		949 64	
2032		546,446	653,874	775 165	011 071		160.01	005 202	202,500	202,500	202,500	202,500		22,300	22,300	22 300		006 66	•
2033		521,146	622 424	741 183	A75 104			249'En2	209,642	209,042	209,842	209,842	122,408	22,593	22,593	27 603	102.00	202,12	
2034	101,108	116,503	591,036	707 178	838,418	1 111 150	1,044,040	217,465	217,465	217,468	217,468	217,468	217,468	22,889	22,889	22,880	22, 889	27 R90	
2035		489,204	563,671	673.214	901541	1 294 188		236,022	225,382	235,382	225,382	225,382	225,382	23,207	23,207	23.207	23 207	100 62	400'77
'						And I have a	-and ann	000'007	80C'SS7	Z33,506	233,586	233,586	233,586	23,542	23,542	23,542	29,542	23,542	23542

#### Nuclear Fuel Cost M\$

#### Fuel Cost M\$ BASE BASE WITH LEE COUNTY Existing Nuclear New Nuclear Existing Nuclear New Nuclear Year 133 147 Ŧ --÷ ------------146 172 2024 1 254 259 2031 437

#### Nuclear Energy GWh

nergy GWh		\SE	BASE V	VITH LEE COUNTY	
	Existing Nuclear	New Nuclear	Existing Nucle	New Nuclear	_
/ear					
2010	22,582	-	22,582		-
2011	23,765	-	23,765		-
2012	25,774	-	25,774		-
2013	26,361	-	26,361	İ	-
2014	27,751	-	27,751		-
2015	27,353	-	27,353		-
2016	27,355	-	27,355		-
2017		-	27,751		÷
2018	27,353	5,150	27,353	5,	15
2019	27,276	8,785	27,276	8,	78
2020	27,831	13,959	27,831	13,	9
2021	27,353	17,569	27,353	17,	56
2022	27,276	22,720	27,276	22,	72
2023	27,751	26,354	27,751	26,	3
2024	27,433	31,576	27,433	31,	.5
2025	27,276	35,138	27,276	35,	
2026	27,751	35,138	27,751	35.	
2027	27,353	35,138	27,353	35.	
2028	27,355	35,235	27,355	35,	2
2029	27,751	35,138	27,751	35,	· ·
2030	27,353	40,289	27,353	40	2
2031	27,276	43,923		43,	,9
2032	27,364	49,194	27,364	49.	,1
2033	27,833	52,707	27,833	52	
2034	28,217	52,707	28,217	52	
2035			27,742	52	